

Open periodic waveguides: Theory and computation Elizaveta Vasilevskaya

▶ To cite this version:

Elizaveta Vasilevskaya. Open periodic waveguides : Theory and computation. Other [q-bio.OT]. Université Sorbonne Paris Cité, 2016. English. NNT : 2016USPCD008 . tel-01697429v1

HAL Id: tel-01697429 https://theses.hal.science/tel-01697429v1

Submitted on 31 Jan 2018 (v1), last revised 31 Jan 2018 (v2)

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

UNIVERSITÉ PARIS 13 Laboratoire Analyse, Géométrie et Applications, UMR 7539

THÈSE

présentée pour obtenir le grade de

Docteur de l'Université Paris 13

Discipline : Mathématiques

présentée et soutenue publiquement par :

Elizaveta VASILEVSKAYA

le 07 juillet 2016

Open periodic waveguides. Theory and computation

JURY

Mme	Bérangère DELOURME	Université Paris 13	co- $encadrante$
Mme	Laurence HALPERN	Université Paris 13	directrice
М.	Patrick JOLY	INRIA	co- $directeur$
М.	Konstantin PANKRASHKIN	Université Paris-Sud	rapporteur
М.	Nicolas POPOFF	Université de Bordeaux	examinateur
М.	Olaf POST	Universität Trier	rapporteur
М.	Karim RAMDANI	INRIA	président

UNIVERSITÉ PARIS 13 Laboratoire Analyse, Géométrie et Applications, UMR 7539

THÈSE

présentée pour obtenir le grade de

Docteur de l'Université Paris 13

Discipline : Mathématiques

présentée et soutenue publiquement par :

Elizaveta VASILEVSKAYA

le 07 juillet 2016

Open periodic waveguides. Theory and computation

JURY

Mme	Bérangère DELOURME	Université Paris 13	co- $encadrante$
Mme	Laurence HALPERN	Université Paris 13	directrice
М.	Patrick JOLY	INRIA	co- $directeur$
М.	Konstantin PANKRASHKIN	Université Paris-Sud	rapporteur
М.	Nicolas POPOFF	Université de Bordeaux	examinateur
М.	Olaf POST	Universität Trier	rapporteur
М.	Karim RAMDANI	INRIA	président

Remerciements

Je tiens tout d'abord à exprimer mes remerciements les plus sincères à mon directeur de thèse, Patrick Joly, de m'avoir donné la possibilité d'effectuer cette thèse et de m'avoir soigneusement guidée tout au long de ces années. Je lui suis particulièrement reconnaissante pour tous ses conseils avisés et lectures méticuleuses des manuscrits, les longues heures passées à préparer mes transparents, l'attention et les encouragements.

Je remercie très chaleureusement Bérangère Delourme qui m'a accompagnée au quotidien, toujours disponible et attentionnée. C'est grâce à nos échanges réguliers, aussi bien personnels que scientifiques, que j'ai pu mener à terme ce travail. Sans toi cette thèse n'aurait jamais vu le jour !

Je remercie également Sonia Fliss pour les nombreuses discussions autour de notre travail, les relectures du manuscrit et surtout pour son aide précieuse dans les aspects numériques de ce travail.

Je souhaiterais exprimer ma gratitude à Konstantin Pankrashkin et Olaf Post qui m'ont fait l'honneur de rapporter cette thèse. Je remercie Laurence Halpern, Nicolas Popoff et Karim Ramdani d'avoir accepté d'être membres de mon jury. Je suis particulièrement reconnaissante envers Laurence Halpern qui m'a soutenue et encouragée dans les moments difficiles et sans qui ce travail n'aurait pas pu être réalisé.

Durant les six années passées au LAGA, j'ai pu bénéficier d'un cadre convivial et épanouissant. Je remercie tous les membres du laboratoire et plus particulièrement l'équipe "Modélisation et calcul scientifique". Un grand merci à Gilles Scarella pour sa disponibilité en cas de problème informatique et surtout d'avoir accepté le travail qui m'a permis d'effectuer des simulations en C++. Je remercie Linda El Alaoui pour les échanges que l'on a eus autour des TD, ainsi que son soutien et sa sollicitude.

Je suis très reconnaissante envers Alain Grigis qui m'a soutenue et encouragée pendant mon année de master et qui n'a cessé de demander de mes nouvelles.

J'ai une pensée toute particulière pour Anne Quéguiner-Mathieu et Frédéric Clerc. Je voudrais vous remercier pour votre disponibilité, votre soutien dans mes aventures et, bien sûr, pour les nombreux échanges et conseils sur la pédagogie, qui m'ont tant apporté, lors de cette dernière année au LAGA. Elle était dure mais passionnante ! Je remercie également Laurent Rigal de m'avoir ouvert cette porte.

Mes remerciements vont aussi à l'équipe administrative du LAGA : Yolande Jimenez, Isabelle Barbotin et Jean-Philippe Domergue. Sans votre disponibilité et votre efficacité les démarches administratives aurait été beaucoup moins évidentes. Je suis également très reconnaissante envers le directeur de l'école doctorale Galilée, Vincent Lorent, pour son soutien.

Pendant ma thèse, j'ai été chaleureusement accueillie au sein de l'UMA (L'Unité de Mathématiques Appliquées de l'ENSTA ParisTech). Je remercie tous les membres du laboratoire POEMS et tout particulièrement Anne-Sophie Bonnet-BenDhia, ainsi que Valentin, Camille, Geoffrey, Marina. C'était un grand plasir et une vraie chance de faire partie d'une équipe si brillante.

Enfin les amis doctorants du B410, dans l'ordre chronologique de nos rencontres. Les "anciens": Nikolaj (c'est toi qui m'as convaincue de venir à Paris 13 et m'as aidée dans mes premiers pas en France), David (Votre majesté), Binh (le prédécesseur), Amine (et son séminaire des doctorants); ensuite Rémi (et ses soirées jeux de société), Julien (et ses chats, ou plutôt les nôtres), Taiwang (tu ne m'as jamais expliqué pourquoi tu me tues tout le temps dans les jeux); enfin les "nouveaux" : Eva (tu t'inquétais tant pour mes soucis), Annalaura (la chef), Bruno (je ne te pardonnerai jamais d'avoir pris ma place !), Pierre (et son C++ ainsi que ses manuels). C'est vous tous qui faites du bureau ce qu'il est : un endroit où on a envie de revenir (quand il n'est pas vide). Je vous en remercie ! Je n'oublie pas Marion (et nos discussions autour de l'enseignement ainsi que ton soutien et ton optimisme), Guillaume (et nos conversations en russe ainsi que nos encouragements mutuels pour les concours), Tom (et nos échanges de mails du dimanche soir pour préparer les TD du lundi matin), Irène, Amal, Kaouther, Alexandre, Guiseppe ($\times 2$), Asma, Didier, Roland et tous ceux avec qui nos chemins se sont croisés.

Je pense également à mes amis russes, Svetlana, Zhenya et Olga. Même si on n'arrivait pas à se voir souvent, sans votre soutien les difficultés de cette aventure à l'étranger aurait été beaucoup plus difficiles à surmonter.

Ces remerciements seraient incomplets sans une pensée particulière pour ma famille. Papa, c'est toi qui m'as transmis le goût des sciences et c'est ton exemple que j'ai suivi. Je ne sais pas si j'ai été à la hauteur de tes attentes, mais j'espère que tu aurais été content si tu avais pu voir ce jour. Grand-mère, tu n'aurais pas pu imaginer que la vie m'amène dans le pays de ton rêve. Je parle ta langue préférée maintenant ! J'aimerais tant que tu puisses être là. Grand-père, tu t'inquiétais pour ma thèse plus que moi. C'est aussi grâce à toi que j'ai persévéré dans ce travail. J'espère que son aboutissement te fait plaisir et compense, au moins en partie, mes longues absences. Enfin maman. Inutile de dire que sans votre affection et le soutien constant que vous m'apportez, toi et Gerald, je n'en serais pas là. Je sais que ces années n'ont pas été faciles pour toi, avec tous les doûtes et inquiétudes. J'espère à travers ce travail, aussi modeste soit-il, te rendre heureuse et fière de moi !

À ma famille

CONTENTS

In	trod	uction		1
	0.1	Descri	ption of the problem	3
1	Tra	pped n	nodes in a locally perturbed periodic ladder: existence results	7
	1.1	Preser	ntation of the problem	7
	1.2	Mathe	ematical formulation of the problem	8
		1.2.1	Determination of the essential spectrum of A^{μ}_{ε}	9
		1.2.2	The essential spectrum of the operator A^{μ}_{ε}	10
		1.2.3	Method of study	11
	1.3	Spectr	cal problem on the graph	12
		1.3.1	The operator A^{μ}	12
		1.3.2	The spectrum of the operator A_s^{μ}	14
		1.3.3	Results for the operator A^{μ}_{as}	29
		1.3.4	The spectrum of the operator A	31
	1.4	Existe	ence of eigenvalues for the non-limit operator	31
		1.4.1	Main result	31
		1.4.2	Method of a pseudo-mode	32
		1.4.3	Construction of a pseudo-mode	33
2	Tra	pped r	nodes in a locally perturbed periodic ladder: asymptotic ex-	
	pan	sions o	of the eigenvalues	37
	2.1	Forma	ll expansions	37
	2.2	Near f	field and far field problems	42
		2.2.1	Formal derivation of the near field problems	42
		2.2.2	Formal derivation of the far field problem	47
		2.2.3	Order 0	49
		2.2.4	Well-posedness of the near field problems	50
		2.2.5	Well-posedness of the far field problem	56
	2.3	Existe	ence of the terms of the asymptotic expansions	60
	2.4	Const	ruction of a pseudo-mode at any order	68

3	Traj	pped modes in a locally perturbed periodic ladder: numerical study 77	7
	3.1	Goals and difficulties	7
	3.2	DtN operator method	8
		3.2.1 Half-band problems	8
		3.2.2 The DtN operators Λ^{\pm}	2
		3.2.3 The interior problem $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots $	4
		3.2.4 Practical construction of the operators $\Lambda^{\pm}(\alpha)$	8
	3.3	Discretization by a conform finite element method	3
		3.3.1 Discretization of the operator A^{μ}_{ε}	4
		3.3.2 The essential spectrum $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots $	9
		3.3.3 Summary of the algorithm $\ldots \ldots \ldots$	1
	3.4	Numerical results	2
		3.4.1 Validation of the computation of the local DtN operators 102	2
		3.4.2 Numerical computation of the essential spectrum	7
		3.4.3 Numerical computation of the discrete spectrum	9
4	Gui	ided modes in open periodic lineic waveguides: the 2D case 110	ŋ
-	4 1	Existence results 12	1
	1.1	4.1.1 The essential spectrum of the limit operator 12	3
		4.1.2 The discrete spectrum of the limit operator 13	1
		4.1.3 Results for the operator $A^{\mu}(\beta)$	5
		4.1.4 A more general geometry \dots 130	6
	4.2	Numerical approach	8
		4.2.1 Numerical results	8
5	Gui	ided modes in open periodic lineic waveguides: the 3D case 143	3
	5.1	1 Geometry and statement of the problem $\ldots \ldots \ldots \ldots \ldots \ldots \ldots$	
	5.2	The limit operator $\ldots \ldots \ldots$	5
		5.2.1 Computation of the essential spectrum $\ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots 14'$	7
		5.2.2 Computation of the discrete spectrum	9
	5.3	The operator $A^{\mu}_{\varepsilon}(\beta)$	9
A	App	pendices 177	1
	A.1	Self-adjointness of the operator A^{μ}	1
	A.2	Quasi-modes method	2
	A.3	Some auxiliary assertions	5
	A.4	Technical results associated with Chapter 5	7
		-	

INTRODUCTION

Motivation

The present work deals with study of periodic media. Periodic media play an important role in applications, such as solid state physics and optics. One talks about a periodic medium when the geometry and the physical characteristics of the problem are periodic functions. The periodicity of materials is observed at the atomic scale. Arranging different materials in repeating patterns can give rise to metamaterials. The properties of such a material are mainly no more determined by the properties of the materials it is composed of but by the way they are assembled. As a consequence, a metamaterial can present properties that do not exist in nature or are difficult to obtain.

Electromagnetic bandgap metamaterials affect light propagation. This is the case of photonic crystals which are structures composed of periodically alternated dielectric regions with high and low dielectric constants. The interest of these structures lies in the existence of intervals of "forbidden" frequences, i.e. frequences for which light cannot propagate in the medium (such intervals are called band gaps). The intervals of "permitted" frequences (for which light propagation is possible) are called spectral bands.

From the mathematical point of view, the existence of band gaps is explained by the spectral properties of the underlying periodic partial differential operators. The spectra of such operators are known to have a band-gap structure (Floquet-Bloch theory [15, 40]).

At the same time, introducing a perturbation to a perfectly periodic medium can lead to appearance of "permitted" frequences inside spectral gaps (which corresponds to the appearance of isolated eigenvalues of finite multiplicity for the underlying operator). One often talks about local perturbations and linear defects (the one considered in the present work). These eigenvalues inside gaps give rise to the so-called "trapped modes" (in the case of a local perturbation) and "guided modes" (in the case of a linear defect), which are, roughly speaking, solutions of the wave equation which are localized in the neighbourhood of the perturbation (a guided mode can be seen as a wave propagating along the defect and confined in the transversal direction). Such localized solutions are particularly interesting for applications such as design of lasers, filters and waveguides (cf. [33, 34]). The present work is devoted to the study of a particular type of waveguides that will be described in detail below. The aim is to create guided modes by introducing a geometrical perturbation of a purely periodic material.

State of the art

Two questions arise naturally when studying periodic media: the existence of gaps and the existence of eigenvalues inside gaps when a perturbation is introduced. Neither of them is completely answered. In the one-dimensional case, the gaps always exist except for constant media ([9]).

Necessary conditions of existence of gaps in higher dimensions are not known (in the onedimensional case it is well-known that a periodic operator has no gaps if and only if it is constant, see [9]). However, some examples of sufficient conditions leading to the presence of gaps have been found in [20, 21, 52, 54, 3, 28, 29, 39] and references therein. According to the Bethe-Sommerfeld conjecture, a periodic operator in higher dimensions can only have a finite number of gaps (this is not true in the one-dimensional case, where, in general, a periodic operator has infinitely many gaps). The Bethe-Sommerfeld conjecture has been completely proved for the periodic Schrödinger operator ([55, 56]), but is still partially open for Maxwell equations ([67]).

For the second question, which is the possibility of creating eigenvalues inside gaps, some examples have been given in [18, 19, 1, 41, 46]. In these works strong contrasts in the properties of the medium are required in order to ensure the existence of eigenvalues. In [48, 49] guided modes are found in periodic lattices.

In the present work we consider open periodic waveguides having the geometry of a fattened grid described in more detail in Section 0.1. As the thickness of the grid tends to zero, the domain shrinks to a graph. We use then the classical approach of asymptotic analysis (used, for instance, in [20, 54]), which consists in approximating the problem in question by a limit one posed on a graph, for which the spectrum is easier to determine.

The convergence of the spectrum of operators in thin domains to the spectrum of the corresponding limit operator defined on a graph has been studied in the literature. In [61, 47] the convergence of the eigenvalues has been established in the case of bounded domains (which implies due to the Floquet-Bloch theory the convergence of the spectrum for periodic domains). In [57] this result has been extended to much more general domains (not necessarily bounded), for which the convergence of all components of the spectrum has been proved. In our case this implies the existence of eigenvalues for ε small enough. We propose though another (a less general but more explicit) proof based on the construction of a quasimode.

A periodic medium being infinite, this presents a difficulty for numerical study of localized modes. Several methods have been developed in order to overcome this difficulty. The most classical one is the "supercell" method which consists in truncating the computation domain far from the perturbation and solving the problem in the truncated domain with periodic boundary conditions. The localized modes being exponentially decaying, the solution of the problem in the truncated domain tends to infinity ([10], [64], [62], [13]). Another method (and this is the one we use in the present work) is based on the Dirichlet-to-Neumann (DtN) approach developed in [24], [22], [23] (see also [38]). This method consists in replacing the initial eigenvalue problem in an infinite domain by a nonlinear eigenvalue problem posed in a neighbourhood of the perturbation (not necessarily big)

via special DtN operators. The DtN operators depending themselves on the frequency, the obtained problem is thus nonlinear. It is solved using fix point type methods.

0.1 Description of the problem

In this present work we study the propagation of acoustic waves described by the scalar wave equation

$$\frac{\partial^2 u}{\partial t^2} = \Delta u, \qquad (0.1.1)$$

and homogeneous Neumann boundary conditions in open waveguides having the geometry shown in figure 1. The propagation domain (in grey) is supposed homogeneous. It can be seen as \mathbb{R}^2 minus an infinite set of periodically spaced rectangular obstacles of size $1 \times L$. The distance between the obstacles, denoted by ε , is supposed to be small.



Figure 1: Propagation domain (grey area)

The principle question that we are interested in is the possibility of creating of guided modes in this type of waveguides. It is known that the existence of guided modes requires the introduction of a perturbation. The perturbation that we consider is a geometric one: without changing the properties of the medium (which is still homogeneous) we introduce a linear defect in its geometry by modifying the thickness of one infinite branch of the domain from ε to $\mu\varepsilon$ with some positive coefficient μ (cf. figure 1b). It turns out that for $\mu < 1$ (i.e. when the domain shrinks) guided modes do appear. We conjecture that for $\mu > 1$ (i.e. when the domain is enlarged) there are no guided modes (at least for ε small enough).

By creating a guided mode we mean, roughly speaking, searching a solution of the equation (0.1.1) with Neumann boundary conditions which propagates along the perturbation and is confined in the transversal direction. As explained in more detail in Chapter 4, this

implies the following form for the solution:

$$u(x, y, t) = e^{i\omega t}v(x, y),$$

where the function v is β -quasiperiodic:

$$v(x, y+L) = e^{-i\beta}v(x, y).$$

For the function v one gets then an eigenvalue problem for the Laplacian in a periodicity band $\mathcal{P}^{\mu}_{\varepsilon}$ of the domain $\Omega^{\mu}_{\varepsilon}$ (cf. figure 2a) with β -quasiperiodic boundary conditions on the upper and lower parts of the boundary:

Find
$$v \in L_2(\mathcal{P}^{\mu}_{\varepsilon})$$
 such that
$$\begin{cases} \Delta v = -\omega^2 v, \\ v|_{\Sigma'} = e^{-i\beta} v|_{\Sigma}, \\ \partial_n v = 0 \quad \text{on the boundaries of the obstacles.} \end{cases}$$

Thus, we have to study the discrete spectrum (the so-called trapped modes) of the Lapla-



cian with β -quasiperiodic conditions in the periodicity band $\mathcal{P}^{\mu}_{\varepsilon}$ that we call a "ladder". It turns out that the spectral properties of the Laplacian with β -quasiperiodic conditions in the "ladder" are very similar to the ones of the Laplacian with Neumann boundary conditions. For this reason, we consider the Neumann Laplacian in the ladder as a model problem and the first three chapters are devoted to the of study this model problem. In Chapter 4 we study the β -quasiperiodic case using the same scheme as the Neumann case.

General methodology

Both in the case of Neumann Laplacian and in the β -quasiperiodic case the method of study consists of three main steps:

- 1. Identification of the (formal) limit problem. As $\varepsilon \to 0$, the periodicity band shown in figure 2a (the "ladder") tends to a periodic graph (cf. figure 2b). Moreover, due to Neumann boundary conditions, a trapped mode in the "ladder" can be approximated by a one-dimensional function defined on the limit graph. This limit function is an eigenfunction of the limit operator defined as the second-order derivative operator on each edge of the graph completed by transmission conditions (called Kirchhoff's conditions) at the vertices of the graph ([17, 11, 47]). Let us mention that quantum graphs have been studied exhaustively in the literature: one can refer, for example, to the surveys [42, 43, 44] as well as the books [6, 58] and the bibliography therein.
- 2. Computation of the spectrum of the limit problem. We first investigate the essential spectrum using the Floquet-Bloch theory. Then, we compute the discrete spectrum using a reduction to a finite difference scheme ([2, 16]).
- 3. Asymptotic analysis. The last step is to show that the formal limit problem is indeed a good approximation of the initial one. In order to prove the existence of guided modes we need precisely the following assertion: if the limit operator has an eigenvalue λ_0 inside a gap $[a_0, b_0]$, then the non-limit operator also has an eigenvalue λ_{ε} inside a gap $[a_{\varepsilon}, b_{\varepsilon}]$ such that $a_{\varepsilon}, b_{\varepsilon}, \lambda_{\varepsilon}$ are close to a_0, b_0, λ_0 respectively for ε small enough. Notice that the convergence of the spectrum of the non-limit operator to the spectrum of the limit one cannot be uniform.

Structure of the work

The present work is organized as follows.

In **Chapter 1** we study the spectrum of the limit problem for the ladder. We prove that the limit operator (more precisely, its symmetric part A_s^{μ}) has infinitely many gaps and one or two simple eigenvalues inside each gap if $\mu < 1$. For $\mu > 1$ the limit operator has no eigenvalues (Proposition 1.3.4 and Theorem 1.3.1). We then deduce by asymptotic analysis that for ε small enough the non-limit operator has arbitrarily many gaps (for any $k \in \mathbb{N}$ there exists $\varepsilon_k > 0$ such that for $\varepsilon < \varepsilon_k$ the operator $A_{\varepsilon,s}^{\mu}$ has at least k gaps and at least one or two eigenvalues in each of these gaps), cf. Theorems 1.4.1, 1.4.3. This last result concerning the eigenvalues is obtained by considering a "naive" quasimode that permits to show the convergence of the eigenvalues of the operator $A_{\varepsilon,s}^{\mu}$ to the ones of the operator A_s^{μ} at order $\sqrt{\varepsilon}$, which is not optimal (the convergence is actually linear in ε).

In Chapter 2 we obtain a full asymptotic expansion of the eigenvalues of the operator $A_{\varepsilon,s}^{\mu}$ which proves at the same time the linear convergence of the eigenvalues. This is done by considering other quasimodes which are constructed using matched asymptotic expansions of the solution in the ladder. Far from the vertices the solution is modelized by one-dimensional functions defined on the graph (the so-called far field expansion). In the neighbourhood of the vertices a rescaling is done (the so-called near field expansion). Finally, both expansions are supposed to be valid in some intermediate areas called matching areas.

In **Chapter 3** the numerical approach to the problem is discussed. We first remind the Dirichlet-to-Neumann (DtN) operator method developed for numerical study of periodic media. More precisely, the initial eigenvalue problem on an unbounded domain can be

reduced to a nonlinear eigenvalue problem posed on a bounded domain. Then, we give some details of implementation of this method using the P1 finite element discretization. Finally, we present numerical results for the problem in the ladder obtained using this method. We compare the results for the eigenvalues obtained via the DtN approach with the ones obtained by computing numerically the first terms of the full asymptotic expansion.

In **Chapter 4** we come back to the waveguide problem. The computations in this chapter are very similar to the ones performed in Chapter 1 for the ladder. Analogues of the results obtained for the operator $A^{\mu}_{\varepsilon,s}$ (existence of gaps and of eigenvalues inside gaps) are established in the β -quesiperiodic case for the operator $A^{\mu}_{\varepsilon}(\beta)$ (Theorems 4.1.2, 4.1.3). This shows the existence of guided modes in the case $\mu < 1$ for ε small enough. A slightly more general geometry is discussed in Section 4.1.4. Varying an additional parameter permits to influence the size of the gaps. In conclusion, numerical results for the β quasiperiodic case are presented. A time-dependent simulation is described that shows the presence of a guided mode (in the temporal regime).

In **Chapter 5** we discuss a 3D generalization of the waveguide studied in Chapter 4. The limit problem on the corresponding graph is studied and analogues of the most part of the results obtained in the 2D case (existence of gaps and of eigenvalues inside gaps) are established, cf. Proposition 5.2.3 and Theorem 5.2.1. The equations describing the 3D case are very similar to the ones describing the 2D case. The conclusions for the non-limit operator are given in Theorems 5.3.1, 5.3.2. The principal difference between the 2D and the 3D case is that in the 3D the equations are less explicit, which makes the analysis somewhat more technical.

CHAPTER 1

TRAPPED MODES IN A LOCALLY PERTURBED PERIODIC LADDER: EXISTENCE RESULTS

1.1 Presentation of the problem

This chapter is devoted to the research of localized modes (also called trapped modes) in a ladder-like periodic domain (cf. figure 1.1). The domain Ω_{ε} is supposed homogeneous and consists of an infinite band of height L minus an infinite set of equispaced rectangular obstacles. The domain is 1-periodic (with respect to x). The distance between two consecutive obstacles and the distance from the obstacles to the boundary of the band is denoted by ε and is supposed to be small:

$$\Omega_{\varepsilon} = \left(\mathbb{R} \times \left] - \frac{L}{2}, \frac{L}{2} \right[\right) \setminus \bigcup_{j \in \mathbb{Z}} \mathcal{S}_{\varepsilon, j}, \qquad \mathcal{S}_{\varepsilon, j} = \left[j + \frac{\varepsilon}{2}, j + 1 - \frac{\varepsilon}{2} \right] \times \left[-\frac{L}{2} + \varepsilon, \frac{L}{2} - \varepsilon \right].$$

By a localized mode we mean a solution of the homogeneous scalar wave equation with



Figure 1.1: Propagation domain (in grey)

Neumann boundary conditions, i.e.,

$$\frac{\partial^2 u}{\partial t^2} = \Delta u \quad \text{in} \quad \Omega_{\varepsilon}, \qquad \frac{\partial u}{\partial n}\Big|_{\partial\Omega_{\varepsilon}} = 0, \tag{1.1.1}$$

which is confined in the x-direction. More precisely, u is supposed to have the following form:

$$u(x, y, t) = v(x, y)e^{i\omega t}, \qquad v \in L_2(\Omega_{\varepsilon}), \tag{1.1.2}$$

where the term $e^{i\omega t}$ shows the harmonic dependence on time whereas the function v (which does not depend on the time) is in some sense confined (since it belongs to $L_2(\Omega_{\varepsilon})$). Plugging (1.1.2) into (1.1.1) leads to the following problem for the function v:

$$\begin{cases} -\Delta v = \omega^2 v & \text{in } \Omega_{\varepsilon}, \\ \frac{\partial v}{\partial n} \Big|_{\partial \Omega_{\varepsilon}} = 0. \end{cases}$$
(1.1.3)

Problem (1.1.3) is an eigenvalue problem posed in the unbounded domain Ω_{ε} . It is wellknown (cf. Theorem XIII.86 in [60], volume IV) that elliptic periodic operators in 2D domains have no eigenvalue. In order to create eigenvalues one needs to introduce a perturbation. We will consider a local perturbation of the geometry of the domain where the width of one vertical edge is modified from ε to $\mu\varepsilon$ with some $\mu > 0$ (see figure 1.2):

$$\Omega_{\varepsilon}^{\mu} = \left(\mathbb{R} \times \left|-\frac{L}{2}, \frac{L}{2}\right|\right) \setminus \left\{\mathcal{S}_{\varepsilon}^{\mu,+} \cup \mathcal{S}_{\varepsilon}^{\mu,-}\right\},$$

$$\mathcal{S}_{\varepsilon}^{\mu,+} = \left(\left[\frac{\mu\varepsilon}{2}, 1-\frac{\varepsilon}{2}\right] \cup \bigcup_{j \in \mathbb{N}^{*}} \left[j+\frac{\varepsilon}{2}, j+1-\frac{\varepsilon}{2}\right]\right) \times \left[-\frac{L}{2}+\varepsilon, \frac{L}{2}-\varepsilon\right],$$

$$\mathcal{S}_{\varepsilon}^{\mu,-} = \left(\left[-1+\frac{\varepsilon}{2}, -\frac{\mu\varepsilon}{2}, \right] \cup \bigcup_{j \in \mathbb{N}^{*}} \left[-j-1+\frac{\varepsilon}{2}, -j-\frac{\varepsilon}{2}\right]\right) \times \left[-\frac{L}{2}+\varepsilon, \frac{L}{2}-\varepsilon\right],$$

$$N$$

$$N$$

$$N$$

$$\Omega_{\varepsilon}^{\mu}$$



Figure 1.2: Perturbed domain $(\mu < 1)$

As we will see, such a perturbation does not change the essential spectrum of the underlying operator but it can introduce a non-empty discrete spectrum, which is exactly what we are interested in (since this discrete spectrum corresponds to trapped modes). A precise mathematical description of the problem is given in the next section.

1.2 Mathematical formulation of the problem

Let us introduce the operator A_{ε}^{μ} in the space $L_2(\Omega_{\varepsilon}^{\mu})$, associated with the eigenvalue problem (1.1.3) in the perturbed domain:

$$A^{\mu}_{\varepsilon}u = -\Delta u, \qquad D(A^{\mu}_{\varepsilon}) = \left\{ u \in H^{1}_{\Delta}(\Omega^{\mu}_{\varepsilon}), \quad \frac{\partial u}{\partial n} \Big|_{\partial \Omega^{\mu}_{\varepsilon}} = 0 \right\}.$$

Here

$$H^1_{\Delta}(\Omega^{\mu}_{\varepsilon}) = \left\{ u \in H^1(\Omega^{\mu}_{\varepsilon}), \quad \Delta u \in L_2(\Omega^{\mu}_{\varepsilon}) \right\}.$$

The operator A^{μ}_{ε} is self-adjoint and positive. We have then to study its spectrum and, more precisely, to find sufficient conditions for existence of eigenvalues.

1.2.1 Determination of the essential spectrum of A^{μ}_{ε}

To determine the essential spectrum of the operator A^{μ}_{ε} , we start by studying the perfectly periodic case ($\mu = 1$). In this case, the domain is Ω_{ε} (figure 1.1) and the corresponding operator A^{1}_{ε} will be denoted by A_{ε} .

According to the Floquet-Bloch theory, periodic elliptic operators do not have discrete spectrum and their essential spectrum has a band-gap structure [15, 60, 40]:

$$\sigma(A_{\varepsilon}) = \sigma_{ess}(A_{\varepsilon}) = \mathbb{R} \setminus \bigcup_{1 \le n \le N} a_n, b_n[.$$
(1.2.1)

The intervals $]a_n, b_n[$ are called spectral gaps. Their number N is conjectured to be finite (Bethe-Sommerfeld, 1933, [55, 56, 67]).

The band-gap structure of the spectrum is a consequence of the following result given by the Floquet-Bloch theory:

$$\sigma(A_{\varepsilon}) = \bigcup_{\theta \in [-\pi,\pi]} \sigma\left(A_{\varepsilon}(\theta)\right), \qquad (1.2.2)$$

where for any $\theta \in [-\pi, \pi]$, $A_{\varepsilon}(\theta)$ is the Laplace operator defined on the periodicity cell $C_{\varepsilon} = \Omega_{\varepsilon} \cap \{x \in \left[-\frac{1}{2}, \frac{1}{2}\right]\}$ (cf. figure 1.3) with θ -quasiperiodic boundary conditions on the lateral boundaries: for $\theta \in [-\pi, \pi]$,

$$A_{\varepsilon}(\theta) : L_2(\mathcal{C}_{\varepsilon}) \longrightarrow L_2(\mathcal{C}_{\varepsilon}), \qquad A_{\varepsilon}(\theta)u = -\Delta u, \qquad (1.2.3)$$

$$D(A_{\varepsilon}(\theta)) = \left\{ u \in H^{1}_{\Delta}(\mathcal{C}_{\varepsilon}), \quad \partial_{n} u |_{\partial \mathcal{C}_{\varepsilon} \cap \left\{ x \in \left] - \frac{1}{2}, \frac{1}{2} \right[\right\}} = 0,$$
(1.2.4)

$$u|_{x=\frac{1}{2}} = e^{-i\theta} u|_{x=-\frac{1}{2}}, \quad \partial_x u|_{x=\frac{1}{2}} = e^{-i\theta} \partial_x u|_{x=-\frac{1}{2}} \bigg\}.$$
 (1.2.5)



Figure 1.3: Periodicity cell

For each $\theta \in [-\pi, \pi]$ the operator $A_{\varepsilon}(\theta)$ is self-adjoint and positive and its resolvent is compact due to the compactness of the embedding $H^1(\mathcal{C}_{\varepsilon}) \subset L_2(\mathcal{C}_{\varepsilon})$. Its spectrum is then a sequence of non-negative eigenvalues of finite multiplicity tending to infinity:

$$0 \leqslant \lambda_1(\varepsilon, \theta) \leqslant \lambda_2(\varepsilon, \theta) \leqslant \dots \leqslant \lambda_n(\varepsilon, \theta) \leqslant \dots, \qquad \lim_{n \to \infty} \lambda_n(\varepsilon, \theta) = +\infty.$$
(1.2.6)

In (1.2.6) the eigenvalues are repeated with their multiplicity. The functions $\theta \mapsto \lambda_n(\varepsilon, \theta)$ are called dispersion curves and are known to be continuous and non-constant (cf. Theorem XIII.86 in [60], volume IV). Thus, (1.2.2) can be rewritten as

$$\sigma(A_{\varepsilon}) = \bigcup_{n \in \mathbb{N}} \lambda_n \left(\varepsilon, \left[-\pi, \pi\right]\right),$$

which gives (1.2.1). The conjecture of Bethe-Sommerfeld means that for n large enough the intervals $\lambda_n (\varepsilon, [-\pi, \pi])$ overlap. The fact that the dispersion curves are non-constant implies that no one of the intervals $\lambda_n (\varepsilon, [-\pi, \pi])$ is reduced to a point and the operator $\underline{A_{\varepsilon}}$ has no eigenvalues. Finally, the dispersion curves are even: indeed, $D(A_{\varepsilon}(-\theta)) = \overline{D(A_{\varepsilon}(\theta))}$ and the operators $A_{\varepsilon}(\theta)$ have real coefficients. Thus, it is sufficient to consider $\theta \in [0, \pi]$ in (1.2.2).

1.2.2 The essential spectrum of the operator A^{μ}_{ε} .

It is well-known that local perturbations of the domain do not change the essential spectrum of the corresponding operator. This is due to Weyl's Theorem (see, for example, Ch.13 Vol. 4 in [60], Ch. 9 in [7], Theorem 1 in [18]). For the sake of completeness we prove this result in our case.

Proposition 1.2.1. $\sigma_{ess}(A^{\mu}_{\varepsilon}) = \sigma_{ess}(A_{\varepsilon}).$

This is a direct consequence of the following assertion.

Lemma 1.2.1. Let $\chi \in C^{\infty}(\Omega_{\varepsilon})$ be a function such that

1.
$$\partial_{\chi}|_{\partial\Omega_{\varepsilon}} = 0,$$

2. $\exists M > 0$ such that $|x| > M \Rightarrow \chi(x, y) = 1$.

If $\{u_j\}_{j\in\mathbb{N}}$ is a singular sequence for the operator A_{ε} corresponding to the value λ , then there exists a subsequence of $\{\chi u_j\}_{j\in\mathbb{N}}$ which is also a singular sequence for the operator A_{ε} corresponding to the value λ .

Proof. By definition, the sequence $\{u_i\}_{i\in\mathbb{N}}$ has the following properties:

1. $u_j \in D(A_{\varepsilon}), \quad j \in \mathbb{N};$ 2. $\inf_j \|u_j\|_{L_2(\Omega_{\varepsilon})} > 0;$ 3. $u_j \xrightarrow{w} 0$ in $L_2(\Omega_{\varepsilon})$ ($\{u_j\}$ is weakly convergent to 0 in $L_2(\Omega_{\varepsilon}));$ 4. $A_{\varepsilon}u_j - \lambda u_j \longrightarrow 0.$

Let us show that there exists a subsequence of $\{\chi u_j\}_{j\in\mathbb{N}}$ which has the same properties as well. The property 1 is verified due to the property 1 of the function χ . To obtain a lower bound for $\|\chi u_j\|_{L_2(\Omega_{\varepsilon})}$ we will show that

$$||u_j||_{L_2(K)} \longrightarrow 0, \quad j \longrightarrow \infty, \quad \text{for all compact } K \subset \Omega_{\varepsilon}.$$
 (1.2.7)

Indeed, from

$$(A_{\varepsilon}u_j - \lambda u_j, u_j)_{L_2(\Omega_{\varepsilon})} = \|\nabla u_j\|_{L_2(\Omega_{\varepsilon})} - \lambda \|u_j\|_{L_2(\Omega_{\varepsilon})},$$

taking into account the properties 2–4 of the sequence $\{u_j\}$ we conclude that $\sup_j ||u_j||_{H^1(\Omega_{\varepsilon})} < C$ for some C > 0. We can then extract a subsequence still denoted by $\{u_j\}$ such that

$$u_j \xrightarrow{w} 0, \qquad j \longrightarrow \infty, \qquad \text{in} \quad H^1(\Omega_{\varepsilon}).$$
 (1.2.8)

Thus, (1.2.7) is verified and, consequently, $\inf_{j} \|\chi u_{j}\|_{L_{2}(\Omega_{\varepsilon})} \ge \inf_{j} \|u_{j}\|_{L_{2}(\Omega_{\varepsilon} \cap \{|x| > M\})} > 0$. The property 3 being obvious the only thing to show is the property 4 for the sequence $\{\chi u_{j}\}_{j \in \mathbb{N}}$. We have:

$$\|A_{\varepsilon}(\chi u_j) - \lambda(\chi u_j)\|_{L_2(\Omega_{\varepsilon})}^2 \leqslant \|\chi(A_{\varepsilon}u_j - \lambda u_j)\|_{L_2(\Omega_{\varepsilon})}^2 + 2\|\nabla\chi\nabla u_j\|_{L_2(\Omega_{\varepsilon})}^2 + \|u_j\Delta\chi\|_{L_2(\Omega_{\varepsilon})}^2.$$

The first and the last terms in the right-hand side tend to zero (due to the property 4 of $\{u_j\}$ and (1.2.7)). Let us estimate the second term.

$$\begin{split} \|\nabla\chi\nabla u_j\|_{L_2(\Omega_{\varepsilon})}^2 &= \int_{\Omega_{\varepsilon}} \nabla u_j \nabla \overline{u}_j |\nabla\chi|^2 d\Omega \\ &= -\int_{supp(\nabla\chi)} u_j \text{div} \left(\nabla \overline{u}_j |\nabla\chi|^2\right) d\Omega + \int_{\partial\Omega_{\varepsilon}} u_j \partial_n \overline{u}_j |\nabla\chi|^2 d\Gamma. \end{split}$$

The last term in the right-hand side vanishes due to the property 1 of $\{u_j\}$. The first term tends to 0 due to (1.2.7), (1.2.8) and the properties 2, 4 of $\{u_j\}$.

Proof of Proposition 1.2.1. It is sufficient to take a function χ in the previous lemma which does not depend on y, vanishes in a neighbourhood of the perturbed edge and such that $\nabla \chi$ vanishes in a neighbourhood of all vertical edges. Then, it follows from Lemma 1.2.1 that any singular sequence of the operator A_{ε} contains a singular sequence of the operator A_{ε}^{μ} and vice versa.

The essential spectrum of the operator $\sigma_{ess}(A^{\mu}_{\varepsilon})$ having a band-gap structure, we will be interested in finding eigenvalues inside gaps (once the existence of gaps is established).

1.2.3 Method of study

Our analysis consists of three main steps.

- First, we find a formal limit of the problem when $\varepsilon \to 0$. It is clear that geometrically as ε goes to zero, the domains Ω_{ε} and $\Omega_{\varepsilon}^{\mu}$ shrink to a graph. The limit problem is then associated with a second-order differential operator defined on the graph. Neumann boundary conditions on the ladder give rise to the so called Kirchhoff's conditions at the vertices of the graph. This limit operator is well-known from the works [61] and Kuchment-Zheng [47]. We will describe it more rigorously in section 1.3.1.
- The second step is an explicit calculation of the spectrum of the limit operator. The essential spectrum is determined using the Floquet-Bloch theory (by solving a set of cell problems) while the discrete spectrum of the perturbed operator is found using a reduction to a finite difference equation (section 1.3.2.2). In particular, we will see that the limit operator has infinitely many gaps. For $\mu < 1$ it also has infinitely many eigenvalues, whereas for $\mu > 1$ it has no discrete spectrum.

• Finally, we have to show that the operator on the ladder also has eigenvalues for $\mu < 1$ when ε is small enough (section 1.4). Despite the fact that this result follows from [57], we will give another proof based on the construction of a quasi-mode (an approximation of the eigenfunction). To go further, we will compute the full asymptotic expansion of the eigenvalues inside the gaps (chapter 2).

To begin with, we will decompose the operator A^{μ}_{ε} into the sum of its symmetric and antisymmetric parts. As it will become clear later, the "true model problem" for the β -quasiperiodic case is actually given by the considering only the symmetric (or only the antisymmetric) part of the operator A^{μ}_{ε} . Let us introduce the following orthogonal decomposition of the space $L_2(\Omega^{\mu}_{\varepsilon})$:

$$L_2(\Omega^{\mu}_{\varepsilon}) = L_{2,s}(\Omega^{\mu}_{\varepsilon}) \oplus L_{2,as}(\Omega^{\mu}_{\varepsilon}),$$

where $L_{2,s}(\Omega_{\varepsilon}^{\mu})$ and $L_{2,as}(\Omega_{\varepsilon}^{\mu})$ are the subspaces consisting of functions respectively symmetric and antisymmetric with respect to the axis y = 0:

$$L_{2,s}(\Omega_{\varepsilon}^{\mu}) = \left\{ u \in L_2(\Omega_{\varepsilon}^{\mu}) / u(x,y) = u(x,-y) \right\},\$$
$$L_{2,as}(\Omega_{\varepsilon}^{\mu}) = \left\{ u \in L_2(\Omega_{\varepsilon}^{\mu}) / u(x,y) = -u(x,-y) \right\}$$

Consequently, the operator A^{μ}_{ε} is decomposed into the orthogonal sum

$$A^{\mu}_{\varepsilon} = A^{\mu}_{\varepsilon,s} \oplus A^{\mu}_{\varepsilon,as}$$

with

$$A^{\mu}_{\varepsilon,s} = \left. A^{\mu}_{\varepsilon} \right|_{L_{2,s}(\Omega^{\mu}_{\varepsilon})}, \qquad A^{\mu}_{\varepsilon,as} = \left. A^{\mu}_{\varepsilon} \right|_{L_{2,as}(\Omega^{\mu}_{\varepsilon})}$$

The spectrum of the operator A^{μ}_{ε} is then given by the union of the spectra of its symmetric and antisymmetric parts:

$$\sigma(A^{\mu}_{\varepsilon}) = \sigma(A^{\mu}_{\varepsilon,s}) \cup \sigma(A^{\mu}_{\varepsilon,as})$$

1.3 Spectral problem on the graph

1.3.1 The operator A^{μ} .

As $\varepsilon \to 0$, the domain Ω_{ε} tends to the periodic graph G represented in figure 1.4. Let us enumerate the vertices of the graph from left to right by an integer index j, the superscripts "+" and "-" corresponding to the upper and the lower vertices respectively. The set of all the vertices of the graph is then

$$V = \left\{ v_j^{\pm} \right\}_{j \in \mathbb{Z}}, \qquad v_j^{\pm} = \left(j, \pm \frac{L}{2} \right).$$

The vertical edge joining the vertices v_j^+ and v_j^- is denoted by e_j and the horizontal edge joining the vertices v_j^{\pm} and v_{j+1}^{\pm} is denoted by $e_{j+\frac{1}{2}}^{\pm}$. Thus, the set of all the edges of the graph is

$$E = \left\{ e_j, e_{j+\frac{1}{2}}^{\pm} \right\}_{j \in \mathbb{Z}}, \qquad e_j = \{j\} \times \left[-\frac{L}{2}, \frac{L}{2} \right], \qquad e_{j+\frac{1}{2}} = [j, j+1] \times \left\{ \pm \frac{L}{2} \right\}.$$

The edge corresponding to the perturbation is e_0 . The set of all the edges of the graph containing the vertex v is denoted by E_v .



Figure 1.4: Limit graph

If u is a function defined on the graph G, we will use the following notation:

$$\mathbf{u}_{j}^{\pm} = u(v_{j}^{\pm}), \qquad u_{j} = u|_{e_{j}}, \qquad u_{j+\frac{1}{2}}^{\pm} = u|_{e_{j+\frac{1}{2}}^{\pm}}$$

We introduce a local coordinate s at each horizontal edge $e_{j+\frac{1}{2}}^{\pm}$ of the graph that varies 0 at v_j^{\pm} to 1 at v_{j+1}^{\pm} . In other words,

$$u_j(y) = u(j, y), \quad y \in \left[-\frac{L}{2}, \frac{L}{2}\right], \qquad u_{j+\frac{1}{2}}^{\pm}(s) = u\left(j+s, \pm \frac{L}{2}\right), \quad s \in [0, 1]$$

Let $w^{\mu}: E \to \mathbb{R}^+$ be a weight function which is equal to μ on the "perturbed edge" e_0 and to 1 on the other edges:

$$\begin{cases} w^{\mu}(e_0) = \mu, \\ w^{\mu}(e) = 1, \quad \forall e \in E, \quad e \neq e_0. \end{cases}$$

Sometimes, to simplify the expressions, we will use the notation

$$w^{\mu}(j) = w_{j}^{\mu} = \begin{cases} \mu, & j = 0, \\ 1, & \forall j \in \mathbb{Z}^{*}. \end{cases}$$
(1.3.1)

Let us now introduce the following function spaces:

$$L_{2}^{\mu}(G) = \left\{ u \mid u \in L_{2}(e), \forall e \in E; \quad \|u\|_{L_{2}^{\mu}(G)}^{2} = \sum_{e \in E} w^{\mu}(e) \|u\|_{L_{2}(e)}^{2} < \infty \right\}, \quad (1.3.2)$$

$$H^{1}(G) = \left\{ u \mid u \in C(G); \quad u \in H^{1}(e), \forall e \in E; \quad \|u\|_{H^{1}(G)}^{2} = \sum_{e \in E} \|u\|_{H^{1}(e)}^{2} < \infty \right\},$$

$$H^{2}(G) = \left\{ u \mid u \in C(G); \quad u \in H^{2}(e), \forall e \in E; \quad \|u\|_{H^{2}(G)}^{2} = \sum_{e \in E} \|u\|_{H^{2}(e)}^{2} < \infty \right\}.$$

$$(1.3.3)$$

Notice that by definition functions belonging to $H^1(G)$ or $H^2(G)$ are continuous at the vertices of the graph.

We define the limit operator A^{μ} in $L_{2}^{\mu}(G)$ as follows:

$$(A^{\mu}u)|_{e} = -(u|_{e})'', \quad \forall e \in E,$$
 (1.3.4)

$$D(A^{\mu}) = \left\{ u \in H^2(G) \ / \ \sum_{e_v \in E_v} w^{\mu}(e_v)(u_{e_v})'_{ext}(v) = 0, \quad \forall v \in V \right\},$$
(1.3.5)

where $(u_{v_e})'_{ext}(v)$ stands for the derivative of the function u at the edge e_v taken at the vertex v in the outgoing direction. The relations in (1.3.5) are called Kirchoff's conditions. We note that the perturbation is only present in the definition of the operator A^{μ} via the Kirchoff's condition at the perturbed edge (i.e. for the vertices v_0^{\pm} since $w^{\mu}(e_0) = \mu$).

The following assertion is proved in [42], section 3.3 (we give its proof reformulated for our particular case in Appendix).

Proposition 1.3.1 (Kuchment). The operator A^{μ} in the space $L_2^{\mu}(G)$ is self-adjoint. The corresponding closed sesqui-linear form has the following form:

$$a^{\mu}[f,g] = (f',g')_{L^{\mu}_{2}(G)}, \quad \forall f,g \in D[a^{\mu}], \qquad D[a^{\mu}] = H^{1}(G).$$

As for the ladder $\Omega^{\mu}_{\varepsilon}$, we introduce the following decomposition of the space $L^{\mu}_{2}(G)$ into the spaces of symmetric and antisymmetric functions:

$$L_{2}^{\mu}(G) = L_{2,s}^{\mu}(G) \oplus L_{2,as}^{\mu}(G),$$
$$L_{2,s}^{\mu}(G) = \left\{ u \in L_{2}(G) / u(x,y) = u(x,-y) \right\},$$
$$L_{2,as}^{\mu}(G) = \left\{ u \in L_{2}(G) / u(x,y) = -u(x,-y) \right\}$$

Again, the operator A^{μ} can be decomposed into the orthogonal sum

$$A^{\mu} = A^{\mu}_s \oplus A^{\mu}_{as},$$

with

$$A_s^{\mu} = A^{\mu} \big|_{L_{2,s}^{\mu}(G)}, \qquad A_{as}^{\mu} = A^{\mu} \big|_{L_{2,as}^{\mu}(G)},$$

which implies

$$\sigma(A^{\mu}) = \sigma(A^{\mu}_s) \cup \sigma(A^{\mu}_{as}).$$

Thus, it is sufficient to study the spectra of the operators A_s^{μ} and A_{as}^{μ} separately. The analysis of these two operators being analogous, we will present a detailed study of A_s^{μ} (section 1.3.2) and state the results for A_{as}^{μ} (section 1.3.3).

1.3.2 The spectrum of the operator A_s^{μ} .

In this section we study the spectrum of the operator A_s^{μ} . Using the notation introduced in the beginning of this section with $u_{j+\frac{1}{2}} = u_{j+\frac{1}{2}}^+ = u_{j+\frac{1}{2}}^-$ and $\mathbf{u}_j = \mathbf{u}_j^+ = \mathbf{u}_j^-$, it can be rewritten as

$$(A_s^{\mu}u)|_{e_{j+\frac{1}{2}}} = -u_{j+\frac{1}{2}}'', \qquad (A_s^{\mu}u)|_{e_j} = -u_j'', \qquad j \in \mathbb{N},$$

$$D(A^{\mu}) = \{ u \in H^2(G) \ / \ u_i'(0) = 0 \qquad i \in \mathbb{N} \}$$

$$(1.3.6)$$

$$\mathcal{D}(A_{s}^{\mu}) = \left\{ u \in H^{2}(G) \, / \, u_{j}(0) = 0, \qquad j \in \mathbb{N}, \\ u_{j+\frac{1}{2}}^{\prime}(0) + w_{j}^{\mu}u_{j}^{\prime}\left(\frac{-L}{2}\right) - u_{j-\frac{1}{2}}^{\prime}(1) = 0, \qquad j \in \mathbb{N} \right\}.$$
(1.3.7)

1.3.2.1 Essential spectrum of A_s^{μ}

First of all, we reduce the study of the essential spectrum of the operator A_s^{μ} to the study of the periodic case ($\mu = 1$). The corresponding operator A_s^1 will be denoted by A_s . Indeed, similarly to Proposition 1.2.1, by introducing a cut-off function which vanishes in a neighbourhood of the perturbed edge of the graph, we can prove that the essential spectrum of A_s^{μ} coincides with the spectrum of A_s :

Proposition 1.3.2.

$$\sigma_{ess}(A_s^{\mu}) = \sigma_{ess}(A_s). \tag{1.3.8}$$

Computation of the spectrum of the operator A_s

As previously explained, the spectrum of the periodic operator A_s can determined using the Floquet-Bloch theory. One has then to study a set of problems posed on the periodicity cell of G. Since we consider the subspace of symmetric functions with respect to the axis y = 0, this permits to reduce the problem to the lower half part of the periodicity cell defined as (see figure 1.5):

$$C_{-} = G \cap \left\{ \begin{bmatrix} -\frac{1}{2}, \frac{1}{2} \end{bmatrix} \times \begin{bmatrix} -\frac{L}{2}, 0 \end{bmatrix} \right\} = \bigcup_{1 \le j \le 3} e_j,$$

$$e_1 = \begin{bmatrix} -\frac{1}{2}, 0 \end{bmatrix} \times \{ -\frac{L}{2} \}, \qquad e_2 = \begin{bmatrix} 0, \frac{1}{2} \end{bmatrix} \times \{ -\frac{L}{2} \}, \qquad e_3 = \{ 0 \} \times \begin{bmatrix} -\frac{L}{2}, 0 \end{bmatrix}$$

$$\begin{array}{c} \bullet \\ \mathsf{y} \\ -\frac{L}{2} \\ \bullet \\ -\frac{1}{2} \\ \bullet \\ \hline \mathsf{e}_1 \\ \bullet \\ \mathsf{e}_2 \\ \hline \mathsf{e}_2 \\$$

Figure 1.5: The half periodicity cell \mathcal{C}_{-}

For a function u defined on \mathcal{C}_- we use the notation $u_j = u|_{e_j}$, $1 \leq j \leq 3$. Let us introduce the spaces $L_2(\mathcal{C}_-)$ and $H^2(\mathcal{C}_-)$ analogously to (1.3.2), (1.3.3):

$$L_2(\mathcal{C}_-) = \{ u \mid u \in L_2(e_j), \ 1 \le i \le 3 \},\$$
$$H^2(\mathcal{C}_-) = \{ u \mid u \in C(\mathcal{C}_-); \ u \in H^2(e_j), \ 1 \le i \le 3 \}.$$

We have then

$$\sigma(A_s) = \bigcup_{\theta \in [0,\pi]} \sigma\left(A_s(\theta)\right), \qquad (1.3.9)$$

where $A_s(\theta)$ is the following operator defined in $L_2(\mathcal{C}_-)$ with θ -quasiperiodic boundary conditions:

$$A_{s}(\theta)u_{i} = -u_{i}'', \qquad 1 \leq i \leq 3,$$

$$D(A_{s}(\theta)) = \left\{ u \in H^{2}(\mathcal{C}_{-}) / \quad u_{3}'(0) = 0, \\ -u_{1}'(0) + u_{2}'(0) + u_{3}'\left(-\frac{L}{2}\right) = 0, \qquad (1.3.10) \\ u_{2}\left(\frac{1}{2}\right) = e^{-i\theta}u_{1}\left(-\frac{1}{2}\right), \quad u_{2}'\left(\frac{1}{2}\right) = e^{-i\theta}u_{1}\left(-\frac{1}{2}\right) \right\}. \qquad (1.3.11)$$

The condition $u'_3(0) = 0$ comes from the fact that the symmetric subspace with respect to the axis y = 0 is considered. The condition (1.3.10) is the Kirchhoff's condition and the relations (1.3.11) are the quasiperiodic conditions. For each $\theta \in [0, \pi]$ the operator $A_s(\theta)$ is self-adjoint and positive and its resolvent is compact due to the compactness of the embedding $H^1(\mathcal{C}_-) \subset L_2(\mathcal{C}_-)$. Consequently, its spectrum is a sequence of non-negative eigenvalues of finite multiplicity tending to infinity:

$$0 \leqslant \lambda_{1,s}(\theta) \leqslant \lambda_{2,s}(\theta) \leqslant \dots \leqslant \lambda_{n,s}(\theta) \leqslant \dots, \qquad \lim_{n \to \infty} \lambda_{n,s}(\theta) = +\infty.$$

In the present case, the eigenvalues can be computed explicitly.

Proposition 1.3.3. For $\theta \in [0, \pi]$, $\lambda^2 \in \sigma(A_s(\theta))$ if and only if λ is a solution of the equation

$$2\cos\left(\frac{\lambda L}{2}\right)\left(\cos\lambda - \cos\theta\right) = \sin\lambda\sin\left(\frac{\lambda L}{2}\right). \tag{1.3.12}$$

Proof. If $\lambda^2 \neq 0$ is an eigenvalue of the operator $A_s(\theta)$ then the corresponding eigenfunction $u = \{u_1, u_2, u_3\}$ is of the form

$$u_1(x) = c_1 e^{i\lambda x} + d_1 e^{-i\lambda x}, \qquad x \in \left[-\frac{1}{2}, 0\right], \qquad (1.3.13)$$

$$u_2(x) = c_2 e^{i\lambda x} + d_2 e^{-i\lambda x}, \qquad x \in [0, \frac{1}{2}],$$
 (1.3.14)

$$u_3(y) = c_3 e^{i\lambda y} + d_3 e^{-i\lambda y}, \qquad y \in \left[-\frac{L}{2}, 0\right].$$
 (1.3.15)

Taking into account that $u \in D(A_s(\theta))$ we arrive at the following linear system of equations for the coefficients $c_i, d_i, 1 \leq i \leq 3$:

$$c_1 + d_1 = c_2 + d_2 = c_3 e^{-\frac{i\lambda L}{2}} + d_3 e^{\frac{i\lambda L}{2}}, \qquad (1.3.16)$$

$$c_3 = d_3, \tag{1.3.17}$$

$$c_2 e^{\frac{i\lambda}{2}} + d_2 e^{-\frac{i\lambda}{2}} = e^{-i\theta} \left(c_1 e^{-\frac{i\lambda}{2}} + d_1 e^{\frac{i\lambda}{2}} \right), \qquad (1.3.18)$$

$$c_2 e^{\frac{i\lambda}{2}} - d_2 e^{-\frac{i\lambda}{2}} = e^{-i\theta} \left(c_1 e^{-\frac{i\lambda}{2}} - d_1 e^{\frac{i\lambda}{2}} \right), \qquad (1.3.19)$$

$$-c_1 + d_1 + c_2 - d_2 + c_3 e^{-\frac{i\lambda L}{2}} - d_3 e^{\frac{i\lambda L}{2}} = 0.$$
(1.3.20)

The relations (1.3.16) express the continuity of the eigenfunction at the vertex $(0, -\frac{L}{2})$. The equation (1.3.17) comes from the condition $u'_3(0) = 0$. The relations (1.3.18), (1.3.19) correspond to the quasiperiodicity conditions (1.3.11) and the equation (1.3.20) corresponds to the Kirchhoff's condition (1.3.10). Let us introduce the notation

$$\alpha = e^{i\theta}, \qquad z = e^{i\lambda}$$

We notice that the relations (1.3.18), (1.3.19) imply

$$c_1 = c_2 \alpha z, \qquad d_1 = d_2 \alpha \overline{z}.$$

We used the fact that $z^{-1} = \overline{z}$ because the operator $A_s(\theta)$ has only real eigenvalues. Thus, the system (1.3.16)–(1.3.20) can be rewritten as follows (we have eliminated d_3 , c_1 and d_1):

$$\begin{cases} c_2 \left(1 - \alpha z\right) + d_2 \left(1 - \alpha \overline{z}\right) = 0, \\ c_2 + d_2 - c_3 \left(z^{\frac{L}{2}} + \overline{z}^{\frac{L}{2}}\right) = 0, \\ c_2 (1 - \alpha z) - d_2 \left(1 - \alpha \overline{z}\right) + c_3 \left(\overline{z}^{\frac{L}{2}} - z^{\frac{L}{2}}\right) = 0. \end{cases}$$
(1.3.21)

The existence of an eigenfunction is equivalent to the condition

$$D(\lambda) := \begin{vmatrix} 1 - \alpha z & 1 - \alpha \overline{z} & 0 \\ 1 & 1 & -\left(z^{\frac{L}{2}} + \overline{z}^{\frac{L}{2}}\right) \\ 1 - \alpha z & -(1 - \alpha \overline{z}) & -z^{\frac{L}{2}} + \overline{z}^{\frac{L}{2}} \end{vmatrix} = 0.$$

One has

$$D(\lambda) = 2\alpha \Re \left(z^{L/2} \left(3z + \overline{z} - 4\Re \alpha \right) \right) = 4e^{i\theta} \left(2\cos\left(\frac{\lambda L}{2}\right) \left(\cos\lambda - \cos\theta \right) - \sin\left(\frac{\lambda L}{2}\right) \sin\lambda \right)$$

which implies (1.3.12) for $\lambda \neq 0$. For $\lambda = 0$ the relations (1.3.13)–(1.3.15) are replaced by

$$u_1(x) = c_1 + d_1 x, \qquad x \in \left[-\frac{1}{2}, 0\right], u_2(x) = c_2 + d_2 x, \qquad x \in \left[0, \frac{1}{2}\right], u_3(y) = c_3 + d_3 y, \qquad y \in \left[-\frac{L}{2}, 0\right].$$

Using the fact that $u \in D(A_s(\theta))$ we have:

$$c_1 = c_2 = c_3,$$
 $d_3 = 0,$ $d_2 = d_1 e^{-i\theta},$ $d_2 = c_1 (e^{-i\theta} - 1),$ $d_1 = d_2.$

Hence, there exists a non-trivial solution if and only if $\theta = 0$. The solution is a constant function $u_i = c_1, 1 \leq i \leq 3$. At the same time, $\lambda = 0$ is a solution of (1.3.12) if and only if $\theta = 0$ which finishes the proof.

Remark 1.3.1. We notice that for $\cos\left(\frac{\lambda L}{2}\right) \neq 0$ the equation (1.3.12) is equivalent to the relation

$$\cos\theta = \cos\lambda - \frac{1}{2}\sin\lambda\tan\left(\frac{\lambda L}{2}\right). \tag{1.3.22}$$

Remark 1.3.2. One can easily see that if $L \in \mathbb{Q}$, the set $\{\lambda : \lambda^2 \in \sigma(A_s)\}$ is periodic. Indeed, this is due to the fact that both the left-hand side and the right-hand side of (1.3.12) are periodic with the same period.

In the rest of this section we will use the following notation:

$$\Sigma = \{\pi n, n \in \mathbb{N}\}, \qquad \Sigma_s = \left\{\frac{2n\pi}{L}, n \in \mathbb{N}\right\} \qquad \Sigma'_s = \left\{\frac{(2n+1)\pi}{L}, n \in \mathbb{N}\right\}.$$

Some properties of the spectrum of the operator A_s

It follows from the decomposition (1.3.9) together with Proposition 1.3.3 that $\lambda^2 \in \sigma(A_s)$ if and only if λ is a solution of the equation (1.3.12) for some $\theta \in [0, \pi]$. This permits to derive some important properties of the spectrum of the operator A_s .

Proposition 1.3.4.

- 1. $\{\lambda^2, \lambda \in \Sigma \cup \Sigma_s\} \subset \sigma(A_s).$
- 2. The operator A_s has infinitely many gaps whose ends tend to infinity.

Proof.

- 1. If $\lambda \in \Sigma \cup \Sigma_s$, the equation (1.3.12) is obviously verified for θ such that $\cos \theta = \cos \lambda$.
- 2. We will actually show that for any point $\lambda \in \Sigma'_s$ there exists some neighbourhood (probably, a punctured neighbourhood) of λ^2 which is included in the resolvent set of the operator A_s . Consider $\lambda_n = \frac{(2n+1)\pi}{L}$, $n \in \mathbb{N}$. There are two possibilities:
 - (i) $\lambda_n \notin \Sigma$: since $\cos\left(\frac{\lambda_n L}{2}\right) = 0$ and $\sin \lambda_n \neq 0$, the left-hand side of the equation (1.3.12) is equal to zero, whereas the right-hand side is different from zero for any $\theta \in [0, \pi]$. Consequently, there exists a gap of the operator A_s containing the point λ_n^2 .
 - (ii) $\lambda_n \in \Sigma$: in this case it follows from the property 1 that $\lambda_n^2 \in \sigma(A_s)$. We are going to show that the point λ_n^2 is an isolated point of the spectrum of the operator A_s , so that there exist gaps to the left and to the right of it. Setting $\lambda = \lambda_n + \delta$ in (1.3.12) we get:

$$-2\sin\left(\frac{\delta L}{2}\right)\left(\cos\delta - \frac{\cos\theta}{\cos\lambda_n}\right) = \sin\delta\cos\left(\frac{\delta L}{2}\right).$$

If δ is small enough (but different from 0) this equation cannot be verified for any θ . Indeed, for $\delta \neq 0$ it can be rewritten as

$$\frac{\cos\theta}{\cos\lambda_n} - \cos\delta = \frac{\sin\delta}{2\sin\left(\frac{\delta L}{2}\right)}\cos\left(\frac{\delta L}{2}\right).$$

The limit of the right-hand side as $\delta \to 0$ is positive (it equal to $\frac{1}{L}$), whereas the limit of the left-hand side is non-positive for any θ (since $|\cos \lambda_n| = 1$) with a uniform bound in θ for δ small enough:

$$\frac{\cos\theta}{\cos\lambda_n} - \cos\delta \leqslant 1 - \cos\delta, \qquad \forall \theta \in [0, \pi].$$

Hence, the equation (1.3.12) has no solution for δ small enough. This proves the existence of gaps of the form $]\lambda_n^2 - l_n^-, \lambda_n^2[,]\lambda_n^2, \lambda_n^2 + l_n^+[$ for some $l_n^-, l_n^+ > 0$.

Remark 1.3.3. In the proof of Proposition 1.3.4 (2) we showed for any $\lambda \in \Sigma'_s$ the existence of a punctured neighbourhood of the point λ^2 that does not belong to the spectrum of the operator A_s . This can be seen as a result of the graph decoration described in [43]. Indeed, considering the the symmetric subspace on the graph G is equivalent to considering the lower half of G (denoted by G_-) with homogeneous Neumann conditions at the vertices (j, 0), cf. figure 1.6.

The graph G_{-} , in turn, can be seen as a decoration of the graph G_0 that consists of the vertices $\{v_j^-\}_{j\in\mathbb{Z}}$ and the edges $\{e_{j+\frac{1}{2}}^-\}_{j\in\mathbb{Z}}$, obtained by attaching a copy of a segment $G_1 = [-\frac{L}{2}, 0]$ to each its vertex. Then, according to Theorem 5 in [43], for any eigenvalue λ^2 of the problem

$$\begin{cases} u'' + \lambda^2 u = 0, \\ u'(0) = 0, \\ u\left(-\frac{L}{2}\right) = 0, \end{cases}$$
(1.3.23)

there exists a punctured neighbourhood that does not belong to the spectrum of the operator A_s . The set of eigenvalues of (1.6) is exactly the set $\{\lambda^2, \lambda \in \Sigma'_s\}$.



Figure 1.6: The graph G_{-} can be obtained by a decoration of the graph G_{0} : at each vertex of G_{0} we attach a copy of the segment G_{1} .

Proposition 1.3.5. The operator A_s has the following set of eigenvalues of infinite multiplicity which are isolated points of the spectrum:

$$\sigma_{pp}(A_s) = \left\{ \lambda^2, \ \lambda \in \Sigma \cup \Sigma'_s \right\}. \tag{1.3.24}$$

Proof. The point λ^2 is an eigenvalue of the operator A_s of infinity multiplicity if and only if it is an eigenvalue of the operator $A_s(\theta)$ for any $\theta \in [0, \pi]$. According to Proposition 1.3.3, this means that the equation (1.3.12) is satisfied for any $\theta \in [0, \pi]$, which is equivalent to the condition $\cos\left(\frac{\lambda L}{2}\right) = \sin \lambda = 0$ (i.e., $\lambda \in \Sigma \cup \Sigma'_s$). The fact that these points are isolated points of the spectrum is shown in the proof of Proposition 1.3.4, property 2 (case (ii)).

Remark 1.3.4. The set (1.3.24) is non-empty if and only if $L = \frac{2m+1}{k}$ an irreducible fraction with $m \in \mathbb{N}, k \in \mathbb{N}^*$. In this case

$$\sigma_{pp}(A_s) = \left\{ ((2n+1)\pi k)^2, \ n \in \mathbb{N} \right\}.$$

Remark 1.3.5. The presence of eigenvalues of infinite multiplicity is not a common feature for elliptic periodic second-order operators in domains of \mathbb{R}^n . The absolute continuity of the spectrum for such operators is proved under some additional assumptions (cf., for example, [60, 65, 40, 8, 25, 45]). As explained in [43] (Section 5), the absence of pure point spectrum is related to the uniqueness of continuation property. This property fails for the graphs. Indeed, one can easily construct a compactly supported eigenfunction on a graph provided it has a loop consisting of commensurable edges (cf. figure 1.7). This is exactly what happens under the assumptions of Proposition 1.3.5: the edges of the graph *G* are all commensurable. Due to the periodicity of the graph, an infinity of eigenfunctions corresponding to the same eigenvalue can be constructed by a translation of one compactly supported eigenfunction. Moreover, it is shown in [43] that the eigenspace corresponding to an eigenvalue of infinite multiplicity is generated by compactly supported functions.

As we have already seen in Proposition 1.3.4, the operator A_s has infinitely many gaps which are separated by the points $\{\lambda^2, \lambda \in \Sigma \cup \Sigma_s\}$. We are going now to study in more detail the location of the gaps via a geometric interpretation of the equation (1.3.12). We will see that two types of gaps can be distinguished (Proposition 1.3.8) which will permit us to characterize the discrete spectrum of the perturbed operator (according to the type of the gap it will contain one or two eigenvalues of the perturbed operator, Theorem



Figure 1.7: A compactly supported eigenfunction corresponding to the eigenvalue $\lambda^2 = (k\pi)^2$ for L = 1.

1.3.1). We start by an auxiliary statement which is an immediate consequence of the decomposition (1.3.9), Proposition 1.3.3 and Proposition 1.3.4 (1).

Proposition 1.3.6. $\lambda^2 \in \sigma(A_s)$ if and only if one of the following possibilities holds:

- (i) $\lambda \in \Sigma \cup \Sigma_s$;
- (ii) $\lambda \notin \Sigma \cup \Sigma_s$ is a solution of the equation

$$\phi_L(\lambda) = f_\theta(\lambda), \tag{1.3.25}$$

for some $\theta \in [0, \pi]$. Here

$$\phi_L(\lambda) = \frac{2}{\tan\left(\frac{\lambda L}{2}\right)}, \qquad \lambda \notin \Sigma_s, \qquad (1.3.26)$$

$$f_{\theta}(\lambda) = \frac{\sin \lambda}{\cos \lambda - \cos \theta}, \qquad \lambda \in \{\lambda / \cos \lambda \neq \cos \theta\}$$

Geometrically, the solutions of the equation (1.3.25) correspond to the abscissas of the intersections of the graph of the function $\phi_L(\lambda)$ with the one of the function $f_{\theta}(\lambda)$. Hence, to obtain the set described in (*ii*) of Proposition 1.3.6, one has to consider the union of the graphs of the functions $f_{\theta}(\lambda)$ for all $\theta \in [0, \pi]$. We introduce then the following set:

$$D = \bigcup_{\theta \in [0,\pi]} \left\{ (x, f_{\theta}(x)) / x \ge 0, \cos x \ne \cos \theta \right\}.$$

Lemma 1.3.1.

$$D = \bigcup_{n \in \mathbb{N}} D_n^{\pm},$$

where

$$D_n^+ = \left\{ (x, y) / \ x \in]\pi n, \pi (n+1)[, \ y \in [f^+(x), +\infty[\right] \cup (\pi n, 0),$$
(1.3.27)
$$D_n^- = \left\{ (x, y) / \ x \in]\pi n, \pi (n+1)[, \ y \in [f^-(x), +\infty[\right] \cup (\pi n, 0),$$
(1.3.28)

$$D_n^- = \{ (x, y) / \ x \in]\pi n, \pi (n+1)[, \ y \in]-\infty, f^-(x)] \},$$
(1.3.28)

and

$$f^{+}(x) = \tan\left(\frac{x-\pi n}{2}\right), \qquad x \in [\pi n, \pi(n+1)], f^{-}(x) = -\cot\left(\frac{x-\pi n}{2}\right), \qquad x \in [\pi n, \pi(n+1)].$$

One has

$$D_n^{\pm} = D_0^{\pm} + (\pi n, 0), \qquad \forall n \in \mathbb{N}.$$

Proof. It is sufficient to notice that

$$\bigcup_{\substack{\theta \in [0,\pi]\\ \cos x \neq \cos \theta}} f_{\theta}(x) = \begin{cases}]-\infty, f^{-}(x)] \cup [f^{+}(x), +\infty[, \quad x \notin \Sigma, \\ \{0\}, \quad x \in \Sigma. \end{cases}$$



Figure 1.8

The domains D_n^{\pm} as well as the functions f^{\pm} are shown in figure 1.8b. It is worth noticing that the functions f^{\pm} are π -periodic, the function f^+ being right-continuous and the function f^- left-continuous. The function f^+ is strictly increasing in each interval $[\pi n, \pi(n+1)[$ and $f^+([\pi n, \pi(n+1)]) = \mathbb{R}^+$. Similarly, the function f^- is strictly increasing in each interval $]\pi n, \pi(n+1)]$ and $f^-(]\pi n, \pi(n+1)]) = \mathbb{R}^-$. As regards the function ϕ_L , it is a $2\pi/L$ -periodic function defined on $\mathbb{R}^+ \setminus \Sigma_s$. In each interval $]2\pi n/L, 2\pi(n+1)/L[$, $n \in \mathbb{N}$, it is continuous and strictly decreasing and takes all the values in \mathbb{R} (see an example in figure 1.8c).

As it was mentioned above, the set described in (ii) of Proposition 1.3.6 is given by the set of the abscissas of the intersections of D with the graph of the function ϕ_L . This is not exactly the image by the function $x \mapsto \sqrt{x}$ of the spectrum of the operator A_s since the discrete set $\Sigma \cup \Sigma_s$ ((i) of Proposition 1.3.6) is missing. This permits us to consider the intersection of the graph of the function ϕ_L with the domain \overline{D} instead of D. Indeed, the domain \overline{D} differs from D only by adding the vertical boundaries:

$$\overline{D} = \bigcup_{n \in \mathbb{N}} \overline{D_n^{\pm}} = D \cup \left\{ \bigcup_{n \in \mathbb{N}} \left\{ \{\pi n\} \times \mathbb{R}^+ \right\} \right\} \cup \left\{ \bigcup_{n \in \mathbb{N}^*} \left\{ \{\pi n\} \times \mathbb{R}^- \right\} \right\}.$$

We have then

$$D_x := \left\{ x : (x, \phi_L(\lambda)) \in \overline{D} \right\} = \left\{ x : (x, \phi_L(\lambda)) \in D \right\} \cup \Sigma \setminus \Sigma_s, \tag{1.3.29}$$

which means that $\lambda \in D_x$ implies $\lambda^2 \in \sigma(A_s)$. In order to get the whole spectrum of the operator A_s , it is actually enough to add the set Σ_s , which is equivalent to taking the closure of D_x . More precisely, the following assertion holds.

Proposition 1.3.7.

$$\lambda^2 \in \sigma(A_s) \quad \Leftrightarrow \quad \lambda \in D_x \cup \Sigma_s \quad \Leftrightarrow \quad \lambda \in \overline{D_x}.$$

Proof. The first equivalence follows immediately from Proposition 1.3.6 and (1.3.29). The inclusion $\{\lambda^2 : \lambda \in \overline{D_x}\} \subset \sigma(A_s)$ is also obvious since $\{\lambda^2 : \lambda \in D_x\} \subset \sigma(A_s)$ and the spectrum is a closed set. It remains then to prove that $\Sigma_s \subset \overline{D_x}$. Suppose that $\lambda_0 \in \Sigma_s$. Then,

$$\exists \delta_0 > 0 \quad \text{s.t.} \quad (\lambda, \phi_L(\lambda)) \in D, \quad \forall \lambda \in]\lambda_0, \lambda_0 + \delta_0[. \tag{1.3.30}$$

Indeed, for δ small enough the function f^+ is continuous in the interval $[\lambda_0, \lambda_0 + \delta]$. Consequently, for A large enough the band $]\lambda_0, \lambda_0 + \delta[\times[A, +\infty[$ is contained in D (due to (1.3.27)). However, $\lim_{\lambda \to \lambda_0 + 0} \phi_L(\lambda) = +\infty$, which implies 1.3.30. This, in turn, means that $]\lambda_0, \lambda_0 + \delta_0[\subset D_x$ for some $\delta_0 > 0$. Hence, $\lambda_0 \in \overline{D_x}$.



Figure 1.9: Examples of intersections of the domain \overline{D} with the function ϕ_L

In the rest of this section we prove the following characterization of the gaps of the operator A_s .

Proposition 1.3.8. An interval]a, b[is a gap of the operator A_s if and only if $[\sqrt{a}, \sqrt{b}] \cap \Sigma_s = \emptyset$ and one of the following possibilities holds:

- **I** There exists $n \in \mathbb{N}$ such that $\pi n < \sqrt{a} < \sqrt{b} < \pi(n+1)$, and $\phi_L(\sqrt{a}) = f^+(\sqrt{a})$, $\phi_L(\sqrt{b}) = f^-(\sqrt{b})$;
- **II** (i) There exists $n \in \mathbb{N}$ such that $\pi n = \sqrt{a} < \sqrt{b} < \pi(n+1)$, and $\phi_L(\sqrt{a}) \leq 0$, $\phi_L(\sqrt{b}) = f^-(\sqrt{b});$
 - (ii) There exists $n \in \mathbb{N}$ such that $\pi n < \sqrt{a} < \sqrt{b} = \pi(n+1)$, and $\phi_L(\sqrt{a}) = f^+(\sqrt{a})$, $\phi_L(\sqrt{b}) \ge 0$.

We start by proving the following characterization of the ends of the gaps.

Lemma 1.3.2. The point λ_0^2 is the lower end of a gap of the operator A_s if and only if one of the following possibilities holds:

- (i) $\lambda_0 \in \mathbb{R}_+ \setminus (\Sigma \cup \Sigma_s)$ and $\phi_L(\lambda_0) = f^+(\lambda_0)$;
- (*ii*) $\lambda_0 \in \Sigma \setminus \Sigma_s$ and $\phi_L(\lambda_0) \leq 0$.

Similarly, the point λ_0^2 is the upper end of a gap of the operator A_s if and only if one of the following possibilities holds:

(iii) $\lambda_0 \in \mathbb{R}_+ \setminus (\Sigma \cup \Sigma_s)$ and $\phi_L(\lambda_0) = f^-(\lambda_0)$;

(iv)
$$\lambda_0 \in \Sigma \setminus \Sigma_s \text{ and } \phi_L(\lambda_0) \ge 0.$$

Proof. Let us study the possible configurations of intersections of the function ϕ_L with the domains $\overline{D_n^{\pm}}$ in the interval $[\pi n, \pi(n+1)]$. We will treat separately the "regular" points (which are neither the end of the interval nor the points of discontinuity of the function ϕ_L).

- Case 1. Regular points: $\lambda_0 \notin \Sigma \cup \Sigma_s$
- (a) The point $(\lambda_0, \phi_L(\lambda_0))$ is an interior point of the domains \overline{D}_n^{\pm} :

$$(\lambda_0, \phi_L(\lambda_0)) \in \operatorname{int}\left(\overline{D_n^{\pm}}\right) \ \Rightarrow \ \exists \delta > 0 \ \text{s.t.} \ (\lambda, \phi_L(\lambda)) \in \overline{D_n^{\pm}}, \ \forall \lambda \in]\lambda_0 - \delta, \lambda_0 + \delta[. \ (1.3.31)$$

This follows immediately from the fact that the function ϕ_L is continuous at λ_0 . We note that due to (1.3.31) the points such that $(\lambda_0, \phi_L(\lambda_0)) \in \operatorname{int}(\overline{D_n^{\pm}})$ correspond to interior points of the spectrum (and not to the ends of gaps).

(b) The point λ_0 satisfies $\phi_L(\lambda_0) = f^+(\lambda_0)$:

$$\begin{aligned} \lambda_0 \in]\pi n, \pi(n+1)[\\ \phi_L(\lambda_0) = f^+(\lambda_0) \end{aligned} \Rightarrow \exists \delta > 0 \quad \text{s.t.} \quad \begin{aligned} & (\lambda, \phi_L(\lambda)) \in \overline{D_n^+}, \quad \forall \lambda \in]\lambda_0 - \delta, \lambda_0], \\ & (\lambda, \phi_L(\lambda)) \notin \overline{D_n^+}, \quad \forall \lambda \in]\lambda_0, \lambda_0 + \delta[. \end{aligned}$$
 (1.3.32)

(c) The point λ_0 satisfies $\phi_L(\lambda_0) = f^-(\lambda_0)$:

$$\begin{aligned} \lambda_0 \in]\pi n, \pi(n+1)[\\ \phi_L(\lambda_0) = f^-(\lambda_0) \end{aligned} \Rightarrow \quad \exists \delta > 0 \quad \text{s.t.} \quad \begin{aligned} & (\lambda, \phi_L(\lambda)) \in \overline{D_n^-}, \quad \forall \lambda \in [\lambda_0, \lambda_0 + \delta[, \\ & (\lambda, \phi_L(\lambda)) \notin \overline{D_n^-}, \quad \forall \lambda \in]\lambda_0 - \delta, \lambda_0[. \end{aligned}$$
 (1.3.33)

This follows from the monotonicity of the functions f^{\pm} defining the boundaries of the domains $\overline{D_n^{\pm}}$ on the one hand and from the monotonicity of the function ϕ_L from the other hand. Indeed, the point $(\lambda_0, f^+(\lambda_0))$ belongs to the boundary of the domain $\overline{D_n^+}$. Hence, the whole band $[\pi n, \lambda_0] \times [f^+(\lambda_0), \infty[$ belongs to $\overline{D_n^+}$ and the band $[\lambda_0, \pi(n+1)[\times] -\infty, f^+(\lambda_0)[$ does not intersect $\overline{D_n^+}$. Taking into account that the function ϕ_L is strictly decreasing and continuous in some neighbourhood of the point λ_0 , we obtain (1.3.32) which means that λ_0^2 is the lower end of a gap. An analogous argument gives (1.3.33) which means that λ_0^2 is the upper end of a gap.

• Case 2. Points of discontinuity of the function ϕ_L (the set Σ_s)

This case has been considered in the proof of Proposition 1.3.7. We showed that there exists $\delta^+ > 0$ such that $]\lambda_0, \lambda_0 + \delta^+[\subset D_x]$. Using an analogous argument one easily checks that there exists $\delta^- > 0$ such that $]\lambda_0 - \delta^-, \lambda_0[\subset D_x]$. Taking into account that $D_x \cup \Sigma_s = \{\lambda : \lambda^2 \in \sigma(A_s)\}$, we conclude that the points of Σ_s correspond to internal points of the spectrum (and not to the ends of gaps).

• Case 3. Ends of the interval

$$\begin{aligned} \lambda_0 &= \pi n \notin \Sigma_s \\ \phi_L(\lambda_0) &\ge 0 \end{aligned} \Rightarrow \begin{aligned} & (\lambda_0, \phi_L(\lambda_0)) \in D_n^+, \\ & \exists \delta > 0 \quad \text{s.t.} \quad (\lambda, \phi_L(\lambda)) \notin \overline{D_n^+}, \quad \forall \lambda \in]\lambda_0 - \delta, \lambda_0[. \end{aligned} \tag{1.3.34}$$

$$\begin{array}{ll}
\lambda_0 = \pi n \notin \Sigma_s \\
\phi_L(\lambda_0) \leqslant 0 \\
\end{array} \Rightarrow \quad \begin{array}{ll}
(\lambda_0, \phi_L(\lambda_0)) \in \overline{D_n^-}, \\
\exists \delta > 0 \quad \text{s.t.} \quad (\lambda, \phi_L(\lambda)) \notin \overline{D_n^-}, \\
\forall \lambda \in]\lambda_0, \lambda_0 + \delta[.
\end{array} \tag{1.3.35}$$

These properties follow from the fact that the function ϕ_L is continuous and strictly decreasing in some neighbourhood of the point λ_0 taking into account that the rays $\{\pi n\} \times \mathbb{R}^+$ and $\{\pi n\} \times \mathbb{R}^-$ are boundaries of the domains $\overline{D_n^+}$ and $\overline{D_n^-}$ respectively.

The properties (1.3.34), (1.3.35) imply that the point $\lambda_0^2 = (\pi n)^2$ is the upper end of a gap if $\phi_L(\lambda_0) \ge 0$ and the lower end of a gap if $\phi_L(\lambda_0) \le 0$. Note that if $\phi_L(\lambda_0) = 0$, there is a gap to the left and to the right of this point, and $\lambda_0^2 = (\pi n)^2$ is an eigenvalue of infinite multiplicity for the operator A_s (which is in accordance with Proposition 1.3.5).

The three cases considered cover all the possible situations, which finishes the proof. \Box

Examples illustrating Lemma 1.3.2 are shown in figure 1.9. We can now prove Proposition 1.3.8.

Proof of Proposition 1.3.8. As it has been shown in the proof of Lemma 1.3.2, case 2, the points of Σ_s correspond to internal points of the spectrum of the operator A_s . This implies that $[\sqrt{a}, \sqrt{b}] \cap \Sigma_s = \emptyset$. Consequently, $\phi_L \in C([\sqrt{a}, \sqrt{b}])$. Since the images of the points of Σ by the function $x \mapsto x^2$ belong to the spectrum of the operator A_s (Proposition 1.3.4(1)), one necessarily has $\pi n \leq \sqrt{a} < \sqrt{b} \leq \pi(n+1)$ for some $n \in \mathbb{N}$.

I Suppose that $\pi n < \sqrt{a} < \sqrt{b} < \pi(n+1)$. It follows from Lemma 1.3.2 that $\phi_L(\sqrt{a}) = f^+(\sqrt{a})$ and $\phi_L(\sqrt{b}) = f^-(\sqrt{b})$.

Π

- (i) Suppose now that $\sqrt{a} = \pi n$. Then, it follows from Lemma 1.3.2 that $\phi_L(\sqrt{a}) \leq 0$. The function ϕ_L being continuous and strictly decreasing in the interval $[\sqrt{a}, \sqrt{b}]$, we conclude that $\phi_L(\sqrt{b}) < 0$. Then, according to Lemma 1.3.2, $\phi_L(\sqrt{b}) = f^-(\sqrt{b})$. Consequently, $f^-(\sqrt{b}) < 0$, which implies that $\sqrt{b} < \pi(n+1)$ (since $f^-(\pi(n+1)) = 0$).
- (ii) The case $\sqrt{b} = \pi(n+1)$ can be considered analogously to the previous case.



Figure 1.10: Particular cases

1.3.2.2 Discrete spectrum of the operator A_s^{μ} .

We now pass to the study of the discrete spectrum of the operator A_s^{μ} . Suppose that $\lambda^2 \neq 0$ is an eigenvalue of the operator A_s^{μ} . Then, an eigenfunction $u \in D(A_s^{\mu})$ corresponding to λ^2 verifies the equation $u'' + \lambda^2 u = 0$ in each edge of the graph G. Hence, it has the form

$$u_{j+\frac{1}{2}}(s) = a_{j+\frac{1}{2}}\sin(\lambda s) + b_{j+\frac{1}{2}}\cos(\lambda s), \qquad s \in [0,1], \qquad \forall j \in \mathbb{Z}, \qquad (1.3.36)$$
$$u_{j}(y) = c_{j}\cos(\lambda y), \qquad \qquad y \in \left[-\frac{L}{2}, \frac{L}{2}\right], \qquad \forall j \in \mathbb{Z}. \qquad (1.3.37)$$

The continuity of the eigenfunction u at the vertices of the graph (which is due to the fact that $u \in D(A_s^{\mu})$) implies that

$$b_{j+\frac{1}{2}} = a_{j-\frac{1}{2}} \sin \lambda + b_{j-\frac{1}{2}} \cos \lambda = c_j \cos \left(\frac{\lambda L}{2}\right), \quad \forall j \in \mathbb{Z}.$$
 (1.3.38)

The Kirchhoff's conditions (1.3.7) for the function u give

$$a_{j+\frac{1}{2}} - a_{j-\frac{1}{2}} \cos \lambda + b_{j-\frac{1}{2}} \sin \lambda + w_j^{\mu} c_j \sin \left(\frac{\lambda L}{2}\right) = 0, \quad \forall j \in \mathbb{Z}.$$
 (1.3.39)

Lemma 1.3.3. If $\lambda \in \Sigma \Delta \Sigma'_s$, then λ^2 is not an eigenvalue of the operator A^{μ}_s for any $\mu > 0$.

Proof.

- (a) Suppose that $\lambda \in \Sigma'_s \setminus \Sigma$. Then, as $\cos\left(\frac{\lambda L}{2}\right) = 0$, the relations (1.3.38) imply that $b_{j+\frac{1}{2}} = 0, \forall j \in \mathbb{Z}$, and $a_{j-\frac{1}{2}} = 0, \forall j \in \mathbb{Z}$, since $\sin \lambda \neq 0$. Finally, taking into account that $\sin\left(\frac{\lambda L}{2}\right) \neq 0$, one finds from the relations (1.3.39) that $c_j = 0, \forall j \in \mathbb{Z}$. Thus, the eigenfunction u is identically zero.
- (b) Suppose that $\lambda \in \Sigma \setminus \Sigma'_s$, $\lambda \neq 0$. Then, since $\sin \lambda = 0$ and $\cos\left(\frac{\lambda L}{2}\right) \neq 0$, the relations (1.3.38) imply that $|b_{j+\frac{1}{2}}| = b$, $\forall j \in \mathbb{Z}$, and $|c_j| = b \left|\cos\left(\frac{\lambda L}{2}\right)\right|^{-1}$, $\forall j \in \mathbb{Z}$. Consequently, if $b \neq 0$, then $u \notin L_2(G)$. If b = 0, then $b_{j+\frac{1}{2}} = 0$ and $c_j = 0$, $\forall j \in \mathbb{Z}$. In this case, one finds from the relations (1.3.39) that $|a_{j+\frac{1}{2}}| = a$, $\forall j \in \mathbb{Z}$. Again, if $a \neq 0$, then $u \notin L_2(G)$, and if a = 0, then the eigenfunction u is identically zero.
(c) Finally, consider the point $\lambda = 0$. Then, the relations (1.3.36)–(1.3.39) are replaced by

$$\begin{split} u_{j+\frac{1}{2}}(s) &= a_{j+\frac{1}{2}}s + b_{j+\frac{1}{2}}, \quad s \in [0,1], \qquad u_j(y) = c_j, \quad y \in \left[-\frac{L}{2}, \frac{L}{2}\right], \qquad \forall j \in \mathbb{Z}, \\ b_{j+\frac{1}{2}} &= a_{j-\frac{1}{2}} + b_{j-\frac{1}{2}} = c_j, \qquad a_{j+\frac{1}{2}} - a_{j-\frac{1}{2}} = 0, \qquad \forall j \in \mathbb{Z}. \end{split}$$

Thus, we have $a_{j+\frac{1}{2}} = a$, $\forall j \in \mathbb{Z}$, and $c_j - c_{j-1} = a$, $\forall j \in \mathbb{Z}$. If $a \neq 0$, then $u \notin L_2(G)$. If a = 0, then $c_j = c_{j-1} = b_{j+\frac{1}{2}} = b_{j-\frac{1}{2}} = c$, $\forall j \in \mathbb{Z}$. Again, either $u \notin L_2(G)$ (if $c \neq 0$), or u = 0.

Remark 1.3.6. Lemma 1.3.3 together with Proposition 1.3.5 imply that the set $\Sigma \cup \Sigma'_s$ can be excluded from the consideration while searching the eigenvalues of the operator A_s^{μ} . Indeed, the points $\lambda \in \Sigma \Delta \Sigma'_s$ do not correspond to eigenvalues whereas the set $\Sigma \cap \Sigma'_s$ corresponds to the eigenvalues of infinite multiplicity.

For $\lambda \notin \Sigma \cup \Sigma'_s$ the coefficients $\{a_{j+\frac{1}{2}}, b_{j+\frac{1}{2}}, c_j\}$ can be expressed in terms of the values $\{\mathbf{u}_j\}$ of the function u at the vertices of the graph:

$$a_{j+\frac{1}{2}} = \frac{1}{\sin\lambda} \left(\mathbf{u}_{j+1} - \cos\lambda\mathbf{u}_j \right), \quad b_{j+\frac{1}{2}} = \mathbf{u}_j, \quad c_j = \frac{\mathbf{u}_j}{\cos\left(\frac{\lambda L}{2}\right)}, \qquad \forall j \in \mathbb{Z}.$$
(1.3.40)

Then, the relations (1.3.36)–(1.3.37) take the form

$$u_{j+\frac{1}{2}}(s) = \mathbf{u}_j \frac{\sin\left(\lambda(1-s)\right)}{\sin\lambda} + \mathbf{u}_{j+1} \frac{\sin\left(\lambda s\right)}{\sin\lambda}, \qquad s \in [0,1], \qquad \forall j \in \mathbb{Z}, \qquad (1.3.41)$$

$$u_j(y) = \mathbf{u}_j \frac{\cos\left(\lambda y\right)}{\cos\left(\frac{\lambda L}{2}\right)}, \qquad \qquad y \in \left[-\frac{L}{2}, \frac{L}{2}\right], \qquad \forall j \in \mathbb{Z}.$$
(1.3.42)

After injecting (1.3.40) into the relation (1.3.39) one obtains the following finite difference equation:

$$\mathbf{u}_{j+1} + 2g(\lambda)\mathbf{u}_j + \mathbf{u}_{j-1} = 0, \qquad j \in \mathbb{Z}^*,$$
 (1.3.43)

$$\mathbf{u}_1 + 2g^{\mu}(\lambda)\mathbf{u}_0 + \mathbf{u}_{-1} = 0, \qquad (1.3.44)$$

where

$$g(\lambda) = -\cos\lambda + \frac{1}{2}\sin\lambda\tan\left(\frac{\lambda L}{2}\right), \qquad (1.3.45)$$

$$g^{\mu}(\lambda) = -\cos\lambda + \frac{\mu}{2}\sin\lambda\tan\left(\frac{\lambda L}{2}\right).$$
(1.3.46)

Thus, the initial problem for a differential operator on the graph reduces to a problem for a finite difference operator acting on sequences $\{\mathbf{u}_j\}_{j\in\mathbb{Z}}$. The characteristic equation associated to the system (1.3.43)-(1.3.44) is

$$r^2 + 2g(\lambda)r + 1 = 0. \tag{1.3.47}$$

We will denote by $r(\lambda)$ the solution of (1.3.47) given by the relation

$$r(\lambda) = -g(\lambda) + sign(g(\lambda))\sqrt{g^2(\lambda) - 1}.$$
(1.3.48)

Clearly, $|r(\lambda)| \leq 1$. Let us first solve the system (1.3.43) for j < 0 and j > 0. If $r(\lambda) \neq r(\lambda)^{-1}$ the general solution of (1.3.43) is

$$\mathbf{u}_j = A_+ r(\lambda)^j + B_+ r(\lambda)^{-j}, \qquad j \ge 0, \tag{1.3.49}$$

$$\mathbf{u}_j = A_- r(\lambda)^j + B_- r(\lambda)^{-j}, \qquad j \leqslant 0.$$
(1.3.50)

In the particular case when $r(\lambda) = r(\lambda)^{-1} = \pm 1$ the general solution is

$$\mathbf{u}_j = C_+ r(\lambda)^j + D_+ r(\lambda)^j j, \qquad j \ge 0, \tag{1.3.51}$$

$$\mathbf{u}_j = C_- r(\lambda)^j + D_- r(\lambda)^j j, \qquad j \leqslant 0.$$
(1.3.52)

Since an eigenfunction has to belong to $L_2(G)$, the value λ^2 can be an eigenvalue of the operator A_s^{μ} only if $|r(\lambda)| < 1$, which is equivalent to $|g(\lambda)| > 1$. If λ^2 is an eigenvalue of the operator A_s^{μ} then $B_+ = A_- = 0$. Moreover, the equations (1.3.49)–(1.3.50) for j = 0 imply that $A_+ = B_- = A$. Thus,

$$\mathbf{u}_j = Ar(\lambda)^{|j|}, \qquad j \in \mathbb{Z}.$$
(1.3.53)

At this point we can remark that all the eigenvalues of the operator A_s^{μ} (if they exist) are simple. Finally, after injecting (1.3.53) into (1.3.44) we find

$$\lambda^2 \in \sigma_d(A_s^\mu) \quad \Leftrightarrow \quad r(\lambda) = -g^\mu(\lambda).$$
 (1.3.54)

Taking into account (1.3.45), (1.3.46) we arrive at the following relation:

$$sign(g(\lambda))\sqrt{g^2(\lambda)-1} = (1-\mu)(g(\lambda)+\cos\lambda).$$

Since $|g(\lambda)| > 1$, the above relation can be rewritten as

$$\mu = F(\lambda), \tag{1.3.55}$$

$$F(\lambda) = 1 - \frac{\sqrt{g^2(\lambda) - 1}}{|g(\lambda) + \cos \lambda|}.$$
(1.3.56)

We arrive then at the following assertion.

Proposition 1.3.9.

$$\lambda^2 \in \sigma_d(A_s^{\mu}) \iff \lambda \text{ is a solution of } (1.3.55).$$

We will now study the existence and the position of the eigenvalues of the operator A_s^{μ} . As it was mentioned before, if λ^2 is an eigenvalue of the operator A_s^{μ} of finite multiplicity, then one necessarily has $|g(\lambda)| > 1$. The following proposition establishes the relation between the absolute value of $g(\lambda)$ and the nature of the point λ^2 .

Proposition 1.3.10. For $\lambda \notin \Sigma'_s$,

$$|g(\lambda)| \leq 1 \quad \Leftrightarrow \quad |r(\lambda)| = 1 \quad \Leftrightarrow \quad \lambda^2 \in \sigma(A_s).$$
 (1.3.57)

Proof. The first equivalence follows immediately from (1.3.48). Next, we remark that $|g(\lambda)| \leq 1$ if and only if there exists $\theta \in [0, \pi]$ such that $g(\lambda) = -\cos \theta$. Due to (1.3.45) we get

$$\cos \lambda - \cos \theta = \frac{1}{2} \sin \lambda \tan \left(\frac{\lambda L}{2}\right).$$

For $\lambda \notin \Sigma \cup \Sigma_s \cup \Sigma'_s$ it is equivalent to (1.3.25). Finally, for $\lambda \in (\Sigma \cup \Sigma_s) \setminus \Sigma'_s$ we have $|g(\lambda)| = 1$ and $\lambda^2 \in \sigma(A_s)$ (cf. Proposition 1.3.6 (i)).

Remark 1.3.7. The relation $|r(\lambda)| = 1$ can be seen as the condition of existence of a generalized eigenfunction of the operator A_s^{μ} for the value λ^2 . By a generalized eigenfunction we mean a solution of the problem $u'' + \lambda^2 u = 0$ in each edge of the graph G which is continuous and verifies the Kirchhoff's conditions and which has at most polynomial growth but does not belong to $L_2(G)$. Indeed, for $\lambda \notin \Sigma \cup \Sigma'_s$ the relations (1.3.49)-(1.3.50) or (1.3.51)-(1.3.52) completed by (1.3.44) give solutions of at most polynomial growth which do not belong to $L_2(G)$ if and only if $|r(\lambda)| = 1$. For $\lambda \in \Sigma \setminus \Sigma'_s$ one has $r(\lambda) = -g(\lambda) = \pm 1$. The existence of a generalized eigenfunction in this case has been established in the proof of Lemma 1.3.3 (b). Thus, Proposition 1.3.10 implies that in our case the existence of a generalized eigenfunction for λ^2 , $\lambda \notin \Sigma'_s$, is equivalent to $\lambda^2 \in \sigma_{ess}(A_s^{\mu})$. The relation between the existence of a generalized eigenfunction for λ and the fact that λ^2 belongs to the essential spectrum of the operator is known as a Schnol's theorem-type result.

It follows from Proposition 1.3.10 that if λ^2 is an eigenvalue of the operator A_s^{μ} of finite multiplicity, it is necessarily in a gap of the operator A_s^{μ} . Thus, the operator A_s^{μ} has no embedded eigenvalues. With the classification of the gaps in two types introduced in Proposition 1.3.8 we can state the following theorem giving the number of eigenvalues inside the gaps of the operator A_s^{μ} .

Theorem 1.3.1. The operator A_s^{μ} has no embedded eigenvalues for any $\mu > 0$. For $0 < \mu < 1$ there exist two simple eigenvalues of the operator A_s^{μ} in each gap of type I and one simple eigenvalue of this operator in each gap of type II. For $\mu \ge 1$ the operator A_s^{μ} has no eigenvalues.

Proof. The absence of embedded eigenvalues has been discussed above. The eigenvalues of the operator A_s^{μ} are characterized by Proposition 1.3.9. Clearly, for $\mu \ge 1$ (1.3.55) has no solutions. Let us consider the case $0 < \mu < 1$. Let]a, b[be a gap of the operator A_s . We will study the behaviour of the function F given by (1.3.56) inside this gap. Using (1.3.45) it can be rewritten as

$$F(\lambda) = 1 - \sqrt{1 - \phi_L(\lambda) \left(\phi_L(\lambda) - \psi(\lambda)\right)}, \qquad (1.3.58)$$

where $\psi(\lambda) = -2/\tan \lambda$ and the function ϕ_L is defined in (1.3.25). The only zeros of the function F in the interval $]\sqrt{a}, \sqrt{b}[$ are given by the zeros of ϕ_L and the zeros of $\varphi = \phi_L - \psi$. Let us investigate the variations of F in the two cases given by Proposition 1.3.8:

I The function g being continuous in $\mathbb{R} \setminus \Sigma'_s$, it follows from (1.3.57) that $|g(\sqrt{a})| = |g(\sqrt{b})| = 1$, and hence, $F(\sqrt{a}) = F(\sqrt{b}) = 1$. In view of Proposition 1.3.8, the strictly decaying function ϕ_L has exactly one zero in $]\sqrt{a}, \sqrt{b}[$ since $\phi_L(\sqrt{a}) = f^+(\sqrt{a}) > 0$, $\phi_L(\sqrt{b}) = f^-(\sqrt{b}) < 0$. We denote this zero by c. Besides, φ is strictly decreasing and one can show that $\varphi(\sqrt{a}) > 0$ and $\varphi(\sqrt{b}) < 0$. Indeed, one easily verifies that $f^-(\lambda) < \psi(\lambda) < f^+(\lambda), \ \forall \lambda \in [\sqrt{a}, \sqrt{b}]$. It follows that φ has a unique zero (denoted by d) in $]\sqrt{a}, \sqrt{b}[$ (see figure 1.11a). As a consequence, the function $F(\lambda)$ is strictly decreasing from 1 to 0 in the interval $[\sqrt{a}, \min(c, d)]$, strictly increasing from 0 to 1 in the interval $[\max(c, d), \sqrt{b}]$ and negative in the interval $]\min(c, d), \max(c, d)[$. Thus, there exists precisely two solutions of (1.3.55) in the interval $]\sqrt{a}, \sqrt{b}[$ for $0 < \mu < 1$.

II Consider, for example, the case (ii), i.e. $\sqrt{b} = \pi(n+1)$ (the case (i) can be considered analogously). As in the case I, one has $F(\sqrt{a}) = 1$. The function ϕ_L does not change the sign in $]\sqrt{a}, \sqrt{b}[$ since $\phi_L(\sqrt{a}) > 0$, $\phi_L(\sqrt{b}) \ge 0$. Hence, the point c from the previous case does not exist. The point d still exists and (see figure 1.11b). In the interval $[\sqrt{a}, d]$ the function $F(\lambda)$ decreases monotonously from 0 to 1 whereas in the interval $[d, \sqrt{b}[$ it is negative. Hence, there exists a unique solution of (1.3.55) in the interval $]\sqrt{a}, \sqrt{b}[$ for $0 < \mu < 1$.



which leads to two eigenvalues in such a gap

(b) Case II: the point d exists but the point c does not, which gives one eigenvalue in this gap

Figure 1.11: Illustration for two types of gaps

1.3.3 Results for the operator A^{μ}_{as}

We will now briefly describe the modifications of the previous considerations in the case of the operator A_{as}^{μ} . The operator corresponding to the periodic case $\mu = 1$ is denoted by A_{as} . The analogue of Proposition 1.3.3 is:

Proposition 1.3.11. For $\theta \in [0, \pi]$, $\lambda^2 \in \sigma(A_{as}(\theta))$ if and only if λ is a solution of the equation

$$2\sin\left(\frac{\lambda L}{2}\right)(\cos\lambda - \cos\theta) = -\sin\lambda\cos\left(\frac{\lambda L}{2}\right), \qquad \lambda \neq 0.$$
(1.3.59)

For $\sin\left(\frac{\lambda L}{2}\right) \neq 0$ the equation (1.3.59) is equivalent to

$$\cos \theta = \cos \lambda - \frac{1}{2} \sin \lambda \tan \left(\frac{\lambda L}{2} + \frac{\pi}{2}\right).$$

The analogues of the properties given in Propositions 1.3.4–1.3.6 hold.

Proposition 1.3.12.

1. $\{\lambda^2, \lambda \in \Sigma \cup \Sigma'_s \setminus \{0\}\} \subset \sigma(A_{as}).$

- 2. The operator A_{as} has infinitely many gaps whose ends tend to infinity.
- 3. The operator A_{as} has the following set of eigenvalues of infinite multiplicity which are isolated points of the spectrum:

$$\sigma_{pp}(A_{as}) = \left\{ \lambda^2, \ \lambda \in \Sigma \cup \Sigma_s \setminus \{0\} \right\}.$$

This set is non-empty if and only if $L = \frac{2m}{k}$ an irreducible fraction with $m \in \mathbb{N}^*$, $k \in \mathbb{N}^*$. In this case, it has the form

$$\sigma_{pp}(A_{as}) = \left\{ \left(n\pi k \right)^2, \ n \in \mathbb{N}^* \right\}.$$

Otherwise $\sigma_{pp}(A_{as}) = \emptyset$.

- 4. $\lambda^2 \in \sigma(A_{as})$ if and only if either one of the following possibilities holds:
 - (i) $\lambda \in \Sigma \cup \Sigma'_{s} \setminus \{0\};$
 - (ii) λ is a solution of the equation

$$-2\tan\left(\frac{\lambda L}{2}\right) = f_{\theta}(\lambda),$$

for some $\theta \in [0, \pi]$.

Next, Proposition 1.3.2 still holds for the operator A_{as}^{μ} :

Proposition 1.3.13.

$$\sigma_{ess}(A^{\mu}_{as}) = \sigma_{ess}(A_{as}).$$

Passing to finite-difference equations for $\lambda \notin \Sigma \cup \Sigma_s$ we find again (1.3.43), (1.3.44) with $g(\lambda), g^{\mu}(\lambda)$ replaced by $g_{as}(\lambda), g^{\mu}_{as}(\lambda)$:

$$g_{as}(\lambda) = -\cos\lambda + \frac{1}{2}\sin\lambda\tan\left(\frac{\lambda L}{2} + \frac{\pi}{2}\right),$$

$$g_{as}^{\mu}(\lambda) = -\cos\lambda + \frac{\mu}{2}\sin\lambda\tan\left(\frac{\lambda L}{2} + \frac{\pi}{2}\right),$$

The characteristic equation (1.3.47) as well as the characterisation (1.3.55)-(1.3.56) of the eigenvalues are still valid with $g(\lambda)$ replaced by $g_{as}(\lambda)$. The relation $|g_{as}(\lambda)| \leq 1 \iff \lambda^2 \in \sigma(A_s)$ for $\lambda \notin \Sigma_s$ analogous to Proposition 1.3.10 also holds. Thus, we can state the analogue of Theorem 1.3.1.

Theorem 1.3.2. The operator A_{as}^{μ} has no embedded eigenvalues for any $\mu > 0$. For $0 < \mu < 1$ there exists either one or two simple eigenvalues of the operator A_{as}^{μ} in each gap of this operator. These eigenvalues are characterised as follows:

$$\lambda^2 \in \sigma_d(A_{as}^{\mu}) \quad \Leftrightarrow \quad \mu = 1 - \frac{\sqrt{g_{as}^2(\lambda) - 1}}{|g_{as}(\lambda) + \cos\lambda|}$$

For $\mu \ge 1$ the operator A^{μ}_{as} has no eigenvalues.

Remark 1.3.8. As for the operator A_s , we could give a more precise version of this theorem by distinguishing two types of gaps of the operator A_{as} analogously to Proposition 1.3.8. For $0 < \mu < 1$ the operator A_{as} has two eigenvalues in the gaps of one type and one eigenvalue in the gaps of the other type.

1.3.4 The spectrum of the operator A.

As we have seen, both of the operators A_s and A_{as} have infinitely many gaps. However, it turns out that the gaps of one operator overlap with spectral bands of the other one, so that the full operator A has no gap.

Proposition 1.3.14.

$$\sigma(A) = \mathbb{R}^+.$$

Proof. Let us suppose that there exists λ such that $\lambda^2 \notin \sigma(A)$ (of course, the same is true for some open neighbourhood of λ). Due to the characterisation of the essential spectrum (1.3.57) and its analogue for A_{as} we have

$$\begin{cases} \left| -\cos\lambda + \frac{1}{2}\sin\lambda\tan\left(\frac{\lambda L}{2}\right) \right| > 1, \\ \left| \cos\lambda + \frac{\sin\lambda}{2\tan\left(\frac{\lambda L}{2}\right)} \right| > 1. \end{cases}$$
(1.3.60)

Let us denote $a = \tan\left(\frac{\lambda L}{2}\right)$. Then, the system (1.3.60) can be rewritten as

$$\int \frac{a^2}{4} \sin^2 \lambda - a \sin \lambda \cos \lambda + \cos^2 \lambda > 1, \qquad (1.3.61)$$

$$\left\{\frac{1}{4a^2}\sin^2\lambda + \frac{1}{a}\sin\lambda\cos\lambda + \cos^2\lambda > 1.\right.$$
(1.3.62)

Multiplying (1.3.62) by a^2 and taking the sum with (1.3.61) we obtain

$$\frac{1}{4}(1+a^2)\sin^2\lambda + (1+a^2)\cos^2\lambda > 1+a^2,$$

which is impossible.

We see that the eigenvalues of the operators A_s^{μ} , A_{as}^{μ} are in fact embedded eigenvalues for the operator A^{μ} .

1.4 Existence of eigenvalues for the non-limit operator

1.4.1 Main result

We return now to the case of the ladder. As it was mentioned above, instead of studying the full operator A^{μ}_{ε} we will study separately the operators $A^{\mu}_{\varepsilon,s}$, $A^{\mu}_{\varepsilon,as}$ for which the existence of eigenvalues inside gaps will be established. The convergence of the essential spectrum of the operator $A_{\varepsilon,s}$ (resp. $A_{\varepsilon,as}$) to the one of the operator A_s (resp. A_{as}) is known since the works [61], [47]. More precisely, the following theorem holds.

Theorem 1.4.1 (Essential spectrum). Let $\{]a_n, b_n[, n \in \mathbb{N}^*\}$ be the gaps of the operator A_s (A_{as}) on the limit graph G. Then, for each $n_0 \in \mathbb{N}^*$ there exists $\varepsilon_0 > 0$ such that if $\varepsilon < \varepsilon_0$ the operator $A_{\varepsilon,s}$ ($A_{\varepsilon,as}$) has at least n_0 gaps $\{]a_{\varepsilon,n}, b_{\varepsilon,n}[, 1 \leq n \leq n_0\}$ such that

$$a_{\varepsilon,n} = a_n + O(\varepsilon), \quad b_{\varepsilon,n} = b_n + O(\varepsilon), \qquad \varepsilon \to 0, \qquad 1 \le n \le n_0.$$

The proof of this fact is based on the Floquet-Bloch theory (which permits to reduce the study of the spectrum of a periodic operator to the study of a family of operators in a bounded domain) and the min-max principle (for studying the eigenvalues of an operator in a bounded domain). Theorem 1.4.1 guarantees the existence of gaps for the operators $A_{\varepsilon,s}$, $A_{\varepsilon,as}$ at least for ε small enough. We fix then one of these gaps and study the existence of eigenvalues inside it. Our principal result is the following.

Theorem 1.4.2 (Discrete spectrum). Let]a, b[be a gap of the operator A_s^{μ} (A_{as}^{μ}) on the limit graph G and $\lambda^{(0)} \in]a, b[$ a (simple) eigenvalue of this operator. Then there exists $\varepsilon_0 > 0$ such that if $\varepsilon < \varepsilon_0$ the operator $A_{\varepsilon,s}^{\mu}$ $(A_{\varepsilon,as}^{\mu})$ has an eigenvalue λ_{ε} inside a gap $]a_{\varepsilon}, b_{\varepsilon}[$ with the following asymptotic expansion at any order n:

$$\lambda_{\varepsilon} = \sum_{k=0}^{n} \lambda^{(k)} \varepsilon^{k} + O\left(\varepsilon^{n+1}\right), \qquad \varepsilon \to 0.$$
(1.4.1)

We prove this theorem in Chapter 2 using Matched Asymptotic Expansions. However, we will give now a proof of a weak version of this theorem using a simpler argument.

Theorem 1.4.3 (Discrete spectrum – weak version). Let]a, b[be a gap of the operator A_s^{μ} (A_{as}^{μ}) on the limit graph G and $\lambda^{(0)} \in]a, b[$ a (simple) eigenvalue of this operator. Then there exists $\varepsilon_0 > 0$ such that if $\varepsilon < \varepsilon_0$ the operator $A_{\varepsilon,s}^{\mu}$ $(A_{\varepsilon,as}^{\mu})$ has an eigenvalue λ_{ε} inside a gap $]a_{\varepsilon}, b_{\varepsilon}[$ such that

$$\lambda_{\varepsilon} = \lambda^{(0)} + O\left(\sqrt{\varepsilon}\right), \qquad \varepsilon \to 0.$$

Though the rate of convergence guaranteed by this theorem is not optimal, the proof given below provides an easy way to establish the existence of eigenvalues for the operators $A^{\mu}_{\varepsilon,s}$, $A^{\mu}_{\varepsilon,as}$. It also illustrates the main ideas that will be used in the proof of Theorem 1.4.2 in a simplified context.

Remark 1.4.1. In [57] O. Post proves the convergence of all the components of the spectrum of the Neumann Laplacian for a large class of graph-like manifolds (in particular, they are not necessarily compact). Applied to our case, these results imply the existence of eigenvalues of the operators $A_{\varepsilon,s}^{\mu}$, $A_{\varepsilon,as}^{\mu}$ inside gaps for ε small enough. They also guarantee that these eigenvalues are simple (which is not established in Theorem 1.4.2). The optimal rate of convergence for the eigenvalues, which is linear in ε , has also been proved in [57]. Thus, we provide an alternative proof of existence of eigenvalues of the operators $A_{\varepsilon,s}^{\mu}$, $A_{\varepsilon,as}^{\mu}$ which permits to obtain a full asymptotic expansion of the eigenvalues (1.4.1).

Combining Propositions 1.3.4 (2), 1.3.12 (2) and Theorems 1.4.1, 1.4.2, 1.3.1, 1.3.2 we can formulate the following result.

Theorem 1.4.4. For any $n_0 \in \mathbb{N}^*$ there exists $\varepsilon_0 > 0$ such that if $\varepsilon < \varepsilon_0$ the operator $A_{\varepsilon,s}$ $(A_{\varepsilon,as})$ has at least n_0 eigenvalues. Moreover, for any A > 0 there exists $\varepsilon_0 > 0$ such that if $\varepsilon < \varepsilon_0$ the operator $A_{\varepsilon,s}$ $(A_{\varepsilon,as})$ has at least one eigenvalue greater than A.

1.4.2 Method of a pseudo-mode

We will now prove Theorem 1.4.3. We consider only the operator $A^{\mu}_{\varepsilon,s}$, the case of the operator $A^{\mu}_{\varepsilon,as}$ being analogous. Our proof is based on the construction of a so-called pseudo-mode, i.e. an approximation of the eigenfunction of the operator $A^{\mu}_{\varepsilon,s}$. More

precisely, we use Lemma A.2.1 that gives an estimate for the distance from the spectrum once such an approximation is found. In application to the operator $A^{\mu}_{\varepsilon,s}$ Lemma A.2.1 means that if there exists a function $u_{\varepsilon} \in H^1_s(\Omega^{\mu}_{\varepsilon})$ (H^1_s standing for the symmetric subspace of H^1) such that

$$\left| \int_{\Omega_{\varepsilon}^{\mu}} \left(\nabla u_{\varepsilon} \nabla v - \lambda^{(0)} u_{\varepsilon} v \right) d\Omega \right| \leqslant C \sqrt{\varepsilon} \| u_{\varepsilon} \|_{H^{1}(\Omega_{\varepsilon}^{\mu})} \| v \|_{H^{1}(\Omega_{\varepsilon}^{\mu})}, \qquad \forall v \in H^{1}_{s}(\Omega_{\varepsilon}^{\mu}), \qquad (1.4.2)$$

then

$$dist(\sigma(A^{\mu}_{\varepsilon,s}),\lambda^{(0)}) \leqslant \tilde{C}\sqrt{\varepsilon}, \qquad (1.4.3)$$

with some constant \widetilde{C} which does not depend on ε . According to Theorem 1.4.1, for ε small enough there exists a constant C such that $\sigma_{ess}(A^{\mu}_{\varepsilon,s}) \cap [a + C\varepsilon, b - C\varepsilon] = \emptyset$. Together with (1.4.3) it means that $\sigma_d(A^{\mu}_{\varepsilon,s}) \cap [\lambda^{(0)} - \widetilde{C}\sqrt{\varepsilon}, \lambda^{(0)} + \widetilde{C}\sqrt{\varepsilon}] \neq \emptyset$ which proves the existence of an eigenvalue of the operator $A^{\mu}_{\varepsilon,s}$ in a neighbourhood of $\lambda^{(0)}$ of order $\sqrt{\varepsilon}$ for ε small enough.

1.4.3 Construction of a pseudo-mode

Let us introduce some notation for the domain $\Omega^{\mu}_{\varepsilon}$ (cf. figure 1.12). We denote by $\mathcal{V}_{j+\frac{1}{2}}^{\varepsilon,\pm}$ its horizontal top and bottom edges,

$$\mathcal{V}_{j+\frac{1}{2}}^{\varepsilon,+} = \left] j + \frac{w_j^{\mu}\varepsilon}{2}, \ j+1 - \frac{w_{j+1}^{\mu}\varepsilon}{2} \right[\times \left] \frac{L}{2} - \varepsilon, \ \frac{L}{2} \right[, \qquad j \in \mathbb{Z},$$

$$\mathcal{V}_{j+\frac{1}{2}}^{\varepsilon,-} = \left] j + \frac{w_j^{\mu}\varepsilon}{2}, \ j+1 - \frac{w_{j+1}^{\mu}\varepsilon}{2} \right[\times \left] - \frac{L}{2}, \ -\frac{L}{2} + \varepsilon \right[, \qquad j \in \mathbb{Z},$$

by $\mathcal{V}_j^{\varepsilon}$ its vertical edges,

$$\mathcal{V}_{j}^{\varepsilon} = \left] j - \frac{w_{j}^{\mu}\varepsilon}{2}, \ j + \frac{w_{j}^{\mu}\varepsilon}{2} \right[\times \left] \frac{L}{2} - \varepsilon, \ \frac{L}{2} + \varepsilon \right[, \qquad j \in \mathbb{Z},$$

and by $K_j^{\varepsilon,\pm}$ the top and bottom junctions,

$$\begin{split} K_{j}^{\varepsilon,+} &= \left] j - \frac{w_{j}^{\mu}\varepsilon}{2}, \ j + \frac{w_{j}^{\mu}\varepsilon}{2} \right[\times \left] \frac{L}{2} - \varepsilon, \ \frac{L}{2} \right[, \qquad \qquad j \in \mathbb{Z}, \\ K_{j}^{\varepsilon,-} &= \left] j - \frac{w_{j}^{\mu}\varepsilon}{2}, \ j + \frac{w_{j}^{\mu}\varepsilon}{2} \right[\times \left] - \frac{L}{2}, \ -\frac{L}{2} + \varepsilon \right[, \qquad \qquad j \in \mathbb{Z}. \end{split}$$

We also introduce a notation for the boundaries separating these subdomains:

$$\Gamma_{j+\frac{1}{2}}^{0,\varepsilon,\pm} = \partial \mathcal{V}_{j+\frac{1}{2}}^{\varepsilon,\pm} \cap \partial K_{j}^{\varepsilon,\pm}, \qquad \Gamma_{j+\frac{1}{2}}^{1,\varepsilon,\pm} = \partial \mathcal{V}_{j+\frac{1}{2}}^{\varepsilon,\pm} \cap \partial K_{j+1}^{\varepsilon,\pm}, \qquad \Gamma_{j}^{\varepsilon,\pm} = \partial \mathcal{V}_{j}^{\varepsilon} \cap \partial K_{j}^{\varepsilon,\pm}, \qquad j \in \mathbb{Z}.$$

We construct the pseudo-mode u_{ε} as follows. Let u_0 be the eigenfunction of the operator A_s^{μ} corresponding to the eigenvalue $\lambda^{(0)}$. Then, u_{ε} is defined on $\Omega_{\varepsilon}^{\mu}$ by "fattening" u_0 (with an appropriate rescaling):

$$u_{\varepsilon}(x,y) = \begin{cases} u_{0,j+\frac{1}{2}}(s_{j+\frac{1}{2}}^{\varepsilon}(x)), & (x,y) \in \mathcal{V}_{j+\frac{1}{2}}^{\varepsilon,\pm}, \\ u_{0,j}(t^{\varepsilon}(y)), & (x,y) \in \mathcal{V}_{j}^{\varepsilon}, \\ \mathbf{u}_{0,j}, & (x,y) \in K_{j}^{\varepsilon,\pm}. \end{cases}$$



Figure 1.12: Construction of a pseudo-mode

Here $s_{j+\frac{1}{2}}^{\varepsilon}$ and t^{ε} are given by the relations

$$s_{j+\frac{1}{2}}^{\varepsilon}(x) = \frac{x - j - w_{j}^{\mu}\varepsilon/2}{1 - \left(w_{j}^{\mu} + w_{j+1}^{\mu}\right)\varepsilon/2}, \qquad t^{\varepsilon}(y) = \frac{y}{1 - 2\varepsilon/L}.$$
(1.4.4)

We note that due to a standard density argument (Meyers-Serrin's theorem) it is sufficient to prove (1.4.2) for any $v \in C^{\infty}(\Omega^{\mu}_{\varepsilon}) \cap H^1_s(\Omega^{\mu}_{\varepsilon})$. Let us estimate the left-hand side of (1.4.2) for $v \in C^{\infty}(\Omega_{\varepsilon}^{\mu}) \cap H^1_s(\Omega_{\varepsilon}^{\mu})$. Integrating by parts and using the fact that u_0 is an eigenfunction on the graph as well as the symmetry of the functions u_0 and v, we get:

$$\left| \int_{\Omega_{\varepsilon}^{\mu}} \left(\nabla u_{\varepsilon} \nabla v - \lambda^{(0)} u_{\varepsilon} v \right) d\Omega \right| \leq 2 \sum_{j \in \mathbb{Z}} \lambda^{(0)} \mathbf{u}_{0,j} \left| \int_{K_{j}^{\varepsilon,-}} v d\Omega \right| + \lambda^{(0)} \|u_{\varepsilon}\|_{L_{2}(\Omega_{\varepsilon}^{\mu})} \|v\|_{L_{2}(\Omega_{\varepsilon}^{\mu})} O(\varepsilon)$$

$$+ 2 \sum_{j \in \mathbb{Z}} \left(-u_{0,j+\frac{1}{2}}'(0) \int_{\Gamma_{j+\frac{1}{2}}^{0,\varepsilon,-}} v(x,y) dy + u_{0,j-\frac{1}{2}}'(1) \int_{\Gamma_{j-\frac{1}{2}}^{1,\varepsilon,-}} v(x,y) dy - u_{0,j}'\left(-\frac{L}{2}\right) \int_{\Gamma_{j}^{\varepsilon,-}} v(x,y) dx \right)$$

$$\times (1 + O(\varepsilon)), \qquad (1.4.5)$$

The terms in (1.4.5) containing $O(\varepsilon)$ appear because of the change of variables (1.4.4). Taking into account (1.3.53) and using Hölder's inequality, the first term in the right-hand side of (1.4.5) can be estimated as follows:

T

$$2\sum_{j\in\mathbb{Z}}\lambda^{(0)}\mathbf{u}_{0,j}\left|\int\limits_{K_{j}^{\varepsilon,-}}vd\Omega\right|\leqslant C\varepsilon\|v\|_{L_{2}(\Omega_{\varepsilon}^{\mu})}.$$
(1.4.6)

Here and in what follows we denote by C all constants (not necessarily the same) which do not depend on ε . Next, if $M_j^{\varepsilon,-}$ is a barycentre of $K_j^{\varepsilon,-}$, then

$$\left|v(x,y)-v(M_{j}^{\varepsilon,-})\right| \leqslant \int_{M_{j}^{\varepsilon,-}}^{(x,y)} |\nabla v| dt,$$

where the integral is taken along the segment joining the points (x, y) and $M_j^{\varepsilon,-}$. Due to Kirchhoff's conditions (1.3.7) verified by the function u_0 we can replace v(x, y) by $v(x, y) - v(M_j^{\varepsilon,-})$ in the integrals over the boundaries in the right-hand side of (1.4.5). We have

$$\left| \int_{\Gamma_{j}^{\varepsilon,-}} \left(v(x,y) - v(M_{j}^{\varepsilon,-}) \right) dx \right| \leqslant \int_{\Gamma_{j}^{\varepsilon,-}} \int_{M_{j}^{\varepsilon,-}}^{(x,y)} |\nabla v| dx dt \leqslant C\varepsilon ||v||_{H^{1}(K_{j}^{\varepsilon,-})}.$$
(1.4.7)

Combining (1.4.5)–(1.4.7) and taking into account (1.3.41), (1.3.42), (1.3.53) we obtain the following estimate:

$$\left| \int_{\Omega_{\varepsilon}^{\mu}} \left(\nabla u_{\varepsilon} \nabla v - \lambda^{(0)} u_{\varepsilon} v \right) d\Omega \right| \leqslant C \varepsilon \|v\|_{H^{1}(\Omega_{\varepsilon}^{\mu})}, \qquad \forall v \in C^{\infty}(\Omega_{\varepsilon}^{\mu}) \cap H^{1}_{s}(\Omega_{\varepsilon}^{\mu}).$$
(1.4.8)

Notice that by definition of u_{ε} one has

$$||u_{\varepsilon}||_{H^{1}(\Omega_{\varepsilon}^{\mu})} \ge C\sqrt{\varepsilon}||u_{0}||_{H^{1}(G)}, \qquad C > 0,$$

which together with (1.4.8) and the density argument mentioned above finishes the proof of (1.4.2) and hence, of Theorem 1.4.3.

CHAPTER 2

TRAPPED MODES IN A LOCALLY PERTURBED PERIODIC LADDER: ASYMPTOTIC EXPANSIONS OF THE EIGENVALUES

In this chapter we show how the proof of Theorem 1.4.3 can be modified in order to obtain the complete asymptotic expansion of the eigenvalue stated in Theorem 1.4.2. It will be done by constructing another pseudo-mode based on the formal asymptotic expansion of the eigenfunction of the operator $A^{\mu}_{\varepsilon,s}$ (here again, we give the proof for the operator $A^{\mu}_{\varepsilon,s}$, the proof for the operator $A^{\mu}_{\varepsilon,as}$ being analogous). To do so, we will use the matched asymptotic expansion method in the spirit of the works [63], [35], [36] (see also [66], [30], [50], [5]). The method consists in distinguishing different areas where the behaviour of the solution is different and imposing formal expansions in these areas. Then, the different expansions have to match in some intermediate zones called matching zones. This leads to matching conditions which, together with the equations satisfied in each zone, permit to determine the terms of the expansions.

The notation in this section is different from the rest of the work: the spectral parameter here is denoted by λ , contrarily to all the others chapters, where it is denoted by λ^2 .

2.1 Formal expansions

Since only symmetric functions are considered when studying the operator $A^{\mu}_{\varepsilon,s}$, it is sufficient to define them on the lower half of the band $\Omega^{\mu}_{\varepsilon}$, which is a comb-shape domain that we denote by C^{μ}_{ε} (see figure 2.2a):

$$\mathcal{C}^{\mu}_{\varepsilon} = \{ (x, y) \in \Omega^{\mu}_{\varepsilon} / y < 0 \} .$$



Figure 2.1

Let u^{ε} and λ^{ε} be a formal solution of the eigenvalue problem

$$\begin{cases} \left. \begin{array}{c} \Delta u^{\varepsilon} + \lambda^{\varepsilon} u^{\varepsilon} = 0 & \text{in } \mathcal{C}^{\mu}_{\varepsilon}, \\ \left. \frac{\partial u^{\varepsilon}}{\partial n} \right|_{\partial \mathcal{C}^{\mu}_{\varepsilon}} = 0, \end{array} \right.$$
(2.1.1)

for which the following expansions are assumed.

Far field expansion

It is valid in the regions situated far from the junctions. When ε is small, the branches of the domain are thin. For this reason, it is natural to model the solution by functions depending on the longitudinal variable only, the dependence on the transversal variable being neglected:

For
$$(x,y) \in \mathcal{V}_{j+\frac{1}{2}}^{\varepsilon,-} \cap \left\{ x \in \left] j + \sqrt{\varepsilon}, \ j+1 - \sqrt{\varepsilon} \right[\right\}, \ j \in \mathbb{Z} :$$

$$u^{\varepsilon}(x,y) \equiv u_{j+\frac{1}{2}}^{\varepsilon}(x,y) = \sum_{k \in \mathbb{N}} \varepsilon^{k} u_{j+\frac{1}{2}}^{(k)}(s) + o(\varepsilon^{\infty}), \qquad s = x - j, \qquad (2.1.2)$$

For
$$(x, y) \in \mathcal{V}_{j}^{\varepsilon} \cap \left\{ y \in \left] - \frac{L}{2} + \sqrt{\varepsilon}, 0 \right] \right\}, \quad j \in \mathbb{Z} :$$

$$u^{\varepsilon}(x, y) \equiv u_{j}^{\varepsilon}(x, y) = \sum_{k \in \mathbb{N}} \varepsilon^{k} u_{j}^{(k)}(y) + o(\varepsilon^{\infty}).$$
(2.1.3)

The functions $u_{j+\frac{1}{2}}^{(k)}$ and $u_j^{(k)}$ are defined on the edges $e_{j+\frac{1}{2}}$ and e_j of the limit graph $G_{\mathcal{C}}$ (which is the lower half of the graph G, cf. figure 2.2b). They are supposed smooth (this

will be shown a posteriori):

$$u_{j+\frac{1}{2}}^{(k)} \in C^{\infty}([0,1]), \quad u_{j}^{(k)} \in C^{\infty}\left(\left[-\frac{L}{2},0\right]\right), \quad \left(u_{j}^{(k)}\right)'(0) = 0, \qquad \forall k \in \mathbb{N}, \quad \forall j \in \mathbb{Z}.$$

The set of functions $\left\{u_{j+\frac{1}{2}}^{(k)}, u_{j}^{(k)}\right\}_{j\in\mathbb{Z}}$ will sometimes be denoted by $u^{(k)}$, which is a function defined on the graph.

Near field expansion

It is valid in the neighbourhood of the junctions. The solution in this region is a function of two rescaled variables X and Y:

For
$$(x,y) \in \mathcal{C}^{\mu}_{\varepsilon} \cap \left\{ x \in \left] j - 2\sqrt{\varepsilon}, \ j + 2\sqrt{\varepsilon} \right[, \ y \in \left] - \frac{L}{2}, \ -\frac{L}{2} + 2\sqrt{\varepsilon} \right] \right\}, \quad j \in \mathbb{Z}:$$

$$u^{\varepsilon}(x,y) \equiv U^{\varepsilon}_{j}(x,y) = \sum_{k \in \mathbb{N}} \varepsilon^{k} U^{(k)}_{j}(X,Y) + o(\varepsilon^{\infty}), \qquad X = \frac{x-j}{\varepsilon}, \qquad Y = \frac{y+L/2}{\varepsilon}.$$

$$(2.1.4)$$

The functions $U_j^{(k)}$ are defined in the rescaled neighbourhoods of the junctions \mathcal{J}_j :

$$U_{j}^{(k)} \in H_{\Delta,loc}^{1}\left(\mathcal{J}_{j}\right), \qquad \forall j \in \mathbb{Z},$$

where

$$\mathcal{J}_j = \begin{cases} \mathcal{J}_*, & j \in \mathbb{Z}^*, \\ \mathcal{J}_0, & j = 0, \end{cases}$$

$$\mathcal{J}_* = \{\mathbb{R} \times]0, 1[\} \cup \{] -\frac{1}{2}, \frac{1}{2} [\times \mathbb{R}\}, \qquad \mathcal{J}_0 = \{\mathbb{R} \times]0, 1[\} \cup \{] -\frac{\mu}{2}, \frac{\mu}{2} [\times \mathbb{R}\}.$$

For $j \neq 0$ the domain $\mathcal{J}_j = \mathcal{J}_*$ is the unperturbed infinite junction whereas for j = 0 the domain \mathcal{J}_0 is the perturbed one (cf. figure 2.2). We will use the following notation:

$$K_{j} = \left] -\frac{w_{j}^{\mu}}{2}, \frac{w_{j}^{\mu}}{2} \right[\times]0, 1[, \qquad B_{j}^{\pm} = \mathcal{J}_{j} \cap \left\{ \pm X > \frac{w_{j}^{\mu}}{2} \right\}, \qquad B_{j}^{0} = \mathcal{J}_{j} \cap \{Y > 1\},$$

$$\Sigma_{j}^{N} = \left[-\frac{w_{j}^{\mu}}{2}, \frac{w_{j}^{\mu}}{2} \right] \times \{0\}, \qquad \Sigma_{j}^{\pm} = \left\{ \pm \frac{w_{j}^{\mu}}{2} \right\} \times [0, 1], \qquad \Sigma_{j}^{0} = \left[-\frac{w_{j}^{\mu}}{2}, \frac{w_{j}^{\mu}}{2} \right] \times \{1\}.$$

The functions $U_j^{(k)}$ are supposed to have at most polynomial growth in the infinite branches B_j^{\pm} , B_j^0 of the junctions \mathcal{J}_j . The set of functions $\left\{U_j^{(k)}\right\}_{j\in\mathbb{Z}}$ will sometimes be denoted by $U^{(k)}$.

Expansion for the eigenvalue:

$$\lambda^{\varepsilon} = \sum_{k \in \mathbb{N}} \varepsilon^k \lambda^{(k)}.$$
 (2.1.5)



Figure 2.2

Far field and near field problems: After injecting the relations (2.1.2), (2.1.3), (2.1.5) into (2.1.1) we obtain the following problem for $u^{(k)}$, $k \in \mathbb{N}$:

$$\begin{cases} \left(u_{j+\frac{1}{2}}^{(k)}\right)''(s) + \lambda^{(0)}u_{j+\frac{1}{2}}^{(k)}(s) = -\sum_{m=0}^{k-1}\lambda^{(k-m)}u_{j+\frac{1}{2}}^{(m)}(s), & s \in [0,1], \quad j \in \mathbb{Z}, \\ \left(u_{j}^{(k)}\right)''(y) + \lambda^{(0)}u_{j}^{(k)}(y) = -\sum_{m=0}^{k-1}\lambda^{(k-m)}u_{j}^{(m)}(y), & y \in \left[-\frac{L}{2},0\right], \quad j \in \mathbb{Z}, \\ \left(u_{j}^{(k)}\right)'(0) = 0, & j \in \mathbb{Z}. \end{cases}$$

$$(2.1.6)$$

Analogously, after injecting (2.1.4), (2.1.5) into (2.1.1) we find a set of problems for $U^{(k)}$, $k \in \mathbb{N}$:

$$\forall j \in \mathbb{Z}, \qquad \begin{cases} \left. \Delta U_j^{(k)}(X,Y) = -\sum_{m=0}^{k-2} \lambda^{(k-m-2)} U_j^{(m)}(X,Y), \qquad (X,Y) \in \mathcal{J}_j, \\ \left. \frac{\partial U_j^{(k)}}{\partial n} \right|_{\partial \mathcal{J}_j} = 0. \end{cases}$$

$$(2.1.7)$$

Clearly, the problems (2.1.6) and (2.1.7) are not well-posed. The far field problem need to be completed by transmission conditions at the vertices of the graph and for the near field terms the behaviour at infinity has to be specified. The missing relations can be found by taking into account the matching conditions in the regions where both far field and near field expansions should hold and match.

Matching conditions: Let us define the intermediate zones (the matching areas), cf. figure 2.3:

$$\mathcal{M}_{j,\varepsilon}^{+} = \mathcal{V}_{j+\frac{1}{2}}^{\varepsilon,-} \cap \left\{ x \in \left] j + \sqrt{\varepsilon}, \ j + 2\sqrt{\varepsilon} \right[\right\}, \quad \mathcal{M}_{j,\varepsilon}^{-} = \mathcal{V}_{j-\frac{1}{2}}^{\varepsilon,-} \cap \left\{ x \in \left] j - 2\sqrt{\varepsilon}, \ j - \sqrt{\varepsilon} \right[\right\},$$
$$\mathcal{M}_{j,\varepsilon}^{0} = \mathcal{V}_{j}^{\varepsilon} \cap \left\{ y \in \left] - \frac{L}{2} + \sqrt{\varepsilon}, -\frac{L}{2} + 2\sqrt{\varepsilon} \right] \right\}, \qquad j \in \mathbb{Z}.$$

As follows from the way we defined the far field and the near field expansions, they should



Figure 2.3: Matching areas

both hold in the matching areas $\mathcal{M}_{j,\varepsilon}^{\pm}$, $\mathcal{M}_{j,\varepsilon}^{0}$:

$$u_{j+\frac{1}{2}}^{\varepsilon}(x,y) = \sum_{k \in \mathbb{N}} \varepsilon^k u_{j+\frac{1}{2}}^{(k)}(x-j) + o(\varepsilon^{\infty}) = \sum_{k \in \mathbb{N}} \varepsilon^k U_j^{(k)}(X,Y) + o(\varepsilon^{\infty}), \qquad (x,y) \in \mathcal{M}_{j,\varepsilon}^+,$$
(2.1.8)

$$u_{j-\frac{1}{2}}^{\varepsilon}(x,y) = \sum_{k \in \mathbb{N}} \varepsilon^k u_{j-\frac{1}{2}}^{(k)}(x+1-j) + o(\varepsilon^{\infty}) = \sum_{k \in \mathbb{N}} \varepsilon^k U_j^{(k)}(X,Y) + o(\varepsilon^{\infty}), \quad (x,y) \in \mathcal{M}_{j,\varepsilon}^-,$$

$$u_j^{\varepsilon}(x,y) = \sum_{k \in \mathbb{N}} \varepsilon^k u_j^{(k)}(y) + o(\varepsilon^{\infty}) = \sum_{k \in \mathbb{N}} \varepsilon^k U_j^{(k)}(X,Y) + o(\varepsilon^{\infty}), \qquad (x,y) \in \mathcal{M}_{j,\varepsilon}^0.$$

The regions $\mathcal{M}_{j,\varepsilon}^{\pm}$ correspond to $x - j \to \pm 0$ and $X \to \pm \infty$. Analogously, the regions $\mathcal{M}_{j,\varepsilon}^{0}$ correspond to $y \to -L/2$ and $Y \to +\infty$. This leads us to studying the behaviour of the far field terms near the vertices of the graph and the behaviour of the near field terms at infinity.

As we will see in the following section, the far field terms have the following behaviour in the infinite branches of the junctions \mathcal{J}_j :

$$U_{j}^{(k)}(X,Y) = P_{j,0,\pm}^{(k)}(X) + \mathcal{E}_{j,\pm}^{(k)}(X,Y), \qquad (X,Y) \in B_{j}^{\pm}, \qquad k \in \mathbb{N}, \qquad j \in \mathbb{Z},$$
$$U_{j}^{(k)}(X,Y) = P_{j,0,0}^{(k)}(Y) + \mathcal{E}_{j,0}^{(k)}(X,Y), \qquad (X,Y) \in B_{j}^{0}, \qquad k \in \mathbb{N}, \qquad j \in \mathbb{Z}.$$

where the terms $\mathcal{E}_{j,\pm}^{(k)}$, $\mathcal{E}_{j,0}^{(k)}$. Consider, for example, the region $\mathcal{M}_{j,\varepsilon}^+$. Since in this region $\mathcal{E}_{j,\pm}^{(k)}$ is exponentially decaying as $X \to \infty$ and $X = O(\varepsilon^{-1/2})$, we can rewrite (2.1.8) as

$$u_{j+\frac{1}{2}}^{\varepsilon}(x,y) = \sum_{k \in \mathbb{N}} \varepsilon^k P_{j,0,\pm}^{(k)}(X) + o(\varepsilon^{\infty}), \qquad (x,y) \in \mathcal{M}_{j,\varepsilon}^+.$$
(2.1.9)

On the other hand, the functions $u_{j+\frac{1}{2}}^{(k)}$ can be decomposed in Taylor series in a neigh-

bourhood of the point s = 0. Then, we get from (2.1.8):

$$u_{j+\frac{1}{2}}^{\varepsilon}(x,y) = \sum_{k \in \mathbb{N}} \sum_{\ell \in \mathbb{N}} \left. \frac{d^{\ell} u_{j+\frac{1}{2}}^{(k)}}{ds^{\ell}} \right|_{s=0} \frac{\varepsilon^{k} (x-j)^{\ell}}{\ell!} + o(\varepsilon^{\infty}) = \sum_{k \in \mathbb{N}} \varepsilon^{k} \sum_{\ell=0}^{k} \left. \frac{d^{\ell} u_{j+\frac{1}{2}}^{(k-\ell)}}{ds^{\ell}} \right|_{s=0} \frac{X^{\ell}}{\ell!} + o(\varepsilon^{\infty}).$$
(2.1.10)

Comparing (2.1.9) with (2.1.10) and identifying the terms with the same powers of ε , we obtain the following expressions for the polynomials $P_{j,0,+}^{(k)}$:

$$P_{j,0,+}^{(k)}(X) = \sum_{\ell=0}^{k} \left. \frac{d^{\ell} u_{j+\frac{1}{2}}^{(k-\ell)}}{ds^{\ell}} \right|_{s=0} \frac{X^{\ell}}{\ell!}, \qquad k \in \mathbb{N}, \qquad j \in \mathbb{Z}.$$
(2.1.11)

2.2 Near field and far field problems

In Section 2.2.1 we obtain (without giving a rigorous argument) the problems for the near field terms posed in a bounded domain. The compatibility conditions for these problems will permit us to obtain the problem for the far field terms (this is done in Section 2.2.2). In Sections 2.2.4 we come back to the near field problems in the domains \mathcal{J}_j and we show that the existence of solutions of the problems posed in the bounded domain implies the existence of near field terms satisfying the problems (2.1.7) with a specified behaviour in the infinite branches. Finally, in Section 2.2.5 we study the well-posedness of the far field problem.

2.2.1 Formal derivation of the near field problems

2.2.1.1 Near field problems in the infinite junctions

In the spirit of the works [36], [35], let us introduce the following orthonormal basis in $L_2([0,1])$ which consists of eigenfunctions of the transverse Laplacian with Neumann boundary conditions:

$$f_0(t) = 1,$$
 $f_p(t) = \sqrt{2}\cos\left(p\pi t\right),$ $p \ge 1,$

Using this basis the functions $U_j^{(k)}$ in the infinite branches B_j^+ can be represented as follows:

$$U_{j}^{(k)}(X,Y) = \sum_{p \in \mathbb{N}} U_{j,p,+}^{(k)}(X) f_{p}(Y), \qquad (X,Y) \in B_{j}^{+}, \quad k \in \mathbb{N}, \qquad j \in \mathbb{Z}.$$
(2.2.1)

After injecting (2.2.1) into (2.1.7) we get the following set of ordinary differential equations for the functions $U_{j,p,+}^{(k)}$:

$$\left(U_{j,p,+}^{(k)}\right)''(X) - \pi^2 p^2 U_{j,p,+}^{(k)}(X) = -\sum_{m=0}^{k-2} \lambda^{(m)} U_{j,p,+}^{(k-2-m)}(X), \qquad k, p \in \mathbb{N}, \quad j \in \mathbb{Z}.$$
(2.2.2)

Making the change of the unknown function in (2.2.2) of the form

$$U_{j,p,+}^{(k)}(X) = P_{j,p,+}^{(k)}(X)e^{-p\pi X},$$
(2.2.3)

we obtain the following recurrence relation:

$$\left(P_{j,p,+}^{(k)}\right)''(X) - 2p\pi \left(P_{j,p,+}^{(k)}\right)'(X) = -\sum_{m=0}^{k-2} \lambda^{(m)} P_{j,p,+}^{(k-2-m)}(X), \qquad k \in \mathbb{N}, \quad p \in \mathbb{N}^*, \quad j \in \mathbb{Z}.$$

As usual, when constructing matched asymptotic expansions, for $p \neq 0$ we search solutions of at most polynomial growth, which implies

$$P_{j,p,+}^{(k)}(X) = -\int_{X}^{\infty} \widetilde{P}_{j,p,+}^{(k-2)}(X') dX' + c_{j,p,+,0}^{(k)}, \qquad k \in \mathbb{N}, \quad p \in \mathbb{N}^*, \quad j \in \mathbb{Z},$$
(2.2.4)

$$\widetilde{P}_{j,p,+}^{(k-2)}(X) = \sum_{m=0}^{k-2} \lambda^{(m)} \int_{X}^{\infty} P_{j,p,+}^{(k-2-m)}(X') e^{2p\pi(X-X')} dX'.$$
(2.2.5)

Hence, by induction in k, the functions $P_{j,p,+}^{(k)}$ are polynomials of degree $\left[\frac{k}{2}\right]$ for $k \in \mathbb{N}$, $p \in \mathbb{N}^*$, $j \in \mathbb{Z}$:

$$P_{j,p,+}^{(k)}(X) = \sum_{\ell=0}^{\left\lfloor \frac{k}{2} \right\rfloor} c_{j,p,+,\ell}^{(k)} X^{\ell}, \qquad k \in \mathbb{N}, \quad p \in \mathbb{N}^*, \quad j \in \mathbb{Z}.$$
(2.2.6)

It follows from (2.2.4), (2.2.5) that for $\ell \neq 0$ the coefficients $c_{j,p,+,\ell}^{(k)}$ are given by the following recurrence relation:

$$c_{j,p,+,\ell}^{(k)} = \sum_{m=0}^{k-2\ell} \sum_{i=\ell-1}^{\left\lfloor \frac{k-m}{2} \right\rfloor - 1} \frac{\lambda^{(m)} i!}{(2p\pi)^{i-\ell+2} \ell!} c_{j,p,+,i}^{(k-2-m)},$$

$$1 \leq \ell \leq \left\lfloor \frac{k}{2} \right\rfloor, \quad k \in \mathbb{N}, \quad p \in \mathbb{N}^*, \quad j \in \mathbb{Z}.$$
(2.2.7)

The functions $U_{j,0,+}^{(k)} = P_{j,0,+}^{(k)}$ have been found in (2.1.11). Let us mention that they satisfy the recurrence relation (2.2.2) for p = 0 due to the fact that the far field terms satisfy the equations (2.1.6). Finally, (2.2.1) can be rewritten as

$$U_{j}^{(k)}(X,Y) = P_{j,0,+}^{(k)}(X) + \sum_{p \in \mathbb{N}^{*}} P_{j,p,+}^{(k)}(X) e^{-p\pi X} f_{p}(Y), \qquad (X,Y) \in B_{j}^{+},$$

$$k \in \mathbb{N}, \qquad j \in \mathbb{Z}, \qquad (2.2.8)$$

with the polynomials $P_{j,p,+}^{(k)}$, $p \in \mathbb{N}^*$, defined in (2.2.6)–(2.2.7). Obviously, a similar argument applies for the domains B_j^- and B_j^0 , and analogues of the relations (2.2.4)–(2.2.8), (2.1.11) can be found:

$$U_{j}^{(k)}(X,Y) = P_{j,0,\pm}^{(k)}(X) + \sum_{p \in \mathbb{N}^{*}} P_{j,p,\pm}^{(k)}(\pm X) e^{\mp p\pi X} f_{p}(Y), \qquad (X,Y) \in B_{j}^{\pm}, \qquad (2.2.9)$$

$$U_{j}^{(k)}(X,Y) = P_{j,0,0}^{(k)}(Y) + \frac{1}{\sqrt{w_{j}^{\mu}}} \sum_{p \in \mathbb{N}^{*}} P_{j,p,0}^{(k)}(Y) e^{-p\pi Y/w_{j}^{\mu}} f_{p}\left(\frac{X}{w_{j}^{\mu}} - \frac{1}{2}\right), \quad (X,Y) \in B_{j}^{0}.$$
(2.2.10)

The functions $P_{j,p,\delta}^{(k)}$ for $p \neq 0$ are polynomials of degree $\left[\frac{k}{2}\right]$ given by the relations

$$P_{j,p,\delta}^{(k)}(s) = \sum_{\ell=0}^{\left\lfloor \frac{k}{2} \right\rfloor} c_{j,p,\delta,\ell}^{(k)} s^{\ell}, \qquad \delta \in \{+,-,0\}, \quad k \in \mathbb{N}, \quad p \in \mathbb{N}^*, \quad j \in \mathbb{Z},$$
(2.2.11)

$$c_{j,p,\pm,\ell}^{(k)} = \sum_{m=0}^{k-2\ell} \sum_{i=\ell-1}^{\left\lfloor\frac{k-m}{2}\right\rfloor - 1} \frac{\lambda^{(m)}i!}{(2p\pi)^{i-\ell+2}\ell!} c_{j,p,\pm,i}^{(k-2-m)}, \quad c_{j,p,0,\ell}^{(k)} = \sum_{m=0}^{k-2\ell} \sum_{i=\ell-1}^{\left\lfloor\frac{k-m}{2}\right\rfloor - 1} \frac{(w_j^{\mu})^{i-\ell+2}\lambda^{(m)}i!}{(2p\pi)^{i-\ell+2}\ell!} c_{j,p,0,i}^{(k-2-m)}, \quad 1 \le \ell \le \left\lfloor\frac{k}{2}\right\rfloor, \quad k \in \mathbb{N}, \quad p \in \mathbb{N}^*, \quad j \in \mathbb{Z}.$$
(2.2.12)

The functions $P_{j,0,\delta}^{(k)}$, $\delta \in \{+, -, 0\}$, are polynomials of degree k for k odd and k + 1 for k even. As we have seen above (cf. (2.1.11)), they can be expressed in terms of the far field terms:

$$P_{j,0,+}^{(k)}(X) = \sum_{\ell=0}^{k} \left. \frac{d^{\ell} u_{j+\frac{1}{2}}^{(k-\ell)}}{ds^{\ell}} \right|_{s=0} \frac{X^{\ell}}{\ell!}, \qquad P_{j,0,-}^{(k)}(X) = \sum_{\ell=0}^{k} \left. \frac{d^{\ell} u_{j-\frac{1}{2}}^{(k-\ell)}}{ds^{\ell}} \right|_{s=1} \frac{X^{\ell}}{\ell!}, \qquad P_{j,0,0}^{(k)}(Y) = \sum_{\ell=0}^{k} \left. \frac{d^{\ell} u_{j}^{(k-\ell)}}{dy^{\ell}} \right|_{y=-\frac{L}{2}} \frac{Y^{\ell}}{\ell!}, \qquad k \in \mathbb{N}, \qquad j \in \mathbb{Z}.$$
(2.2.13)

Thus, we end up with the following problems for the near field terms:

Find
$$U \in H^1_{loc}(\mathcal{J}_j)$$
 satisfying (2.1.7) and (2.2.9) - (2.2.13). $(\mathcal{P}_j^{(k)})$

Remark 2.2.1. It is proved in [36] that a solution of (2.1.7) of at most polynomial growth admits the modal decomposition (2.2.9)–(2.2.10) with the coefficients of the polynomials $P_{j,p,\delta}^{(k)}$, $p \neq 0$, satisfying the recurrence relations (2.2.12) and the polynomials $P_{j,0,\delta}^{(k)}$ satisfying the recurrence relation (2.2.2) with p = 0.

2.2.1.2 Towards a bounded domain

We will now introduce some auxiliary objects that will permit us to reduce the problems $(\mathcal{P}_j^{(k)})$ to problems posed on the bounded domains K_j . For this, we introduce the operators T (which are in fact DtN operators). Then, we formulate a problem on the domain K_j with boundary conditions containing the operators T. As we will see, the near field problems will be reduced to problems having this form. Finally, we consider two particular cases of the problem in question that we will need in the sequel.

Operators T. Let us define the following linear operators:

$$T\varphi = \sum_{p \ge 1} p\pi(\varphi, f_p)_{L_2} f_p, \quad \forall \varphi \in H^{1/2} \left([0, 1] \right),$$

$$(T_j \varphi) \left(X \right) = \sum_{p \ge 1} \frac{p\pi}{w_j^{\mu}} \left(\varphi, f_p \left(\frac{\bullet}{w_j^{\mu}} - \frac{1}{2} \right) \right)_{L_2} f_p \left(\frac{\bullet}{w_j^{\mu}} - \frac{1}{2} \right),$$

$$\forall \varphi \in H^{1/2} \left(\left[-\frac{w_j^{\mu}}{2}, \frac{w_j^{\mu}}{2} \right] \right), \quad j \in \mathbb{Z}.$$

$$(2.2.14)$$

We will use the notation

$$\forall j \in \mathbb{Z}, \qquad T_{j,\delta} = \begin{cases} T, & \delta \in \{+, -\}, \\ T_j, & \delta = 0. \end{cases}$$

It is shown in [63] that

$$T \in \mathcal{L}\left(H^{1/2}([0,1]), H^{-1/2}([0,1])\right), T_j \in \mathcal{L}\left(H^{1/2}\left(\left[-\frac{w_j^{\mu}}{2}, \frac{w_j^{\mu}}{2}\right]\right), H^{-1/2}\left(\left[-\frac{w_j^{\mu}}{2}, \frac{w_j^{\mu}}{2}\right]\right)\right), \qquad j \in \mathbb{Z},$$

and the operators T, T_j are positive and symmetric.

A problem in the domain K_j . Let us consider the following problem for $j \in \mathbb{Z}$:

$$\begin{cases} \Delta U(X,Y) = \Phi(X,Y), & (X,Y) \in K_j, \\ \frac{\partial U}{\partial n}\Big|_{\Sigma_j^N} = 0, \\ \frac{\partial U}{\partial n}\Big|_{\Sigma_j^\delta} + T_{j,\delta} U\Big|_{\Sigma_j^\delta} = g_\delta, \quad \delta \in \{+,-,0\}. \end{cases}$$
(2.2.16)

The well-posedness result for the problems of this type can be found in [63] (Lemma 2.3.2). Adapted to our geometry, it implies that the following assertion holds.

Lemma 2.2.1. For any $j \in \mathbb{Z}$, $\Phi \in L_2(K_j)$, $g_{\pm} \in H^{-1/2}(\Sigma_j^{\pm})$, $g_0 \in H^{-1/2}(\Sigma_j^0)$ there exists a unique modulo an additive constant solution $U \in H^1(K_j)$ of the problem (2.2.22) if and only if the following compatibility condition is satisfied:

$$\langle g_+, 1 \rangle_{\Sigma_j^+} + \langle g_-, 1 \rangle_{\Sigma_j^-} + \langle g_0, 1 \rangle_{\Sigma_j^0} = \int_{K_j} \Phi.$$
 (2.2.17)

Here $\langle g_{\delta}, f \rangle_{\Sigma_{j}^{\delta}}$ denotes the duality brackets for $g_{\delta} \in H^{-1/2}(\Sigma_{j}^{\delta}), f \in H^{1/2}(\Sigma_{j}^{\delta}), \delta \in \{+, -, 0\}.$

The functions W_j^{\pm} . For any $j \in \mathbb{Z}$, we consider the following two problems:

$$\begin{cases} \left. \Delta W_{j}^{-}(X,Y) = 0, \quad (X,Y) \in K_{j}, \\ \left. \frac{\partial W_{j}^{-}}{\partial n} \right|_{\Sigma_{j}^{N}} = 0, \\ \left. \frac{\partial W_{j}^{-}}{\partial n} \right|_{\Sigma_{j,\delta}} + T_{j,\delta} W_{j}^{-} \right|_{\Sigma_{j,\delta}} = g_{j,\delta}^{(W^{-})}, \qquad \delta \in \{+,-,0\}, \end{cases}$$

$$(2.2.18)$$

with

$$g_{j,+}^{(W^-)} = 0, \qquad g_{j,-}^{(W^-)} = 1, \qquad g_{j,0}^{(W^-)} = -\frac{1}{w_j^{\mu}}, \qquad j \in \mathbb{Z}.$$

and

$$\begin{cases} \Delta W_{j}^{+}(X,Y) = 0, \quad (X,Y) \in K_{j}, \\ \left. \frac{\partial W_{j}^{+}}{\partial n} \right|_{\Sigma_{j}^{N}} = 0, \\ \left. \frac{\partial W_{j}^{+}}{\partial n} \right|_{\Sigma_{j,\delta}} + T_{j,\delta} W_{j}^{+} \right|_{\Sigma_{j,\delta}} = g_{j,\delta}^{(W^{+})}, \quad \delta \in \{+,-,0\}, \end{cases}$$

$$(2.2.19)$$

with

$$g_{j,+}^{(W^+)} = -1, \qquad g_{j,-}^{(W^+)} = 0, \qquad g_{j,0}^{(W^+)} = \frac{1}{w_j^{\mu}}, \qquad j \in \mathbb{Z}$$

The problems (2.2.18), (2.2.19) are of type (2.2.16) and one easily verifies that the compatibility condition (2.2.17) is satisfied for these problems. We denote by W_j^- (resp. W_j^+) the unique (modulo an additive constant) solution in $H^1(K_j)$ of the problem (2.2.18) (resp. (2.2.19)). Note that

$$W_{j}^{+}(X,Y) = -W_{j}^{-}(-X,Y), \qquad j \in \mathbb{Z}.$$
 (2.2.20)

Remark 2.2.2. Obviously, there are only two different functions W^- : W_0^- and W_j^- , $j \neq 0$. Similarly, there are only two different functions W^+ : W_0^+ and W_j^+ , $j \neq 0$. In other words,

$$W_i^{\pm} = W_j^{\pm}, \qquad \forall i, j \in \mathbb{Z}^*.$$
(2.2.21)

2.2.1.3 Near field problems in a bounded domain

We can now reduce the near field problems to problems set in the domains K_j . Indeed, if $U_j^{(k)}$ is a solution of (2.1.7) satisfying (2.2.8) then

$$\frac{\partial U_j^{(k)}}{\partial X}\bigg|_{\Sigma_j^+} + T U_j^{(k)}\bigg|_{\Sigma_j^+} = \left(P_{j,0,+}^{(k)}\right)' \left(\frac{w_j^{\mu}}{2}\right) + \sum_{p \in \mathbb{N}^*} \left(P_{j,p,+}^{(k)}\right)' \left(\frac{w_j^{\mu}}{2}\right) e^{-\frac{p\pi w_j^{\mu}}{2}} f_p.$$

Analogous relations can be found on the boundaries Σ_j^- and Σ_j . This yields a problem of type (2.2.16) for $U_j^{(k)}$, $j \in \mathbb{Z}$, $k \in \mathbb{N}$:

$$\begin{cases} \Delta U_{j}^{(k)}(X,Y) = \Phi_{j}^{(k-2)}(X,Y), \quad (X,Y) \in K_{j}, \\ \frac{\partial U_{j}^{(k)}}{\partial n} \bigg|_{\Sigma_{j}^{N}} = 0, \\ \frac{\partial U_{j}^{(k)}}{\partial n} \bigg|_{\Sigma_{j}^{\delta}} + T_{j,\delta} U_{j}^{(k)} \bigg|_{\Sigma_{j}^{\delta}} = g_{j,\delta}^{(k-1)}, \quad \delta \in \{+,-,0\}, \end{cases}$$
(2.2.22)

where

$$\Phi_j^{(k-2)} = -\sum_{m=0}^{k-2} \lambda^{(k-m-2)} U_j^{(m)}, \qquad k \ge 2, \qquad \Phi_j^{(-2)} = \Phi_j^{(-1)} = 0, \qquad (2.2.23)$$

$$g_{j,+}^{(k-1)} = \sum_{l=0}^{k-1} \frac{\left(w_{j}^{\mu}\right)^{l}}{2^{l}l!} \left. \frac{d^{l+1}}{ds^{l+1}} u_{j+\frac{1}{2}}^{(k-l-1)} \right|_{s=0} + \sum_{p \in \mathbb{N}^{*}} P_{j,p,+}^{(k)} \left(\frac{w_{j}^{\mu}}{2}\right) e^{-\frac{p\pi w_{j}^{\mu}}{2}} f_{p}, \qquad k \ge 1, \qquad (2.2.24)$$

$$g_{j,-}^{(k-1)} = -\sum_{l=0}^{k-1} \frac{\left(-w_{j}^{\mu}\right)^{l}}{2^{l}l!} \left. \frac{d^{l+1}}{ds^{l+1}} u_{j-\frac{1}{2}}^{(k-l-1)} \right|_{s=1} + \sum_{p \in \mathbb{N}^{*}} P_{j,p,-}^{(k)} \left(\frac{w_{j}^{\mu}}{2} \right) e^{-\frac{p\pi w_{j}^{\mu}}{2}} f_{p}, \quad k \ge 1, \quad (2.2.25)$$

$$g_{j,0}^{(k-1)} = \sum_{l=0}^{k-1} \frac{1}{l!} \left. \frac{d^{l+1}}{dy^{l+1}} u_j^{(k-l-1)} \right|_{y=-\frac{L}{2}} + \sum_{p \in \mathbb{N}^*} P_{j,p,0}^{(k)}{}'(1) e^{-\frac{p\pi}{w_j^{\mu}}} f_p\left(\frac{\bullet}{w_j^{\mu}} - \frac{1}{2}\right) \quad k \ge 1, \qquad (2.2.26)$$

$$g_{j,+}^{(-1)} = g_{j,-}^{(-1)} = g_{j,0}^{(-1)} = 0.$$
(2.2.27)

As follows from Lemma 2.2.1, the problems (2.2.22) define the near field terms in the domains K_j modulo additive constants. In order to fix these constants, we impose the following conditions:

$$\int_{\Sigma_{j}^{+}} U_{j}^{(k)} = P_{j,0,+}^{(k)} \left(\frac{w_{j}^{\mu}}{2}\right), \qquad k \in \mathbb{N}, \qquad j \in \mathbb{Z}.$$
(2.2.28)

These relations are the average trace continuity conditions for the functions $U_j^{(k)}$ (cf. (2.2.8)). Taking into account the definition (2.1.11) of the polynomials $P_{j,0,+}^{(k)}$, the relations (2.2.28) can be rewritten as

$$\int_{\Sigma_{j}^{+}} U_{j}^{(k)} = \sum_{\ell=0}^{k} \frac{\left(w_{j}^{\mu}\right)^{\ell}}{2^{\ell} \ell!} \left. \frac{d^{\ell} u_{j+\frac{1}{2}}^{(k-\ell)}}{ds^{\ell}} \right|_{s=0}, \qquad k \in \mathbb{N}, \qquad j \in \mathbb{Z}.$$
(2.2.29)

Remark 2.2.3. We choose to impose the average trace continuity condition on the boundary Σ_j^+ . Obviously, we could have imposed analogous conditions on the boundaries Σ_j^- , Σ_j^0 instead. However, it turns out that all these conditions are equivalent as soon as the far field terms satisfy the jump conditions that will be obtained in the next section (cf. Lemma (2.2.3)).

Finally, we introduce the following problem for the near field terms K_i :

Find
$$V \in H^1(K_j)$$
 satisfying (2.2.22) and (2.2.29). $(\mathcal{P}_j^{(k)})$

2.2.2 Formal derivation of the far field problem

We will now derive the problems for the far field terms. The following assertion is a direct consequence of Lemma 2.2.1.

Lemma 2.2.2. Let $j \in \mathbb{Z}$, $k \in \mathbb{N}$. Suppose that $\Phi^{(k-1)} \in L_2(K_j)$, $g_{\pm}^{(k)} \in H^{-1/2}(\Sigma_j^{\pm})$, $g_0^{(k)} \in H^{-1/2}(\Sigma_j^0)$. Suppose also that the following relation is satisfied:

$$\left(u_{j+\frac{1}{2}}^{(k)}\right)'(0) - \left(u_{j-\frac{1}{2}}^{(k)}\right)'(1) + w_{j}^{\mu}\left(u_{j}^{(k)}\right)'\left(-\frac{L}{2}\right) = \Xi_{j}^{(k-1)}, \qquad (2.2.30)$$

where

$$\Xi_{j}^{(k-1)} = -\sum_{\ell=1}^{k} \left(\frac{\left(w_{j}^{\mu}\right)^{\ell}}{2^{\ell}\ell!} \frac{d^{\ell+1}}{ds^{\ell+1}} u_{j+\frac{1}{2}}^{(k-\ell)} \Big|_{s=0} - \frac{\left(-w_{j}^{\mu}\right)^{\ell}}{2^{\ell}\ell!} \frac{d^{\ell+1}}{ds^{\ell+1}} u_{j-\frac{1}{2}}^{(k-\ell)} \Big|_{s=1} + w_{j}^{\mu} \frac{1}{\ell!} \left. \frac{d^{\ell+1}}{dy^{\ell+1}} u_{j}^{(k-\ell)} \right|_{y=-\frac{L}{2}} \right) + \int_{K_{j}} \Phi_{j}^{(k-1)}, \qquad k \ge 1, \qquad (2.2.31)$$

 $\Xi_j^{(-1)} = 0.$

Then, problem $\widetilde{\mathcal{P}}_{j}^{(k+1)}$ has a unique solution.

The relation (2.2.30) gives **non-homogeneous Kirchhoff's conditions** for the function $u^{(k)}$ provided the functions $\{u^{(m)}, U^{(m)}\}_{m=0}^{k-1}$ are known. We need to complete the problem for $u^{(k)}$ by **jump conditions** at the vertices. The trick consists in performing the following integration by parts:

$$\int_{K_{j}} \left(U_{j}^{(k)} \Delta W_{j}^{-} - W_{j}^{-} \Delta U_{j}^{(k)} \right) = \int_{\partial K_{j}} \left(\frac{\partial W^{-}{}_{j}}{\partial n} U_{j}^{(k)} - W^{-}{}_{j} \frac{\partial U_{j}^{(k)}}{\partial n} \right) \\
= \int_{\Sigma_{j}^{-}} U_{j}^{(k)} - \int_{\Sigma_{j}^{0}} \frac{U_{j}^{(k)}}{w_{j}^{\mu}} - \sum_{\delta \in \{+, -, 0\}} \left\langle g_{j,\delta}^{(k-1)}, W_{j}^{-} |_{\Sigma_{j}^{\delta}} \right\rangle_{\Sigma_{j}^{\delta}} \\
= \sum_{\ell=0}^{k} \frac{\left(-w_{j}^{\mu} \right)^{\ell}}{2^{\ell} \ell!} \left. \frac{d^{\ell} u_{j-\frac{1}{2}}^{(k-\ell)}}{ds^{\ell}} \right|_{s=1} - \sum_{\ell=0}^{k} \frac{1}{\ell!} \left. \frac{d^{\ell} u_{j}^{(k-\ell)}}{dy^{\ell}} \right|_{y=-\frac{L}{2}} - \sum_{\delta \in \{+, -, 0\}} \left\langle g_{j,\delta}^{(k-1)}, W_{j}^{-} |_{\Sigma_{j}^{\delta}} \right\rangle_{\Sigma_{j}^{\delta}}. \tag{2.2.32}$$

In the last equality we used the assumption that the near field terms satisfy the average trace continuity conditions of type (2.2.29) on the boundaries $\Sigma_{j,+}$, $\Sigma_{j,0}$. The relation (2.2.32) yields the following conditions on the jumps of the function $u^{(k)}$:

$$u_{j-\frac{1}{2}}^{(k)}(1) - u_{j}^{(k)}\left(-\frac{L}{2}\right) = \Delta_{j,-}^{(k-1)}, \qquad j \in \mathbb{Z},$$
(2.2.33)

where

$$\Delta_{j,-}^{(k-1)} = \sum_{\ell=1}^{k} \left(\frac{1}{\ell!} \left. \frac{d^{\ell} u_{j}^{(k-\ell)}}{dy^{\ell}} \right|_{y=-\frac{L}{2}} - \frac{\left(-w_{j}^{\mu}\right)^{\ell}}{2^{\ell}\ell!} \left. \frac{d^{\ell} u_{j-\frac{1}{2}}^{(k-\ell)}}{ds^{\ell}} \right|_{s=1} \right) + \sum_{\delta \in \{+,-,0\}} \left\langle g_{j,\delta}^{(k-1)}, W_{j}^{-}|_{\Sigma_{j}^{\delta}} \right\rangle_{\Sigma_{j}^{\delta}} - \int_{K_{j}} \Phi_{j}^{(k-2)} W_{j}^{-}, \qquad k \ge 1, \qquad (2.2.34)$$

 $\Delta_{j,-}^{(-1)} = 0.$

In a similar way, computing $\int_{K_j} \left(U_j^{(k)} \Delta W_j^+ - W_j^+ \Delta U_j^{(k)} \right)$ and assuming that the average

trace continuity for the near field terms on the boundaries $\Sigma_{j,0}$, $\Sigma_{j,+}$ are satisfied, we get another set of jump conditions for function $u^{(k)}$:

$$u_{j}^{(k)}\left(-\frac{L}{2}\right) - u_{j+\frac{1}{2}}^{(k)}(0) = \Delta_{j,+}^{(k-1)}, \qquad j \in \mathbb{Z},$$
(2.2.35)

where

$$\Delta_{j,+}^{(k-1)} = \sum_{l=1}^{k} \left(\frac{\left(w_{j}^{\mu} \right)^{l}}{2^{l} l!} \frac{d^{l} u_{j+\frac{1}{2}}^{(k-l)}}{ds^{l}} \bigg|_{s=0} - \frac{1}{l!} \frac{d^{l} u_{j}^{(k-l)}}{dy^{l}} \bigg|_{y=-\frac{L}{2}} \right) + \sum_{\delta \in \{+,-,0\}} \left\langle g_{j,\delta}^{(k-1)}, W_{j}^{+} |_{\Sigma_{j}^{\delta}} \right\rangle_{\Sigma_{j}^{\delta}} - \int_{K_{j}} \Phi_{j}^{(k-2)} W_{j}^{+}, \qquad k \ge 1, \qquad (2.2.36)$$

 $\Delta_{j,+}^{(-1)} = 0.$

Finally, combining (2.1.6), (2.2.30), (2.2.33), (2.2.35) we get up with the following set of problems for the far field terms $u^{(k)}$, $k \in \mathbb{N}$:

$$\begin{cases} \left(u_{j+\frac{1}{2}}^{(k)}\right)''(s) + \lambda^{(0)}u_{j+\frac{1}{2}}^{(k)}(s) = -\lambda^{(k)}u_{j+\frac{1}{2}}^{(0)}(s) - f_{j+\frac{1}{2}}^{(k-1)}(s), \quad s \in [0,1], \quad j \in \mathbb{Z}, \\ \left(u_{j}^{(k)}\right)''(y) + \lambda^{(0)}u_{j}^{(k)}(y) = -\lambda^{(k)}u_{j}^{(0)}(y) - f_{j}^{(k-1)}(y), \quad y \in \left[-\frac{L}{2}, 0\right], \quad j \in \mathbb{Z}, \\ \left(u_{j}^{(k)}\right)'(0) = 0, \quad j \in \mathbb{Z}, \quad (\mathcal{P}_{u}^{(k)}) \\ u_{j-\frac{1}{2}}^{(k)}(1) - u_{j}^{(k)}\left(-\frac{L}{2}\right) = \Delta_{j,-}^{(k)}, \quad u_{j}^{(1)}\left(-\frac{L}{2}\right) - u_{j+\frac{1}{2}}^{(k)}(0) = \Delta_{j,+}^{(k)}, \quad j \in \mathbb{Z}, \\ \left(u_{j+\frac{1}{2}}^{(k)}\right)'(0) - \left(u_{j-\frac{1}{2}}^{(k)}\right)'(1) + w_{j}^{\mu}\left(u_{j}^{(k)}\right)'\left(-\frac{L}{2}\right) = \Xi_{j}^{(k-1)}, \quad j \in \mathbb{Z}. \end{cases}$$

where

$$\begin{aligned} f_{j+\frac{1}{2}}^{(-1)} &= -\lambda^{(0)} u_{j+\frac{1}{2}}^{(0)}, \qquad f_{j}^{(-1)} &= -\lambda^{(0)} u_{j}^{(0)}, \qquad j \in \mathbb{Z}, \\ f_{j+\frac{1}{2}}^{(0)} &= 0, \qquad f_{j}^{(0)} &= 0, \qquad j \in \mathbb{Z}, \end{aligned}$$

$$(2.2.37)$$

$$f_{j+\frac{1}{2}}^{(k-1)} = \sum_{m=1}^{k-1} \lambda^{(k-m)} u_{j+\frac{1}{2}}^{(m)}, \qquad f_j^{(k-1)} = \sum_{m=1}^{k-1} \lambda^{(k-m)} u_j^{(m)}, \qquad j \in \mathbb{Z}, \quad k \ge 2.$$
(2.2.38)

2.2.3 Order 0

Before studying the well-posedness of the near-field and the far field problems at any order formulated in the previous sections, let us look at the order 0.

Far field problem. Putting k = 0 in $(\mathcal{P}_u^{(k)})$, we get

$$\begin{cases} \left(u_{j+\frac{1}{2}}^{(0)}\right)''(s) + \lambda^{(0)}u_{j+\frac{1}{2}}^{(0)}(s) = 0, & s \in [0,1], & j \in \mathbb{Z}, \\ \left(u_{j}^{(0)}\right)''(y) + \lambda^{(0)}u_{j}^{(0)}(y) = 0, & y \in \left[-\frac{L}{2},0\right], & j \in \mathbb{Z}, \\ \left(u_{j}^{(0)}\right)'(0) = 0, & j \in \mathbb{Z}, \\ u_{j-\frac{1}{2}}^{(0)}(1) = u_{j}^{(0)}\left(-\frac{L}{2}\right) = u_{j+\frac{1}{2}}^{(0)}(0), & j \in \mathbb{Z}, \\ \left(u_{j+\frac{1}{2}}^{(0)}\right)'(0) - \left(u_{j-\frac{1}{2}}^{(0)}\right)'(1) + w_{j}^{\mu}\left(u_{j}^{(0)}\right)'\left(-\frac{L}{2}\right) = 0, & j \in \mathbb{Z}. \end{cases}$$
(2.2.39)

This is exactly the eigenvalue problem for the operator A_s^{μ} (cf. (1.3.6)) which has been studied in Section 1.3.2. From now on $\lambda^{(0)}$ will stand for an eigenvalue of the operator A_s^{μ} and $u^{(0)}$ for the corresponding eigenfunction (all the eigenvalues of the operator A_s^{μ} are simple). To fix the eigenfunction in a unique way, we impose the condition $\mathbf{u}_0 = 1$. The expressions for the eigenfunction $u^{(0)}$ have been found in (1.3.41), (1.3.42), (1.3.53), (1.3.48):

$$u_{j+\frac{1}{2}}^{(0)}(s) = \mathbf{u}_j \frac{\sin\left(\sqrt{\lambda^{(0)}}(1-s)\right)}{\sin\sqrt{\lambda^{(0)}}} + \mathbf{u}_{j+1} \frac{\sin\left(\sqrt{\lambda^{(0)}}s\right)}{\sin\sqrt{\lambda^{(0)}}}, \quad s \in [0,1], \quad \forall j \in \mathbb{Z},$$
(2.2.40)

$$u_j^{(0)}(y) = \mathbf{u}_j \frac{\cos\left(\sqrt{\lambda^{(0)}}y\right)}{\cos\left(\sqrt{\lambda^{(0)}}L/2\right)}, \qquad y \in \left[-\frac{L}{2}, \frac{L}{2}\right], \quad \forall j \in \mathbb{Z},$$
(2.2.41)

where

$$\mathbf{u}_j = r^{|j|}, \qquad j \in \mathbb{Z}, \qquad r = r(\sqrt{\lambda^{(0)}}).$$
 (2.2.42)

We remind that in this chapter the spectral parameter is denoted by λ , whereas in Chapter 1 it was denoted by λ^2 . This is the reason of $\sqrt{\lambda^{(0)}}$ appearing in the expressions for the eigenfunction.

Near field problems. Consider now the near field problems $(\mathcal{P}_j^{(k)})$ for k = 0. The relations (2.2.11), (2.2.13) take the form

$$P_{j,p,\delta}^{(0)}(s) = c_{j,p,\delta,0}^{(0)}, \qquad \delta \in \{+, -, 0\}, \quad p \in \mathbb{N}^*, \quad j \in \mathbb{Z},$$

$$P_{j,0,+}^{(0)}(X) = u_{j+\frac{1}{2}}^{(0)}(0), \qquad P_{j,0,-}^{(0)}(X) = u_{j-\frac{1}{2}}^{(0)}(1), \qquad P_{j,0,0}^{(0)}(Y) = u_{j}^{(0)}\left(-\frac{L}{2}\right), \qquad j \in \mathbb{Z}.$$

Taking into account that $u_{j+\frac{1}{2}}^{(0)}(0) = u_{j-\frac{1}{2}}^{(0)}(1) = u_j^{(0)}\left(-\frac{L}{2}\right) = \mathbf{u}_j$, we can rewrite the problems $(\mathcal{P}_j^{(k)})$ for k = 0 as

$$\begin{cases}
\left. \Delta U_{j}^{(0)}(X,Y) = 0, \quad (X,Y) \in \mathcal{J}_{j}, \\
\left. \frac{\partial U_{j}^{(0)}}{\partial n} \right|_{\partial \mathcal{J}_{j}} = 0, \\
U_{j}^{(0)}(X,Y) = \mathbf{u}_{j}^{(0)} + \sum_{p \in \mathbb{N}^{*}} c_{j,p,\pm,0}^{(0)} e^{\mp p\pi X} f_{p}(Y), \quad (X,Y) \in B_{j}^{\pm}, \\
U_{j}^{(0)}(X,Y) = \mathbf{u}_{j}^{(0)} + \frac{1}{\sqrt{w_{j}^{\mu}}} \sum_{p \in \mathbb{N}^{*}} c_{j,p,0,0}^{(0)} e^{-p\pi Y/w_{j}^{\mu}} f_{p}\left(\frac{X}{w_{j}^{\mu}} - \frac{1}{2}\right), \quad (X,Y) \in B_{j}^{0}.
\end{cases}$$
(2.2.43)

Consequently, the constants

$$U_j^{(0)}(X,Y) = \mathbf{u}_j^{(0)}, \qquad (X,Y) \in \mathcal{J}_j, \qquad \forall j \in \mathbb{Z},$$
(2.2.44)

are obviously solutions of (2.2.43) with

$$c_{j,p,\delta,0}^{(0)} = 0, \qquad \delta \in \{+, -, 0\}, \qquad p \in \mathbb{N}^*, \qquad j \in \mathbb{Z}.$$
 (2.2.45)

2.2.4 Well-posedness of the near field problems

In this section we suppose that the far fields up to the order k are constructed, i.e., there exist functions $\{u^{(m)}, 0 \leq m \leq k\}$ satisfying the problems $\{(\mathcal{P}^{(m)}), 0 \leq m \leq k\}$. The near field terms are supposed constructed up to the order k-1, i.e., there exist functions $\{U_j^{(m)}, j \in \mathbb{Z}, 0 \leq m \leq k-1\}$ that solve the problems $\{(\mathcal{P}_j^{(m)}), j \in \mathbb{Z}, 0 \leq m \leq k-1\}$.

We also suppose that at order k the near field terms are constructed on a bounded domain, i.e., there exist functions $\left\{V_{j}^{(k)}\right\}_{j\in\mathbb{Z}}$ solving the problems $\left\{(\widetilde{\mathcal{P}}_{j}^{(k)})\right\}_{j\in\mathbb{Z}}$. We will show that under these assumptions the near filed terms at order k can be continued to the whole infinite junctions, i.e., there exist functions $\left\{U_{j}^{(k)}\right\}_{j\in\mathbb{Z}}$ solving the problems $\left\{(\mathcal{P}_{j}^{(k)})\right\}_{j\in\mathbb{Z}}$.

We start by some auxiliary assertions. The following lemma that establishes the equivalence between the average trace continuity conditions imposed on different boundaries of the domain K_j mentioned in Remark 2.2.3.

Lemma 2.2.3. Suppose that there exist functions $\{u^{(m)}, 0 \leq m \leq k\}$ satisfying the problems $\{(\mathcal{P}^{(m)}), 0 \leq m \leq k\}$ as well as functions $\{U_j^{(m)}, j \in \mathbb{Z}, 0 \leq m \leq k-1\}$ satisfying the problems $\{(\mathcal{P}_j^{(m)}), j \in \mathbb{Z}, 0 \leq m \leq k-1\}$ for some $k \in \mathbb{N}$. Suppose also that the function V satisfies the problem (2.2.22) for some $j \in \mathbb{Z}$. Then, the following three relations are equivalent:

$$(i) \int_{\Sigma_{j}^{+}} V = P_{j,0,+}^{(k)} \left(\frac{w_{j}^{\mu}}{2}\right); \qquad (ii) \int_{\Sigma_{j}^{-}} V = P_{j,0,-}^{(k)} \left(-\frac{w_{j}^{\mu}}{2}\right); \qquad (iii) \int_{\Sigma_{j}^{0}} \frac{V}{w_{j}^{\mu}} = P_{j,0,0}^{(k)} \left(1\right).$$

Proof. Let us show that the relations (ii) implies (iii). All the other implications can be shown in a similar way. If W_i^- is the function introduced in Section 2.2.1.2, then

$$\int_{K_j} \left(V \Delta W_j^- - W_j^- \Delta V \right) = \int_{\partial K_j} \left(\frac{\partial W_j^-}{\partial n} V - W_j^- \frac{\partial V}{\partial n} \right) = \int_{\Sigma_j^-} V - \int_{\Sigma_j^0} \frac{V}{w_j^\mu} - \sum_{\delta \in \{+, -, 0\}} \left\langle g_{j,\delta}^{(k-1)}, W_j^- |_{\Sigma_j^\delta} \right\rangle_{\Sigma_j^\delta}$$

which implies

$$\begin{split} \int_{\Sigma_{j}^{0}} \frac{V}{w_{j}^{\mu}} &= \int_{\Sigma_{j}^{-}} V - \sum_{\delta \in \{+,-,0\}} \left\langle g_{j,\delta}^{(k-1)}, W_{j}^{-}|_{\Sigma_{j}^{\delta}} \right\rangle_{\Sigma_{j}^{\delta}} + \int_{K_{j}} \Phi_{j}^{(k-2)} W_{j}^{-} \\ &= \sum_{\ell=0}^{k} \frac{\left(-w_{j}^{\mu}\right)^{\ell}}{2^{\ell}\ell!} \left. \frac{d^{\ell} u_{j-\frac{1}{2}}^{(k-\ell)}}{ds^{\ell}} \right|_{s=1} - \sum_{\delta \in \{+,-,0\}} \left\langle g_{j,\delta}^{(k-1)}, W_{j}^{-}|_{\Sigma_{j}^{\delta}} \right\rangle_{\Sigma_{j}^{\delta}} + \int_{K_{j}} \Phi_{j}^{(k-2)} W_{j}^{-}. \end{split}$$

In the last equality we used the assumption (ii) and the definition (2.2.13) of the polynomial $P_{j,0,-}^{(k)}$. Taking into account the definition (2.2.34) of $\Delta_{j,-}^{(k-1)}$, we get

$$\int_{\Sigma_j^0} \frac{V}{w_j^{\mu}} = \sum_{\ell=1}^k \frac{1}{\ell!} \left. \frac{d^{\ell} u_j^{(k-\ell)}}{dy^{\ell}} \right|_{y=-\frac{L}{2}} - \Delta_{j,-}^{(k-1)} + u_{j-\frac{1}{2}}^{(k)}(1).$$

Finally, using the fact that $u^{(k)}$ satisfies the jump condition (2.2.33), we obtain

$$\int_{\Sigma_{j}^{0}} \frac{V}{w_{j}^{\mu}} = \sum_{\ell=0}^{k} \frac{1}{\ell!} \left. \frac{d^{\ell} u_{j}^{(k-\ell)}}{dy^{\ell}} \right|_{y=-\frac{L}{2}} = P_{j,0,0}^{(k)}\left(1\right).$$

The same computation can be used to show that (iii) implies (ii). The equivalence between (i) and (iii) can be proved using an analogous argument with W_j^- replaced by W_j^+ .

The following assertion concerns the behaviour of the coefficients $c_{j,p\delta,l}^{(k)}$ as $p \to \infty$ assuming the existence of solutions of the near field problems $(\mathcal{P}_j^{(k)})$.

Lemma 2.2.4. Suppose that for some $j \in \mathbb{Z}$ there exist functions $U_j^{(m)} \in H^1_{loc}(\mathcal{J}_j)$, $0 \leq m \leq n$, which are solutions of the problems $(\mathcal{P}_j^{(m)})$ for $0 \leq m \leq n$. Then,

$$\sum_{p \in \mathbb{N}^*} p\left(c_{j,p,\pm,\ell}^{(m)}\right)^2 e^{-p\pi w_j^{\mu}} < \infty, \qquad \sum_{p \in \mathbb{N}^*} p\left(c_{j,p,0,\ell}^{(m)}\right)^2 e^{-\frac{2p\pi}{w_j^{\mu}}} < \infty, \quad 0 \leqslant \ell \leqslant \left[\frac{m}{2}\right], \quad 0 \leqslant m \leqslant n$$

$$(2.2.46)$$

Proof. We will give the proof for $\delta = +$, the proof for $\delta \in \{-, 0\}$ being analogous. The proof is done by induction in m. Indeed, for m = 0 the result holds due to (2.2.45). Suppose that $1 \leq q \leq n$ and the relations (2.2.46) are satisfied for $0 \leq m \leq q - 1$. Let us show that they are satisfied for m = q as well. First, the recurrence relation for the coefficients c (2.2.12) implies that (2.2.46) is satisfied for m = q, $1 \leq \ell \leq \left[\frac{q}{2}\right]$. Then, due to (2.2.9) for m = q and the fact that $U_i^{(q)} \in H^1_{loc}(\mathcal{J}_j)$ we have

$$\sum_{p \in \mathbb{N}^*} P_{j,p,+}^{(q)} \left(\frac{w_j^{\mu}}{2}\right) e^{-\frac{p\pi w_j^{\mu}}{2}} f_p \in H^{1/2} \left(\Sigma_j^+\right).$$

Taking into account (2.2.11) and (2.2.46) for m = q, $1 \leq \ell \leq \left\lfloor \frac{q}{2} \right\rfloor$ we conclude that

$$\sum_{p \in \mathbb{N}^*} c_{j,p,+,0}^{(q)} e^{-\frac{p\pi w_j^{\mu}}{2}} f_p \in H^{1/2}\left(\Sigma_j^+\right),$$

which means that (2.2.46) is also satisfied for m = q, $\ell = 0$.

We will now pass to the construction of a continuation of the near field terms from a bounded domain to the infinite junction.

Proposition 2.2.1. Suppose that there exist functions $\{u^{(m)}, 0 \le m \le k\}$ satisfying the problems $\{(\mathcal{P}^{(m)}), 0 \le m \le k\}$ as well as functions $\{U_j^{(m)}, j \in \mathbb{Z}, 0 \le m \le k-1\}$ satisfying the problems $\{(\mathcal{P}_j^{(m)}), j \in \mathbb{Z}, 0 \le m \le k-1\}$ for some $k \in \mathbb{N}^*$. Suppose also that the function $V_j^{(k)} \in H^1(K_j)$ solves the problem $(\widetilde{\mathcal{P}}_j^{(k)})$ for some $j \in \mathbb{Z}$. Then, there exists a function $U_j^{(k)} \in H^1_{loc}(\mathcal{J}_j)$ solving the problem $(\mathcal{P}_j^{(k)})$ and such that $U_j^{(k)}|_{K_j} = V_j^{(k)}$.

Proof. The proof consists of several steps. First, we define functions in the bands B_j^{δ} that will be shown to be the extensions of V_j . These function are constructed in such a way that the have the form (2.2.8), (2.2.9), (2.2.10). In these relations everything is defined by the previous orders except from the coefficients $c_{j,p,\delta,0}^{(k)}$. These coefficients are found from the traces of V_j on the boundaries Σ_j^{δ} . Then, we show that the constructed functions in the bands B_j^{δ} belong to $H_{loc}^1(B_j^{\delta})$, that their traces on Σ_j^{δ} coincide with those of V_j , that they solve the Laplacian equation (first line of (2.1.7)) in B_j^{δ} and finally that their normal derivatives on Σ_j^{δ} are the opposite of those of V_j and the homogeneous boundary conditions are satisfied on the other boundaries. This permits to conclude that the constructed extension solves $(\mathcal{P}_j^{(k)})$.

Construction of an extension: Let us denote by $\varphi_{j,\delta}^{(k)}$ the traces of the function $V_j^{(k)}$ on the boundaries Σ_j^{δ} :

$$\varphi_{j,\delta}^{(k)} = V_j^{(k)} \Big|_{\Sigma_j^{\delta}}, \qquad \varphi_{j,\delta}^{(k)} \in H^{1/2}\left(\Sigma_j^{\delta}\right), \qquad \delta \in \{+, -, 0\}.$$
(2.2.47)

These traces can be decomposed in Fourier series:

$$\varphi_{j,\pm}^{(k)} = \sum_{p \in \mathbb{N}} \varphi_{j,p,\pm}^{(k)} f_p, \qquad \varphi_{j,0}^{(k)} = \sum_{p \in \mathbb{N}} \varphi_{j,p,0}^{(k)} f_p\left(\frac{\bullet}{w_j^{\mu}} - \frac{1}{2}\right), \qquad (2.2.48)$$

and due to (2.2.47) we have

$$\sum_{p \in \mathbb{N}} p\left(\varphi_{j,p,\delta}^{(k)}\right)^2 < \infty, \qquad \delta \in \{+,-,0\}.$$
(2.2.49)

Note that the average trace condition (2.2.29) for $V_j^{(k)}$ together with Lemma 2.2.3 which gives analogous conditions on the other boundaries, imply that

$$\varphi_{j,0,\pm}^{(k)} = P_{j,0,\pm}^{(k)} \left(\pm \frac{w_j^{\mu}}{2} \right), \qquad \varphi_{j,0,0}^{(k)} = P_{j,0,0}^{(k)} \left(1 \right) \qquad j \in \mathbb{Z}.$$
(2.2.50)

We construct the functions $\widetilde{U}_{j,\delta}^{(k)}$ in the bands B_j^{δ} for $\delta \in \{+, -, 0\}$ as follows:

$$\widetilde{U}_{j,\pm}^{(k)}(X,Y) = P_{j,0,\pm}^{(k)}(X) + \sum_{p \in \mathbb{N}^*} \xi_{j,p,\pm}^{(k)}(X,Y), \qquad (X,Y) \in B_j^{\pm},$$
(2.2.51)

$$\widetilde{U}_{j,0}^{(k)}(X,Y) = P_{j,0,0}^{(k)}(Y) + \sum_{p \in \mathbb{N}^*} \xi_{j,p,0}^{(k)}(X,Y), \qquad (X,Y) \in B_j^0,$$
(2.2.52)

$$\xi_{j,p,\pm}^{(k)}(X,Y) = \left(\varphi_{j,p,\pm}^{(k)} e^{\frac{p\pi w_j^{\mu}}{2}} + \sum_{\ell=1}^{\left\lfloor\frac{k}{2}\right\rfloor} c_{j,p,\pm,\ell}^{(k)} \left((\pm X)^{\ell} - \left(\frac{w_j^{\mu}}{2}\right)^{\ell}\right)\right) e^{\mp p\pi X} f_p(Y), \quad p \in \mathbb{N}^*,$$
(2.2.53)

$$\xi_{j,p,0}^{(k)}(X,Y) = \left(\varphi_{j,p,0}^{(k)}e^{\frac{p\pi}{w_j^{\mu}}} + \frac{1}{\sqrt{w_j^{\mu}}}\sum_{\ell=1}^{\left\lfloor\frac{k}{2}\right\rfloor}c_{j,p,0,\ell}^{(k)}\left(Y^{\ell}-1\right)\right)e^{-\frac{p\pi Y}{w_j^{\mu}}}f_p\left(\frac{X}{w_j^{\mu}}-\frac{1}{2}\right), \quad p \in \mathbb{N}^*,$$
(2.2.54)

where the polynomials $P_{j,0,\delta}^{(k)}$ are defined in (2.2.13) and the coefficients $c_{j,p,\delta,\ell}^{(k)}$ are defined in (2.2.12). We note that all the objects appearing in the definition of $P_{j,0,\delta}^{(k)}$ and $c_{j,p,\delta,\ell}^{(k)}$, i.e., the far field terms up to the order k and the coefficients c for the previous orders, are already constructed by assumption. We will perform the rest of the analysis for the function $\tilde{U}_{j,+}^{(k)}$, the analysis for the functions $\tilde{U}_{j,0}^{(k)}$, $\tilde{U}_{j,0}^{(k)}$ being analogous. The function $\widetilde{U}_{j,+}^{(k)}$ belongs to $H_{loc}^1\left(B_j^+\right)$: We have:

$$\left\| \xi_{j,p,+}^{(k)} \right\|_{H^1\left(B_j^+\right)}^2 \leq C(k) \left(p^2 \left(\varphi_{j,p,+}^{(k)} \right)^2 e^{p\pi w_j^{\mu}} \int_{\frac{w_j^{\mu}}{2}}^{\infty} e^{-2p\pi X} dX + \sum_{\ell=1}^{\left\lfloor \frac{k}{2} \right\rfloor} p^2 \left(c_{j,p,+,\ell}^{(k)} \right)^2 \int_{\frac{w_j^{\mu}}{2}}^{\infty} X^{2\ell} e^{-2p\pi X} dX \right).$$

Using the fact that

$$\int_{\frac{w_j^{\mu}}{2}}^{\infty} X^{2\ell} e^{-2p\pi X} dX \leqslant C(\ell) p^{-1} e^{-p\pi w_j^{\mu}},$$

we get

$$\left\|\xi_{j,p,+}^{(k)}\right\|_{H^1\left(B_j^+\right)}^2 \leqslant C(k) \left(p\left(\varphi_{j,p,+}^{(k)}\right)^2 + \sum_{\ell=1}^{\left\lfloor\frac{k}{2}\right\rfloor} p\left(c_{j,p,+,\ell}^{(k)}\right)^2 e^{-p\pi w_j^{\mu}}\right)$$

As follows from Lemma 2.2.4, the inequalities (2.2.46) hold for $0 \leq m \leq k-1$. Taking into account the definition (2.2.12) of the coefficients $c_{j,p,+,\ell}^{(k)}$ as well as the relation (2.2.49), we conclude that

$$\sum_{p \in \mathbb{N}^*} \left\| \xi_{j,p}^{(k)} \right\|_{H^1\left(B_j^+\right)}^2 < \infty, \tag{2.2.55}$$

and consequently, $\widetilde{U}_{j,+}^{(k)} \in H^1_{loc}\left(B_j^+\right), \ j \in \mathbb{Z}.$

Trace continuity: We can now find the trace of the function $\widetilde{U}_{j,+}^{(k)}$ on the surface Σ_j^+ . It follows from the definition of the function $\widetilde{U}_{j,+}^{(k)}$ ((2.2.51)–(2.2.53)) that

$$\widetilde{U}_{j,+}^{(k)}\Big|_{\Sigma_{j}^{+}} = P_{j,0,+}^{(k)}\left(\frac{w_{j}^{\mu}}{2}\right) + \sum_{p \in \mathbb{N}^{*}} \varphi_{j,p,+}^{(k)} f_{p}.$$

Comparing it with the trace of the function $V_j^{(k)}$ ((2.2.47), (2.2.48), (2.2.50)), we see that

$$\widetilde{U}_{j,+}^{(k)}\Big|_{\Sigma_j^+} = V_j^{(k)}\Big|_{\Sigma_j^+}, \qquad j \in \mathbb{Z}.$$
(2.2.56)

The Laplacian equation for $\widetilde{U}_{j,+}^{(k)}$: Using the relation (2.2.12) between the coefficients c of the polynomials P, we find from (2.2.53)

$$\Delta\xi_{j,p,+}^{(k)} = -\sum_{m=0}^{k-2} \lambda^{(m)} P_{j,p,+}^{(k-2-m)}(X) e^{-p\pi X} f_p(Y), \qquad (X,Y) \in B_j^+, \qquad p \in \mathbb{N}^+.$$
(2.2.57)

Since

$$\sum_{p=1}^{N} \xi_{j,p}^{(k)} \xrightarrow[L_2(B_j^+)]{} \widetilde{U}_{j,+}^{(k)} - P_{j,0,+}^{(k)}, \qquad N \to \infty,$$

...

(as follows from (2.2.51), (2.2.55)), and

$$\sum_{p=1}^{N} \Delta \xi_{j,p,+}^{(k)} \xrightarrow{L_2(B_j^+)} -\sum_{m=0}^{k-2} \lambda^{(m)} \left(U_j^{(k-2-m)} - P_{j,0,+}^{(k-2-m)} \right), \qquad N \to \infty,$$

(as follows from (2.2.57) and the relations (2.2.9) satisfied by the functions $U_j^{(k-2-m)}$, $0 \leq m \leq k-2$), we can conclude that $\Delta \widetilde{U}_{j,+}^{(k)} \in L_2(B_j^+)$ and

$$\Delta \widetilde{U}_{j,+}^{(k)} = \left(P_{j,0,+}^{(k)}\right)'' - \sum_{m=0}^{k-2} \lambda^{(m)} \left(U_j^{(k-2-m)} - P_{j,0,+}^{(k-2-m)}\right) = -\sum_{m=0}^{k-2} \lambda^{(m)} U_j^{(k-2-m)}$$

The last equality follows from the definition (2.2.13) of the polynomials $P_{j,0,+}^{(k)}$ as well as the fact that the far field terms verify the equations (2.1.6) (as was mentioned above, this implies that the polynomials $P_{j,0,+}^{(k)}$ satisfy the recurrence relation (2.2.2) for p = 0).

Continuity of the normal derivative: We have $\partial_n \widetilde{U}_{j,+}^{(k)} \in H^{-1/2} \left(\partial B_j^+ \right)$ and $\partial_n \widetilde{U}_{j,+}^{(k)} = 0$ on $\partial B_j^+ \setminus \Sigma_j^+$,

$$\partial_n \widetilde{U}_{j,+}^{(k)} \Big|_{\Sigma_j^+} = -\left(P_{j,0,+}^{(k)}\right)' + \sum_{p \in \mathbb{N}^*} p \pi \varphi_{j,p,+}^{(k)} f_p - \sum_{p \in \mathbb{N}^*} \sum_{\ell=1}^{\left\lfloor \frac{k}{2} \right\rfloor} c_{j,p,+,\ell}^{(k)} \ell\left(\frac{w_j^{\mu}}{2}\right)^{\ell-1} e^{-\frac{p \pi w_j^{\mu}}{2}} f_p = T \varphi_{j,+}^{(k)} - g_{j,+}^{(k-1)}.$$

The last equality follows from the definitions of the function $g_{j,+}^{(k-1)}$ and the operator T (cf. (2.2.24), (2.2.14)) as well as the definition of the polynomials P (cf. (2.2.13), (2.2.11)). Finally, taking into account the fact that $V_j^{(k)}$ satisfies the problem (2.2.22), we get

$$\partial_n \widetilde{U}_{j,+}^{(k)}\Big|_{\Sigma_j^+} = - \left.\partial_n V_j^{(k)}\right|_{\Sigma_j^+}.$$

Conclusion: The function $U_j^{(k)}$ constructed as

$$U_{j}^{(k)}(X,Y) = \begin{cases} V_{j}^{(k)}, & (X,Y) \in K_{j}, \\ \widetilde{U}_{j,\delta}^{(k)}, & (X,Y) \in B_{j}^{\delta}, \quad \delta \in \{+,-,0\}, \end{cases}$$
(2.2.58)

satisfies the problem $(\mathcal{P}_{i}^{(k)})$.

Remark 2.2.4. As follows from (2.2.51)–(2.2.54), the coefficients $c_{p,j,\delta,0}^{(k)}$ are given by the following relation:

$$c_{j,p,\pm,0}^{(k)} = \varphi_{j,p,\pm}^{(k)} e^{\frac{p\pi w_j^{\mu}}{2}} - \sum_{\ell=1}^{\left[\frac{k}{2}\right]} c_{j,p,\pm,\ell}^{(k)} \left(\frac{w_j^{\mu}}{2}\right)^{\ell}, \qquad c_{j,p,0,0}^{(k)} = \varphi_{j,p,0}^{(k)} e^{\frac{p\pi}{w_j^{\mu}}} - \sum_{\ell=1}^{\left[\frac{k}{2}\right]} c_{j,p,0,\ell}^{(k)}, \qquad j \in \mathbb{Z}, \qquad p \in \mathbb{N}^*.$$
(2.2.59)

To find them, we need to know the coefficients c at the orders up to k-1 and the function $V_j^{(k)}$ (cf. (2.2.47)–(2.2.48)).

Remark 2.2.5. In view of Remark 2.2.1, the uniqueness of the solution of the near field problem $(\widetilde{\mathcal{P}}_{j}^{(k)})$ in K_{j} implies the uniqueness of the solution of the problem (2.2.2) of at most polynomial growth satisfying the matching conditions (2.2.13).

2.2.5 Well-posedness of the far field problem

In this section we prove the following well-posedness result for the problem $(\mathcal{P}_u^{(k)})$ with $k \ge 1$:

$$\begin{pmatrix} u_{j+\frac{1}{2}}^{(k)} \end{pmatrix}''(s) + \lambda^{(0)} u_{j+\frac{1}{2}}^{(k)}(s) = -\lambda^{(k)} u_{j+\frac{1}{2}}^{(0)}(s) - f_{j+\frac{1}{2}}^{(k-1)}(s), \quad s \in [0,1], \quad j \in \mathbb{Z}, \\ (u_{j}^{(k)})''(y) + \lambda^{(0)} u_{j}^{(k)}(y) = -\lambda^{(k)} u_{j}^{(0)}(y) - f_{j}^{(k-1)}(y), \quad y \in \left[-\frac{L}{2}, 0\right], \quad j \in \mathbb{Z}, \\ (u_{j}^{(k)})'(0) = 0, \quad j \in \mathbb{Z}, \\ u_{j-\frac{1}{2}}^{(k)}(1) - u_{j}^{(k)}\left(-\frac{L}{2}\right) = \Delta_{j,-}^{(k)}, \quad u_{j}^{(1)}\left(-\frac{L}{2}\right) - u_{j+\frac{1}{2}}^{(k)}(0) = \Delta_{j,+}^{(k)}, \quad j \in \mathbb{Z}, \\ (u_{j+\frac{1}{2}}^{(k)})'(0) - \left(u_{j-\frac{1}{2}}^{(k)}\right)'(1) + w_{j}^{\mu}\left(u_{j}^{(k)}\right)'\left(-\frac{L}{2}\right) = \Xi_{j}^{(k-1)}, \quad j \in \mathbb{Z}.$$

Proposition 2.2.2. For $k \in \mathbb{N}^*$, $f^{(k-1)} \in L_2^{\mu}(G_{\mathcal{C}})$, $\left\{\Delta_{j,\pm}^{(k-1)}\right\}_{j\in\mathbb{Z}}$, $\left\{\Xi_j^{(k-1)}\right\}_{j\in\mathbb{Z}} \in \ell_2(\mathbb{Z})$, the problem $(\mathcal{P}_u^{(k)})$ is has a solution in $L_2(G_{\mathcal{C}})$ if and only if

$$\lambda^{(k)} = \left\| u^{(0)} \right\|_{L_2^{\mu}(G_{\mathcal{C}})}^{-2} \left(\sum_{j \in \mathbb{Z}} \widetilde{\Xi}_j^{(k-1)} \mathbf{u}_j^{(0)} - \left(f^{(k-1)}, u^{(0)} \right)_{L_2^{\mu}(G_{\mathcal{C}})} \right),$$
(2.2.60)

where

$$\widetilde{\Xi}_{j}^{(k-1)} = \Xi_{j}^{(k-1)} - \frac{\sqrt{\lambda^{(0)}}}{\sin\sqrt{\lambda^{(0)}}} \left(\Delta_{j+1,-}^{(k-1)} - \Delta_{j-1,+}^{(k-1)} + \cos\sqrt{\lambda^{(0)}} \left(\Delta_{j,+}^{(k-1)} - \Delta_{j,-}^{(k-1)} \right) \right), \qquad j \in \mathbb{Z}.$$
(2.2.61)

The solution is unique in $L_2(G_c)/span\left\{ \left. u^{(0)} \right|_{G_c} \right\}$.

Remark 2.2.6. As we see, the far field terms are not defined in a unique way. We will explain in the proof of Lemma 2.3.1 how we fix them. This choice is completely arbitrary and could be done in a different way. We could show by an explicit computation that the choice of the far field terms does not influence the values $\lambda^{(k)}$ defined by the relation (2.2.60). However, this will be justified a posteriori by proving Theorem 1.4.2.

To prove Proposition 2.2.2, we start with the following assertion for the problem with zero jump conditions at the vertices.

Proposition 2.2.3. Let $\lambda^{(0)}$ be an eigenvalue of the operator A_s^{μ} and $u^{(0)}$ the corresponding eigenfunction. For $f \in L_2^{\mu}(G_c)$, $\{\Xi_j\}_{j \in \mathbb{Z}} \in \ell_2(\mathbb{Z})$ the problem

$$\begin{cases} u_{j+\frac{1}{2}}''(s) + \lambda^{(0)} u_{j+\frac{1}{2}}(s) = f_{j+\frac{1}{2}}(s), \quad s \in [0,1], \quad j \in \mathbb{Z}, \\ u_{j}''(y) + \lambda^{(0)} u_{j}(y) = f_{j}(y), \quad y \in \left[-\frac{L}{2}, 0\right], \quad j \in \mathbb{Z}, \\ u_{j}'(0) = 0, \quad j \in \mathbb{Z}, \\ u_{j-\frac{1}{2}}(1) = u_{j}\left(-\frac{L}{2}\right) = u_{j+\frac{1}{2}}(0), \quad j \in \mathbb{Z}, \\ u_{j+\frac{1}{2}}'(0) - u_{j-\frac{1}{2}}'(1) + w_{j}^{\mu}u_{j}'\left(-\frac{L}{2}\right) = \Xi_{j}, \quad j \in \mathbb{Z}, \end{cases}$$

$$(2.2.62)$$

has a solution in $L_2(G_{\mathcal{C}})$ if and only if the following compatibility condition is verified:

$$(f, u^{(0)})_{L_2^{\mu}(G_{\mathcal{C}})} + \sum_{j \in \mathbb{Z}} \Xi_j \mathbf{u}_j^{(0)} = 0.$$

The solution is unique in $L_2(G_c)/span\left\{ u^{(0)} \big|_{G_c} \right\}$.

Proof. The uniqueness of the solution in $L_2(G_c)/span\left\{ u^{(0)} \Big|_{G_c} \right\}$ is obvious. The proof of the existence will be analogous to the proof of Theorem 4.10 from [51]. Passing to the variational formulation of the problem (2.2.62) and considering it as a problem in the symmetric subspace $L_{2,s}^{\mu}(G)$ on the whole graph G, we will search a function $u \in H_s^1(G)$ such that

$$-(u',v')_{L_{2}^{\mu}(G)} + \lambda^{(0)}(u,v)_{L_{2}^{\mu}(G)} = (f,v)_{L_{2}^{\mu}(G)} + 2\sum_{j\in\mathbb{Z}}\Xi_{j}\mathbf{v}_{j}, \qquad \forall v\in H_{s}^{1}(G), \quad (2.2.63)$$

where $\mathbf{v}_j = v\left(j, -\frac{L}{2}\right)$. Due to the estimate

$$\left| - (u', v')_{L_{2}^{\mu}(G)} + \lambda^{(0)} (u, v)_{L_{2}^{\mu}(G)} \right| \leq \max\left\{ 1, \lambda^{(0)} \right\} \|u\|_{H^{1}(G)} \|v\|_{H^{1}(G)}, \qquad \forall u, v \in H^{1}_{s}(G),$$

we can then define a bounded linear self-adjoint operator $A : H^1_s(G) \to H^1_s(G)$ such that

$$(Au, v)_{H^1(G)} = -(u', v')_{L_2^{\mu}(G)} + \lambda^{(0)}(u, v)_{L_2^{\mu}(G)}, \qquad \forall u, v \in H_s^1(G).$$
(2.2.64)

In the same manner, we have the estimate

$$\left| (f,v)_{L_{2}^{\mu}(G)} + 2\sum_{j \in \mathbb{Z}} \Xi_{j} \mathbf{v}_{j} \right| \leqslant \left(\|f\|_{L_{2}^{\mu}(G)} + C\left(\sum_{j \in \mathbb{Z}} \Xi_{j}^{2}\right)^{1/2} \right) \|v\|_{H^{1}(G)}, \qquad \forall v \in H_{s}^{1}(G),$$

with some constant C depending only on the geometry of the graph. Hence, there exists a unique function $F \in H^1_s(G)$ such that

$$(F, v)_{H^1(G)} = (f, v)_{L_2^{\mu}(G)} + 2\sum_{j \in \mathbb{Z}} \Xi_j \mathbf{v}_j, \qquad \forall v \in H_s^1(G).$$
(2.2.65)

Thus, combining (2.2.63), (2.2.64) and (2.2.65) we end up with the equation

$$Au = F, (2.2.66)$$

which is equivalent to the problem (2.2.62). Indeed, considering in (2.2.63) functions v that vanish at the vertices of the graph, we get the first two lines of (2.2.62) and considering functions v that do not vanish at only one vertex of the graph we recover the last line (Kirchhoff's conditions). Taking F = 0 we get then

$$KerA = span\{u^{(0)}\}.$$
 (2.2.67)

We will now use Lemma 2.2.5 proved below which states that ImA is closed. The operator A being self-adjoint, Lemma 2.2.5 together with (2.2.67) imply that the problem (2.2.66) is solvable if and only if $(F, u^{(0)})_{H^1(G)} = 0$ (cf. Theorem 5, Chapter 3 in [7]). Together with (2.2.65) this finishes the proof (the coefficient 2 disappears when we pass to the graph $G_{\mathcal{C}}$).

Lemma 2.2.5. The image of the operator A defined in (2.2.64) is closed:

 $ImA = \overline{ImA}.$

Proof. Suppose that $ImA \neq \overline{ImA}$. Then, there exists a singular sequence $\{u_n\}_{n \in \mathbb{N}} \subset H^1_s(G)$ such that

- (a) $||u_n||_{H^1(G)} = 1$,
- (b) $u_n \xrightarrow{w} 0$ in $H^1_s(G)$,
- (c) $||Au_n||_{H^1_s(G)} \longrightarrow 0.$

Let us show that there exists $n_0 \in \mathbb{N}$ and $\delta > 0$ such that

$$\|u_n\|_{L^{\mu}_2(G)} \ge \delta, \qquad \forall n \ge n_0.$$
(2.2.68)

Indeed,

$$-\|u_n'\|_{L_2^{\mu}(G)}^2 + \lambda^{(0)} \|u_n\|_{L_2^{\mu}(G)}^2 = (Au_n, u_n)_{H^1(G)} \leqslant \|Au_n\|_{H_s^1(G)} \to 0, \qquad n \to \infty$$

If there was a subsequence of $\{u_{n_k}\}_{k\in\mathbb{N}}$ of $\{u_n\}_{n\in\mathbb{N}}$ such that $||u_{n_k}||_{L_2^{\mu}(G)} \to 0$, $k \to \infty$, it would mean that $||u'_{n_k}||_{L_2^{\mu}(G)} \to 0$, and hence, $||u_{n_k}||_{H^1(G)} \to 0$ which contradicts the property (a) of the sequence $\{u_n\}$. Next,

$$(u_n, (A_s^{\mu} - \lambda^{(0)}I)v)_{L_2^{\mu}(G)} = -(Au_n, v)_{H^1(G)} \quad \forall v \in D(A_s^{\mu})$$

Since $\lambda^{(0)}$ is a simple (isolated) eigenvalue of the self-adjoint operator A_s^{μ} in $L_{2,s}^{\mu}(G)$, we have $Im\left(A_s^{\mu}-\lambda^{(0)}I\right) = \left\{u^{(0)}\right\}^{\perp_{L_{2,s}^{\mu}(G)}}$ and the operator $B = \left(A_s^{\mu}-\lambda^{(0)}I\right)^{-1}|_{\left\{u^{(0)}\right\}^{\perp_{L_{2,s}^{\mu}(G)}}}$ is a continuous operator from $L_{2,s}^{\mu}(G)$ to $H_s^1(G)$. Hence,

$$(u_n, w)_{L_2^{\mu}(G)} = -(Au_n, Bw)_{H^1(G)}, \qquad \forall w \in \left\{u^{(0)}\right\}^{\perp_{L_{2,s}^{\mu}(G)}}.$$
 (2.2.69)

Let us introduce the notation

$$u_n^{\perp} = u_n - \frac{\left(u_n, u^{(0)}\right)_{L_2^{\mu}(G)}}{\left\|u^{(0)}\right\|_{L_2^{\mu}(G)}^2} u^{(0)}, \qquad n \in \mathbb{N}.$$
(2.2.70)

Obviously, $u_n^{\perp} \in \left\{u^{(0)}\right\}^{\perp_{L_{2,s}^{\mu}(G)}}$. Then, applying (2.2.69) to u_n^{\perp} we obtain:

$$\left\|u_{n}^{\perp}\right\|_{L_{2}^{\mu}(G)}^{2} = \left(u_{n}, u_{n}^{\perp}\right)_{L_{2}^{\mu}(G)} \leqslant \|Au_{n}\|_{H_{s}^{1}(G)}\|B\|_{L_{2,s}^{\mu}(G) \to H_{s}^{1}(G)} \left\|u_{n}^{\perp}\right\|_{L_{2}^{\mu}(G)}, \qquad n \in \mathbb{N}$$

Taking into account the property (c) of the sequence $\{u_n\}$ we conclude that $\|u_n^{\perp}\|_{L^{\mu}_2(G)}^2 \to 0$ as $n \to \infty$. The property (b) implies that $u_n \xrightarrow{w} 0$ in $L^{\mu}_{2,s}(G)$. Consequently, $(u_n, u^{(0)})_{L^{\mu}_2(G)} \to 0$. Thus, it follows from the relation (2.2.70) that $\|u_n\|_{L^{\mu}_2(G)}^2 \to 0$ which contradicts (2.2.68).

We can now deduce Proposition 2.2.2 from Proposition 2.2.3.

Proof of Proposition 2.2.2. Let us introduce the following function $v^{(k)}$ for $k \in \mathbb{N}$:

$$v_{j+\frac{1}{2}}^{(k)}(s) = -\Delta_{j,+}^{(k-1)} \frac{\sin\left(\sqrt{\lambda^{(0)}}(1-s)\right)}{\sin\sqrt{\lambda^{(0)}}} + \Delta_{j+1,-}^{(k-1)} \frac{\sin\left(\sqrt{\lambda^{(0)}}s\right)}{\sin\sqrt{\lambda^{(0)}}}, \qquad s \in [0,1], \qquad j \in \mathbb{Z},$$
$$v_{j}^{(k)}(y) = 0, \qquad y \in \left[-\frac{L}{2}, \frac{L}{2}\right], \qquad j \in \mathbb{Z}.$$

Then, the function $\widetilde{u}^{(k)} = u^{(k)} - v^{(k)}, \ k \in \mathbb{N}$, satisfies the following problem:

$$\begin{cases} \left(\widetilde{u}_{j+\frac{1}{2}}^{(k)}\right)''(s) + \lambda^{(0)}\widetilde{u}_{j+\frac{1}{2}}^{(k)}(s) = -\lambda^{(k)}u_{j+\frac{1}{2}}^{(0)}(s) - f_{j+\frac{1}{2}}^{(k-1)}(s), & s \in [0,1], \\ \left(\widetilde{u}_{j+\frac{1}{2}}^{(k)}\right)''(u) + \lambda^{(0)}\widetilde{u}_{j+\frac{1}{2}}^{(k)}(u) = -\lambda^{(k)}u_{j+\frac{1}{2}}^{(0)}(u) - f_{j+\frac{1}{2}}^{(k-1)}(u), & u \in [-\frac{L}{2}, 0] \\ i \in \mathbb{Z}, \end{cases}$$

$$(u_j^{(k)})'(y) + \lambda^{(*)}u_j^{(*)}(y) = -\lambda^{(*)}u_j^{(*)}(y) - J_j^{(*)}(y), \qquad y \in [-\frac{1}{2}, 0], \qquad j \in \mathbb{Z},$$

$$(u_j^{-\gamma})(0) = 0, \qquad \qquad j \in \mathbb{Z},$$

$$\widetilde{u}_{j-\frac{1}{2}}^{(k)}(1) = \widetilde{u}_{j}^{(k)}\left(-\frac{L}{2}\right) = \widetilde{u}_{j+\frac{1}{2}}^{(k)}(0), \qquad j \in \mathbb{Z},$$

$$(\widetilde{z}_{j}^{(k)})'(0) = (\widetilde{z}_{j}^{(k)})'(1) + \omega^{\mu}(\widetilde{z}_{j}^{(k)})'(-L) = \widetilde{z}_{j}^{(k-1)}$$

$$(\widetilde{u}_{j+\frac{1}{2}}^{(k)})'(0) - (\widetilde{u}_{j-\frac{1}{2}}^{(k)})'(1) + w_j^{\mu} (\widetilde{u}_j^{(k)})' \left(-\frac{L}{2}\right) = \Xi_j^{(k-1)}, \qquad j \in \mathbb{Z},$$

$$(2.2.71)$$

Applying Proposition 2.2.3 to the problem (2.2.71) we finish the proof.

We finish this section with the following lemma that establishes a symmetry property of a solution of $(\mathcal{P}_{u}^{(k)})$ provided the right-hand terms are symmetric.

Lemma 2.2.6. Suppose that for some $k \in \mathbb{N}^*$, $f^{(k-1)} \in L_2^{\mu}(G_{\mathcal{C}})$, $\left\{\Delta_{j,\pm}^{(k-1)}\right\}_{j\in\mathbb{Z}} \in \ell_2(\mathbb{Z})$, $\left\{\Xi_j^{(k-1)}\right\}_{j\in\mathbb{Z}} \in \ell_2(\mathbb{Z})$ and the condition (2.2.60) is satisfied. Suppose also that $f_{-j-\frac{1}{2}}^{(k-1)}(1-s) = f_{j+\frac{1}{2}}^{(k-1)}(s)$, $s \in [0,1]$, $f_{-j}^{(k-1)}(y) = f_j^{(k-1)}(y)$, $y \in [-\frac{L}{2},0]$, $j \in \mathbb{Z}$,

and

$$\Delta_{j,+}^{(k-1)} = -\Delta_{-j,-}^{(k-1)}, \qquad \Delta_{j,-}^{(k-1)} = -\Delta_{-j,+}^{(k-1)}, \qquad \Xi_j^{(k-1)} = \Xi_{-j}^{(k-1)}, \qquad \forall j \in \mathbb{Z}.$$

If $u^{(k)}$ is a solution of $(\mathcal{P}_u^{(k)})$, then

$$u_{-j-\frac{1}{2}}^{(k)}(1-s) = u_{j+\frac{1}{2}}^{(k)}(s), \quad s \in [0,1], \qquad u_{-j}^{(k)}(y) = u_{j}^{(k)}(y), \quad y \in [-\frac{L}{2},0], \qquad j \in \mathbb{Z}.$$

Proof. Let us introduce the function $\hat{u}^{(k)}$ which is obtained from $u^{(k)}$ by the reflection with respect to the axis x = 0:

$$\widehat{u}_{j+\frac{1}{2}}^{(k)}(s) = u_{-j-\frac{1}{2}}^{(k)}(1-s), \quad s \in [0,1], \qquad \widehat{u}_{j}^{(k)}(s) = u_{-j}^{(k)}(s), \quad y \in \left[-\frac{L}{2}, 0\right], \qquad j \in \mathbb{Z}.$$

Then, the function $w^{(k)}$ which is defined as the difference $w^{(k)} = u^{(k)} - \hat{u}^{(k)}$, solves the homogeneous problem (2.2.39). Consequently, $w^{(k)} = cu^{(0)}$ for some $c \in \mathbb{R}$. On the other hand, $\mathbf{w}_0^{(k)} = 0$, which implies that $w^{(k)} = 0$. This finishes the proof.

2.3 Existence of the terms of the asymptotic expansions

In this section we will prove the existence of all the terms of the formal asymptotic expansions considered in Section 2.1 and, consequently, justify these expansions. We proceed by induction assuming that both the far field and the near field terms are constructed up to some order n and showing that they can be constructed at order n + 1. We remind that the terms of order 0 have been constructed in Section 2.2.3.

We start with an auxiliary result which will permit us to conclude that if both the far field and the near field terms are exponentially decaying up to the order n, then the far field terms at order n + 1 are also exponentially decaying.

Lemma 2.3.1. Let $k \in \mathbb{N}^*$. Suppose that $f^{(k-1)} \in L_2^{\mu}(G_{\mathcal{C}})$ and $\left\{\Delta_{j,\pm}^{(k-1)}\right\}_{j\in\mathbb{Z}} \in \ell_2(\mathbb{Z})$, $\left\{\Xi_j^{(k-1)}\right\}_{j\in\mathbb{Z}} \in \ell_2(\mathbb{Z})$, and that the compatibility condition (2.2.60) is satisfied. Suppose also that there exists a family of polynomials $\left\{a_\ell^{(k-1)}, b_\ell^{(k-1)}, c_\ell^{(k-1)}, d_\ell^{(k-1)}\right\}$ of degree $k - 1 - \ell$, $0 \leq \ell \leq k - 1$, and three polynomials $q_{\Delta_+}, q_{\Delta_-}, q_{\Sigma}$ of degree k - 1, such that

$$F_{j+\frac{1}{2}}(s) = r^{j} \sum_{\ell=0}^{k-1} s^{\ell} \left(a_{\ell}^{(k-1)}(j) \cos\left(\sqrt{\lambda_{0}}s\right) + b_{\ell}^{(k-1)}(j) \sin\left(\sqrt{\lambda_{0}}s\right) \right), \quad s \in [0,1], \quad j \in \mathbb{N},$$
(2.3.1)

$$F_{j}(y) = r^{j} \sum_{\ell=0}^{k-1} y^{\ell} \left(c_{\ell}^{(k-1)}(j) \cos\left(\sqrt{\lambda_{0}}y\right) + d_{\ell}^{(k-1)}(j) \sin\left(\sqrt{\lambda_{0}}y\right) \right), \quad y \in \left[-\frac{L}{2}, 0\right], \quad j \in \mathbb{N}^{*},$$
(2.3.2)

$$\Delta_{j,+}^{(k-1)} = r^j q_{\Delta_+}(j), \qquad \Delta_{j,-}^{(k-1)} = r^j q_{\Delta_-}(j), \qquad \Xi_j^{(k-1)} = r^j q_{\Sigma}(j), \qquad j \in \mathbb{N}^*, \qquad (2.3.3)$$

where

$$F_{j+\frac{1}{2}} = -\lambda^{(k)} u_{j+\frac{1}{2}}^{(0)} - f_{j+\frac{1}{2}}^{(k-1)}, \qquad F_j = -\lambda^{(k)} u_j^{(0)} - f_j^{(k-1)}, \qquad j \in \mathbb{N}.$$

Let $u^{(k)}$ be a solution of $(\mathcal{P}_u^{(k)})$. Then, there exist polynomials $\left\{\widetilde{a}_{\ell}^{(k)}, \widetilde{b}_{\ell}^{(k)}, \widetilde{c}_{\ell}^{(k)}, \widetilde{d}_{\ell}^{(k)}\right\}$ of degree $k - \ell$, $0 \leq \ell \leq k$, such that

$$u_{j+\frac{1}{2}}^{(k)}(s) = r^{j} \sum_{\ell=0}^{k} s^{\ell} \left(\widetilde{a}_{\ell}^{(k)}(j) \cos\left(\sqrt{\lambda_{0}}s\right) + \widetilde{b}_{\ell}^{(k)}(j) \sin\left(\sqrt{\lambda_{0}}s\right) \right), \qquad s \in [0,1], \qquad j \in \mathbb{N},$$
(2.3.4)

$$u_j^{(k)}(y) = r^j \sum_{\ell=0}^k y^\ell \left(\widetilde{c}_\ell^{(k)}(j) \cos\left(\sqrt{\lambda_0}y\right) + \widetilde{d}_\ell^{(k)}(j) \sin\left(\sqrt{\lambda_0}y\right) \right), \qquad y \in \left[-\frac{L}{2}, 0\right], \quad j \in \mathbb{N}^*.$$

$$(2.3.5)$$

Proof. First of all, we notice that the first two lines of $(\mathcal{P}_{u}^{(k)})$ together with the assumptions (2.3.1), (2.3.2) for the right-hand sides imply the forms (2.3.4), (2.3.5) of the solutions $u_{j+\frac{1}{2}}^{(k)}$, $u_{j}^{(k)}$ with some coefficients $\left\{\widetilde{a}_{\ell}^{(k)}(j), \widetilde{b}_{\ell}^{(k)}(j), \widetilde{c}_{\ell}^{(k)}(j), \widetilde{d}_{\ell}^{(k)}(j)\right\}_{\ell=0}^{k}$ (not necessarily polynomial with respect to j). We will start by establishing a recurrence relations for the coefficients with $\ell > 0$ which will permit us to show that they are polynomials. Then,

using the transmission conditions we will get a system of equations for the coefficients with $\ell = 0$ which will yield a finite difference equation for $\tilde{a}_0^{(k)}$. The study of this finite difference equation will show that the coefficients with $\ell = 0$ are also polynomials.

Recurrence relations for the coefficients with $\ell \ge 1$: In order to determine the dependence of these coefficients on j we plug the relations (2.3.4), (2.3.1) into the first line of $(\mathcal{P}_{u}^{(k)})$. We get then:

$$2(\ell+1)\sqrt{\lambda_0}\widetilde{b}_{\ell+1}^{(k)}(j) + (\ell+1)(\ell+2)\widetilde{a}_{\ell+2}^{(k)}(j) = a_{\ell}^{(k-1)}(j), \qquad 0 \le \ell \le k-1, \\ -2(\ell+1)\sqrt{\lambda_0}\widetilde{a}_{\ell+1}^{(k)}(j) + (\ell+1)(\ell+2)\widetilde{b}_{\ell+2}^{(k)}(j) = b_{\ell}^{(k-1)}(j), \qquad 0 \le \ell \le k-1,$$

where, by convention, $\widetilde{a}_{k+1}^{(k)}(j) = \widetilde{b}_{k+1}^{(k)}(j) = 0$. Hence,

$$\widetilde{b}_{k}^{(k)}(j) = \frac{a_{k-1}^{(k-1)}(j)}{2k\sqrt{\lambda_{0}}}, \qquad \widetilde{a}_{k}^{(k)}(j) = -\frac{b_{k-1}^{(k-1)}(j)}{2k\sqrt{\lambda_{0}}}, \qquad (2.3.6)$$

and the following recurrence relations define all the other coefficients except $\tilde{a}_0^{(k)}(j)$, $\tilde{b}_0^{(k)}(j)$ which remain undetermined:

$$\widetilde{b}_{\ell}^{(k)}(j) = \frac{a_{\ell-1}^{(k-1)}(j) - \ell(\ell+1)\widetilde{a}_{\ell+1}^{(k)}(j)}{2\ell\sqrt{\lambda_0}}, \qquad \widetilde{a}_{\ell}^{(k)}(j) = -\frac{b_{\ell-1}^{(k-1)}(j) - \ell(\ell+1)\widetilde{b}_{\ell+1}^{(k)}(j)}{2\ell\sqrt{\lambda_0}}, \qquad 1 \le \ell \le k-1.$$
(2.3.7)

One can prove by induction with respect to ℓ that the coefficients $\tilde{a}_{\ell}^{(k)}(j)$, $\tilde{b}_{\ell}^{(k)}(j)$ are polynomials (with respect to j) of degree $k - \ell$ for $1 \leq \ell \leq k$. Indeed, taking into account the assumptions for the coefficients $\left\{a_{\ell}^{(k-1)}(j), b_{\ell}^{(k-1)}(j)\right\}_{\ell=0}^{k-1}$ we see from (2.3.6) that for $\ell = k$ the coefficients $\tilde{a}_{k}^{(k)}(j)$, $\tilde{b}_{k}^{(k)}(j)$ are constants in j. If we suppose that $\tilde{a}_{\ell}^{(k)}(j)$, $\tilde{b}_{\ell}^{(k)}(j)$ are polynomials (with respect to j) of degree $k - \ell$ for all ℓ such that $m \leq \ell \leq k$ for some $1 < m \leq k$ then the relations (2.3.7) imply that $\tilde{a}_{m-1}^{(k)}(j)$, $\tilde{b}_{m-1}^{(k)}(j)$ are polynomials in j of degree k - m + 1. Thus, it remains only to determine the behaviour of the coefficients $\tilde{a}_{0}^{(k)}(j)$, $\tilde{b}_{0}^{(k)}(j)$ with respect to j for $j \geq 0$.

Repeating the same argument applied to the coefficients $\left\{ \tilde{c}_{\ell}^{(k)}(j), \tilde{d}_{\ell}^{(k)}(j) \right\}_{\ell=0}^{k}$, i. e., injecting (2.3.5), (2.3.2) into the second line of $(\mathcal{P}_{u}^{(k)})$ we get the analogues of the relations (2.3.6), (2.3.7) with $a_{\ell}^{(k-1)}(j)$, $b_{\ell}^{(k-1)}(j)$, $\tilde{a}_{\ell}^{(k)}(j)$, $\tilde{b}_{\ell}^{(k)}(j)$ replaced by $c_{\ell}^{(k-1)}(j)$, $d_{\ell}^{(k-1)}(j)$, $\tilde{c}_{\ell}^{(k)}(j)$, $\tilde{d}_{\ell}^{(k)}(j)$ respectively. However, this time there is an additional relation coming from the third line of $(\mathcal{P}_{u}^{(k)})$ (Neumann boundary condition):

$$\tilde{c}_1^{(k)}(j) + \sqrt{\lambda_0} \tilde{d}_0^{(k)}(j) = 0.$$
(2.3.8)

Hence, in the same manner as before we see that the coefficients $\tilde{c}_{\ell}^{(k)}(j)$, $\tilde{d}_{\ell}^{(k)}(j)$ are polynomials with respect to j (for $j \ge 1$) of degree $k - \ell$ for $1 \le \ell \le k$ and the coefficient $\tilde{d}_{0}^{(k)}(j)$ is a polynomial of degree k - 1.
The coefficients $\tilde{a}_0^{(k)}(j)$, $\tilde{b}_0^{(k)}(j)$, $\tilde{c}_0^{(k)}(j)$: Let us now establish the dependence of the coefficients $\tilde{a}_0^{(k)}(j)$, $\tilde{b}_0^{(k)}(j)$, $\tilde{c}_0^{(k)}(j)$ on j using the transmission conditions (last two lines of $(\mathcal{P}_u^{(k)})$). We get for $j \ge 1$:

$$r^{j-1}\cos\sqrt{\lambda_0}\widetilde{a}_0^{(k)}(j-1) + r^{j-1}\sin\sqrt{\lambda_0}\widetilde{b}_0^{(k)}(j-1) - r^j\cos\left(\frac{\sqrt{\lambda_0}L}{2}\right)\widetilde{c}_0^{(k)}(j) = t_1^{(k)}(j), \quad (2.3.9)$$

$$r^{j}\cos\left(\frac{\nabla A_{0}L}{2}\right)\widetilde{c}_{0}^{(k)}(j) - r^{j}\widetilde{a}_{0}^{(k)}(j) = t_{2}^{(k)'}(j), \qquad (2.3.10)$$

$$r^{j}\widetilde{b}_{0}^{(k)}(j) + r^{j-1}\sin\sqrt{\lambda_{0}}\widetilde{a}_{0}^{(k)}(j-1) - r^{j-1}\cos\sqrt{\lambda_{0}}\widetilde{b}_{0}^{(k)}(j-1) + r^{j}\sin\left(\frac{\sqrt{\lambda_{0}}L}{2}\right)\widetilde{c}_{0}^{(k)}(j) = t_{3}^{(k)}(j), \qquad (2.3.11)$$

where

$$\begin{split} t_1^{(k)}(j) &= \Delta_{j,-}^{(k-1)} - r^{j-1} \sum_{\ell=1}^k \left(\cos \sqrt{\lambda_0} \tilde{a}_\ell^{(k)}(j-1) + \sin \sqrt{\lambda_0} \tilde{b}_\ell^{(k)}(j-1) \right) \\ &+ r^j \sum_{\ell=1}^k \left(-\frac{L}{2} \right)^\ell \left(\cos \left(\frac{\sqrt{\lambda_0} L}{2} \right) \tilde{c}_\ell^{(k)}(j) - \sin \left(\frac{\sqrt{\lambda_0} L}{2} \right) \tilde{d}_\ell^{(k)}(j) \right) + \frac{r^j}{\sqrt{\lambda_0}} \sin \left(\frac{\sqrt{\lambda_0} L}{2} \right) \tilde{c}_1^{(k)}(j), \\ t_2^{(k)}(j) &= \Delta_{j,+}^{(k-1)} - r^j \sum_{\ell=1}^k \left(-\frac{L}{2} \right)^\ell \left(\cos \left(\frac{\sqrt{\lambda_0} L}{2} \right) \tilde{c}_\ell^{(k)}(j) - \sin \left(\frac{\sqrt{\lambda_0} L}{2} \right) \tilde{d}_\ell^{(k)}(j) \right) \\ &- \frac{r^j}{\sqrt{\lambda_0}} \sin \left(\frac{\sqrt{\lambda_0} L}{2} \right) \tilde{c}_1^{(k)}(j), \end{split}$$

$$\begin{split} t_3^{(k)}(j) &= r^{j-1} \sum_{\ell=1}^k \left(\left(\ell \frac{\cos \sqrt{\lambda_0}}{\sqrt{\lambda_0}} - \sin \sqrt{\lambda_0} \right) \widetilde{a}_\ell^{(k)}(j-1) + \left(\ell \frac{\sin \sqrt{\lambda_0}}{\sqrt{\lambda_0}} + \cos \sqrt{\lambda_0} \right) \widetilde{b}_\ell^{(k)}(j-1) \right) \\ &- r^j \sum_{\ell=1}^k \left(-\frac{L}{2} \right)^\ell \left(\sin \left(\frac{\sqrt{\lambda_0}L}{2} \right) - \frac{2\ell}{L\sqrt{\lambda_0}} \cos \left(\frac{\sqrt{\lambda_0}L}{2} \right) \right) \widetilde{c}_\ell^{(k)}(j) \\ &- r^j \sum_{\ell=1}^k \left(-\frac{L}{2} \right)^\ell \left(\cos \left(\frac{\sqrt{\lambda_0}L}{2} \right) + \frac{2\ell}{L\sqrt{\lambda_0}} \sin \left(\frac{\sqrt{\lambda_0}L}{2} \right) \right) \widetilde{d}_\ell^{(k)}(j) \\ &+ \frac{\Xi_j^{(k-1)}}{\sqrt{\lambda_0}} - \frac{r^j}{\sqrt{\lambda_0}} \left(\widetilde{a}_1^{(k)}(j) - \cos \left(\frac{\sqrt{\lambda_0}L}{2} \right) \widetilde{c}_1^{(k)}(j) \right). \end{split}$$

The expressions above imply that the right-hand terms of the equations (2.3.9)–(2.3.11) have the form

$$t_i^{(k)}(j) = r^j p_{t,i}^{(k)}(j), \qquad 1 \le i \le 3,$$
(2.3.12)

where $\left\{p_{t,i}^{(k)}\right\}_{i=1}^{3}$ are polynomials in j of degree k-1. The system (2.3.9)–(2.3.11) reduces to the following one:

$$\widetilde{c}_{0}^{(k)}(j) = \frac{1}{\cos\left(\frac{\sqrt{\lambda_{0}}L}{2}\right)} \left(\widetilde{a}_{0}^{(k)}(j) + p_{t,2}^{(k)}(j) \right), \qquad j \ge 1,$$
(2.3.13)

$$\widetilde{b}_{0}^{(k)}(j) = \frac{1}{\sin\sqrt{\lambda_{0}}} \left(r \widetilde{a}_{0}^{(k)}(j+1) - \widetilde{a}_{0}^{(k)}(j) \cos\sqrt{\lambda_{0}} + r \left(p_{t,1}^{(k)}(j+1) + p_{t,2}^{(k)}(j+1) \right) \right), \quad j \ge 0.$$
(2.3.14)

$$r^{2}\widetilde{a}_{0}^{(k)}(j+2) + 2rg\left(\sqrt{\lambda_{0}}\right)\widetilde{a}_{0}^{(k)}(j+1) + \widetilde{a}_{0}^{(k)}(j) = p_{a}^{(k)}(j), \qquad j \ge 0, \qquad (2.3.15)$$

where the function g is defined in (1.3.45) and

$$p_a^{(k)}(j) = r \sin \sqrt{\lambda_0} p_{t,3}^{(k)}(j+1) - r^2 \left(p_{t,1}^{(k)}(j+2) + p_{t,2}^{(k)}(j+2) \right) - r \sin \sqrt{\lambda_0} \tan \left(\frac{\sqrt{\lambda_0}L}{2} \right) p_{t,2}^{(k)}(j+1) + r \cos \sqrt{\lambda_0} \left(p_{t,1}^{(k)}(j+1) + p_{t,2}^{(k)}(j+1) \right).$$

The function $p_a^{(k)}(j)$ being a polynomial in j of degree k-1, we introduce a notation for its coefficients:

$$p_a^{(k)}(j) = \sum_{m=0}^{k-1} \rho_m j^m.$$
(2.3.16)

Study of the finite difference equation (2.3.15): First, recalling that r and r^{-1} are solutions of the equation (1.3.47), we find the general (real) solution of the corresponding homogeneous equation:

$$\widetilde{a}_0^{(k)}(j) = C + Dr^{-2j}, \qquad C, \ D \in \mathbb{R}, \qquad j \geqslant 0.$$

However, the condition $u^{(k)} \in L_2^{\mu}(G_{\mathcal{C}})$ implies that D = 0. Next, the right-hand side of the equation (2.3.15) being a polynomial of degree k - 1, we search a particular solution as a polynomial of degree k:

$$\widetilde{a}_0^{(k)}(j) = \sum_{m=0}^k \alpha_m j^m$$

Together with (2.3.16) it gives the following system for the coefficients $\{\gamma_m\}_{m=0}^k$:

$$\sum_{i=m}^{k} C_i^m r^2 \alpha_i + 2g\left(\sqrt{\lambda^{(0)}}\right) r \alpha_m + \sum_{i=m}^{k} (-1)^{i-m} C_i^m r^2 \alpha_i = \rho_m, \qquad 0 \leqslant m \leqslant k,$$

where $\rho_k = 0$. Since r is a solution of (1.3.47), the equation for m = k is satisfied automatically and the other equations take the form

$$\sum_{i=m+1}^{k} C_i^m \left(\left(2^{i-m} - 1 \right) r^2 - 1 \right) \alpha_i = \rho_m, \qquad 0 \le m \le k - 1, \tag{2.3.17}$$

which is a system with an upper-diagonal matrix with non-zero elements (|r| < 1) for the coefficients $\{\alpha_m\}_{m=1}^k$. The coefficient α_0 cannot be determined, which corresponds to the general solution $\tilde{a}_0^{(k)}(j) = C$, $j \ge 0$. We can fix it by choosing, for example, $\alpha_0 = 0$. Hence, the coefficient $\tilde{a}_0^{(k)}(j)$ is a polynomial in j of degree k for $j \ge 0$ as well as the coefficients $\tilde{b}_0^{(k)}(j)$ for $j \ge 0$ and $\tilde{c}_0^{(k)}(j)$ for $j \ge 1$ (cf. (2.3.13), (2.3.14)). This finishes the proof.

We are now able to prove the existence of the terms of the asymptotic expansions (2.1.2)–(2.1.5).

Proposition 2.3.1. There exist functions $\{u^{(k)}, U^{(k)}\}_{k \in \mathbb{N}}$ and real numbers $\{\lambda^{(k)}\}_{k \in \mathbb{N}}$ having the following properties:

1. For any $k \in \mathbb{N}$, $u^{(k)}$ is a solution of the far field problem $(\mathcal{P}_u^{(k)})$. Moreover,

$$u_{j+\frac{1}{2}}^{(k)} \in C^{\infty}([0,1]), \qquad u_{j}^{(k)} \in C^{\infty}\left(\left[-\frac{L}{2},0\right]\right), \qquad j \in \mathbb{Z}, \qquad k \in \mathbb{N}, \qquad (2.3.18)$$

and

$$u_{-j-\frac{1}{2}}^{(k)}(1-s) = u_{j+\frac{1}{2}}^{(k)}(s), \quad s \in [0,1], \qquad u_{-j}^{(k)}(y) = u_{j}^{(k)}(y), \quad y \in [-\frac{L}{2},0],$$
$$j \in \mathbb{Z}, \qquad k \in \mathbb{N}.$$
(2.3.19)

2. For $k \in \mathbb{N}$, $j \in \mathbb{Z}$, $U_j^{(k)} \in H^1_{loc}(\mathcal{J}_j)$ is a solution of the problem $(\mathcal{P}_j^{(k)})$. Moreover, $U_{-j}^{(k)}(-X,Y) = U_j^{(k)}(X,Y), \qquad j \in \mathbb{Z}, \qquad k \in \mathbb{N}.$ (2.3.20)

- 3. The number $\lambda^{(k)}$ satisfies (2.2.60).
- 4. For any $k \in \mathbb{N}$ the relations (2.3.4), (2.3.5) for the far field terms $u^{(k)}$ are satisfied. Moreover, there exist functions $\mathcal{U}_q^{(k)} \in H^1_{loc}(\mathcal{J}_*)$ such that the near field terms satisfy the following relation:

$$U_{j}^{(k)} = r^{j} \sum_{q=0}^{k} j^{q} \mathcal{U}_{q}^{(k)}, \qquad j \in \mathbb{N}^{*}.$$
 (2.3.21)

The last property means that the far field terms $u_j^{(k)}$ and the near field terms $U_j^{(k)}$ are exponentially decaying as |j| tends to infinity.

Proof. First, it is clear that the the functions $u^{(0)}$, $U^{(0)}$ and the number $\lambda^{(0)}$ found in Section 2.2.3 satisfy the properties 1–4. Suppose that for some $n \ge 1$ functions $\{u^{(k)}, U^{(k)}\}_{k=0}^{n}$ and real numbers $\{\lambda^{(k)}\}_{k=0}^{n}$ having the required properties are constructed. Let us show the existence of functions $u^{(n+1)}$, $U^{(n+1)}$ and a real number $\lambda^{(n+1)}$ satisfying the properties 1–4 as well.

Construction of the right-hand terms of the problems $\mathcal{P}_{u}^{(n+1)}$ and $\mathcal{P}_{i}^{(n+1)}$

• The function $f^{(n)}$ is given by the relations (2.2.37), (2.2.38) with k replaced by n + 1. Since the functions $\{u^{(k)}\}_{k=1}^n$ satisfy the property 1, one has $f^{(n)} \in L_2(G_{\mathcal{C}})$ and

$$f_{j+\frac{1}{2}}^{(n)} \in C^{\infty}([0,1]), \qquad f_{j}^{(n)} \in C^{\infty}\left(\left[-\frac{L}{2},0\right]\right), \qquad j \in \mathbb{Z},$$
 (2.3.22)

$$f_{-j-\frac{1}{2}}^{(n)}(1-s) = f_{j+\frac{1}{2}}^{(n)}(s), \quad s \in [0,1], \qquad f_{-j}^{(n)}(y) = f_{j}^{(n)}(y), \quad y \in [-\frac{L}{2},0], \qquad j \in \mathbb{Z}.$$
(2.3.23)

Moreover, the functions $\{u^{(k)}\}_{k=1}^{n}$ satisfying the property 4, the function $f^{(n)}$ is also of the form (2.3.4), (2.3.5). In other words, there exist polynomials $\{a_{f,\ell}^{(n)}, b_{f,\ell}^{(n)}, c_{f,\ell}^{(n)}, d_{f,\ell}^{(n)}\}$ of degree $n - \ell$, $0 \leq \ell \leq n$, such that

$$f_{j+\frac{1}{2}}^{(n)}(s) = r^{j} \sum_{\ell=0}^{n} s^{\ell} \left(a_{f,\ell}^{(n)}(j) \cos\left(\sqrt{\lambda_{0}}s\right) + b_{f,\ell}^{(n)}(j) \sin\left(\sqrt{\lambda_{0}}s\right) \right), \quad s \in [0,1], \qquad j \in \mathbb{N},$$
(2.3.24)

$$f_{j}^{(n)}(y) = r^{j} \sum_{\ell=0}^{n} y^{\ell} \left(c_{f,\ell}^{(n)}(j) \cos\left(\sqrt{\lambda_{0}}y\right) + d_{f,\ell}^{(n)}(j) \sin\left(\sqrt{\lambda_{0}}y\right) \right), \quad y \in \left[-\frac{L}{2}, 0\right], \quad j \in \mathbb{N}^{*}.$$
(2.3.25)

• The coefficients $\left\{c_{j,p,\delta,\ell}^{(n+1)}, j \in \mathbb{Z}, p \in \mathbb{N}^*, \delta \in \{+, -, 0\}, 1 \leq \ell \leq \left[\frac{n+1}{2}\right]\right\}$ are computed via the recurrence relation (2.2.12). Since the functions $\left\{U^{(k)}\right\}_{k=0}^n$ verify the property 2, Lemma 2.2.4 applies which gives the convergence result (2.2.46) for the coefficients c up to the order n, but also for the coefficients of order n + 1 for $\ell \geq 1$:

$$\sum_{p \in \mathbb{N}^*} p\left(c_{j,p,\pm,\ell}^{(n+1)}\right)^2 e^{-p\pi w_j^{\mu}} < \infty, \quad \sum_{p \in \mathbb{N}^*} p\left(c_{j,p,0,\ell}^{(n+1)}\right)^2 e^{-\frac{2p\pi}{w_j^{\mu}}} < \infty, \quad 1 \leqslant \ell \leqslant \left[\frac{n+1}{2}\right].$$
(2.3.26)

Moreover, the following property can be easily shown by induction in k:

$$c_{-j,p,\pm,\ell}^{(k)} = c_{j,p,\mp,\ell}^{(k)}, \qquad c_{-j,p,0,\ell}^{(k)} = c_{j,p,0,\ell}^{(k)}, \qquad 0 \leqslant k \leqslant n, \quad j \in \mathbb{Z}, \quad p \in \mathbb{N}^*, \quad 0 \leqslant \ell \leqslant \left[\frac{k}{2}\right].$$

and there exist real numbers $\check{c}_{q,p,\delta,\ell}^{(k)}$, $0 \leq k \leq n$, $0 \leq q \leq k$, $p \in \mathbb{N}^*$, $\delta \in \{+, -, 0\}$, $0 \leq \ell \leq \left\lfloor \frac{k}{2} \right\rfloor$, such that

$$c_{j,p,\delta,\ell}^{(k)} = r^j \sum_{q=0}^k j^q \check{c}_{q,p,\delta,\ell}^{(k)}, \quad 0 \leqslant k \leqslant n, \quad j \in \mathbb{N}^*, \quad p \in \mathbb{N}^*, \quad \delta \in \{+,-,0\}, \quad 0 \leqslant \ell \leqslant \left[\frac{k}{2}\right].$$

In the base case k = 0 these properties obviously hold since all the coefficients are equal to zero (cf. (2.2.45)). The proof of the inductive step relies on the recurrence relation (2.2.12) for $\ell \ge 1$ and the relation (2.2.59) for $\ell = 0$, taking into account the properties 2 and 4 of the functions $\{U^{(k)}\}_{k=0}^{n}$.

Again, due to the recurrence relation (2.2.12), the same is true for the coefficients of order n + 1 for $\ell \ge 1$:

$$c_{-j,p,\pm,\ell}^{(n+1)} = c_{j,p,\pm,\ell}^{(n+1)}, \qquad c_{-j,p,0,\ell}^{(n+1)} = (-1)^p c_{j,p,0,\ell}^{(n+1)}, \qquad j \in \mathbb{Z}, \quad p \in \mathbb{N}^*, \quad 1 \leqslant \ell \leqslant \left[\frac{n+1}{2}\right],$$
(2.3.27)

nd there exist real numbers $\check{c}^{(n+1)} = 0 \leqslant a \leqslant n+1, \quad n \in \mathbb{N}^*, \quad \delta \in \{+, -0\}$

and there exist real numbers $\check{c}_{q,p,\delta,\ell}^{(n+1)}$, $0 \leq q \leq n+1$, $p \in \mathbb{N}^*$, $\delta \in \{+,-,0\}$, $1 \leq \ell \leq \left[\frac{n+1}{2}\right]$, such that

$$c_{j,p,\delta,\ell}^{(n+1)} = r^j \sum_{q=0}^n j^q \check{c}_{q,p,\delta,\ell}^{(n+1)}, \quad j \in \mathbb{N}^*, \quad p \in \mathbb{N}^*, \quad \delta \in \{+, -, 0\}, \quad 1 \le \ell \le \left[\frac{n+1}{2}\right].$$
(2.3.28)

• The functions $\left\{g_{j,\delta}^{(n)}\right\}_{j\in\mathbb{Z}}$, $\delta \in \{+, -, 0\}$, are defined by the relations (2.2.24)–(2.2.26). The polynomials $P_{j,p,\delta}^{(n+1)'}$ appearing in this relations are completely defined at this point as follows since

$$P_{j,p,\delta}^{(k)}{}' = \sum_{\ell=1}^{\left[\frac{k}{2}\right]} c_{j,p,\delta,\ell}^{(k)} \ell s^{\ell-1}, \qquad \delta \in \{+,-,0\}, \quad k \in \mathbb{N}, \quad p \in \mathbb{N}^*, \quad j \in \mathbb{Z}.$$

The relations (2.3.26) permit to conclude that

$$g_{j,\delta}^{(n)} \in H^{-1/2}\left(\Sigma_j^{\pm}\right), \qquad \delta \in \{+, -, 0\}.$$
 (2.3.29)

Moreover, the relations (2.3.27) together with the property 1 satisfied by the functions $\{u^{(k)}\}_{k=0}^{n}$ imply that

$$g_{j,\pm}^{(n)} = g_{-j,\mp}^{(n)}, \qquad g_{j,0}^{(n)}(X) = g_{-j,0}^{(n)}(-X), \qquad j \in \mathbb{Z},$$
 (2.3.30)

where we identify the spaces $H^{-1/2}(\Sigma_j^+)$ and $H^{-1/2}(\Sigma_j^-)$. The relations (2.3.28) together with the property 4 satisfied by the functions $\{u^{(k)}\}_{k=0}^n$ imply that there exist functions $\check{g}_{q,\delta}^{(n)} \in H^{-1/2}(\Sigma_j^{\delta}), \ 0 \leq q \leq n, \ \delta \in \{+, -, 0\}$, such that

$$g_{j,\delta}^{(n)} = r^j \sum_{q=0}^n j^q \check{g}_{q,\delta}^{(n)}, \qquad j \in \mathbb{N}^*, \qquad \delta \in \{+, -, 0\}.$$
(2.3.31)

• The functions

$$\Phi_j^{(n-1)}, \, \Phi_j^{(n)} \in H^1_{loc}(\mathcal{J}_j), \qquad j \in \mathbb{Z},$$

$$(2.3.32)$$

are defined by the relation (2.2.23). From the properties 2 and 4 (symmetry and exponential decay) satisfied by the functions $\{U^{(k)}\}_{k=0}^{n}$ it follows that

$$\Phi_{-j}^{(k)}(-X,Y) = \Phi_{j}^{(k)}(X,Y), \qquad j \in \mathbb{Z}, \qquad 0 \le k \le n,$$
(2.3.33)

and there exist functions $\check{\Phi}_q^{(k)} \in H^1_{loc}(\mathcal{J}_*), \ 0 \leq k \leq n, \ 0 \leq q \leq k$, such that

$$\Phi_j^{(k)} = r^j \sum_{q=0}^k j^q \check{\Phi}_q^{(k)}, \qquad \forall j \in \mathbb{N}^*, \qquad 0 \leqslant k \leqslant n.$$
(2.3.34)

• The sequence $\left\{\widetilde{\Xi}_{j}^{(n)}\right\}_{j\in\mathbb{Z}}$ is defined in (2.2.61) with k replaced by n + 1, and the sequences $\left\{\Delta_{j,\pm}^{(n)}\right\}_{j\in\mathbb{Z}}$, $\left\{\Xi_{j}^{(n)}\right\}_{j\in\mathbb{Z}}$ are defined in (2.2.36), (2.2.34), (2.2.31). The symmetry properties of the functions $\left\{u^{(k)}\right\}_{k=0}^{n}$ (property 1), $g_{j,\delta}^{(n)}$ (2.3.30), $\Phi_{j}^{(n)}$, $\Phi_{j}^{(n-1)}$ (2.3.33) and W_{j}^{\pm} (2.2.20) imply that

$$\Delta_{j,+}^{(n)} = -\Delta_{-j,-}^{(n)}, \qquad \Delta_{j,-}^{(n)} = -\Delta_{-j,+}^{(n)}, \qquad \Xi_j^{(n)} = \Xi_{-j}^{(n)}, \qquad j \in \mathbb{Z}.$$
 (2.3.35)

Similarly, the decay properties of the functions $\{u^{(k)}\}_{k=0}^{n}$ (property 4), $g_{j,\delta}^{(n)}$ (2.3.31), $\Phi_{j}^{(n)}$, $\Phi_{j}^{(n-1)}$ (2.3.34) and the relations (2.2.21) for W_{j}^{\pm} imply that there exist polynomials $q_{\Sigma}^{(n)}$, $q_{\Delta\pm}^{(n)}$ of degree n such that

$$\Xi_{j}^{(n)} = r^{j} q_{\Sigma}^{(n)}(j), \qquad \Delta_{j,\pm}^{(n)} = r^{j} q_{\Delta_{\pm}}^{(n)}(j), \qquad j \in \mathbb{N}^{*}.$$
(2.3.36)

Construction of the number $\lambda^{(n+1)}$ and the far field terms $\boldsymbol{u}^{(n+1)}$

Due to the decay properties of $f^{(n)}$, $\left\{\Xi_{j}^{(n)}\right\}_{j\in\mathbb{Z}}$, $\left\{\Delta_{j,\pm}^{(n)}\right\}_{j\in\mathbb{Z}}$ (cf. (2.3.24), (2.3.25), (2.3.23), (2.3.36), (2.3.35)), we have $f^{(n)} \in L_2(G_{\mathcal{C}})$, $\left\{\Xi_{j}^{(n)}\right\}_{j\in\mathbb{Z}} \in \ell_2(\mathbb{Z})$, $\left\{\Delta_{j,\pm}^{(n)}\right\}_{j\in\mathbb{Z}} \in \ell_2(\mathbb{Z})$. This permits to apply Proposition 2.2.2 for k = n + 1. The number $\lambda^{(n+1)}$ is defined by the relation (2.2.60) and the function $u^{(n+1)}$ is a corresponding solution of the problem $(\mathcal{P}_{u}^{(n+1)})$. To fix it in a unique way, one can, for example, impose the condition $\mathbf{u}_{0}^{(n+1)}$. The smoothness (2.3.18) of the functions $u_{j+\frac{1}{2}}^{(n+1)}$, $u_{j}^{(n+1)}$ at each vertex of the graph $G_{\mathcal{C}}$ is guaranteed by the smoothness of the functions $f_{j+\frac{1}{2}}^{(n)}$, $f_{j}^{(n)}$ (2.3.22). Due to the symmetry properties of $f^{(n)}$, $\left\{\Xi_{j}^{(n)}\right\}_{j\in\mathbb{Z}}$ and $\left\{\Delta_{j,\pm}^{(n)}\right\}_{j\in\mathbb{Z}}$ (cf. (2.3.23), (2.3.35)), Lemma 2.2.6 applies and we get the symmetry property (2.3.19) for the function $u^{(n+1)}$. Finally, to get the decay property 4 for $u^{(n+1)}$, we apply Lemma 2.3.1. Indeed, the hypothesis of the Lemma are verified due to the relations (2.3.24), (2.3.25), (2.3.36).

At this point, we can remark that the polynomials (2.2.13) satisfy the following symmetry property:

$$P_{j,0,\pm}^{(n+1)}(X) = P_{-j,0,\mp}^{(n+1)}(-X), \qquad P_{j,0,0}^{(n+1)}(Y) = P_{-j,0,0}^{(n+1)}(Y), \qquad j \in \mathbb{Z}.$$
 (2.3.37)

Moreover, there exists real numbers $\left\{\check{p}_q^{(n+1)}\right\}_{q=0}^{n+1}$ such that

$$P_{j,0,+}^{(n+1)}\left(\frac{1}{2}\right) = r^j \sum_{q=0}^{n+1} \check{p}_q^{(n+1)} j^q, \qquad j \in \mathbb{N}^*.$$

Construction of the near field terms $U^{(n+1)}$

• In a bounded domain: for any $j \in \mathbb{Z}$, we start by constructing the solution $V_j^{(n+1)}$ of the problem $(\widetilde{\mathcal{P}}_j^{(n+1)})$ in a bounded domain. The existence (and uniqueness) of such a solution is guaranteed by Lemma 2.2.2. Indeed, the right-hand sides belong to the appropriate spaces (cf. (2.3.32), (2.3.29)). The compatibility condition (2.2.30) is satisfied since the function $u^{(n)}$ solves the problem $(\mathcal{P}_u^{(n)})$. Notice that the average trace condition (2.2.29) requires the knowledge of the function $u^{(n+1)}$. That is why at a given order the far field problem should be solved first.

Let us show that the functions $\left\{V_{j}^{(n+1)}\right\}_{j\in\mathbb{Z}}$ verify the following symmetry property:

$$V_{-j}^{(n+1)}(-X,Y) = V_j^{(n+1)}(X,Y), \qquad j \in \mathbb{Z}.$$
(2.3.38)

Indeed, if $V_{-j}^{(n+1)}$ satisfies the problem $(\widetilde{\mathcal{P}}_{-j}^{(n+1)})$, then the function $V_{-j}^{(n+1)}(-X,Y)$ satisfies the problem $(\widetilde{\mathcal{P}}_{j}^{(n+1)})$. This follows from the symmetry properties of the right-hand sides $\Phi_{j}^{(n-1)}$, $g_{j,\delta}^{(n)}$ (cf. (2.3.33), (2.3.30)) together with Lemma 2.2.3 and the symmetry property 1 of the functions $\{u^{(k)}\}_{k=0}^{n+1}$ which ensure the equivalence of the trace average conditions in the problems $(\widetilde{\mathcal{P}}_{j}^{(n+1)})$ and the one satisfied by the function $V_{-j}^{(n+1)}(-X,Y)$. The solution of the problem $(\widetilde{\mathcal{P}}_{j}^{(n+1)})$ being unique, we get (2.3.38).

Finally, let us show that there exist functions $\mathcal{V}_q^{(n+1)} \in H^1(K_*), \ 0 \leq q \leq n+1$, such that

$$V_j^{(n+1)} = r^j \sum_{q=0}^{n+1} j^q \mathcal{V}_q^{(n+1)}, \qquad j \in \mathbb{N}^*.$$
(2.3.39)

To do this, we introduce the following set of problems for $0 \leq q \leq n+1$:

Find
$$V \in H^1(K_*)$$
 satisfying (2.2.16) with $\Phi = \check{\Phi}_q^{(n-1)}$,
 $g_{\delta} = \check{g}_{q,\delta}^{(n)}, \quad \delta \in \{+, -, 0\},$ and such that $\int_{\Sigma_*^+} V = \check{p}_q^{(n+1)}.$ $(\check{\mathcal{P}}_q^{(n+1)})$

Here $K_* = \left] -\frac{1}{2}, \frac{1}{2} \right[\times]0, 1[, \Sigma_*^+ = \left\{ \frac{1}{2} \right\} \times [0, 1]$ (which are just K_j and Σ_j^+ for $j \in \mathbb{Z}^*$), and $\check{\mathbf{X}}_j^{(n-1)} = \check{\mathbf{X}}_j^{(n-1)} = 0$ (2.2.40)

$$\check{\Phi}_{n}^{(n-1)} = \check{\Phi}_{n+1}^{(n-1)} = 0, \qquad \check{g}_{n+1,\delta}^{(n)} = 0, \qquad \delta \in \{+, -, 0\}.$$
(2.3.40)

The compatibility condition (2.2.17) is satisfied for the problem $\check{\mathcal{P}}_q^{(n+1)}$ for any $0 \leq q \leq n+1$. Indeed, we have already seen that it is satisfied for $\Phi = \Phi_j^{(n-1)}$, $g_{\delta} = g_{j,\delta}^{(n)}$, $\delta \in \{+, -, 0\}$ with any $j \in \mathbb{N}^*$ (these are the problems for the functions $V_j^{(n+1)}$). Taking into account the relations (2.3.34), (2.3.31) completed by (2.3.40), we get

$$\sum_{q=0}^{n+1} j^q \left(\left\langle \check{g}_{q,+}^{(n)}, 1 \right\rangle_{\Sigma_j^+} + \left\langle \check{g}_{q,-}^{(n)}, 1 \right\rangle_{\Sigma_j^-} + \left\langle \check{g}_{q,0}^{(n)}, 1 \right\rangle_{\Sigma_j^0} - \int_{K_j} \check{\Phi}_q^{(n-1)} \right) = 0, \qquad j \in \mathbb{N}^*.$$

This yields the compatibility condition for the problem $\check{\mathcal{P}}_q^{(n+1)}$ for any $0 \leq q \leq n+1$. Let us denote by $\mathcal{V}_q^{(n+1)}$ the unique solution of the problem $\check{\mathcal{P}}_q^{(n+1)}$, $0 \leq q \leq n+1$. Then, we get the relations (2.3.39) for the functions $\left\{V_j^{(n+1)}\right\}_{i\in\mathbb{N}^*}$.

• Extension to the unbounded domain \mathcal{J}_j : applying Proposition 2.2.1, we construct the near field terms $U_j^{(n+1)} \in H^1_{loc}(\mathcal{J}_j)$ that solve the problems $(\mathcal{P}_j^{(n+1)})$. We have already proved the symmetry property (2.3.38) for the functions $\left\{V_j^{(n+1)}\right\}_{j\in\mathbb{Z}}$. The symmetry property 2 for the functions $\left\{U_j^{(n+1)}\right\}_{j\in\mathbb{Z}}$ follows immediately from the symmetry property (2.3.38) of the functions $\left\{V_j^{(n+1)}\right\}_{j\in\mathbb{Z}}$. Similarly, the decay property 4 follows from (2.3.39) by taking $\mathcal{U}_q^{(n+1)}$ as the continuation to \mathcal{J}_* of the function $\mathcal{V}_q^{(n+1)}$ for any $0 \leq q \leq n+1$.

2.4 Construction of a pseudo-mode at any order

In this section we construct an appropriate pseudo-mode that will permit to prove Theorem 1.4.2.

Let us introduce a function $\chi \in C^{\infty}(\mathbb{R})$ such that

$$0 \leqslant \chi_0(x) \leqslant 1, \quad \forall x \in \mathbb{R}, \qquad \chi_0(x) = \begin{cases} 0, & x \ge 1, \\ 1, & x \leqslant 2. \end{cases}$$

We construct the pseudo-mode at order n as follows:

$$\mathcal{U}_{\varepsilon}(x,y) = \begin{cases} \mathcal{U}_{\varepsilon}^{(1)}(x) + \mathcal{U}_{\varepsilon}^{(2)}(y) + \mathcal{U}_{\varepsilon}^{(3)}(x,y), & (x,y) \in \mathcal{C}_{\varepsilon}^{\mu}, \\ \mathcal{U}_{\varepsilon}(x,-y), & (x,y) \in \Omega_{\varepsilon}^{\mu} \setminus \mathcal{C}_{\varepsilon}^{\mu}, \end{cases}$$
(2.4.1)

where

$$\mathcal{U}_{\varepsilon}^{(1)}(x) = \sum_{j \in \mathbb{Z}} \left(\sum_{k=0}^{n} \varepsilon^{k} u_{j+\frac{1}{2}}^{(k)}(x-j) \right) \chi\left(\frac{x-j}{\varepsilon^{\alpha}}\right) \chi\left(\frac{j+1-x}{\varepsilon^{\alpha}}\right), \qquad (2.4.2)$$

$$\mathcal{U}_{\varepsilon}^{(2)}(y) = \sum_{j \in \mathbb{Z}} \left(\sum_{k=0}^{n} \varepsilon^{k} u_{j}^{(k)}(y) \right) \chi\left(\frac{2y+L}{2\varepsilon^{\alpha}}\right), \qquad (2.4.3)$$

$$\mathcal{U}_{\varepsilon}^{(3)}(x,y) = \sum_{j \in \mathbb{Z}} \left(\sum_{k=0}^{n} \varepsilon^{k} U_{j}^{(k)} \left(\frac{x-j}{\varepsilon}, \frac{2y+L}{2\varepsilon} \right) \right) \left(1 - \chi \left(\frac{x-j}{\varepsilon^{\alpha}} \right) \right) \left(1 - \chi \left(\frac{j-x}{\varepsilon^{\alpha}} \right) \right) \left(1 - \chi \left(\frac{2y+L}{2\varepsilon^{\alpha}} \right) \right).$$

$$(2.4.4)$$



Figure 2.4: Construction of a pseudo-mode

For ε small enough we have the following estimate:

$$\|\mathcal{U}_{\varepsilon}\|_{L_{2}(\Omega_{\varepsilon}^{\mu})}^{2} \geqslant C_{n} \sum_{j \in \mathbb{Z}} \left(\int_{\frac{1}{4}}^{\frac{3}{4}} \int_{-\frac{L}{2}}^{\frac{L}{2} + \varepsilon} \left| u_{j+\frac{1}{2}}^{(0)}(s,y) \right|^{2} ds dy + \int_{-\frac{w_{j}^{\mu}\varepsilon}{2}}^{\frac{w_{j}^{\mu}\varepsilon}{2}} \int_{-\frac{L}{4}}^{0} \left| u_{j}^{(0)}(x,y) \right|^{2} dx dy \right) \geqslant C_{n}\varepsilon.$$

$$(2.4.5)$$

Here and in what follows we denote by C_n all the constants that do not depend on ε . In the rest of the section we prove the following estimate.

Proposition 2.4.1. For any $0 < \alpha < 1$ and $n \in \mathbb{N}$ there exist $\varepsilon(n, \alpha) > 0$ and $C(n, \alpha) > 0$ such that

$$\left| \int_{\Omega_{\varepsilon}^{\mu}} \left(\nabla \mathcal{U}_{\varepsilon} \nabla v - \lambda_{\varepsilon, n} \mathcal{U}_{\varepsilon} v \right) d\Omega \right| \leqslant C(n, \alpha) \varepsilon^{\alpha n + \frac{1}{2}} \|v\|_{H^{1}(\Omega_{\varepsilon}^{\mu})}, \quad \forall v \in H^{1}_{s}(\Omega_{\varepsilon}^{\mu}), \quad 0 < \varepsilon < \varepsilon(n, \alpha),$$

with

$$\lambda_{\varepsilon,n} = \sum_{k=0}^{n} \varepsilon^k \lambda^{(k)}.$$

Proof. Let us denote

$$\mathcal{I}_{\mathcal{U}_{\varepsilon}}(v) = \int_{\mathcal{C}_{\varepsilon}^{\mu}} \left(\nabla \mathcal{U}_{\varepsilon} \nabla v - \lambda_{\varepsilon, n} \mathcal{U}_{\varepsilon} v \right) d\Omega = \frac{1}{2} \int_{\Omega_{\varepsilon}^{\mu}} \left(\nabla \mathcal{U}_{\varepsilon} \nabla v - \lambda_{\varepsilon, n} \mathcal{U}_{\varepsilon} v \right) d\Omega.$$
(2.4.6)

After injecting (2.4.1)–(2.4.4) in (2.4.6) the value $\mathcal{I}_{\mathcal{U}_{\varepsilon}}(v)$ can be rewritten as

$$\mathcal{I}_{\mathcal{U}_{\varepsilon}}(v) = \mathcal{I}_{\mathcal{U}_{\varepsilon}^{(1)}}(v) + \mathcal{I}_{\mathcal{U}_{\varepsilon}^{(2)}}(v) + \mathcal{I}_{\mathcal{U}_{\varepsilon}^{(3)}}(v),$$

with

$$\begin{split} \mathcal{I}_{\mathcal{U}_{\varepsilon}^{(1)}}(v) &= \sum_{j \in \mathbb{Z}} \int_{\mathcal{C}_{\varepsilon}^{\mu}} \left(\sum_{k=0}^{n} \varepsilon^{k} \left(u_{j+\frac{1}{2}}^{(k)} \right)' (x-j) \right) \chi \left(\frac{x-j}{\varepsilon^{\alpha}} \right) \chi \left(\frac{j+1-x}{\varepsilon^{\alpha}} \right) \partial_{x} v(x,y) d\Omega \\ &\quad - \lambda_{\varepsilon,n} \sum_{j \in \mathbb{Z}} \int_{\mathcal{C}_{\varepsilon}^{\mu}} \left(\sum_{k=0}^{n} \varepsilon^{k} u_{j+\frac{1}{2}}^{(k)} (x-j) \right) \chi \left(\frac{x-j}{\varepsilon^{\alpha}} \right) \chi \left(\frac{j+1-x}{\varepsilon^{\alpha}} \right) v(x,y) d\Omega \\ &\quad + \varepsilon^{-\alpha} \sum_{j \in \mathbb{Z}} \int_{\mathcal{C}_{\varepsilon}^{\mu}} \left(\sum_{k=0}^{n} \varepsilon^{k} u_{j+\frac{1}{2}}^{(k)} (x-j) \right) \left(\chi' \left(\frac{x-j}{\varepsilon^{\alpha}} \right) - \chi' \left(\frac{j+1-x}{\varepsilon^{\alpha}} \right) \right) \partial_{x} v(x,y) d\Omega \\ &= -\sum_{j \in \mathbb{Z}} \sum_{k=n+1}^{2n} \sum_{p=k-n}^{n} \varepsilon^{k} \int_{\mathcal{C}_{\varepsilon}^{\mu}} \lambda^{(k-p)} u_{j+\frac{1}{2}}^{(p)} (x-j) \chi \left(\frac{x-j}{\varepsilon^{\alpha}} \right) \chi \left(\frac{j+1-x}{\varepsilon^{\alpha}} \right) v(x,y) d\Omega \\ &\quad - \varepsilon^{-\alpha} \sum_{j \in \mathbb{Z}} \int_{\mathcal{C}_{\varepsilon}^{\mu}} \left(\sum_{k=0}^{n} \varepsilon^{k} \left(\left(u_{j+\frac{1}{2}}^{(k)} \right)' (x-j) v(x,y) - u_{j+\frac{1}{2}}^{(k)} (x-j) \partial_{x} v(x,y) \right) \right) \right) \chi' \left(\frac{x-j}{\varepsilon^{\alpha}} \right) d\Omega \\ &\quad \varepsilon^{-\alpha} \sum_{j \in \mathbb{Z}} \int_{\mathcal{C}_{\varepsilon}^{\mu}} \left(\sum_{k=0}^{n} \varepsilon^{k} \left(\left(u_{j+\frac{1}{2}}^{(k)} \right)' (x-j) v(x,y) - u_{j+\frac{1}{2}}^{(k)} (x-j) \partial_{x} v(x,y) \right) \right) \right) \chi' \left(\frac{y+1-x}{\varepsilon^{\alpha}} \right) d\Omega, \end{split}$$

$$\mathcal{I}_{\mathcal{U}_{\varepsilon}^{(2)}}(v) = -\sum_{j\in\mathbb{Z}}\sum_{k=n+1}^{2n}\sum_{p=k-n}^{n}\varepsilon^{k}\int_{\mathcal{C}_{\varepsilon}^{\mu}}\lambda^{(k-p)}u_{j}^{(p)}(y)\chi\left(\frac{2y+L}{2\varepsilon^{\alpha}}\right)v(x,y)d\Omega$$
$$-\varepsilon^{-\alpha}\sum_{j\in\mathbb{Z}}\int_{\mathcal{C}_{\varepsilon}^{\mu}}\left(\sum_{k=0}^{n}\varepsilon^{k}\left(\left(u_{j}^{(k)}\right)'(y)v(x,y)-u_{j}^{(k)}(y)\partial_{y}v(x,y)\right)\right)\chi'\left(\frac{2y+L}{2\varepsilon^{\alpha}}\right)d\Omega,$$

$$\begin{split} \mathcal{I}_{\mathcal{U}_{\varepsilon}^{(3)}}(v) &= \\ -\sum_{j\in\mathbb{Z}}\sum_{k=n-1}^{n}\sum_{p=0}^{k}\varepsilon_{\mathcal{C}_{\varepsilon}^{\mu}}^{k}\lambda^{(k-p)}U_{j}^{(p)}\left(\frac{x-j}{\varepsilon},\frac{2y+L}{2\varepsilon}\right)\left(1-\chi\left(\frac{x-j}{\varepsilon^{\alpha}}\right)\right)\left(1-\chi\left(\frac{j-x}{\varepsilon^{\alpha}}\right)\right)\left(1-\chi\left(\frac{2y+L}{2\varepsilon^{\alpha}}\right)\right)v(x,y)d\Omega \\ -\sum_{j\in\mathbb{Z}}\sum_{k=n+1}^{2n}\sum_{p=k-n}^{n}\varepsilon_{\mathcal{C}_{\varepsilon}^{\mu}}^{k}\lambda^{(k-p)}U_{j}^{(p)}\left(\frac{x-j}{\varepsilon},\frac{2y+L}{2\varepsilon}\right)\left(1-\chi\left(\frac{x-j}{\varepsilon^{\alpha}}\right)\right)\left(1-\chi\left(\frac{j-x}{\varepsilon^{\alpha}}\right)\right)\left(1-\chi\left(\frac{2y+L}{2\varepsilon^{\alpha}}\right)\right)v(x,y)d\Omega \\ +\varepsilon^{-\alpha}\sum_{j\in\mathbb{Z}}\int_{\mathcal{C}_{\varepsilon}^{\mu}}\left(\sum_{k=0}^{n}\varepsilon^{k}\left(\partial_{x}U_{j}^{(k)}\left(\frac{x-j}{\varepsilon},\frac{2y+L}{2\varepsilon}\right)v(x,y)-U_{j}^{(k)}\left(\frac{x-j}{\varepsilon},\frac{2y+L}{2\varepsilon}\right)\partial_{x}v(x,y)\right)\right)\chi'\left(\frac{x-j}{\varepsilon^{\alpha}}\right)d\Omega \\ -\varepsilon^{-\alpha}\sum_{j\in\mathbb{Z}}\int_{\mathcal{C}_{\varepsilon}^{\mu}}\left(\sum_{k=0}^{n}\varepsilon^{k}\left(\partial_{x}U_{j}^{(k)}\left(\frac{x-j}{\varepsilon},\frac{2y+L}{2\varepsilon}\right)v(x,y)-U_{j}^{(k)}\left(\frac{x-j}{\varepsilon},\frac{2y+L}{2\varepsilon}\right)\partial_{x}v(x,y)\right)\right)\chi'\left(\frac{j-x}{\varepsilon^{\alpha}}\right)d\Omega \end{split}$$

$$+\varepsilon^{-\alpha}\sum_{j\in\mathbb{Z}}\int\limits_{\mathcal{C}_{\varepsilon}^{\mu}}\left(\sum_{k=0}^{n}\varepsilon^{k}\left(\partial_{y}U_{j}^{(k)}\left(\frac{x-j}{\varepsilon},\frac{2y+L}{2\varepsilon}\right)v(x,y)-U_{j}^{(k)}\left(\frac{x-j}{\varepsilon},\frac{2y+L}{2\varepsilon}\right)\partial_{y}v(x,y)\right)\right)\chi'\left(\frac{2y+L}{2\varepsilon^{\alpha}}\right)d\Omega.$$

Summing up these expressions we can regroup the terms in the following way:

$$\mathcal{I}_{\mathcal{U}_{\varepsilon}}(v) = \mathcal{I}_{\varepsilon,F}(v) + \mathcal{I}_{\varepsilon,N}(v) + \mathcal{I}_{\varepsilon,M}(v),$$

where

$$\mathcal{I}_{\varepsilon,F}(v) = \mathcal{I}_{\varepsilon,F}^{(1)}(v) + \mathcal{I}_{\varepsilon,F}^{(2)}(v),$$

$$\begin{aligned} \mathcal{I}_{\varepsilon,F}^{(1)}(v) &= -\sum_{j\in\mathbb{Z}} \sum_{k=n+1}^{2n} \sum_{p=k-n}^{n} \varepsilon^k \int\limits_{\mathcal{C}_{\varepsilon}^{\mu}} \lambda^{(k-p)} u_{j+\frac{1}{2}}^{(p)}(x-j) \chi\left(\frac{x-j}{\varepsilon^{\alpha}}\right) \chi\left(\frac{j+1-x}{\varepsilon^{\alpha}}\right) v(x,y) d\Omega, \\ \mathcal{I}_{\varepsilon,F}^{(2)}(v) &= -\sum_{j\in\mathbb{Z}} \sum_{k=n+1}^{2n} \sum_{p=k-n}^{n} \varepsilon^k \int\limits_{\mathcal{C}_{\varepsilon}^{\mu}} \lambda^{(k-p)} u_j^{(p)}(y) \chi\left(\frac{2y+L}{2\varepsilon^{\alpha}}\right) v(x,y) d\Omega, \end{aligned}$$

$$\begin{split} \mathcal{I}_{\varepsilon,N}(v) &= \\ -\sum_{j\in\mathbb{Z}}\sum_{k=n-1}^{n}\sum_{p=0}^{k}\varepsilon_{\mathcal{C}_{\varepsilon}^{\mu}}^{k}\lambda^{(k-p)}U_{j}^{(p)}\left(\frac{x-j}{\varepsilon},\frac{2y+L}{2\varepsilon}\right)\left(1-\chi\left(\frac{x-j}{\varepsilon^{\alpha}}\right)\right)\left(1-\chi\left(\frac{j-x}{\varepsilon^{\alpha}}\right)\right)\left(1-\chi\left(\frac{2y+L}{2\varepsilon^{\alpha}}\right)\right)v(x,y)d\Omega \\ -\sum_{j\in\mathbb{Z}}\sum_{k=n+1}^{2n}\sum_{p=k-n}^{n}\varepsilon_{\mathcal{C}_{\varepsilon}^{\mu}}^{k}\lambda^{(k-p)}U_{j}^{(p)}\left(\frac{x-j}{\varepsilon},\frac{2y+L}{2\varepsilon}\right)\left(1-\chi\left(\frac{x-j}{\varepsilon^{\alpha}}\right)\right)\left(1-\chi\left(\frac{j-x}{\varepsilon^{\alpha}}\right)\right)\left(1-\chi\left(\frac{2y+L}{2\varepsilon^{\alpha}}\right)\right)v(x,y)d\Omega, \\ \mathcal{I}_{\varepsilon,M}(v) &=\sum_{\delta\in\{+,-,0\}}\mathcal{I}_{\varepsilon,M}^{\delta}(v), \qquad \mathcal{I}_{\varepsilon,M}^{\delta}(v) = \mathcal{I}_{\varepsilon,M}^{\delta,1}(v) + \mathcal{I}_{\varepsilon,M}^{\delta,2}(v), \qquad \delta\in\{+,-,0\}, \\ \mathcal{I}_{\varepsilon,M}^{+,1}(v) &= -\varepsilon^{-\alpha}\sum_{j\in\mathbb{Z}}\sum_{k=0}^{n}\varepsilon^{k}\int_{\mathcal{C}_{\varepsilon}^{\mu}}\left(\left(u_{j+\frac{1}{2}}^{(k)}\right)'(x-j) - \partial_{x}U_{j}^{(k)}\left(\frac{x-j}{\varepsilon},\frac{2y+L}{2\varepsilon}\right)\right)\chi'\left(\frac{x-j}{\varepsilon^{\alpha}}\right)v(x,y)d\Omega, \end{split}$$

$$\mathcal{I}_{\varepsilon,M}^{+,2}(v) = \varepsilon^{-\alpha} \sum_{j \in \mathbb{Z}} \sum_{k=0}^{n} \varepsilon^{k} \int_{\mathcal{C}_{\varepsilon}^{\mu}} \left(u_{j+\frac{1}{2}}^{(k)}(x-j) - U_{j}^{(k)}\left(\frac{x-j}{\varepsilon}, \frac{2y+L}{2\varepsilon}\right) \right) \chi'\left(\frac{x-j}{\varepsilon^{\alpha}}\right) \partial_{x} v(x,y) d\Omega,$$

$$\mathcal{I}_{\varepsilon,M}^{-,1}(v) = \varepsilon^{-\alpha} \sum_{j \in \mathbb{Z}} \sum_{k=0}^{n} \varepsilon^{k} \int_{\mathcal{C}_{\varepsilon}^{\mu}} \left(\left(u_{j-\frac{1}{2}}^{(k)} \right)' (x+1-j) - \partial_{x} U_{j}^{(k)} \left(\frac{x-j}{\varepsilon}, \frac{2y+L}{2\varepsilon} \right) \right) \chi' \left(\frac{j-x}{\varepsilon^{\alpha}} \right) v(x,y) d\Omega,$$
$$\mathcal{I}_{\varepsilon,M}^{-,2}(v) = -\varepsilon^{-\alpha} \sum_{j \in \mathbb{Z}} \sum_{k=0}^{n} \varepsilon^{k} \int_{\mathcal{C}_{\varepsilon}^{\mu}} \left(u_{j-\frac{1}{2}}^{(k)} (x+1-j) - U_{j}^{(k)} \left(\frac{x-j}{\varepsilon}, \frac{2y+L}{2\varepsilon} \right) \right) \chi' \left(\frac{j-x}{\varepsilon^{\alpha}} \right) \partial_{x} v(x,y) d\Omega,$$

$$\begin{aligned} \mathcal{I}_{\varepsilon,M}^{0,1}(v) &= -\varepsilon^{-\alpha} \sum_{j \in \mathbb{Z}} \sum_{k=0}^{n} \varepsilon^{k} \int_{\mathcal{C}_{\varepsilon}^{\mu}} \left(\left(u_{j}^{(k)} \right)'(y) - \partial_{y} U_{j}^{(k)}\left(\frac{x-j}{\varepsilon}, \frac{2y+L}{2\varepsilon} \right) \right) \chi'\left(\frac{2y+L}{2\varepsilon^{\alpha}} \right) v(x,y) d\Omega, \\ \mathcal{I}_{\varepsilon,M}^{0,2}(v) &= \varepsilon^{-\alpha} \sum_{j \in \mathbb{Z}} \sum_{k=0}^{n} \varepsilon^{k} \int_{\mathcal{C}_{\varepsilon}^{\mu}} \left(u_{j}^{(k)}(y) - U_{j}^{(k)}\left(\frac{x-j}{\varepsilon}, \frac{2y+L}{2\varepsilon} \right) \right) \chi'\left(\frac{2y+L}{2\varepsilon^{\alpha}} \right) \partial_{y} v(x,y) d\Omega. \end{aligned}$$

The term $\mathcal{I}_{\varepsilon,F}(v)$ corresponds to the regions where the far field expansion holds and the term $\mathcal{I}_{\varepsilon,N}(v)$ corresponds to the regions where the near field expansion holds. Roughly speaking, these terms measure the error between the true eigenfunction and the constructed quasimode in the corresponding regions. The term $\mathcal{I}_{\varepsilon,M}(v)$ corresponds to the regions of matching. It can be seen as a measure of the difference between the two expansions in these regions.

• Estimation of the term $\mathcal{I}_{\varepsilon,F}(v)$

Using the Cauchy-Schwarz inequality, we get the following estimate:

$$\begin{aligned} \left| \mathcal{I}_{\varepsilon,F}^{(1)}(v) \right| \\ \leqslant C \sum_{k=n+1}^{2n} \sum_{p=k-n}^{n} \varepsilon^{k} \lambda^{(k-p)} \sum_{j \in \mathbb{Z}} \left(\int_{0}^{1} \int_{-\frac{L}{2}}^{-\frac{L}{2}+\varepsilon} \left| u_{j+\frac{1}{2}}^{(p)}(s) \right|^{2} ds dy \right)^{1/2} \left(\int_{j}^{j+1} \int_{-\frac{L}{2}}^{j+1-\frac{L}{2}+\varepsilon} |v(x,y)|^{2} dx dy \right)^{1/2} \\ \leqslant C \sum_{k=n+1}^{2n} \sum_{p=k-n}^{n} \varepsilon^{k+\frac{1}{2}} \lambda^{(k-p)} \left\| u^{(p)} \right\|_{L_{2}(G_{\varepsilon})} \left\| v \right\|_{L_{2}\left(\mathcal{C}_{\varepsilon}^{\mu}\right)} \leqslant C \varepsilon^{n+\frac{3}{2}} \| v \|_{H^{1}\left(\mathcal{C}_{\varepsilon}^{\mu}\right)}. \end{aligned}$$

$$(2.4.7)$$

The term $\mathcal{I}^{(2)}_{\varepsilon,F}(v)$ can be estimated in the same way.

• Estimation of the term $\mathcal{I}_{\varepsilon,N}(v)$

In order to estimate the term $\mathcal{I}_{\varepsilon,N}(v)$ we will use Lemma A.3.1 (cf. Annexe). First, assuming that $\alpha < 1$ we can estimate $\mathcal{I}_{\varepsilon,N}(v)$ as follows:

$$|\mathcal{I}_{\varepsilon,N}(v)| \leq C \left(\mathcal{I}_{\varepsilon,N}^{(0)}(v) + \mathcal{I}_{\varepsilon,N}^{(1)}(v) + \mathcal{I}_{\varepsilon,N}^{(2)}(v) + \mathcal{I}_{\varepsilon,N}^{(3)}(v) \right),$$

where

$$\begin{split} \mathcal{I}_{\varepsilon,N}^{(0)}(v) &= \sum_{j \in \mathbb{Z}} \int_{Q_{\varepsilon}^{j}} \left| S_{\varepsilon}^{j}(x,y)v(x,y) \right| d\Omega, \qquad Q_{\varepsilon}^{j} = \left[j - w_{j}^{\mu}\varepsilon, j + w_{j}^{\mu}\varepsilon \right] \times \left[-\frac{L}{2}, -\frac{L}{2} + 2\varepsilon \right], \\ \mathcal{I}_{\varepsilon,N}^{(1)}(v) &= \sum_{j \in \mathbb{Z}} \int_{j+\varepsilon}^{j+\varepsilon} \int_{-\frac{L}{2}}^{j+\varepsilon} \left| S_{\varepsilon}^{j}(x,y)v(x,y) \right| d\Omega, \\ \mathcal{I}_{\varepsilon,N}^{(2)}(v) &= \sum_{j \in \mathbb{Z}} \int_{j-\varepsilon^{\alpha}}^{j-\varepsilon} \int_{-\frac{L}{2}}^{j+\varepsilon} \left| S_{\varepsilon}^{j}(x,y)v(x,y) \right| d\Omega, \\ \mathcal{I}_{\varepsilon,N}^{(3)}(v) &= \sum_{j \in \mathbb{Z}} \int_{j-\frac{w_{j}^{\mu}\varepsilon}{2}}^{j+\frac{w_{j}^{\mu}\varepsilon}{2}} \int_{-\frac{L}{2}+2\varepsilon}^{j+\varepsilon^{\alpha}} \left| S_{\varepsilon}^{j}(x,y)v(x,y) \right| d\Omega, \\ S_{\varepsilon}^{j}(x,y) &= \sum_{k=n-1}^{n} \sum_{p=0}^{k} \varepsilon^{k} \lambda^{(k-p)} U_{j}^{(p)}\left(\frac{x-j}{\varepsilon}, \frac{2y+L}{2\varepsilon} \right) + \sum_{k=n+1}^{2n} \sum_{p=k-n}^{n} \varepsilon^{k} \lambda^{(k-p)} U_{j}^{(p)}\left(\frac{x-j}{\varepsilon}, \frac{2y+L}{2\varepsilon} \right). \end{split}$$

For $\mathcal{I}_{\varepsilon,N}^{(0)}(v)$ we have:

$$\mathcal{I}_{\varepsilon,N}^{(0)}(v) \leqslant \sum_{j\in\mathbb{Z}} \left(\sum_{k=n-1}^{n} \sum_{p=0}^{k} \varepsilon^{k+1} \left| \lambda^{(k-p)} \right| \left\| U_{j}^{(p)} \right\|_{L_{2}(K_{j})} + \sum_{k=n+1}^{2n} \sum_{p=k-n}^{n} \varepsilon^{k+1} \left| \lambda^{(k-p)} \right| \left\| U_{j}^{(p)} \right\|_{L_{2}(K_{j})} \right) \|v\|_{L_{2}(Q_{\varepsilon}^{j})} \\
\leqslant C\varepsilon^{n} \sum_{p=0}^{n} \left(\sum_{j\in\mathbb{Z}} \left\| U_{j}^{(p)} \right\|_{L_{2}(K_{j})}^{2} \right)^{1/2} \left(\sum_{j\in\mathbb{Z}} \|v\|_{L_{2}(Q_{\varepsilon}^{j})}^{2} \right)^{1/2} \leqslant C\varepsilon^{n+\frac{1}{2}} \|v\|_{H^{1}(\mathcal{C}_{\varepsilon}^{\mu})}. \quad (2.4.8)$$

Here we used the estimate (A.3.2) for the function v, fact that $U_j^{(k)} \in H^1_{loc}(\mathcal{J}_j), \forall j \in \mathbb{Z}, k \in \mathbb{N}$ as well as the exponential decay property (2.3.21) for the functions $\{U^{(p)}\}_{p=0}^n$.

In order to estimate the term $\mathcal{I}_{\varepsilon,N}^{(1)}(v)$ we notice that the behaviour of the functions $\{U^{(p)}\}_{p=0}^{n}$ in the band B_{j}^{+} (cf. (2.2.9), (2.2.11), (2.2.13)) implies the following inequality:

$$\begin{split} \left| S_{\varepsilon}^{j}(x,y) \right| \leqslant C_{n}(j) \left(\sum_{k=n-1}^{n} \sum_{p=0}^{k} \varepsilon^{k+p(\alpha-1)} + \sum_{k=n+1}^{2n} \sum_{p=k-n}^{n} \varepsilon^{k+p(\alpha-1)} \right) \leqslant C_{n}(j) \varepsilon^{(n-1)\alpha}, \\ (x,y) \in]j + \varepsilon, \ j + \varepsilon^{\alpha} [\times] - \frac{L}{2}, -\frac{L}{2} + \varepsilon [.$$

Due (A.3.1) and the exponential decay of $\{U^{(p)}\}_{p=0}^{n}$ (2.3.21) (which implies the exponential decay of constants $C_n(j)$ in j) we get:

$$\mathcal{I}_{\varepsilon,N}^{(1)}(v) \leqslant C_n \varepsilon^{n\alpha + \frac{1}{2}} \|v\|_{H^1(\mathcal{C}_{\varepsilon}^{\mu})}.$$
(2.4.9)

The terms $\mathcal{I}_{\varepsilon,N}^{(2)}(v)$ and $\mathcal{I}_{\varepsilon,N}^{(3)}(v)$ can be estimated analogously to $\mathcal{I}_{\varepsilon,N}^{(1)}(v)$.

• Estimation of the term $\mathcal{I}_{\varepsilon,M}(v)$

We will estimate the term $\mathcal{I}_{\varepsilon,M}^+(v)$, the estimation of the terms $\mathcal{I}_{\varepsilon,M}^-(v)$, $\mathcal{I}_{\varepsilon,M}^0(v)$ being analogous.

Using the relation (2.2.9) giving the behaviour of the functions $\{U^{(k)}\}_{k=0}^{n}$ in the band B_{j}^{+} , we get

$$\begin{aligned} \left|\mathcal{I}_{\varepsilon,M}^{+,1}(v)\right| \\ \leqslant C_n \varepsilon^{-\alpha} \left(\sum_{j \in \mathbb{N}} \int_{j+\varepsilon^{\alpha}}^{j+2\varepsilon^{\alpha}} \int_{-\frac{L}{2}}^{-\frac{L}{2}+\varepsilon} \left|\mathcal{P}_{\varepsilon}^j(x)v(x,y)\right| dxdy + \sum_{j \in \mathbb{N}} \int_{j+\varepsilon^{\alpha}}^{j+2\varepsilon^{\alpha}} \int_{-\frac{L}{2}}^{-\frac{L}{2}+\varepsilon} \left|\mathcal{E}_{\varepsilon}^j(x,y)v(x,y)\right| dxdy \right), \end{aligned}$$

where

$$\mathcal{P}^{j}_{\varepsilon}(x) = \sum_{k=0}^{n} \varepsilon^{k} \left(u^{(k)}_{j+\frac{1}{2}} \right)' (x-j) - \sum_{k=0}^{n} \varepsilon^{k-1} \left(P^{(k)}_{j,0,+} \right)' \left(\frac{x-j}{\varepsilon} \right),$$
$$\mathcal{E}^{j}_{\varepsilon}(x,y) = \sum_{k=0}^{n} \varepsilon^{k} \partial_{x} \left(\sum_{p \in \mathbb{N}^{*}} P^{(k)}_{j,p,+} \left(\frac{x-j}{\varepsilon} \right) e^{-\frac{p\pi(x-j)}{\varepsilon}} f_{p} \left(\frac{2y+L}{2\varepsilon} \right) \right).$$

Taking into account the definition (2.2.13) of the polynomials $P_{j,0,+}^{(k)}$ we can rewrite $\mathcal{P}_{\varepsilon}^{j}$ as

$$\mathcal{P}_{\varepsilon}^{j}(x) = \sum_{k=0}^{n} \varepsilon^{k} \left(\left(u_{j+\frac{1}{2}}^{(k)} \right)'(x-j) - \sum_{\ell=0}^{n-k-1} \left. \frac{d^{\ell} \left(u_{j+\frac{1}{2}}^{(k)} \right)'}{ds^{l}} \right|_{s=0} \frac{(x-j)^{\ell}}{\ell!} \right).$$

Hence,

$$\left|\mathcal{P}_{\varepsilon}^{j}(x)\right| \leqslant \sum_{k=0}^{n} \varepsilon^{k} \left\| \frac{d^{n-k+1} u_{j+\frac{1}{2}}^{(k)}}{ds^{n-k+1}} \right\|_{L_{\infty}([0,1])} (2\varepsilon^{\alpha})^{n-k} \leqslant C_{n}(j) \varepsilon^{\alpha n}, \qquad x \in [j+\varepsilon^{\alpha}, \, j+2\varepsilon^{\alpha}],$$

and using (A.3.1) we get

$$\sum_{j\in\mathbb{N}}\int_{j+\varepsilon^{\alpha}}\int_{-\frac{L}{2}}\int_{-\frac{L}{2}}|\mathcal{P}_{\varepsilon}^{j}(x)v(x,y)|\,dxdy\leqslant C_{n}\varepsilon^{\alpha(n+1)+\frac{1}{2}}\|v\|_{H^{1}(\mathcal{C}_{\varepsilon}^{\mu})}$$

Next, we obviously have the following estimate for the other term due to the decaying exponentials ($\alpha < 1$):

$$\sum_{j\in\mathbb{N}}\int_{j+\varepsilon^{\alpha}}\int_{-\frac{L}{2}}\int_{-\frac{L}{2}}|\mathcal{E}_{\varepsilon}^{j}(x,y)v(x,y)|\,dxdy \leqslant C_{n}(N)\varepsilon^{N}\|v\|_{H^{1}(\mathcal{C}_{\varepsilon}^{\mu})},\qquad\forall n\in\mathbb{N}.$$
(2.4.10)

Thus,

$$\left|\mathcal{I}_{\varepsilon,M}^{+,1}(v)\right| \leqslant C_n \varepsilon^{\alpha n + \frac{1}{2}} \|v\|_{H^1(\Omega_{\varepsilon}^{\mu,-})}.$$
(2.4.11)

Finally, for $\mathcal{I}^{+,2}_{\varepsilon,M}(v)$ we get

$$\begin{split} \left|\mathcal{I}_{\varepsilon,M}^{+,2}(v)\right| \leqslant \\ C\varepsilon^{-\alpha} \left(\sum_{j\in\mathbb{N}} \int\limits_{j+\varepsilon^{\alpha}}^{j+2\varepsilon^{\alpha}} \int\limits_{-\frac{L}{2}}^{-\frac{L}{2}+\varepsilon} \left|\mathcal{T}_{\varepsilon}^{j}(x)\partial_{x}v(x,y)\right| dxdy + \sum_{j\in\mathbb{N}} \int\limits_{j+\varepsilon^{\alpha}}^{j+2\varepsilon^{\alpha}} \int\limits_{-\frac{L}{2}}^{-\frac{L}{2}+\varepsilon} \left|\mathcal{R}_{\varepsilon}^{j}(x,y)\partial_{x}v(x,y)\right| dxdy\right), \end{split}$$

where

$$\mathcal{T}^{j}_{\varepsilon}(x) = \sum_{k=0}^{n} \varepsilon^{k} u^{(k)}_{j+\frac{1}{2}}(x-j) - \sum_{k=0}^{n} \varepsilon^{k} P^{(k)}_{j,0,+}\left(\frac{x-j}{\varepsilon}\right),$$
$$\mathcal{R}^{j}_{\varepsilon}(x,y) = \sum_{k=0}^{n} \sum_{p \in \mathbb{N}^{*}} \varepsilon^{k} P^{(k)}_{j,p,+}\left(\frac{x-j}{\varepsilon}\right) e^{-\frac{p\pi(x-j)}{\varepsilon}} f_{p}\left(\frac{2y+L}{2\varepsilon}\right).$$

Using the definition (2.2.13) of the polynomials $P_{j,0,+}^{(k)}$, the term $\mathcal{T}_{\varepsilon}^{j}$ can be rewritten as

$$\mathcal{T}_{\varepsilon}^{j}(x) = \sum_{k=0}^{n} \varepsilon^{k} \left(u_{j+\frac{1}{2}}^{(k)}(x-j) - \sum_{\ell=0}^{n-k} \left. \frac{d^{\ell} u_{j+\frac{1}{2}}^{(k)}}{ds^{\ell}} \right|_{s=0} \frac{(x-j)^{\ell}}{\ell!} \right),$$

which implies

$$\left|\mathcal{T}_{\varepsilon}^{j}(x)\right| \leqslant \sum_{k=0}^{n} \varepsilon^{k} \left\| \frac{d^{n-k+1}u_{j+\frac{1}{2}}^{(k)}}{ds^{n-k+1}} \right\|_{L_{\infty}([0,1])} (2\varepsilon^{\alpha})^{n-k} \leqslant C(j) \varepsilon^{\alpha n}, \qquad x \in [j+\varepsilon^{\alpha}, \, j+2\varepsilon^{\alpha}].$$

Thus, using Cauchy-Schwartz inequality we get:

$$\sum_{j\in\mathbb{N}}\int_{j+\varepsilon^{\alpha}}\int_{-\frac{L}{2}}^{-\frac{L}{2}+\varepsilon} \left|\mathcal{T}_{\varepsilon}^{j}(x)\partial_{x}v(x,y)\right| dxdy \leqslant C\varepsilon^{\alpha(n+1)+\frac{1}{2}} \|v\|_{H^{1}(\mathcal{C}_{\varepsilon}^{\mu})}.$$

Analogously to (2.4.10) we can estimate the term containing the decaying exponentials as follows:

$$\sum_{j\in\mathbb{N}}\int_{j+\varepsilon^{\alpha}}\int_{-\frac{L}{2}}^{\frac{j+2\varepsilon^{\alpha}}{2}-\frac{L}{2}+\varepsilon} \left|\mathcal{R}_{\varepsilon}^{j}(x,y)v(x,y)\right| dxdy \leqslant C_{n}(N)\varepsilon^{N} \|v\|_{H^{1}(\mathcal{C}_{\varepsilon}^{\mu})}, \qquad \forall N\in\mathbb{N}.$$

Finally,

$$\left|\mathcal{I}_{\varepsilon,M}^{+,2}(v)\right| \leqslant C_n \varepsilon^{\alpha n + \frac{1}{2}} \|v\|_{H^1(\mathcal{C}_{\varepsilon}^{\mu})}.$$
(2.4.12)

Summing up the estimates for all the terms (cf. (2.4.7), (2.4.8), (2.4.9), (2.4.11), (2.4.12)) we can conclude that for $0 < \alpha < 1$ there exists a constant $C(n, \alpha) > 0$ such that

$$|\mathcal{I}_{\mathcal{U}_{\varepsilon}}(v)| \leqslant C(n,\alpha)\varepsilon^{\alpha n + \frac{1}{2}} \|v\|_{H^{1}(\mathcal{C}_{\varepsilon}^{\mu})}, \qquad \forall v \in H^{1}_{s}(\Omega_{\varepsilon}^{\mu}),$$

for ε small enough. This finishes the proof.

Proof of Theorem 1.4.2. Proposition 2.4.1 together with the estimate (2.4.5) imply due to Lemma A.2.1 that

$$dist\left(\sigma\left(A_{\varepsilon}^{\mu}\right),\lambda_{\varepsilon,n}\right)\leqslant C(n,\alpha)\varepsilon^{\alpha n}.$$

for ε small enough. This is not exactly the estimate we need in order to prove Theorem 1.4.2. However, going up to the order n + 2, we would get

$$dist\left(\sigma\left(A_{\varepsilon}^{\mu}\right),\lambda_{\varepsilon,n+2}\right)\leqslant C(n+2,\alpha)\varepsilon^{\alpha(n+2)}\leqslant C(n+2)\varepsilon^{n+1},$$

if α is chosen close enough to 1. On the other hand,

$$\lambda_{\varepsilon,n+2} - \lambda_{\varepsilon,n} = \lambda^{(n+1)} \varepsilon^{n+1} + \lambda^{(n+2)} \varepsilon^{n+2}.$$

Hence, due to the triangle inequality, we finally obtain

$$dist\left(\sigma\left(A_{\varepsilon}^{\mu}\right),\lambda_{\varepsilon,n}\right)\leqslant C(n)\varepsilon^{n+1}$$

This finishes the proof of Theorem 1.4.2.

CHAPTER 3

TRAPPED MODES IN A LOCALLY PERTURBED PERIODIC LADDER: NUMERICAL STUDY

3.1 Goals and difficulties

In the previous chapter we dealt with an eigenvalue problem in an unbounded domain Ω_{ε} . This requires some special methods when trying to find numerical approximations for this problem. One of the most common methods is the Supercell method ([64, 10, 62]) which consists in considering a big bounded domain with periodic boundary conditions. The solution in the truncated domain converges to the exact one exponentially if the defect mode is exponentially decaying. However, this method appears to be costly, especially when the mode is not well confined.

We apply here another method developed by S. Fliss ([24]) which is based on the construction of an appropriate Dirichlet-to-Neumann (DtN) operator. In the framework of this method the initial eigenvalue problem set in an unbounded domain is replaced by a nonlinear eigenvalue problem set in a bounded domain containing the defect (this time the domain does not have to be big). This nonlinear eigenvalue problem can be discretized by finite element method and a Newton type algorithm can be applied to solve the discrete problem.

In section 3.2 we explain how to reduce the initial problem to a problem posed in a bounded domain and in section 3.3 we show how this last problem is discretized by finite element method.

3.2 DtN operator method

Let us remind the initial problem. It consists in finding values λ_{ε} such that there exists a non-trivial function $u_{\varepsilon} \in H^1_{\Delta}(\Omega^{\mu}_{\varepsilon})$ solving the problem

$$\begin{cases} -\Delta u_{\varepsilon} = \lambda_{\varepsilon}^{2} u_{\varepsilon} & \text{in } \Omega_{\varepsilon}^{\mu}, \\ \frac{\partial u_{\varepsilon}}{\partial n} \Big|_{\partial \Omega_{\varepsilon}^{\mu}} = 0. \end{cases}$$
(3.2.1)

To simplify the notation we will write in this section u_{ε} , λ_{ε} , keeping in mind that these values depend on μ . The idea of the DtN operator method is to find a problem which is equivalent to (3.2.1) but posed in a neighbourhood of the perturbation (the perturbed cell). The boundary conditions on the boundaries separating the perturbed cell from the rest of the domain (two half-bands) will contain the DtN operators which are defined using the corresponding problems in these half-bands.

3.2.1 Half-band problems

We denote $B_{\varepsilon}^{\pm} = \Omega_{\epsilon}^{\mu} \cap \{\pm x > 1\}, \Gamma_{\varepsilon}^{\pm} = \{\pm 1\} \times \left] - \frac{L}{2}, \frac{L}{2} \right[$ (cf. figure 3.1). Let us introduce



Figure 3.1: The half-bands B_{ε}^+ and B_{ε}^-

the following function spaces:

$$H^{1}_{\Delta}\left(B^{\pm}_{\varepsilon}\right) = \left\{ u \in H^{1}\left(B^{\pm}_{\varepsilon}\right), \quad \Delta u \in L_{2}\left(B^{\pm}_{\varepsilon}\right) \right\},$$
$$H^{1}_{\Delta,N}\left(B^{\pm}_{\varepsilon}\right) = \left\{ u \in H^{1}_{\Delta}\left(B^{\pm}_{\varepsilon}\right), \quad \frac{\partial u}{\partial n} \Big|_{\partial B^{\pm}_{\varepsilon} \setminus \Gamma^{\pm}_{\varepsilon}} = 0 \right\}$$

We denote by γ_0^{\pm} and γ_1^{\pm} the trace maps on $\Gamma_{\varepsilon}^{\pm}$:

$$\begin{split} \gamma_{0}^{\pm} &\in \mathcal{L}\left(H^{1}\left(B_{\varepsilon}^{\pm}\right), H^{1/2}\left(\Gamma_{\varepsilon}^{\pm}\right)\right): \qquad \forall u \in H^{1}\left(B_{\varepsilon}^{\pm}\right), \quad \gamma_{0}^{\pm}u = \left.u\right|_{\Gamma_{\varepsilon}^{\pm}}, \\ \gamma_{1}^{\pm} &\in \mathcal{L}\left(H_{\Delta}^{1}\left(B_{\varepsilon}^{\pm}\right), H^{-1/2}\left(\Gamma_{\varepsilon}^{\pm}\right)\right): \qquad \forall u \in H_{\Delta}^{1}\left(B_{\varepsilon}^{\pm}\right), \quad \gamma_{1}^{\pm}u = \mp \left.\frac{\partial u}{\partial x}\right|_{\Gamma_{\varepsilon}^{\pm}} \end{split}$$

Let us consider the operators A_{ε}^{\pm} defined by the relations

$$A_{\varepsilon}^{\pm}u = -\Delta u, \qquad D(A_{\varepsilon}^{\pm}) = \left\{ u \in H_{\Delta,N}^{1} \left(B_{\varepsilon}^{\pm} \right), \quad u|_{\Gamma_{\varepsilon}^{\pm}} = 0 \right\}.$$
(3.2.2)

Obviously, the two operators are unitarily-equivalent. Let \mathcal{U} be the unitary operator from $L_2(B_{\varepsilon}^+)$ to $L_2(B_{\varepsilon}^-)$ defined on smooth functions by the relation

$$(\mathcal{U}u)(x,y) = u(-x,y), \qquad (x,y) \in B_{\varepsilon}^{-},$$

and extended by continuity to $L_2(B_{\varepsilon}^+)$. Then,

$$D(A_{\varepsilon}^{-}) = \mathcal{U}D(A_{\varepsilon}^{+}), \qquad A_{\varepsilon}^{-} = \mathcal{U}A_{\varepsilon}^{+}\mathcal{U}^{-1}.$$

In fact, we do not have to distinguish between the two operators since the difference is purely geometric. Even if we still keep the symbols \pm in the notation in order to point out this geometric difference, all the results will be proved for the operator A_{ε}^+ , the analogous results for the operator A_{ε}^- being obvious. We will use the notation

$$d^{\pm}(\alpha) = \operatorname{dist}\left(\alpha^2, \sigma(A_{\varepsilon}^{\pm})\right).$$

As before, A_{ε} stands for the non-perturbed operator defined in Section 1.2.1 with empty discrete spectrum (cf. (1.2.1)).

Lemma 3.2.1.

$$\sigma(A_{\varepsilon}^{\pm}) = \sigma(A_{\varepsilon}). \tag{3.2.3}$$

Proof. Let \mathcal{P} be the operator of continuation by antisymmetry from $L_2(B_{\varepsilon}^+)$ to $L_2(\Omega_{\varepsilon})$ defined on smooth functions as

$$\left(\mathcal{P}u\right)(x,y) = \begin{cases} u(x,y), & (x,y) \in B_{\varepsilon}^{+}, \\ -u(1-x,y), & (x,y) \in \Omega_{\varepsilon} \setminus B_{\varepsilon}^{+} \end{cases}$$

and extended by continuity to $L_2(B_{\varepsilon}^+)$. Then, the operator \mathcal{P} transforms an eigenfunction of the operator A_{ε}^+ into an eigenfunction of the operator A_{ε} and any singular sequence of the operator A_{ε}^+ into a singular sequence of the operator A_{ε} , which proves the inclusion $\sigma(A_{\varepsilon}^+) \subset \sigma(A_{\varepsilon})$. Conversely, the operator $\widetilde{\mathcal{P}}$ from $L_2(\Omega_{\varepsilon})$ to $L_2(B_{\varepsilon}^+)$ defined on smooth functions as

$$(\mathcal{P}u)(x,y) = u(x,y) - u(1-x,y), \qquad (x,y) \in B_{\varepsilon}^+,$$

and extended by continuity to $L_2(\Omega_{\varepsilon})$ transforms an eigenfunction of the operator A_{ε} into an eigenfunction of the operator A_{ε}^+ and any singular sequence of the operator A_{ε} into a singular sequence of the operator A_{ε}^+ . This proves the inclusion $\sigma(A_{\varepsilon}) \subset \sigma(A_{\varepsilon}^+)$.

For any $\alpha \in \mathbb{R}$, $\varphi \in H^{1/2}(\Gamma_{\varepsilon}^{\pm})$ we consider the following half-band problems for $u_{\varepsilon}^{\pm} \in H^{1}_{\Delta,N}(B_{\varepsilon}^{\pm})$:

$$\begin{cases} \Delta u_{\varepsilon}^{\pm} + \alpha^2 u_{\varepsilon}^{\pm} = 0 & \text{in } B_{\varepsilon}^{\pm}, \\ u_{\varepsilon}^{\pm}|_{\Gamma_{\varepsilon}^{\pm}} = \varphi. \end{cases}$$
(3.2.4)

The following statement establishes the well-posedness of these problems.

Proposition 3.2.1. For any $\alpha^2 \notin \sigma(A_{\varepsilon})$, $\varphi \in H^{1/2}(\Gamma_{\varepsilon}^{\pm})$ the problems (3.2.4) have unique solutions. Moreover, the following estimates for the norms of the solutions hold:

$$\|u_{\varepsilon}^{\pm}\|_{H^{1}\left(B_{\varepsilon}^{\pm}\right)} \leqslant \left(C_{1}(\alpha) + \frac{C_{2}(\alpha)}{d^{\pm}(\alpha)}\right) \|\varphi\|_{H^{1/2}\left(\Gamma_{\varepsilon}^{\pm}\right)},\tag{3.2.5}$$

where $C_1(\alpha)$, $C_2(\alpha)$ are continuous functions of α depending only on the domains B_{ε}^{\pm} .

Proof. Let us denote by $\widehat{\varphi}$ a lift function belonging to $H^1(B_{\varepsilon}^+)$ such that $\widehat{\varphi}|_{\Gamma_{\varepsilon}^+} = \varphi$ and

$$\|\widehat{\varphi}\|_{H^1\left(B_{\varepsilon}^+\right)} \leqslant C_+ \|\varphi\|_{H^{1/2}\left(\Gamma_{\varepsilon}^+\right)},\tag{3.2.6}$$

where the constant C_+ depends only on the domain B_{ε}^+ . We also introduce the following function space:

$$H_D^1\left(B_{\varepsilon}^+\right) = \left\{ u \in H^1\left(B_{\varepsilon}^+\right), \quad u|_{\Gamma_{\varepsilon}^+} = 0 \right\},$$

which we equip with H^1 -norm. Multiplying the first line of (3.2.4) by a test function $v \in H^1_D(B^+_{\varepsilon})$ and changing the unknown function by

$$\widehat{u} = u_{\varepsilon}^{+} - \widehat{\varphi}, \qquad (3.2.7)$$

we get the following variational problem for $\widehat{u} \in H_D^1(B_{\varepsilon}^+)$:

$$\left(\nabla\widehat{u},\nabla v\right)_{L_2\left(B_{\varepsilon}^+\right)} - \alpha^2\left(\widehat{u},v\right)_{L_2\left(B_{\varepsilon}^+\right)} = -\left(\nabla\widehat{\varphi},\nabla v\right)_{L_2\left(B_{\varepsilon}^+\right)} + \alpha^2\left(\widehat{\varphi},v\right)_{L_2\left(B_{\varepsilon}^+\right)}, \quad \forall v \in H_D^1\left(B_{\varepsilon}^+\right).$$

$$(3.2.8)$$

The right-hand side of (3.2.8) is a continuous linear functional in v with respect to the norm H^1 :

$$\left| - \left(\nabla \widehat{\varphi}, \nabla v \right)_{L_2\left(B_{\varepsilon}^+\right)} + \alpha^2 \left(\widehat{\varphi}, v \right)_{L_2\left(B_{\varepsilon}^+\right)} \right| \leqslant C_{\varphi}(\alpha) \left\| v \right\|_{H^1\left(B_{\varepsilon}^+\right)}, \qquad \forall v \in H_D^1\left(B_{\varepsilon}^+\right),$$

where

$$C_{\varphi}(\alpha) = (1 + \alpha^2) \left\| \widehat{\varphi} \right\|_{H^1(B_{\varepsilon}^+)}.$$
(3.2.9)

Hence, there exists a function $f \in H_D^1(B_{\varepsilon}^+)$ such that

$$-\left(\nabla\widehat{\varphi},\nabla v\right)_{L_2\left(B_{\varepsilon}^+\right)} + \alpha^2\left(\widehat{\varphi},v\right)_{L_2\left(B_{\varepsilon}^+\right)} = (f,v)_{H^1\left(B_{\varepsilon}^+\right)}, \qquad \forall v \in H_D^1\left(B_{\varepsilon}^+\right). \tag{3.2.10}$$

Besides, $\|f\|_{H^1(B_{\varepsilon}^+)} \leq C_{\varphi}(\alpha)$. Consequently, the problem (3.2.8) can be rewritten as

$$(\widehat{u}, v)_{H^1\left(B_{\varepsilon}^+\right)} - (1 + \alpha^2) \left(\widehat{u}, v\right)_{L_2\left(B_{\varepsilon}^+\right)} = (f, v)_{H^1\left(B_{\varepsilon}^+\right)}, \qquad \forall v \in H_D^1\left(B_{\varepsilon}^+\right).$$
(3.2.11)

Due to the estimate

$$\left| \left(u, v \right)_{L_2\left(B_{\varepsilon}^+ \right)} \right| \leq \left\| u \right\|_{H^1\left(B_{\varepsilon}^+ \right)} \left\| v \right\|_{H^1\left(B_{\varepsilon}^+ \right)}, \qquad \forall u, v \in H_D^1\left(B_{\varepsilon}^+ \right),$$

we can introduce the bounded self-adjoint operator $K(\alpha) \in \mathcal{L}(H_D^1(B_{\varepsilon}^+))$ defined by the relation

$$(\widehat{u}, v)_{H^1\left(B_{\varepsilon}^+\right)} - (1 + \alpha^2) \left(\widehat{u}, v\right)_{L_2\left(B_{\varepsilon}^+\right)} = (K(\alpha)\widehat{u}, v)_{H^1\left(B_{\varepsilon}^+\right)}, \qquad \forall v \in H_D^1\left(B_{\varepsilon}^+\right). \quad (3.2.12)$$

Finally, the problem (3.2.11) takes the form

$$K(\alpha)\widehat{u} = f. \tag{3.2.13}$$

We remark that this problem is equivalent to the initial problem (3.2.4). Indeed, we have just shown that if u_{ε}^+ is a solution of (3.2.4), then the function \hat{u} defined in (3.2.7) is a solution of (3.2.13) with f defined in (3.2.10). Conversely, if \hat{u} is a solution of (3.2.13) with f defined in (3.2.10), then, repeating the argument in the other sense, we arrive at the variational formulation (3.2.8) which is equivalent to (3.2.4) with the change of unknown function (3.2.7). Therefore, in order to establish the well-posedness of (3.2.4) it is enough to prove the well-posedness of (3.2.13). Let us denote $d_K(\alpha) = \text{dist}(0, \sigma(K(\alpha)))$. We will use the result proved in Lemma 3.2.2 below which states that

$$d_K(\alpha) \ge \frac{d(\alpha)}{d(\alpha) + 1 + \alpha^2}.$$
(3.2.14)

Since by assumption $\alpha^2 \notin \sigma(A_{\varepsilon})$, due to Lemma 3.2.1 we have $\alpha^2 \notin \sigma(A_{\varepsilon}^+)$. Hence, $d(\alpha) > 0$ and (3.2.14) implies that $d_K(\alpha) > 0$. This shows that the problem (3.2.13) is well-posed. Moreover, the norm of its solution can be estimated as follows:

$$\|\widehat{u}\|_{H^{1}(B_{\varepsilon}^{+})} \leqslant \|K^{-1}\|_{H^{1}(B_{\varepsilon}^{+}) \to H^{1}(B_{\varepsilon}^{+})} \|f\|_{H^{1}(B_{\varepsilon}^{+})} \leqslant \frac{C_{\varphi}(\alpha)}{d_{K}(\alpha)}.$$
(3.2.15)

Putting together the relations (3.2.7), (3.2.6), (3.2.9), (3.2.14), (3.2.15) we get:

$$\|u_{\varepsilon}^{+}\|_{H^{1}(B_{\varepsilon}^{+})} \leqslant \left(C_{+}+1+\alpha^{2}+\frac{(1+\alpha^{2})^{2}}{d(\alpha)}\right)\|\varphi\|_{H^{1/2}(\Gamma_{\varepsilon}^{+})}.$$

This is exactly the estimate (3.2.5) with $C_1(\alpha) = C_+ + 1 + \alpha^2$, $C_2(\alpha) = (1 + \alpha^2)^2$. \Box

Remark 3.2.1. The uniqueness of solutions of the problems (3.2.4) is obvious since Lemma 3.2.1 implies that the operators A_{ε}^{\pm} have no eigenvalues. The goal of the above proof is to establish the existence. However, the argument used in the proof establishes the existence and the uniqueness simultaneously (the reason for which we talk about well-posedness).

Remark 3.2.2. The proof of Proposition 3.2.1 would be more straightforward for $\varphi \in H^{3/2}(\Gamma_{\varepsilon}^{\pm})$. In this case it would be unnecessary to pass to the weak formulation and the well-posedness would be directly guaranteed by the fact that $\alpha^2 \notin \sigma(A_{\varepsilon}^{\pm})$. However, we need to consider $\varphi \in H^{1/2}(\Gamma_{\varepsilon}^{\pm})$ since the problem will be discretized by P1 finite elements (cf. section 3.3) which implies working with H^1 functions, and, consequently, $H^{1/2}$ on the boundary.

Lemma 3.2.2. In the notation of the proof of Proposition 3.2.1 the estimate (3.2.14) holds.

Proof. If $d_K(\alpha) \ge 1$ the result is obvious. Suppose that $d_K(\alpha) < 1$. Then, since $\sigma(K(\alpha))$ is a closed subset of \mathbb{R} , there exists $\gamma \in \mathbb{R}$ such that $|\gamma| = d_k(\alpha)$ and $\gamma \in \sigma(K(\alpha))$ (in other words, either $d_K(\alpha)$ or $-d_k(\alpha)$ belongs to $\sigma(K(\alpha))$). The operator $K(\alpha)$ being self-adjoint, this means that there exists a singular sequence $\{u_n\}_{n\in\mathbb{N}} \subset H^1_D(B^+_{\varepsilon})$ such that

1.
$$||u_n||_{H^1(B_{\varepsilon}^+)} = 1, \quad n \in \mathbb{N},$$

2.
$$\|(K(\alpha) - \gamma I) u_n\|_{H^1(B_c^+)} \to 0, \quad n \to \infty,$$

where I is the identity operator in $H^1(B_{\varepsilon}^+)$. This implies that for any $\delta > 0$ there exists $u_{\delta} \in H^1_D(B_{\varepsilon}^+)$, $\|u_{\delta}\|_{H^1(B_{\varepsilon}^+)} = 1$, such that $\|(K(\alpha) - \gamma I) u_{\delta}\|_{H^1(B_{\varepsilon}^+)} \leq \delta$. Hence,

$$\left| \left((K(\alpha) - \gamma I) \, u_{\delta}, v \right)_{H^{1}(B_{\varepsilon}^{+})} \right| \leq \left\| (K(\alpha) - \gamma I) \, u_{\delta} \right\|_{H^{1}(B_{\varepsilon}^{+})} \| v \|_{H^{1}(B_{\varepsilon}^{+})} \\ \leq \delta \| u_{\delta} \|_{H^{1}(B_{\varepsilon}^{+})} \| v \|_{H^{1}(B_{\varepsilon}^{+})}, \quad \forall v \in H^{1}_{D}(B_{\varepsilon}^{+}).$$
(3.2.16)

Using the definition (3.2.12) of the operator $K(\alpha)$, we get:

$$\left| (u_{\delta}, v)_{H^{1}(B_{\varepsilon}^{+})} - \frac{1 + \alpha^{2}}{1 - \gamma} (u_{\delta}, v)_{L_{2}(B_{\varepsilon}^{+})} \right| \leqslant \frac{\delta \|u_{\delta}\|_{H^{1}(B_{\varepsilon}^{+})} \|v\|_{H^{1}(B_{\varepsilon}^{+})}}{1 - \gamma}, \qquad \forall v \in H^{1}_{D}(B_{\varepsilon}^{+})$$

The sesquilinear form $a[u, v] = (u_{\delta}, v)_{H^{1}(B_{\varepsilon}^{+})}$ defined on $D[a] = H_{D}^{1}(B_{\varepsilon}^{+})$ is the unique sesquilinear form corresponding to the operator $A_{\varepsilon}^{+} + I$. It obviously satisfies the estimate $a[u, u] \ge ||u||_{L_{2}(B_{\varepsilon}^{+})}^{2}$, $\forall u \in D[a]$, which permits us to apply Lemma A.2.1. Thus, we can conclude that

dist
$$\left(\sigma\left(A_{\varepsilon}^{+}+I\right), \frac{1+\alpha^{2}}{1-\gamma}\right) < C\delta, \quad \forall \delta > 0.$$

Since δ can be chosen arbitrarily small, one gets $\frac{1+\alpha^2}{1-\gamma} \in \sigma(A_{\varepsilon}^++I)$, which is equivalent to $\frac{\gamma+\alpha^2}{1-\gamma} \in \sigma(A_{\varepsilon}^+)$. This means that $d \leq \frac{|\gamma|(1+\alpha^2)}{1-\gamma} \leq \frac{d_K(\alpha)(1+\alpha^2)}{1-d_K(\alpha)}$. This yields (3.2.14).

From now on $u_{\varepsilon}^{\pm}(\alpha, \varphi)$ will stand for the unique solutions of the problems (3.2.4) for $\alpha^2 \notin \sigma(A_{\varepsilon}), \ \varphi \in H^{1/2}(\Gamma_{\varepsilon}^{\pm}).$

3.2.2 The DtN operators Λ^{\pm}

We define the DtN operators $\Lambda^{\pm}: H^{1/2}(\Gamma_{\varepsilon}^{\pm}) \to H^{-1/2}(\Gamma_{\varepsilon}^{\pm})$ for $\alpha^2 \notin \sigma(A_{\varepsilon})$ as follows:

$$\left\langle \Lambda^{\pm}(\alpha)\varphi,\psi\right\rangle = \int_{B_{\varepsilon}^{\pm}} \nabla u_{\varepsilon}^{\pm}(\alpha,\varphi)\nabla u_{\varepsilon}^{\pm}(\alpha,\psi) - \alpha^{2} \int_{B_{\varepsilon}^{\pm}} u_{\varepsilon}^{\pm}(\alpha,\varphi)u_{\varepsilon}^{\pm}(\alpha,\psi), \qquad \forall\varphi,\psi\in H^{1/2}\left(\Gamma_{\varepsilon}^{\pm}\right).$$
(3.2.17)

Here $u_{\varepsilon}^{\pm}(\alpha, \varphi)$ is the solution of the problem (3.2.4) and $\langle \cdot, \cdot \rangle$ stands for the duality brackets between $H^{-1/2}(\Gamma_{\varepsilon}^{\pm})$ and $H^{1/2}(\Gamma_{\varepsilon}^{\pm})$. In other words,

$$\Lambda^{\pm}(\alpha)\varphi = \left.\frac{\partial u_{\varepsilon}^{\pm}(\alpha,\varphi)}{\partial n}\right|_{\Gamma_{\varepsilon}^{\pm}}, \qquad \forall \varphi \in H^{1/2}\left(\Gamma_{\varepsilon}^{\pm}\right).$$
(3.2.18)

The following assertion which states the norm continuity of the DtN operators with respect to α will be used in Proposition 3.2.5 in order to show the continuity of the functions to which a Newton type algorithm will be applied.

Proposition 3.2.2. The operators $\Lambda^{\pm}(\alpha)$ defined by (3.2.17) are continuous from $H^{1/2}(\Gamma_{\varepsilon}^{\pm})$ to $H^{-1/2}(\Gamma_{\varepsilon}^{\pm})$ and norm-continuous with respect to α :

$$\left\|\Lambda^{\pm}(\alpha_1) - \Lambda^{\pm}(\alpha_2)\right\|_{H^{1/2}\left(\Gamma_{\varepsilon}^{\pm}\right) \to H^{-1/2}\left(\Gamma_{\varepsilon}^{\pm}\right)} \leqslant C \left|\alpha_1^2 - \alpha_2^2\right|.$$
(3.2.19)

Proof. Due to the estimate (3.2.5) one finds

$$\left|\left\langle\Lambda^{\pm}(\alpha)\varphi,\psi\right\rangle\right| \leqslant (1+\alpha^2) \left(C_1(\alpha) + \frac{C_2(\alpha)}{d^{\pm}(\alpha)}\right)^2 \|\varphi\|_{H^{1/2}\left(\Gamma_{\varepsilon}^{\pm}\right)} \|\psi\|_{H^{1/2}\left(\Gamma_{\varepsilon}^{\pm}\right)}.$$

Consequently,

$$\left\|\Lambda^{\pm}(\alpha)\right\|_{H^{1/2}\left(\Gamma_{\varepsilon}^{\pm}\right)\to H^{-1/2}\left(\Gamma_{\varepsilon}^{\pm}\right)} \leqslant (1+\alpha^{2})\left(C_{1}(\alpha)+\frac{C_{2}(\alpha)}{d^{\pm}(\alpha)}\right)^{2}.$$

If $\alpha_1, \alpha_2 \notin \sigma(A_{\varepsilon}^{\pm})$ then one can estimate the norm of the difference $\Lambda^{\pm}(\alpha_1) - \Lambda^{\pm}(\alpha_2)$. Let us denote $\widetilde{u}_{\varepsilon}^{\pm}(\varphi) = u_{\varepsilon}^{\pm}(\alpha_1, \varphi) - u_{\varepsilon}^{\pm}(\alpha_2, \varphi)$. Then for any $\varphi, \psi \in H^{1/2}(\Gamma_{\varepsilon}^{\pm})$ we have:

$$\left\langle \left(\Lambda^{\pm}(\alpha_{1}) - \Lambda^{\pm}(\alpha_{2}) \right) \varphi, \psi \right\rangle = \int_{B_{\varepsilon}^{\pm}} \left(\nabla \widetilde{u}_{\varepsilon}^{\pm}(\varphi) \nabla u_{\varepsilon}^{\pm}(\alpha_{1}, \psi) + \nabla u_{\varepsilon}^{\pm}(\alpha_{2}, \varphi) \nabla \widetilde{u}_{\varepsilon}^{\pm}(\psi) \right) - \alpha_{1}^{2} \int_{B_{\varepsilon}^{\pm}} \left(\widetilde{u}_{\varepsilon}^{\pm}(\varphi) u_{\varepsilon}^{\pm}(\alpha_{1}, \psi) + u_{\varepsilon}^{\pm}(\alpha_{2}, \varphi) \widetilde{u}_{\varepsilon}^{\pm}(\psi) \right) + \left(\alpha_{2}^{2} - \alpha_{1}^{2} \right) \int_{B_{\varepsilon}^{\pm}} u_{\varepsilon}^{\pm}(\alpha_{2}, \varphi) u_{\varepsilon}^{\pm}(\alpha_{2}, \psi).$$

$$(3.2.20)$$

Remark that $\widetilde{u}_{\varepsilon}^{\pm}(\varphi)$ solves the problem

$$A_{\varepsilon}^{\pm} \widetilde{u}_{\varepsilon}^{\pm}(\varphi) - \alpha_{1}^{2} \widetilde{u}_{\varepsilon}^{\pm}(\varphi) = \left(\alpha_{1}^{2} - \alpha_{2}^{2}\right) u_{\varepsilon}^{\pm}(\alpha_{2}, \varphi).$$
(3.2.21)

Therefore, its L_2 -norm can be estimated as follows:

$$\left\|\widetilde{u}_{\varepsilon}^{\pm}(\varphi)\right\|_{L_{2}(B_{\varepsilon}^{\pm})} \leqslant \frac{\left|\alpha_{1}^{2} - \alpha_{2}^{2}\right| \left\|u_{\varepsilon}^{\pm}(\alpha_{2},\varphi)\right\|_{L_{2}(B_{\varepsilon}^{\pm})}}{d^{\pm}(\alpha_{1})}.$$
(3.2.22)

Multiplying (3.2.21) by $\widetilde{u}_{\varepsilon}^{\pm}(\varphi)$ and integrating by parts yields

$$\left\|\nabla \widetilde{u}_{\varepsilon}^{\pm}(\varphi)\right\|_{L_{2}(B_{\varepsilon}^{\pm})}^{2} = \alpha_{1}^{2} \left\|\widetilde{u}_{\varepsilon}^{\pm}(\varphi)\right\|_{L_{2}(B_{\varepsilon}^{\pm})}^{2} + \left(\alpha_{1}^{2} - \alpha_{2}^{2}\right) \left(u_{\varepsilon}^{\pm}(\alpha_{2},\varphi), \widetilde{u}_{\varepsilon}^{\pm}(\varphi)\right)_{L_{2}(B_{\varepsilon}^{\pm})}.$$

Thus, taking into account (3.2.22), on obtains:

$$\left\|\widetilde{u}_{\varepsilon}^{\pm}(\varphi)\right\|_{H^{1}(B_{\varepsilon}^{\pm})} \leqslant \left(\alpha_{1}^{2}+1+d^{\pm}(\alpha_{1})\right)^{1/2} \frac{\left|\alpha_{1}^{2}-\alpha_{2}^{2}\right| \left\|u_{\varepsilon}^{\pm}(\alpha_{2},\varphi)\right\|_{L_{2}(B_{\varepsilon}^{\pm})}}{d^{\pm}(\alpha_{1})}.$$

In other terms, the operator $(A_{\varepsilon}^{\pm} - \alpha_1^2)^{-1}$ is continuous from $L_2(B_{\varepsilon}^{\pm})$ to $H^1(B_{\varepsilon}^{\pm})$. Finally, using (3.2.5) we conclude that

$$\left\|\widetilde{u}_{\varepsilon}^{\pm}(\varphi)\right\|_{H^{1}(B_{\varepsilon}^{\pm})} \leqslant C(\alpha_{1},\alpha_{2}) \frac{\left|\alpha_{1}^{2}-\alpha_{2}^{2}\right| \left\|\varphi\right\|_{H^{1/2}\left(\Gamma_{\varepsilon}^{\pm}\right)}}{d^{\pm}(\alpha_{1})d^{\pm}(\alpha_{2})}, \qquad \forall \varphi \in H^{1/2}\left(\Gamma_{\varepsilon}^{\pm}\right)$$

where $C(\alpha_1, \alpha_2)$ is a continuous function of α_1, α_2 . Notice that it depends on $d^{\pm}(\alpha_1)$, $d^{\pm}(\alpha_2)$. Coming back to (3.2.20) we get the following estimate for any $\varphi, \psi \in H^{1/2}(\Gamma_{\varepsilon}^{\pm})$:

$$\begin{split} \left| \left\langle \left(\Lambda^{\pm}(\alpha_1) - \Lambda^{\pm}(\alpha_2) \right) \varphi, \psi \right\rangle \right| \\ \leqslant C(\alpha_1, \alpha_2) \frac{|\alpha_1^2 - \alpha_2^2| \, \|\varphi\|_{H^{1/2}\left(\Gamma_{\varepsilon}^{\pm}\right)} \, \|\psi\|_{H^{1/2}\left(\Gamma_{\varepsilon}^{\pm}\right)}}{d^{\pm}(\alpha_1) d^{\pm}(\alpha_2)} \left(1 + \frac{1}{d^{\pm}(\alpha_1)} + \frac{1}{d^{\pm}(\alpha_2)} \right). \end{split}$$

Here $C(\alpha_1, \alpha_2)$ stands for some continuous function of α_1 , α_2 (not necessarily the same as above). The obtained estimate implies that if $d^{\pm}(\alpha_1), d^{\pm}(\alpha_2) \ge c_0 > 0$ then the inequality (3.2.19) holds.

We will need the following technical result.

Lemma 3.2.3 (Gårding's inequality for the DtN operators).

For $\alpha^2 \notin \sigma(A_{\varepsilon})$ there exist a constant C_1 and a continuous function $C_2(\alpha)$ depending only on the geometry of the domain such that

$$\langle \Lambda^{\pm}(\alpha)\varphi,\varphi\rangle \geqslant C_1 \|\varphi\|_{H^{1/2}\left(\Gamma_{\varepsilon}^{\pm}\right)}^2 - C_2(\alpha) \|\varphi\|_{L_2\left(\Gamma_{\varepsilon}^{\pm}\right)}^2, \qquad \forall \varphi \in H^{1/2}\left(\Gamma_{\varepsilon}^{\pm}\right).$$
(3.2.23)

Proof. From (3.2.17) we get:

$$\langle \Lambda^{\pm}(\alpha)\varphi,\varphi\rangle = \left\| u_{\varepsilon}^{\pm}(\alpha,\varphi) \right\|_{H^{1}\left(B_{\varepsilon}^{\pm}\right)}^{2} - (1+\alpha^{2}) \left\| u_{\varepsilon}^{\pm}(\alpha,\varphi) \right\|_{L_{2}\left(B_{\varepsilon}^{\pm}\right)}^{2}.$$

The trace applications γ_0^{\pm} being continuous, this implies that

$$\left\langle \Lambda^{\pm}(\alpha)\varphi,\varphi\right\rangle \geqslant C_1 \left\|\varphi\right\|_{H^{1/2}\left(\Gamma_{\varepsilon}^{\pm}\right)}^2 - (1+\alpha^2) \left\|u_{\varepsilon}^{\pm}(\alpha,\varphi)\right\|_{L_2\left(B_{\varepsilon}^{\pm}\right)}^2.$$
(3.2.24)

In order to estimate the L_2 -norm of the solutions $u_{\varepsilon}^{\pm}(\alpha, \varphi)$ let us consider the following problems with $u_{\varepsilon}^{\pm}(\alpha, \varphi)$ being source terms:

$$A^{\pm}_{\varepsilon}v^{\pm} - \alpha^2 v^{\pm} = u^{\pm}_{\varepsilon}(\alpha, \varphi).$$
(3.2.25)

Since $\alpha^2 \notin \sigma(A_{\varepsilon}^{\pm})$ this problems have unique solutions in $D(A_{\varepsilon}^{\pm})$. Multiplying (3.2.25) by $u_{\varepsilon}^{\pm}(\alpha, \varphi)$ and using Green's formula we get:

$$\left\| u_{\varepsilon}^{\pm}(\alpha,\varphi) \right\|_{L_{2}\left(B_{\varepsilon}^{\pm}\right)}^{2} = -\left\langle \frac{\partial v^{\pm}}{\partial n},\varphi \right\rangle.$$
(3.2.26)

The domains B_{ε}^{\pm} are not convex, which makes it impossible to use the argument of global regularity of weak solutions. Nevertheless, it is still possible to use the local regularity near the boundaries that do not contain reentrant angles. More precisely, let K^{\pm} stand for the rectangles $]1, 1 + \frac{\varepsilon}{8} [\times] - \frac{L}{2}, \frac{L}{2} [$ and $]-1 - \frac{\varepsilon}{8}, -1, [\times] - \frac{L}{2}, \frac{L}{2} [$ respectively. Then there exists a continuous function $C(\alpha)$ depending only on the geometry of the domain such that

$$\left\|v^{\pm}\right\|_{H^{2}(K^{\pm})} \leqslant C(\alpha) \left\|u^{\pm}_{\varepsilon}\right\|_{L_{2}\left(B^{\pm}_{\varepsilon}\right)}.$$
(3.2.27)

We detail the proof of this fact in Lemma A.3.3 given in Annexe. Then, the continuity of the trace applications $v \to \left. \frac{\partial v}{\partial n} \right|_{\Gamma_{\varepsilon}^{\pm}}$ defined as operators from $H^2(K^{\pm})$ to $H^{1/2}(\Gamma_{\varepsilon}^{\pm})$, implies that

implies that

$$\left\|\frac{\partial v^{\pm}}{\partial n}\right\|_{H^{1/2}\left(\Gamma_{\varepsilon}^{\pm}\right)} \leqslant C(\alpha) \left\|u_{\varepsilon}^{\pm}\right\|_{L_{2}\left(B_{\varepsilon}^{\pm}\right)}.$$

Here we denote by $C(\alpha)$ any continuous function that depends only on the geometry of the domain (without changing the notation even if its value changes). Thus, coming back to (3.2.26), we get:

$$\left\| u_{\varepsilon}^{\pm}(\alpha,\varphi) \right\|_{L_{2}\left(B_{\varepsilon}^{\pm}\right)} \leqslant C(\alpha) \left\| \varphi \right\|_{H^{1/2}\left(\Gamma_{\varepsilon}^{\pm}\right)}.$$

Injecting this estimate in (3.2.24) yields (3.2.23).

3.2.3 The interior problem

We can now state the following problem in C^{μ}_{ε} (which is a bounded domain corresponding to the perturbed periodicity cell, cf. figure 3.2). It consists in finding values λ_{ε} such that there exists a function $u^0_{\varepsilon} \in H^1_{\Delta}(C^{\mu}_{\varepsilon})$ solving the problem

$$\begin{cases} -\Delta u_{\varepsilon}^{0} = \lambda_{\varepsilon}^{2} u_{\varepsilon}^{0} \text{ in } \mathcal{C}_{\varepsilon}^{\mu}, \\ \frac{\partial u_{\varepsilon}^{0}}{\partial n}\Big|_{\Gamma_{\varepsilon}^{\pm}} + \Lambda^{\pm}(\lambda_{\varepsilon}) u_{\varepsilon}^{0}\Big|_{\Gamma_{\varepsilon}^{\pm}} = 0, \\ \frac{\partial u_{\varepsilon}^{0}}{\partial n}\Big|_{\partial \mathcal{C}_{\varepsilon}^{\mu} \setminus \{\Gamma_{\varepsilon}^{+} \cup \Gamma_{\varepsilon}^{-}\}} = 0. \end{cases}$$
(3.2.28)



Figure 3.2: The interior domain C^{μ}_{ε}

Proposition 3.2.3. If $\lambda_{\varepsilon}^2 \notin \sigma(A_{\varepsilon})$ then the problem (3.2.28) is equivalent to the problem (3.2.1).

Proof. Suppose first that u_{ε} solves the problem (3.2.1) for some λ_{ε} . Let us denote $u_{\varepsilon}^{\pm} = u_{\varepsilon}|_{B_{\varepsilon}^{\pm}}$, $u_{\varepsilon}^{0} = u_{\varepsilon}|_{C_{\varepsilon}^{\mu}}$. Then, u_{ε}^{\pm} are the unique solutions of (3.2.4) for φ replaced by $u_{\varepsilon}|_{\Gamma_{\varepsilon}^{\pm}}$ and $\alpha = \lambda_{\varepsilon}$. Therefore, $\Lambda^{\pm}(\lambda_{\varepsilon}) u_{\varepsilon}^{\pm}|_{\Gamma_{\varepsilon}^{\pm}} = \frac{\partial u_{\varepsilon}^{\pm}}{\partial n}\Big|_{\Gamma_{\varepsilon}^{\pm}}$ by definition of the DtN operators (cf. (3.2.18)). On the other hand, since $u_{\varepsilon} \in H_{\Delta}^{1}(\Omega_{\varepsilon}^{\mu})$, both its traces and the traces of its normal derivative are continuous: $u_{\varepsilon}^{\pm}|_{\Gamma_{\varepsilon}^{\pm}} = u_{\varepsilon}^{0}|_{\Gamma_{\varepsilon}^{\pm}}, \frac{\partial u_{\varepsilon}^{\pm}}{\partial n}\Big|_{\Gamma_{\varepsilon}^{\pm}} = -\frac{\partial u_{\varepsilon}^{0}}{\partial n}\Big|_{\Gamma_{\varepsilon}^{\pm}}$. This implies that u_{ε}^{0} satisfies the second line of (3.2.28). It also satisfies the first and the last lines (the equation and Neumann boundary condition) since it is a restriction of u_{ε} which solves the initial problem (3.2.1).

Conversely, suppose that u_{ε}^{0} is a solution of (3.2.28). Let us denote by u_{ε}^{\pm} the unique solutions of (3.2.4) with α replaced by λ_{ε} and φ replaced by $u_{\varepsilon}^{0}|_{\Gamma_{\varepsilon}^{\pm}}$. Then, the function constructed as

$$u_{\varepsilon}(x,y) = \begin{cases} u_{\varepsilon}^{\pm}(x,y), & (x,y) \in B_{\varepsilon}^{\pm}, \\ u_{\varepsilon}^{0}(x,y), & (x,y) \in C_{\varepsilon}^{\mu}, \end{cases}$$

solves (3.2.1) for $\alpha = \lambda_{\varepsilon}$. Indeed, its traces on $\Gamma_{\varepsilon}^{\pm}$ are continuous by definition of u_{ε}^{\pm} . The continuity of the normal derivative across $\Gamma_{\varepsilon}^{\pm}$ is guaranteed by definition of the DtN operators. Hence, u_{ε} constructed in such a way appears to be a function in $H^1_{\Delta}(\mathcal{C}_{\varepsilon}^{\mu})$. It satisfies the equation in first line of (3.2.1) since the same equation is satisfied by $u_{\varepsilon}^0, u_{\varepsilon}^{\pm}$. Finally, it satisfies Neumann boundary conditions on $\partial \Omega_{\varepsilon}^{\mu}$, which follows from Neumann boundary conditions for $u_{\varepsilon}^0, u_{\varepsilon}^{\pm}$ on the respective boundaries.

We see that the initial eigenvalue problem (3.2.1) posed in the (unbounded) domain $\Omega^{\mu}_{\varepsilon}$ can be replaced by an equivalent problem (3.2.28) posed in the (bounded) domain C^{μ}_{ε} . However, this problem is a nonlinear one since the DtN operators appearing in the boundary conditions on $\Gamma^{\pm}_{\varepsilon}$ depend themselves on the spectral parameter.

Let us now study the nonlinear problem (3.2.28). For $\alpha^2 \notin \sigma(A_{\varepsilon})$ we introduce the operator $A^0_{\varepsilon}(\alpha)$ defined as follows:

$$A^{0}_{\varepsilon}(\alpha)u = -\Delta u,$$

$$D\left(A^{0}_{\varepsilon}(\alpha)\right) = \left\{ u \in H^{1}_{\Delta}\left(\mathcal{C}^{\mu}_{\varepsilon}\right), \quad \frac{\partial u}{\partial n}\Big|_{\Gamma^{\pm}_{\varepsilon}} + \Lambda(\alpha)^{\pm} \left. u \right|_{\Gamma^{\pm}_{\varepsilon}} = 0, \quad \frac{\partial u}{\partial n}\Big|_{\partial\mathcal{C}^{\mu}_{\varepsilon}\setminus\left\{\Gamma^{+}_{\varepsilon}\cup\Gamma^{-}_{\varepsilon}\right\}} = 0 \right\}.$$

(3.2.29)

We will show some important properties of this operator.

Proposition 3.2.4. For $\alpha^2 \notin \sigma(A_{\varepsilon})$ the operator $A^0_{\varepsilon}(\alpha)$ defined in (3.2.29) is self-adjoint and bounded from below. Moreover, its resolvent is compact.

Proof. The fact that the operator $A^0_{\varepsilon}(\alpha)$ is self-adjoint follows directly from the fact that the DtN operators are symmetric. Let us show that it is bounded from below. Indeed, for any $u \in D(A^0_{\varepsilon}(\alpha))$ one has:

$$\begin{aligned} \left(A^{0}_{\varepsilon}(\alpha)u,u\right)_{L_{2}\left(\mathcal{C}^{\mu}_{\varepsilon}\right)} &= \left\|\nabla u\right\|^{2}_{L_{2}\left(\mathcal{C}^{\mu}_{\varepsilon}\right)} + \left\langle\Lambda^{+}(\alpha) \ u|_{\Gamma^{+}_{\varepsilon}}, \ u|_{\Gamma^{+}_{\varepsilon}}\right\rangle + \left\langle\Lambda^{-}(\alpha) \ u|_{\Gamma^{-}_{\varepsilon}}, \ u|_{\Gamma^{-}_{\varepsilon}}\right\rangle \\ &\geqslant \left\|u\right\|^{2}_{H^{1}\left(\mathcal{C}^{\mu}_{\varepsilon}\right)} - \left\|u\right\|^{2}_{L_{2}\left(\mathcal{C}^{\mu}_{\varepsilon}\right)} - C_{2}(\alpha) \left(\left\|u|_{\Gamma^{+}_{\varepsilon}}\right\|^{2}_{L_{2}\left(\Gamma^{+}_{\varepsilon}\right)} + \left\|u|_{\Gamma^{-}_{\varepsilon}}\right\|^{2}_{L_{2}\left(\Gamma^{-}_{\varepsilon}\right)}\right), \end{aligned}$$

where $C_2(\alpha)$ is the constant from Lemma 3.2.3. The L_2 -norms of the traces of u on $\Gamma_{\varepsilon}^{\pm}$ can be estimated using Lemma A.3.2 (Annexe) applied to the rectangles $]-1, -1 + \frac{\varepsilon}{4} [\times] -\frac{L}{2}, \frac{L}{2} [$ and $]1 - \frac{\varepsilon}{4}, 1 [\times] -\frac{L}{2}, \frac{L}{2} [$ respectively. Then, for $\delta > 0$ we get:

$$\left(A^{0}_{\varepsilon}(\alpha)u, u \right)_{L_{2}\left(\mathcal{C}^{\mu}_{\varepsilon}\right)} \geqslant \left(1 - C_{2}(\alpha)\delta \right) \left\| u \right\|^{2}_{H^{1}\left(\mathcal{C}^{\mu}_{\varepsilon}\right)} - \left(1 + C_{2}(\alpha)C(\delta) \right) \left\| u \right\|^{2}_{L_{2}\left(\mathcal{C}^{\mu}_{\varepsilon}\right)},$$

$$\forall u \in D\left(A^{0}_{\varepsilon}(\alpha) \right).$$

Choosing $\delta < 1/C_2(\alpha)$ one obtains the boundedness from below for the operator $A^0_{\varepsilon}(\alpha)$. In fact, we got even a stronger inequality than just a lower bound. More precisely, if we fix δ in an appropriate way and put $C(\alpha) = 1 - C_2(\alpha)\delta$, $m(\alpha) = 1 + C_2(\alpha)C(\delta)$, then

$$\left(A^{0}_{\varepsilon}(\alpha)u, u\right)_{L_{2}\left(\mathcal{C}^{\mu}_{\varepsilon}\right)} \geq C(\alpha) \left\|u\right\|^{2}_{H^{1}\left(\mathcal{C}^{\mu}_{\varepsilon}\right)} - m(\alpha) \left\|u\right\|^{2}_{L_{2}\left(\mathcal{C}^{\mu}_{\varepsilon}\right)}, \qquad \forall u \in D\left(A^{0}_{\varepsilon}(\alpha)\right), \quad (3.2.30)$$

where $C(\alpha)$, $m(\alpha) > 0$ are continuous functions of α . It is now easy to see that the resolvent of the operator $A^0_{\varepsilon}(\alpha)$ is compact. Indeed, let f be a function in $L_2(\mathcal{C}^{\mu}_{\varepsilon})$ and $u = (A^0_{\varepsilon}(\alpha) + m(\alpha)I)^{-1} f$. Then, the estimate (3.2.30) implies that

$$C(\alpha) \|u\|_{H^1(\mathcal{C}_{\varepsilon}^{\mu})}^2 \leqslant (f, u)_{L_2(\mathcal{C}_{\varepsilon}^{\mu})}.$$

This, in turn, implies that the resolvent $(A^0_{\varepsilon}(\alpha) + m(\alpha)I)^{-1}$ is a continuous operator from $L_2(\mathcal{C}^{\mu}_{\varepsilon})$ to $H^1(\mathcal{C}^{\mu}_{\varepsilon})$. Due to the compactness of the embedding $H^1(\mathcal{C}^{\mu}_{\varepsilon}) \subset L_2(\mathcal{C}^{\mu}_{\varepsilon})$ we conclude that the resolvent of the operator $A^0_{\varepsilon}(\alpha)$ is a compact operator in $L_2(\mathcal{C}^{\mu}_{\varepsilon})$. \Box

It follows from the previous theorem that for $\alpha^2 \notin \sigma(A_{\varepsilon})$ the spectrum of the operator $A_{\varepsilon}^0(\alpha)$ is discrete and consists of a sequence of eigenvalues of finite multiplicity tending to infinity:

$$\varkappa_1(\alpha) \leqslant \varkappa_2(\alpha) \leqslant \cdots \leqslant \varkappa_n(\alpha) \leqslant \dots, \qquad \varkappa_n(\alpha) \xrightarrow[n \to \infty]{} +\infty.$$

Let us consider the following positively defined operator:

$$\tilde{A}^0_{\varepsilon}(\alpha) = A^0_{\varepsilon}(\alpha) + m(\alpha)I.$$

Due to (3.2.30), the boundedness of the DtN operators and the boundedness of the trace operators from $H^1(\mathcal{C}^{\mu}_{\varepsilon})$ to $H^{1/2}(\Gamma^{\pm}_{\varepsilon})$ one has:

$$C(\alpha) \|u\|_{H^1(\mathcal{C}^{\mu}_{\varepsilon})}^2 \leqslant \left(\widetilde{A}^0_{\varepsilon}(\alpha)u, u\right)_{L_2(\mathcal{C}^{\mu}_{\varepsilon})} \leqslant \widetilde{C}(\alpha) \|u\|_{H^1(\mathcal{C}^{\mu}_{\varepsilon})}^2, \qquad \forall u \in D\left(A^0_{\varepsilon}(\alpha)\right),$$

with some constant $\widetilde{C}(\alpha)$. This implies that the sesquilinear form $\widetilde{a}^0_{\varepsilon}(\alpha)$ corresponding to the operator $\widetilde{A}^0_{\varepsilon}(\alpha)$ is defined on $H^1(\mathcal{C}^{\mu}_{\varepsilon})$ and the norm defined by this form is equivalent to the H^1 -norm:

$$D[\widetilde{a}^{0}_{\varepsilon}(\alpha)] = H^{1}(\mathcal{C}^{\mu}_{\varepsilon}),$$

$$\widetilde{a}^{0}_{\varepsilon}(\alpha)[u,v] = (\nabla u, \nabla v)_{L_{2}(\mathcal{C}^{\mu}_{\varepsilon})} + \sum_{\delta \in \{+,-\}} \left\langle \Lambda^{\delta}(\alpha) \ u|_{\Gamma^{\delta}_{\varepsilon}}, \ v|_{\Gamma^{\delta}_{\varepsilon}} \right\rangle + m(\alpha)(u,v)_{L_{2}(\mathcal{C}^{\mu}_{\varepsilon})},$$

$$C(\alpha) \|u\|^{2}_{H^{1}(\mathcal{C}^{\mu}_{\varepsilon})} \leqslant \widetilde{a}^{0}_{\varepsilon}(\alpha)[u,u] \leqslant \widetilde{C}(\alpha) \|u\|^{2}_{H^{1}(\mathcal{C}^{\mu}_{\varepsilon})}, \qquad \forall u \in H^{1}(\mathcal{C}^{\mu}_{\varepsilon}).$$
(3.2.31)

This permits to characterize the eigenvalues $\{\widetilde{\varkappa}_n(\alpha)\}_{n\in\mathbb{N}}$ of the operator $\widetilde{A}^0_{\varepsilon}(\alpha)$ using the min-max principle as follows:

$$\widetilde{\varkappa}_{n}(\alpha) = \inf_{\substack{\mathcal{M} \subset H^{1}(\mathcal{C}^{\mu}_{\varepsilon}) \\ \dim \mathcal{M} = n}} \sup_{\substack{u \in \mathcal{M} \\ \|u\|_{L_{2}(\mathcal{C}^{\mu}_{\varepsilon})} = 1}} \widetilde{a}^{0}_{\varepsilon}(\alpha)[u, u].$$
(3.2.32)

The following assertion states the continuity of these eigenvalues and, as a consequence, of the eigenvalues $\varkappa_n(\alpha)$, with respect to the parameter α .

Proposition 3.2.5. The functions $\varkappa_n(\alpha)$ are continuous for $\alpha^2 \notin \sigma(A_{\varepsilon})$, $n \in \mathbb{N}$.

Proof. First, we notice that the spectrum of the operator $A^0_{\varepsilon}(\alpha)$ can be obtained by a translation by $m(\alpha)$ of the spectrum of the operator $\widetilde{A}^0_{\varepsilon}(\alpha)$. Hence, the function $m(\alpha)$ being continuous, it is enough to prove the continuity of the functions $\widetilde{\varkappa}_n(\alpha)$. For this we will use (3.2.32). Let us consider the difference $\widetilde{a}^0_{\varepsilon}(\alpha_1)[u, u] - \widetilde{a}^0_{\varepsilon}(\alpha_2)[u, u]$ for $\alpha^2_1, \alpha^2_2 \notin \sigma(A_{\varepsilon})$ and $u \in H^1(\mathcal{C}^{\mu}_{\varepsilon})$ such that $||u||_{L_2(\mathcal{C}^{\mu}_{\varepsilon})} = 1$. One gets:

$$\widetilde{a}^{0}_{\varepsilon}(\alpha_{1})[u,u] - \widetilde{a}^{0}_{\varepsilon}(\alpha_{2})[u,u] = \sum_{\delta \in \{+,-\}} \left\langle \left(\Lambda^{\delta}(\alpha_{1}) - \Lambda^{\delta}(\alpha_{2})\right)\varphi^{\delta}, \varphi^{\delta} \right\rangle + \left(m(\alpha_{1}) - m(\alpha_{2})\right) \|u\|^{2}_{L_{2}(\mathcal{C}^{\mu}_{\varepsilon})},$$

where $\varphi^{\pm} = u|_{\Gamma_{\varepsilon}^{\pm}}$. Hence, the estimate (3.2.19) implies that if $d^{\pm}(\alpha_1), d^{\pm}(\alpha_2) \ge c_0 > 0$ then

$$\left|\widetilde{a}_{\varepsilon}^{0}(\alpha_{1})[u,u] - \widetilde{a}_{\varepsilon}^{0}(\alpha_{2})[u,u]\right| \leq \left(C\left|\alpha_{1}^{2} - \alpha_{2}^{2}\right| + \left|m(\alpha_{1}) - m(\alpha_{2})\right|\right) \|u\|_{H^{1}(\mathcal{C}_{\varepsilon}^{\mu})}^{2}.$$

The H^1 -norm of u can be estimated using (3.2.31) with $\alpha = \alpha_2$, which yields

$$\widetilde{a}^{0}_{\varepsilon}(\alpha_{2})[u,u](1-f(\alpha_{1},\alpha_{2})) \leqslant \widetilde{a}^{0}_{\varepsilon}(\alpha_{1})[u,u] \leqslant \widetilde{a}^{0}_{\varepsilon}(\alpha_{2})[u,u](1+f(\alpha_{1},\alpha_{2})),$$

$$\forall u \in H^{1}(\mathcal{C}^{\mu}_{\varepsilon}), \qquad (3.2.33)$$

where $f(\alpha_1, \alpha_2) = (C |\alpha_1^2 - \alpha_2^2| + |m(\alpha_1) - m(\alpha_2)|) / C(\alpha_2)$ and $C(\alpha_2)$ is the constant in the left-hand side of (3.2.31) which is strictly positive. This implies that the function f is continuous and vanishes on the diagonal: $f(\alpha_1, \alpha_2) \to 0$ when $\alpha_1 \to \alpha_2$. Taking the upper bound of all the terms in (3.2.33) over all unit vectors belonging to some *n*-dimensional subspace of $H^1(\mathcal{C}_{\varepsilon}^{\mu})$ and then the lower bound over all such subspaces we get:

$$\widetilde{\varkappa}_n(\alpha_2)(1-f(\alpha_1,\alpha_2)) \leqslant \widetilde{\varkappa}_n(\alpha_1) \leqslant \widetilde{\varkappa}_n(\alpha_2)(1+f(\alpha_1,\alpha_2)), \quad \forall n \in \mathbb{N}.$$

This together with the properties of the function f mentioned above proves the continuity of the functions $\widetilde{\varkappa}_n(\alpha), n \in \mathbb{N}$.

The following theorem is an immediate consequence of Proposition 3.2.3.

Theorem 3.2.1. A number λ_{ε}^2 is an eigenvalue of the operator A_{ε}^{μ} if and only if λ_{ε} is a solution of the equation

$$\alpha^2 = \varkappa_n(\alpha), \tag{3.2.34}$$

for some $n \in \mathbb{N}$.

The functions \varkappa_n being continuous, it's the equation (3.2.34) that we solve numerically using a Newton type algorithm.

3.2.4 Practical construction of the operators $\Lambda^{\pm}(\alpha)$

Let us now give a procedure of construction of the DtN operators $\Lambda^{\pm}(\alpha)$ that avoids solving the problems (3.2.4) posed in unbounded domains and consequently can be used in numerical computations.

3.2.4.1 The local DtN operators

We introduce a periodicity cell C_{ε} such that the band B_{ε}^+ is a union of translated periodicity cells (fig. 3.3):

$$\overline{B_{\varepsilon}^{+}} = \bigcup_{n \in \mathbb{N}} \overline{\mathcal{C}_{\varepsilon,n}}, \qquad \mathcal{C}_{\varepsilon,n} = \mathcal{C}_{\varepsilon} + (n,0), \qquad \mathcal{C}_{\varepsilon} = B_{\varepsilon}^{+} \cap \{1 < x < 2\}.$$

We also denote the vertical boundaries of the cells $C_{\varepsilon,n}$ by $\Gamma_{\varepsilon,n} = \{n+1\} \times \left] -\frac{L}{2}, \frac{L}{2} \right[, n \in \mathbb{N}.$



Figure 3.3: The cells $\mathcal{C}_{\varepsilon,n}$ and the boundaries $\Gamma_{\varepsilon,n}$

Function spaces on $\mathcal{C}_{\varepsilon,n}$ and $\Gamma_{\varepsilon,n}$ for different *n* will be often identified. Let us consider the following cell problems for $\alpha^2 \notin \sigma(A_{\varepsilon}), \ \delta \in \{0, 1\}$:

$$\begin{pmatrix}
-\Delta e_{\delta} - \alpha^{2} e_{\delta} = 0 & \text{in } \mathcal{C}_{\varepsilon}, \\
\frac{\partial e_{\delta}}{\partial n} \Big|_{\partial \mathcal{C}_{\varepsilon} \setminus \{\Gamma_{\varepsilon,0} \cup \Gamma_{\varepsilon,1}\}} = 0, \\
e_{\delta}|_{\Gamma_{\varepsilon,\delta}} = \varphi, \quad e_{\delta}|_{\Gamma_{\varepsilon,1-\delta}} = 0.
\end{cases}$$
(3.2.35)

Proposition 3.2.6. For any $\alpha^2 \notin \sigma_D$ and $\varphi \in H^{1/2}(\Gamma_{\varepsilon,\delta})$ the problems (3.2.35) have unique solutions in $H^1_{\Delta}(\mathcal{C}_{\varepsilon})$, where σ_D is a countable set.

Proof. We will give the proof for $\delta = 0$, the proof for $\delta = 1$ being analogous. The argument is very close to the one used in the proof of Proposition 3.2.1. We repeat it for the sake of completeness. Let us $\widehat{\varphi}$ be a function in $H^1(\mathcal{C}_{\varepsilon})$ such that $\widehat{\varphi}|_{\Gamma_{\varepsilon,0}} = \varphi$, $\widehat{\varphi}|_{\Gamma_{\varepsilon,1}} = 0$ and $\|\widehat{\varphi}\|_{H^1(\mathcal{C}_{\varepsilon})} \leq C \|\varphi\|_{H^{1/2}(\Gamma_{\varepsilon,0})}$, the constant C depending only on the domain $\mathcal{C}_{\varepsilon}$. After multiplication of the first line of (3.2.35) by a test function $v \in H^1_D(\mathcal{C}_{\varepsilon})$, where

$$H_D^1\left(\mathcal{C}_{\varepsilon}\right) = \left\{ u \in H^1\left(\mathcal{C}_{\varepsilon}\right), \quad u|_{\Gamma_{\varepsilon,0}} = 0, \quad u|_{\Gamma_{\varepsilon,1}} = 0 \right\},$$

and changing the unknown function by $\hat{e}_0 = e_0 - \hat{\varphi}$, we get the following variational problem for $\hat{e}_0 \in H^1_D(\mathcal{C}_{\varepsilon})$:

$$(\nabla \widehat{e}_{0}, \nabla v)_{L_{2}(\mathcal{C}_{\varepsilon})} - \alpha^{2} (\widehat{e}_{0}, v)_{L_{2}(\mathcal{C}_{\varepsilon})} = - (\nabla \widehat{\varphi}, \nabla v)_{L_{2}(\mathcal{C}_{\varepsilon})} + \alpha^{2} (\widehat{\varphi}, v)_{L_{2}(\mathcal{C}_{\varepsilon})}, \forall v \in H_{D}^{1} (\mathcal{C}_{\varepsilon}).$$
(3.2.36)

This problem can be reduced to the form

$$(I - (1 + \alpha^2)P) \hat{e}_0 = f,$$
 (3.2.37)

where the bounded self-adjoint operator $P \in \mathcal{L}(H^1_D(\mathcal{C}_{\varepsilon}))$ is defined by the relation

$$(u,v)_{L_2(\mathcal{C}_{\varepsilon})} = (Pu,v)_{H^1(\mathcal{C}_{\varepsilon})}, \qquad \forall u,v \in H^1_D(\mathcal{C}_{\varepsilon}), \qquad (3.2.38)$$

and $f \in H_D^1(\mathcal{C}_{\varepsilon})$ is defined by the relation

$$-\left(\nabla\widehat{\varphi},\nabla v\right)_{L_2(\mathcal{C}_{\varepsilon})} + \alpha^2 \left(\widehat{\varphi},v\right)_{L_2(\mathcal{C}_{\varepsilon})} = (f,v)_{H^1(\mathcal{C}_{\varepsilon})}, \qquad \forall v \in H^1_D\left(\mathcal{C}_{\varepsilon}\right).$$

The operator P is compact. Indeed, it is bounded as an operator from $L_2(\mathcal{C}_{\varepsilon})$ to $H^1(\mathcal{C}_{\varepsilon})$ since taking in (3.2.38) v = Pu one has:

$$\|Pu\|_{H^1(\mathcal{C}_{\varepsilon})}^2 = (u, Pu)_{L_2(\mathcal{C}_{\varepsilon})} \leqslant \|Pu\|_{H^1(\mathcal{C}_{\varepsilon})} \|u\|_{L_2(\mathcal{C}_{\varepsilon})}, \qquad \forall u \in H_D^1(\mathcal{C}_{\varepsilon}).$$

Due to the compactness of the embedding $H^1(\mathcal{C}_{\varepsilon}) \subset L_2(\mathcal{C}_{\varepsilon})$ one concludes that the operator $P \in \mathcal{L}(H_D^1(\mathcal{C}_{\varepsilon}))$ is compact. Hence, its spectrum $\sigma(P)$ consists of eigenvalues of finite multiplicity that can only accumulate to 0. The problem (3.2.37) is then well-posed if and only if $\alpha^2 \notin \{1/\gamma - 1, \gamma \in \sigma(P)\}$, which is a discrete set accumulating only to infinity that we denote by σ_D .

The unique solutions of the problems (3.2.35) for $\alpha^2 \notin \sigma_D$, $\varphi \in H^{1/2}(\Gamma_{\varepsilon,\delta})$ will be denoted by $e_{\delta}(\alpha, \varphi)$, $\delta \in \{0, 1\}$. Notice that repeating the same argument as the one used in the proof of Proposition 3.2.1 one can see that

$$\|e_{\delta}(\alpha,\varphi)\|_{H^{1}(\mathcal{C}_{\varepsilon})} \leqslant C(\alpha)\|\varphi\|_{H^{1/2}(\Gamma_{\varepsilon,\delta})}, \qquad \forall \varphi \in H^{1/2}(\Gamma_{\varepsilon,\delta}), \quad \delta \in \{0,1\}, \qquad (3.2.39)$$

where the function $C(\alpha)$ is continuous and depends only on the geometry of the domain. We can now introduce the local DtN operators $T_{\gamma\delta}(\alpha) \in \mathcal{L}\left(H^{1/2}(\Gamma_{\varepsilon,\gamma}), H^{-1/2}(\Gamma_{\varepsilon,\delta})\right), \gamma, \delta \in \{0, 1\}$, as follows:

$$T_{\gamma\delta}(\alpha)\varphi = \left.\frac{\partial e_{\gamma}(\alpha,\varphi)}{\partial n}\right|_{\Gamma_{\varepsilon,\delta}}, \qquad \gamma,\delta \in \{0,1\},$$
(3.2.40)

where the normal derivative is taken along the outside normal to the domain C_{ε} .



Figure 3.4: The local DtN operators

Lemma 3.2.4. The operators $T_{\gamma\delta}(\alpha)$ are symmetric for any $\gamma, \delta \in \{0, 1\}$, $\alpha^2 \notin \sigma_D$. *Proof.* Indeed, we have

$$\langle T_{\gamma\delta}(\alpha)\varphi,\psi\rangle = \int_{\mathcal{C}_{\varepsilon}} \Delta e_{\gamma}(\alpha,\varphi)e_{\delta}(\alpha,\psi) - \int_{\mathcal{C}_{\varepsilon}} e_{\gamma}(\alpha,\varphi)\Delta e_{\delta}(\alpha,\psi) + \int_{\Gamma_{\varepsilon,\gamma}} e_{\gamma}(\alpha,\varphi)\frac{\partial e_{\delta}(\alpha,\psi)}{\partial n}$$
$$= \langle T_{\delta\gamma}(\alpha)\psi,\varphi\rangle \,.$$

Since in our case of a symmetric cell the operators T_{01} and T_{10} coincide, this finishes the proof.

3.2.4.2 The propagation operator

Let us introduce the propagation operator $P(\alpha)$ for $\alpha^2 \notin \sigma(A_{\varepsilon})$ as follows:

$$P(\alpha) \in \mathcal{L}\left(H^{1/2}\left(\Gamma_{\varepsilon,0}\right), H^{1/2}\left(\Gamma_{\varepsilon,1}\right)\right), \qquad P(\alpha)\varphi = \left.u_{\varepsilon}^{+}(\alpha,\varphi)\right|_{\Gamma_{\varepsilon,1}}.$$
(3.2.41)

The spectral radius of the propagation operator will play an important role in the sequel. It will be denoted by $\rho(P(\alpha))$.

Lemma 3.2.5. For $\alpha^2 \notin \sigma(A_{\varepsilon})$ the operator $P(\alpha)$ is compact. Moreover, $\rho(P(\alpha)) < 1$.

Proof. Let φ be a function in $H^{1/2}(\Gamma_{\varepsilon,0})$. We will use Lemma 3.2.6 proved below, according to which $\|P(\alpha)u_{\varepsilon}^{+}(\alpha,\varphi)\|_{H^{1}(\Gamma_{\varepsilon,1})} \leq C(\alpha)\|\varphi\|_{H^{1/2}(\Gamma_{\varepsilon,0})}$. Recalling the definition (3.2.41) of the operator $P(\alpha)$ and using the compactness of the embedding $H^{1}(\Gamma_{\varepsilon,1}) \subset H^{1/2}(\Gamma_{\varepsilon,1})$ we conclude that the propagation operator is compact. Consequently, its spectrum is a sequence of isolated eigenvalues of finite multiplicity with the only possible accumulation point at 0 (in this case $0 \in \sigma_{c}(P(\alpha))$). If λ_{1} is the eigenvalue of $P(\alpha)$ with the biggest absolute value then $|\lambda_{1}| = \rho(P(\alpha))$. Suppose that $|\lambda_{1}| \ge 1$ and $\varphi_{1} \in H^{1/2}(\Gamma_{\varepsilon,1})$ is a corresponding eigenfunction. Then, $u_{\varepsilon}^{+}(\alpha,\varphi_{1})|_{\Gamma_{\varepsilon,n}} = P(\alpha)^{n}\varphi_{1} = \lambda_{1}^{n}\varphi_{1}, \forall n \in \mathbb{N}$. On the other hand, due to the uniqueness of solutions of (3.2.4) and the translation invariance of the domain B_{ε}^{+} we have

$$u_{\varepsilon}^{+}(\alpha,\varphi_{1})\big|_{B_{\varepsilon,n}} = \mathcal{T}_{n}u_{\varepsilon}^{+}(\alpha,\lambda_{1}^{n}\varphi_{1}) = \lambda_{1}^{n}\mathcal{T}_{n}u_{\varepsilon}^{+}(\alpha,\varphi_{1}), \qquad \forall n \in \mathbb{N},$$
(3.2.42)

where

$$B_{\varepsilon,n} = B_{\varepsilon}^+ \cap \{X > n+1\}, \qquad n \in \mathbb{N},$$

and \mathcal{T}_n is the translation operator from $L_2(B_{\varepsilon}^+)$ to $L_2(B_{\varepsilon,n})$ defined on smooth functions by the relation

$$(\mathcal{T}_n u)(x,y) = u(x-n,y), \qquad (x,y) \in B_{\varepsilon,n}$$

and extended by continuity to $L_2(B_{\varepsilon}^+)$. It follows from (3.2.42) that

$$\left\|u_{\varepsilon}^{+}(\alpha,\varphi_{1})\right\|_{B_{\varepsilon,n}}\left\|_{L_{2}(B_{\varepsilon,n})}=\left|\lambda_{1}^{n}\right|\left\|u_{\varepsilon}^{+}(\alpha,\varphi_{1})\right\|_{L_{2}\left(B_{\varepsilon}^{+}\right)}\geqslant\left\|u_{\varepsilon}^{+}(\alpha,\varphi_{1})\right\|_{L_{2}\left(B_{\varepsilon}^{+}\right)},\qquad\forall n\in\mathbb{N}.$$

However, $u_{\varepsilon}^{+}(\alpha, \varphi_{1}) \in L_{2}(B_{\varepsilon}^{+})$ which implies $\left\| u_{\varepsilon}^{+}(\alpha, \varphi_{1}) \right\|_{B_{\varepsilon,n}} \left\|_{L_{2}(B_{\varepsilon,n})} \xrightarrow[n \to \infty]{} 0$. This is possible only if $u_{\varepsilon}^{+}(\alpha, \varphi_{1})$ identically zero which contradicts the fact φ_{1} is an eigenfunction of the operator $P(\alpha)$.

The following Lemma used in the proof of Lemma 3.2.5 states in fact the interior regularity of the solutions u_{ε}^{\pm} of (3.2.4). For the sake of completeness we will give the argument in the case of our geometry that permits to apply the regularity result (Theorems 2.3.7 and 2.4.3 from [27]).

Lemma 3.2.6. For any $\alpha^2 \notin \sigma(A_{\varepsilon})$ and $\varphi \in H^{1/2}(\Gamma_{\varepsilon,0})$ one has $P(\alpha)u_{\varepsilon}^+(\alpha,\varphi) \in H^1(\Gamma_{\varepsilon,1})$. Moreover, there exists a continuous function $C(\alpha)$ depending only on the geometry of the domain such that

$$\left\| P(\alpha) u_{\varepsilon}^{+}(\alpha, \varphi) \right\|_{H^{1}(\Gamma_{\varepsilon, 1})} \leqslant C(\alpha) \|\varphi\|_{H^{1/2}(\Gamma_{\varepsilon, 0})}, \qquad \forall \varphi \in H^{1/2}\left(\Gamma_{\varepsilon, 0}\right).$$

Proof. The argument is very close to the one used in the proof of Lemma A.3.3. Let us introduce a cut-off function $\chi \in C^{\infty}(\mathbb{R})$ such that

$$\begin{cases} \chi(x) = 1, & x \in \left[2 - \frac{\varepsilon}{8}, 2 + \frac{\varepsilon}{8}\right], \\ \chi(x) = 0, & x \in \left] -\infty, \ 2 - \frac{\varepsilon}{4}\right] \cup \left[2 + \frac{\varepsilon}{4}, \ \infty\right[, \\ 0 \leqslant \chi(x) \leqslant 1, & \forall x \in \mathbb{R}. \end{cases}$$

Let $\mathcal{K}_{\varepsilon}$ stand for the rectangle $]2 - \frac{\varepsilon}{4}, 2 + \frac{\varepsilon}{4} [\times] - \frac{L}{2}, \frac{L}{2} [$. Then, the function $\widetilde{u} = \chi u_{\varepsilon}^{+}(\alpha, \varphi)$ solves the following problem in $\mathcal{K}_{\varepsilon}$:

$$\begin{cases} -\Delta \widetilde{u} = \alpha^2 u_{\varepsilon}^+(\alpha,\varphi)\chi - u_{\varepsilon}^+(\alpha,\varphi)\Delta\chi - 2\nabla u_{\varepsilon}^+(\alpha,\varphi)\nabla\chi & \text{in } \mathcal{K}_{\varepsilon}, \\ \widetilde{u}|_{\left\{2-\frac{\varepsilon}{4}\right\}\times\left]-\frac{L}{2},\frac{L}{2}\right[} = 0, \quad \widetilde{u}|_{\left\{2+\frac{\varepsilon}{4}\right\}\times\left]-\frac{L}{2},\frac{L}{2}\right[} = 0, \\ \frac{\partial \widetilde{u}}{\partial n}\Big|_{\left[2-\frac{\varepsilon}{4},2+\frac{\varepsilon}{4}\right[\times\left\{\frac{L}{2}\right\}\right]} = 0, \quad \frac{\partial \widetilde{u}}{\partial n}\Big|_{\left[2-\frac{\varepsilon}{4},2+\frac{\varepsilon}{4}\right[\times\left\{-\frac{L}{2}\right\}\right]} = 0. \end{cases}$$

Then, applying the regularity result ([27]), we conclude that $\widetilde{u} \in H^2(\mathcal{K}_{\varepsilon})$ and

$$\|\widetilde{u}\|_{H^{2}(\mathcal{K}_{\varepsilon})} \leqslant C \|\alpha^{2} u_{\varepsilon}^{+}(\alpha,\varphi)\chi - u_{\varepsilon}^{+}(\alpha,\varphi)\Delta\chi - 2\nabla u_{\varepsilon}^{+}(\alpha,\varphi)\nabla\chi\|_{L_{2}(\mathcal{K}_{\varepsilon})} \leqslant C(\alpha)\|\varphi\|_{H^{1/2}(\Gamma_{\varepsilon,0})},$$

where we took into account (3.2.5). Then, using the continuity of the trace application from $H^2(K)$ to $H^1(\partial K)$ in a Lipschitz domain $K = \left]2, 2 + \frac{\varepsilon}{4}\right[\times \left]-\frac{L}{2}, \frac{L}{2}\right[$ (cf. [14]) we get

$$\left\| u \right\|_{\Gamma_{\varepsilon,1}} \left\|_{H^1(\Gamma_{\varepsilon,1})} = \left\| \widetilde{u} \right\|_{\Gamma_{\varepsilon,1}} \right\|_{H^1(\Gamma_{\varepsilon,1})} \leqslant \| \widetilde{u} \|_{H^2(K)} \leqslant C(\alpha) \| \varphi \|_{H^{1/2}(\Gamma_{\varepsilon,0})}.$$

Remark 3.2.3. The propagation operator $P(\alpha)$ is injective.

Proof. Indeed, suppose that $P(\alpha)\varphi_0 = 0$ for some $\varphi_0 \in H^{1/2}(\Gamma_{\varepsilon,0})$. Then, $u_{\varepsilon}^+(\alpha,\varphi)|_{B_{\varepsilon,1}}$ solves the problem (3.2.4) in $B_{\varepsilon,1}$ with $\varphi = 0$, which implies that $u_{\varepsilon}^+(\alpha,\varphi)|_{B_{\varepsilon,1}} = 0$. From the unique continuation property it follows that $u_{\varepsilon}^+(\alpha,\varphi) = 0$ and, consequently, $\varphi_0 = 0$.

3.2.4.3 Another characterization of the propagation operator. Riccati equation

Due to the translation invariance of the domain B_{ε}^+ one has

$$u_{\varepsilon}^{+}(\alpha,\varphi)\big|_{\Gamma_{\varepsilon,n}} = (P(\alpha))^{n} \varphi, \quad \forall n \in \mathbb{N}, \quad \forall \varphi \in H^{1/2}(\Gamma_{\varepsilon,0}).$$

Then, the restriction of $u_{\varepsilon}^+(\alpha,\varphi)$ to the cell $\mathcal{C}_{\varepsilon,n}$ can be computed as follows:

$$u_{\varepsilon}^{+}(\alpha,\varphi)\big|_{\mathcal{C}_{\varepsilon,n}} = e_0\left(\alpha, \left(P(\alpha)\right)^{n-1}\varphi\right) + e_1\left(\alpha, \left(P(\alpha)\right)^n\varphi\right), \qquad \forall n \in \mathbb{N}^*, \quad \forall \varphi \in H^{1/2}\left(\Gamma_{\varepsilon,0}\right).$$

The continuity of the normal derivative of $u_{\varepsilon}^+(\alpha, \varphi)$ across $\Gamma_{\varepsilon,1}$ yields, in view of (3.2.40), that

$$T_{10}(\alpha) \left(P(\alpha)\right)^2 + \left(T_{00}(\alpha) + T_{11}(\alpha)\right) P(\alpha) + T_{01}(\alpha) = 0.$$
(3.2.43)

This relation can be seen as the stationary Riccati equation for $P(\alpha)$. It turns out that it can be used to determine the propagation operator $P(\alpha)$ without solving the problems (3.2.4) once the local DtN operators are constructed.

Proposition 3.2.7. For $\alpha^2 \notin \sigma(A_{\varepsilon}) \cup \sigma_D$ the propagation operator $P(\alpha)$ is the unique solution of the Riccati equation (3.2.43) in the set of operators in $H^{1/2}(\Gamma_{\varepsilon,0})$ with spectral radius smaller than 1.

Proof. Suppose that there exists another compact operator \widetilde{P} in $H^{1/2}(\Gamma_{\varepsilon,0})$ with spectral radius smaller than 1 satisfying (3.2.43). Then, for any $\varphi \in H^{1/2}(\Gamma_{\varepsilon,0})$ one can construct the following function $u \in L_2(B_{\varepsilon}^+)$:

$$u|_{\mathcal{C}_{\varepsilon,n}} = e_0\left(\alpha, \widetilde{P}^{n-1}\varphi\right) + e_1\left(\alpha, \widetilde{P}^n\varphi\right), \qquad \forall n \in \mathbb{N}^*.$$
(3.2.44)

By construction, its trace on the boundary $\Gamma_{\varepsilon,0}$ is φ and u is continuous across the boundaries $\Gamma_{\varepsilon,n}$: $\left(u|_{\mathcal{C}_{\varepsilon,n}}\right)\Big|_{\Gamma_{\varepsilon,n}} = \left(u|_{\mathcal{C}_{\varepsilon,n+1}}\right)\Big|_{\Gamma_{\varepsilon,n}} = \widetilde{P}^n \varphi, \ \forall n \in \mathbb{N}^*.$ Consequently, $u \in H^1_{loc}(B^+_{\varepsilon})$. Moreover, its normal derivative is also continuous. Indeed,

$$\begin{split} \left. \left(\frac{\partial}{\partial n} \left(u |_{\mathcal{C}_{\varepsilon,n}} \right) \right) \right|_{\Gamma_{\varepsilon,n}} &= T_{01}(\alpha) \widetilde{P}^{n-1} \varphi + T_{11}(\alpha) \widetilde{P}^{n} \varphi, \\ \left. \left(\frac{\partial}{\partial n} \left(u |_{\mathcal{C}_{\varepsilon,n+1}} \right) \right) \right|_{\Gamma_{\varepsilon,n}} &= T_{00}(\alpha) \widetilde{P}^{n} \varphi + T_{10}(\alpha) \widetilde{P}^{n+1} \varphi = -T_{01}(\alpha) \widetilde{P}^{n-1} \varphi - T_{11}(\alpha) \widetilde{P}^{n} \varphi, \end{split}$$

where we used the fact that the operator \widetilde{P} solves the equation (3.2.43). Hence,

$$\left(\frac{\partial}{\partial n}\left(u|_{\mathcal{C}_{\varepsilon,n}}\right)\right)\Big|_{\Gamma_{\varepsilon,n}} = -\left.\left(\frac{\partial}{\partial n}\left(u|_{\mathcal{C}_{\varepsilon,n+1}}\right)\right)\Big|_{\Gamma_{\varepsilon,n}}, \qquad \forall n \in \mathbb{N}^*$$

which implies that $u \in H^1_{\Delta, loc}(B^+_{\varepsilon})$. Let us show that in fact $u \in H^1_{\Delta}(B^+_{\varepsilon})$. Since $\rho(\widetilde{P}) = \lim_{n \to +\infty} \|\widetilde{P}^n\|^{1/n}_{\mathcal{L}(H^{1/2}(\Gamma_{\varepsilon,0}))} < 1$, there exists $N \in \mathbb{N}$ and $0 < c_0 < 1$ such that

$$\|\widetilde{P}^n\|_{\mathcal{L}(H^{1/2}(\Gamma_{\varepsilon,0}))} < c_0^n, \quad \forall n \ge N.$$

Therefore,

$$\sum_{n=N+1}^{\infty} \left\| u |_{\mathcal{C}_{\varepsilon,n}} \right\|_{H^1(\mathcal{C}_{\varepsilon,n})}^2 \leqslant C(\alpha) \|\varphi\|_{H^{1/2}(\Gamma_{\varepsilon,0})} \sum_{n=N+1}^{\infty} c_0^{2n} < \infty,$$

where we used (3.2.44) and (3.2.39). This proves that $u \in H^1_{\Delta}(B^+_{\varepsilon})$ and $-\Delta u - \alpha^2 u = 0$ in B^+_{ε} since. Finally, it means that u solves (3.2.4) in B^+_{ε} , i.e. $u = u^+_{\varepsilon}(\alpha, \varphi)$. Then, from (3.2.44) for n = 1 it follows that $u^+_{\varepsilon}(\alpha, \varphi)|_{\Gamma_{\varepsilon,1}} = \widetilde{P}\varphi$ for any $\varphi \in H^{1/2}(\Gamma_{\varepsilon,0})$. Comparing this with the definition of the propagation operator (3.2.41) one conclude that $\widetilde{P} = P(\alpha)$.

Proposition 3.2.7 permits to determine the propagation operator by solving Riccati equation (3.2.43) instead of solving the problems (3.2.4). It is now easy to construct the DtN operators $\Lambda^{\pm}(\alpha)$. Indeed, comparing the definition of the DtN operators (3.2.18) with the definition of the local DtN operators (3.2.40) one finds that

$$\Lambda^{+}(\alpha) = T_{00}(\alpha) + T_{10}(\alpha)P(\alpha).$$
(3.2.45)

Thus, we do not need any more to consider unbounded domains in order to solve the eigenvalue problem (3.2.1).

Remark 3.2.4. The spectral radius of the propagation operator can be used to characterize the essential spectrum of the operator A_{ε} as well. It turns out that

$$\alpha^2 \notin \sigma(A_{\varepsilon}) \Leftrightarrow \rho(P(\alpha)) < 1, \qquad \alpha^2 \in \sigma(A_{\varepsilon}) \Leftrightarrow \rho(P(\alpha)) = 1.$$

This property can be used in the numerical computation of the essential spectrum of the operator A^{μ}_{ε} , but we will mostly use dispersion curves method as discussed below.

3.3 Discretization by a conform finite element method

From a practical point of view, the numerical method used consists in solving the nonlinear equation

$$\alpha^2 = \varkappa_n^h(\alpha) \tag{3.3.1}$$

by a Newton type algorithm. The equation (3.3.1) corresponds to the equation (3.2.34) where the eigenvalues \varkappa_n of the operator A^{μ}_{ε} are replaced by those of its approximation $A^{\mu,h}_{\varepsilon}$ discretized by a standard finite element method. The principal difficulty in the construction of the operator $A^{\mu,h}_{\varepsilon}$ consists in the discretization of the DtN operators $\Lambda^{\pm}(\alpha)$ that we detail in Section 3.3.1. In the sequel we explain how to apply a Newton type algorithm to the equation (3.3.1). Finally, we briefly describe how to compute numerically the essential spectrum of the operator A^{μ}_{ε} .

3.3.1 Discretization of the operator A^{μ}_{ε}

In this section we will describe in detail how we perform the discretization of the problem (3.2.28) without giving the numerical analysis which can be found in [22]. The first step is the construction of the DtN operators using the relation (3.2.45). Thus, the local DtN operators $T_{00}(\alpha)$, $T_{10}(\alpha)$ as well as the propagation operator $P(\alpha)$ have to be constructed. The construction of the operators $T_{00}(\alpha)$, $T_{10}(\alpha)$ requires the resolution of cell problems that are solved using a standard P_1 Lagrange finite element method.

3.3.1.1 Cell problems

Let $\mathcal{T}_h = \{T_\ell\}_{\ell=1}^L$ be a triangular mesh of the domain $\mathcal{C}_{\varepsilon}$. The set of all the vertices of the mesh \mathcal{T}_h will be denoted by \mathcal{M}_h : $\mathcal{M}_h = \{M_j\}_{j=1}^N$. Let us introduce the following function space:

$$V_{h} = \left\{ v_{h} \in C\left(\overline{\mathcal{C}_{\varepsilon}}\right), \quad v_{h}|_{T_{\ell}} \text{ is linear, } 1 \leq \ell \leq L \right\}.$$

$$(3.3.2)$$

Clearly, V_h is a subspace of $H^1(\mathcal{C}_{\varepsilon})$ of dimension N. As usual, we consider the basis $\mathcal{B}_h = \{w_j\}_{j=1}^N$ in V_h , which consists of continuous piecewise linear functions w_j , such that

$$w_j(M_i) = \delta_{ij}, \qquad 1 \leqslant i, j \leqslant N. \tag{3.3.3}$$

We will use the space V_h (and its analogue for the domain C_{ε}^{μ}) to construct internal approximations (also called Galerkin approximations) of all variational problems that we have to solve in our algorithm. The spaces of the traces of the functions in V_h on the boundaries $\Gamma_{\varepsilon,0}$, $\Gamma_{\varepsilon,1}$ will be denoted by

$$S_h^0 = Span\left\{ \left. w_j \right|_{\Gamma_{\varepsilon,0}}, \ 1 \leqslant j \leqslant N \right\}, \qquad S_h^1 = Span\left\{ \left. w_j \right|_{\Gamma_{\varepsilon,1}}, \ 1 \leqslant j \leqslant N \right\}.$$

To simplify the practical implementation of the DtN operator, we make the assumption that the meshes of the boundaries $\Gamma_{\varepsilon,0}$, $\Gamma_{\varepsilon,1}$ coincide. Moreover, if N_J is the number of vertices in the mesh of $\Gamma_{\varepsilon,0}$, then we suppose that the vertices situated on the boundary $\Gamma_{\varepsilon,0}$ are those with the index $1 \leq j \leq N_J$ (going from up to down) and the vertices situated on the boundary $\Gamma_{\varepsilon,1}$ are those with the index $N_J + 1 \leq j \leq 2N_J$ (going from up to down as well), cf. figure 3.5. For the functions w_j corresponding to the vertices situated on the boundaries $\Gamma_{\varepsilon,0}$, $\Gamma_{\varepsilon,1}$ their traces on these boundaries will be denoted by

$$\varphi_i^0 = w_i|_{\Gamma_{\varepsilon,0}}, \qquad \varphi_i^1 = w_{N_J+i}|_{\Gamma_{\varepsilon,1}}, \qquad 1 \leqslant i \leqslant N_J.$$

Notice that all the other traces (of the functions w_j with $1 \leq j \leq N_J$ on $\Gamma_{\varepsilon,1}$, of the functions w_j with $N_J + 1 \leq j \leq 2N_J$ on $\Gamma_{\varepsilon,0}$ and of the functions w_j with $j \geq 2N_J + 1$ on both boundaries) are zero. Hence,

$$S_h^0 = Span\left\{\varphi_i^0, \ 1 \leqslant i \leqslant N_J\right\}, \qquad S_h^1 = Span\left\{\varphi_i^1, \ 1 \leqslant i \leqslant N_J\right\}.$$

Let us start by constructing the solutions of the discretized problem (3.2.35) for $\varphi \in S_h^0 \cup S_h^1$. Recall the variational formulation of the problem (3.2.35) (cf. (3.2.36)):

$$(\nabla \widehat{e}_{\delta}(\alpha,\varphi), \nabla v)_{L_{2}(\mathcal{C}_{\varepsilon})} - \alpha^{2} \left(\widehat{e}_{\delta}(\alpha,\varphi), v\right)_{L_{2}(\mathcal{C}_{\varepsilon})} = -\left(\nabla u_{\delta}(\varphi), \nabla v\right)_{L_{2}(\mathcal{C}_{\varepsilon})} + \alpha^{2} \left(u_{\delta}(\varphi), v\right)_{L_{2}(\mathcal{C}_{\varepsilon})}, \qquad \forall v \in H_{D}^{1}\left(\mathcal{C}_{\varepsilon}\right), \qquad (3.3.4)$$



Figure 3.5: The mesh \mathcal{T}_h

where $\widehat{e}_{\delta}(\alpha, \varphi) = e_{\delta}(\alpha, \varphi) - u_{\delta}(\varphi) \in H^1_D(\mathcal{C}_{\varepsilon})$ and $u_{\delta}(\varphi)$ is a lift function in $H^1(\mathcal{C}_{\varepsilon})$ such that

$$u_{\delta}(\varphi)|_{\Gamma\varepsilon,\delta} = \varphi, \qquad u_{\delta}(\varphi)|_{\Gamma\varepsilon,1-\delta} = 0.$$
 (3.3.5)

Discretizing of (3.3.4) consists in considering its restriction to the space $V_h^D = V_h \cap H_D^1(\mathcal{C}_{\varepsilon})$. Thus, for $\varphi_h \in S_h^{\delta}$, $\delta \in \{0, 1\}$, we search $\hat{e}_{\delta}^h(\alpha, \varphi_h) \in V_h^D$ such that

$$\left(\nabla \widehat{e}^{h}_{\delta}(\alpha,\varphi_{h}), \nabla v \right)_{L_{2}(\mathcal{C}_{\varepsilon})} - \alpha^{2} \left(\widehat{e}^{h}_{\delta}(\alpha,\varphi_{h}), v \right)_{L_{2}(\mathcal{C}_{\varepsilon})} = - \left(\nabla u_{\delta}(\varphi_{h}), \nabla v \right)_{L_{2}(\mathcal{C}_{\varepsilon})} + \alpha^{2} \left(u_{\delta}(\varphi_{h}), v \right)_{L_{2}(\mathcal{C}_{\varepsilon})}, \qquad \forall v \in V_{h}^{D}.$$
(3.3.6)

If $\varphi_h = \sum_{i=1}^{N_J} c_j \varphi_i^{\delta}$, then by linearity $\hat{e}_{\delta}^h(\alpha, \varphi_h) = \sum_{i=1}^{N_J} c_j \hat{e}_{\delta,i}^h(\alpha)$, where $\hat{e}_{\delta,i}^h(\alpha) = \hat{e}_{\delta}^h(\alpha, \varphi_i^{\delta})$, $1 \leq i \leq N_J, \ \delta \in \{0, 1\}$. Remarking that one can choose $u_{\delta}(\varphi_i^{\delta}) = w_{i+\delta N_J}$. Therefore, it is enough to compute the $2N_J$ functions $\hat{e}_{\delta,i}^h(\alpha), \ \delta \in \{0, 1\}, \ 1 \leq i \leq N_J$ solutions to the following system of linear equations:

$$\left(\nabla \widehat{e}^{h}_{\delta,i}(\alpha), \nabla w_{j} \right)_{L_{2}(\mathcal{C}_{\varepsilon})} - \alpha^{2} \left(\widehat{e}^{h}_{\delta,i}(\alpha), w_{j} \right)_{L_{2}(\mathcal{C}_{\varepsilon})} = - \left(\nabla w_{i+\delta N_{J}}, \nabla w_{j} \right)_{L_{2}(\mathcal{C}_{\varepsilon})} + \alpha^{2} \left(w_{i+\delta N_{J}}, w_{j} \right)_{L_{2}(\mathcal{C}_{\varepsilon})} + \alpha^{2} \left(w_{i+\delta N_{J}},$$

Let us decompose the functions $\hat{e}^{h}_{\delta,i}(\alpha)$ in the basis \mathcal{B}_{h} : $\hat{e}^{h}_{\delta,i}(\alpha) = \sum_{j=1}^{N} \hat{E}^{i}_{\delta,j}(\alpha) w_{j}$. Clearly, $\hat{E}^{i}_{\delta,j}(\alpha) = 0$ for $j \ge 2N_{J} + 1$, $1 \le i \le N_{J}$, $\delta \in \{0,1\}$. Then, the problems (3.3.7) take the form

$$\left(K^D(\alpha) - \alpha^2 M^D(\alpha)\right) \widehat{E}^i_{\delta}(\alpha) = F^i_{\delta}(\alpha), \qquad (3.3.8)$$

where

$$K_{i,j}^{D} = \begin{cases} K_{i,j}, & i,j \ge 2N_{J} + 1, \\ \delta_{i,j}, & 1 \le i,j \le 2N_{J}, \\ 0, & \text{otherwise}, \end{cases} \qquad M_{i,j}^{D} = \begin{cases} M_{i,j}, & i,j \ge 2N_{J} + 1, \\ \delta_{i,j}, & 1 \le i,j \le 2N_{J}, \\ 0, & \text{otherwise}, \end{cases}$$

 $K = (K_{i,j})$ and $M = (M_{i,j})$ are the rigidity and the mass matrices respectively defined as

$$K_{i,j} = (\nabla w_i, \nabla w_j)_{L_2(\mathcal{C}_{\varepsilon})}, \qquad M_{i,j} = (w_i, w_j)_{L_2(\mathcal{C}_{\varepsilon})}, \qquad 1 \le i, j \le N,$$
(3.3.9)

and the right-hand side $F^i_{\delta}(\alpha)$ is the $(N \times 1)$ -vector defined as

$$F_{\delta,j}^{i}(\alpha) = \begin{cases} -(K_{i+\delta N_{J},j} - \alpha^{2} M_{i+\delta N_{J},j}), & j \ge 2N_{J} + 1, \\ 0, & 1 \le j \le 2N_{J}. \end{cases}$$

Finally, the vector $E^i_{\delta}(\alpha)$ of the coordinates of the solution $e_{\delta}(\alpha, \varphi^{\delta}_i)$ in the basis \mathcal{B}_h is

$$E^i_{\delta}(\alpha) = \widehat{E}^i_{\delta}(\alpha) + I^i_{\delta}, \qquad 1 \leqslant i \leqslant N_J, \qquad \delta \in 0, 1,$$

where I_{δ}^{i} is the vector with the coordinates $I_{\delta,i+\delta N_{J}}^{i} = 1$ and $I_{\delta,j}^{i} = 0$ for $j \neq i + \delta N_{J}$.

3.3.1.2 The local DtN operators

Recalling the definition (3.2.40) and using the fact that $e_{\gamma}(\alpha, \varphi)$ solves the problem (3.2.35) with δ replaced by γ one easily gets

$$\langle T_{\gamma\delta}(\alpha)\varphi,\psi\rangle = \left(\nabla e_{\gamma}\left(\alpha,\varphi\right),\nabla u_{\delta}\left(\psi\right)\right)_{L_{2}(\mathcal{C}_{\varepsilon})} - \alpha^{2}\left(e_{\gamma}\left(\alpha,\varphi\right),u_{\delta}\left(\psi\right)\right)_{L_{2}(\mathcal{C}_{\varepsilon})}, \\ \forall\varphi\in H^{1/2}\left(\Gamma_{\varepsilon,\gamma}\right), \qquad \forall\psi\in H^{1/2}\left(\Gamma_{\varepsilon,\delta}\right), \quad (3.3.10)$$

where $u_{\delta}(\psi)$ is a function introduced in (3.3.5). We now define the approximated local DtN operators $T^{h}_{\gamma\delta}(\alpha)$ as $(N_J \times N_J)$ -matrices the (i, j) matrix elements of which approach the matrix elements of the operator $T_{\gamma\delta}(\alpha)$ computed on the functions from S^0_h, S^1_h :

$$T^h_{\gamma\delta}(\alpha)(i,j) \simeq \langle T_{\gamma\delta}(\alpha)\varphi^{\gamma}_i, \varphi^{\delta}_j \rangle, \qquad 1 \leqslant i,j \leqslant N_J.$$

More precisely, they are obtained if one replaces in (3.3.10) φ and ψ by φ_i^{γ} and φ_j^{δ} respectively, $e_{\gamma}(\alpha, \varphi)$ by $e_{\gamma,i}^{h}(\alpha)$ and $u_{\delta}(\psi)$ by $w_{J_{\delta}(j)}$:

$$T^{h}_{\gamma\delta}(\alpha)(i,j) = \left(\nabla e^{h}_{\gamma,i}(\alpha), \nabla w_{j+\delta N_{J}}\right)_{L_{2}(\mathcal{C}_{\varepsilon})} - \alpha^{2} \left(e^{h}_{\gamma,i}(\alpha), w_{j+\delta N_{J}}\right)_{L_{2}(\mathcal{C}_{\varepsilon})} \\ = \left(\left(K(\alpha) - \alpha^{2}M(\alpha)\right) E^{i}_{\gamma}(\alpha)\right)(j+\delta N_{J}), \qquad 1 \leq i, j \leq N_{J}, \qquad \gamma, \delta \in \{0,1\}.$$

The next step is now to construct the approximated propagation operator $P^{h}(\alpha)$ using the discretised analogue of the Riccati equation (3.2.43).

3.3.1.3 The propagation operator

We search the approximated propagation operator $P^h(\alpha) \in M_{N_J}(\mathbb{R})$, which is the propagation operator corresponding to the discretized problem, as the solution of the equation

$$T_{10}^{h}(\alpha) \left(P^{h}(\alpha) \right)^{2} + \left(T_{00}^{h}(\alpha) + T_{11}^{h}(\alpha) \right) P^{h}(\alpha) + T_{01}^{h}(\alpha) = 0,$$

completed by the condition

$$\rho\left(P^h(\alpha)\right) < 1. \tag{3.3.11}$$

As described in [22], this equation is solved using Newton's method in Banach spaces (in our case the space $M_{N_I}(\mathbb{R})$). Let F be the following mapping in $M_{N_I}(\mathbb{R})$:

$$F : X \mapsto T_{10}^{h}(\alpha)X^{2} + \left(T_{00}^{h}(\alpha) + T_{11}^{h}(\alpha)\right)X + T_{01}^{h}(\alpha).$$

Then, given the *n*-th iteration X_n the (n + 1)-st iteration is found from the relation

$$F'(X_n)(X_n - X_{n+1}) = F(X_n), \qquad (3.3.12)$$

where $F'(X_n)$ is the Fréchet derivative of the mapping computed F at X_n . It is defined as follows:

$$F'(X) : h \mapsto T_{10}^h(\alpha) (Xh + hX) + (T_{00}^h(\alpha) + T_{11}^h(\alpha)) h, \quad \forall X, h \in M_{N_J}(\mathbb{R}).$$

Combining this with (3.3.12) we finally obtain the equation for $\Delta_{n+1} = X_{n+1} - X_n$:

$$T_{10}^{h}(\alpha) \left(X_{n} \Delta X_{n+1} + \Delta_{n+1} X_{n} \right) + \left(T_{00}^{h}(\alpha) + T_{11}^{h}(\alpha) \right) \Delta_{n+1} = -F(X_{n})$$

This, in turn, can be rewritten as

$$A\Delta_{n+1}B^T + C\Delta_{n+1}D^T = E,$$

where

$$A = (T_{00}^{h}(\alpha) + T_{11}^{h}(\alpha))^{-1} T_{10}^{h}(\alpha), \qquad B = X_{n}^{T}, \qquad C = AB^{T} + I,$$

$$D = I, \qquad E = -(B^{T} + A(B^{T})^{2} + (T_{00}^{h}(\alpha) + T_{11}^{h}(\alpha))^{-1} T_{01}^{h}(\alpha)).$$

This equation is solved using the method described in [26]. The constraint (3.3.11) is taken into account by projecting of the obtained matrix at each iteration on the set of matrices with the spectral radius smaller than 1. More precisely, if $\rho(X_{n+1}) > 1$, then it is replaced by $X_{n+1}/\rho(X_{n+1})$.

3.3.1.4 The DtN operators and the interior problem

The approximated DtN operators $\Lambda^{+,h}(\alpha) = \Lambda^{-,h}(\alpha) = \Lambda^{h}(\alpha)$ are now obtained using the discrete analogue of the relation (3.2.45):

$$\Lambda^h(\alpha) = T^h_{00}(\alpha) + T^h_{10}(\alpha)P^h(\alpha).$$

The discretised version of the interior problem (3.2.28) is obtained from the weak formulation by replacing the DtN operators $\Lambda^{\pm}(\alpha) = \Lambda(\alpha)$ by their discrete analogue $\Lambda^{h}(\alpha)$. We denote by V_{h}^{μ} the analogue of the space (3.3.2) for the interior domain C_{ε}^{μ} :

$$V_h^{\mu} = \left\{ v_h \in C(\overline{\mathcal{C}_{\varepsilon}^{\mu}}), \quad v_h|_{T_{\ell}^{\mu}} \text{ is linear, } 1 \leqslant \ell \leqslant L^{\mu} \right\},$$

where L^{μ} is the number of triangles T_{ℓ}^{μ} in the mesh \mathcal{T}_{h}^{μ} of $\mathcal{C}_{\varepsilon}^{\mu}$. The set of the vertices of this mesh is denoted by $\mathcal{M}_{h}^{\mu} = \{M_{j}^{\mu}\}_{j=1}^{N^{\mu}}$ and the basis $\mathcal{B}_{h}^{\mu} = \{w_{j}^{\mu}\}_{j=1}^{N^{\mu}}$ is constructed analogously to (3.3.3). To be able to match this mesh with the mesh \mathcal{T}_{h} from the right we need the number of vertices in the mesh of the boundary Γ_{ε}^{+} to be equal to the number of vertices in the mesh of Γ_{0} (which is N_{J}). Similarly, the number of vertices in the mesh Γ_{ε}^{-} should be the same as the number of vertices in the mesh of Γ_{1} (which is also N_{J}). We suppose for simplicity that the vertices that belong to the boundary Γ_{ε}^{-} are those with the index $1 \leq N_{J}$ (enumerated from up to down) and the vertices that belong to the boundary Γ_{ε}^{+} are those with the index $N_{J} + 1 \leq 2N_{J}$ (enumerated from up to down as well). Then, we have to solve the following problem:

$$\left(\nabla u_{\varepsilon}^{0,h}, \nabla v\right) + \left\langle \Lambda^{h}\left(\lambda_{\varepsilon}^{h}\right) u_{\varepsilon}^{0,h}, v \right\rangle \Big|_{\Gamma_{\varepsilon}^{+}} + \left\langle \Lambda^{h}\left(\lambda_{\varepsilon}^{h}\right) u_{\varepsilon}^{0,h}, v \right\rangle \Big|_{\Gamma_{\varepsilon}^{-}} = \left(\lambda_{\varepsilon}^{h}\right)^{2} \left(u_{\varepsilon}^{0,h}, v\right), \qquad \forall v \in V_{h}^{\mu}.$$
If $U_{\varepsilon}^{0,h}$ is the vector of coordinates of $u_{\varepsilon}^{0,h}$ in the basis \mathcal{B}_{h}^{μ} , then this problem can be rewritten in a matrix form as

$$\left(K^{\mu} + L\left(\lambda_{\varepsilon}^{h}\right)\right) U_{\varepsilon}^{0,h} = \left(\lambda_{\varepsilon}^{h}\right)^{2} M^{\mu} U_{\varepsilon}^{0,h}, \qquad (3.3.13)$$

where $L(\alpha)$ is the $(N^{\mu} \times N^{\mu})$ -matrix such that

$$L_{i,j}(\alpha) = \begin{cases} \left(\Lambda^{h}(\alpha)\right)_{(J^{+})^{-1}(i),(J^{+})^{-1}(j)}, & i, j \in \mathcal{J}^{+}, \\ \left(\Lambda^{h}(\alpha)\right)_{(J^{-})^{-1}(i),(J^{-})^{-1}(j)}, & i, j \in \mathcal{J}^{-}, \\ 0, & \text{otherwise}, \end{cases}$$

and K^{μ} , M^{μ} are the analogues of the matrices (3.3.9) for the mesh \mathcal{T}^{μ}_{h} . The problem (3.3.13) is a nonlinear generalized eigenvalue problem (the nonlinearity is contained in the dependence of the matrix L on the spectral parameter). It is solved using the false position method described below (cf. for, example, [59]).

3.3.1.5 False position method

Let $]a^h_{\varepsilon}, b^h_{\varepsilon}[$ be an approximation of a gap of the operator A^{μ}_{ε} obtained as described in section 3.3.2. In order to solve the problem (3.3.13) we introduce the functions $\varkappa_n(\alpha)$, $n \in \mathbb{N}$, for $\alpha^2 \in]a^h_{\varepsilon}, b^h_{\varepsilon}[$ that correspond to *n*-th biggest generalized eigenvalue of the problem

$$(K^{\mu} + L(\alpha))U = \varkappa^{h}(\alpha)M^{\mu}U.$$
(3.3.14)

Thus, $\lambda_{\varepsilon}^{h}$ is a solution of the nonlinear equation

$$\varkappa_n^h(\alpha) = \alpha^2$$

for some $n \in \mathbb{N}$. Let $\varkappa^*(\alpha)$ be the nearest to α^2 eigenvalue:

$$\varkappa^{h,*}(\alpha) = \varkappa^h_n(\alpha), \qquad \left|\varkappa^h_n(\alpha) - \alpha^2\right| = \min_{j \in \mathbb{N}} \left|\varkappa^h_j(\alpha) - \alpha^2\right|.$$

Then, the problem is reduced to searching the roots of the function

$$f(\alpha) = \varkappa^{h,*}(\alpha) - \alpha^2. \tag{3.3.15}$$

This function is piecewise continuous on the interval $]a_{\varepsilon}^{h}, b_{\varepsilon}^{h}[$. In order to find its roots we shall first determine empirically the intervals where it is continuous, monotone and has a unique root and then apply the false position method on each of this intervals (it is impossible to apply the standard Newton's algorithm since we do not have analytical expressions for the derivative of the function $\varkappa^{h,*}(\alpha)$). Let $[\xi, \eta]$ be such an interval. Then, we initialize ξ_1 by ξ , η_1 by η and put at *n*-th iteration

$$x_n = \frac{\xi_n f(\eta_n) - \eta_n f(\xi_n)}{f(\eta_n) - f(\xi_n)},$$
(3.3.16)

$$\xi_{n+1} = \xi_n, \quad \eta_{n+1} = x_n, \quad \text{if} \quad f(\xi_n) f(x_n) < 0, \quad (3.3.17)$$

$$\xi_{n+1} = x_n, \quad \eta_{n+1} = \eta_n, \qquad \text{if} \qquad f(\eta_n) f(x_n) < 0.$$
 (3.3.18)

The limit of the sequence $\{x_n\}$ is the value λ_{ε}^h which gives an approximation of λ_{ε} . The initial interval $[\xi, \eta]$ can be chosen for example by plotting the function $f(\alpha)$. We can also note that we expect to find one or two such intervals depending on the type of the corresponding gap of the operator A^{μ} on the graph (cf. Theorem 1.3.1) at least for ε small enough. The knowledge of the eigenvalues of the operator A^{μ} also gives an idea about the location of the ones of the operator A_{ε}^{μ} and can help to choose the intervals in question.

3.3.2 The essential spectrum

There are two ways of determining the essential spectrum of the operator A_{ε}^{μ} . The first one is based on the computation of the spectral radius of the propagation operator (cf. Remark 3.2.4). Discretizing the interval of interest and computing the spectral radius of the propagation operator at each point of the grid one finds if this point belongs to the essential spectrum or not. However, this method requires the computation of the propagation operator, and, consequently, the solution of the Riccati equation, for each value of α . For this reason we privilege a more standard method which is based on the Floquet-Bloch decomposition and the determination of dispersion curves. As it was mentioned in Section 1.2.1, the essential spectrum of the operator A_{ε}^{μ} can be decomposed as

$$\sigma_{ess}\left(A_{\varepsilon}^{\mu}\right) = \bigcup_{n \in \mathbb{N}} \lambda_{n}(\varepsilon, [0, \pi]),$$

where $\lambda_n(\varepsilon, \theta)$ is the *n*-th biggest eigenvalue of the operator $A_{\varepsilon}(\theta)$ defined in (1.2.3). Recall that for fixed ε and *n* it is a continuous function of θ . Hence, the above characterisation of the spectrum can be rewritten as

$$\sigma_{ess}\left(A_{\varepsilon}^{\mu}\right) = \bigcup_{n \in \mathbb{N}} \left[\min_{\theta \in [0,\pi]} \lambda_{n}(\varepsilon,\theta), \ \max_{\theta \in [0,\pi]} \lambda_{n}(\varepsilon,\theta) \right].$$

Thus, it is sufficient to discretize the interval $[0, \pi]$ by the points $0 < \theta_1 < \theta_2 \dots \theta_K < \pi$ and compute the first Q eigenvalues of the operator $A_{\varepsilon}^h(\theta_i)$ (obtained after a discretization of the operator $A_{\varepsilon}(\theta_i)$) for each of this points. This will yield an approximation of the beginning of the essential spectrum of the operator A_{ε}^{μ} :

$$\sigma_{ess}^{K,Q}(A_{\varepsilon}^{\mu}) = \bigcup_{1 \leq n \leq Q} \left[\min_{1 \leq i \leq K} \lambda_n(\varepsilon, \theta_i), \ \max_{1 \leq i \leq K} \lambda_n(\varepsilon, \theta_i) \right].$$
(3.3.19)

In order to find approximations of the eigenvalues of the operator $A_{\varepsilon}(\theta)$ for some $\theta \in [0, \pi]$ we use the mesh \mathcal{T}_h described in section 3.3.1.1. However, we shall now take into account the θ -quasiperiodic boundary conditions on the boundaries $\Gamma_{\varepsilon,0}$, $\Gamma_{\varepsilon,1}$. It can be expressed by the fact that the space V_h should be replaced by its θ -quasiperiodic subspace

$$V_h^{\theta} = \left\{ v_h \in V_h, \quad v_h|_{\Gamma_{\varepsilon,1}} = e^{-i\theta} v_h|_{\Gamma_{\varepsilon,0}} \right\}.$$

Then, the weak formulation for the eigenvalue problem for the operator $A_{\varepsilon}(\theta)$ implies solving the following discretised problem for $\lambda_{\varepsilon}^{h}(\theta)$ and $u \in V_{h}^{\theta}$:

$$(\nabla u, \nabla v)_{L_2(\mathcal{C}_{\varepsilon})} = \lambda_{\varepsilon}^h(\theta)(u, v)_{L_2(\mathcal{C}_{\varepsilon})}, \qquad \forall v \in V_h^{\theta}.$$

From a practical point of view, the space V_h^{θ} is obtained from V_h by considering the basis $\mathcal{B}_h^{\theta} = \{\widetilde{w}_j, j \in [1, N_J] \cup [2N_J + 1, N]\}$ in the space V_h^{θ} , where

$$\widetilde{w}_j = \begin{cases} w_j, & j \ge 2N_J + 1, \\ w_j + e^{-i\theta} w_{j+N_J}, & 1 \le j \le N_J. \end{cases}$$

Decomposing u in this basis as $u = \sum_{j=1}^{N_J} U_j \widetilde{w}_j + \sum_{j=2N_J+1}^{N} U_j \widetilde{w}_j$ and denoting by U the $(N - N_J) \times 1$ vector of coordinates U_j (where the coordinates with $N_J + 1 \leq j \leq 2N_J$ are excluded), we end up with the following generalised eigenvalue problem:

$$\widetilde{K}U = \lambda_{\varepsilon}^{h}(\theta)\widetilde{M}U. \tag{3.3.20}$$

Here \widetilde{K} and \widetilde{M} are $(N - N_J) \times (N - N_J)$ with matrix elements defined as follows:

$$\widetilde{K}_{i,j} = (\nabla \widetilde{w}_i, \nabla \widetilde{w}_j)_{L_2(\mathcal{C}_{\varepsilon})}, \quad \widetilde{M}_{i,j} = (\widetilde{w}_i, \widetilde{w}_j)_{L_2(\mathcal{C}_{\varepsilon})}, \qquad j \in [1, N_J] \cup [2N_J + 1, N].$$

Once the matrices K, M are known, the matrices \widetilde{K} , \widetilde{M} can be constructed using the relations

$$\widetilde{K}_{i,j} = \begin{cases} K_{i,j}, & i, j \ge 2N_J + 1 \\ K_{i,j} + e^{-i\theta} K_{i+N_J,j}, & 1 \le i \le N_J, & j \ge 2N_J + 1, \\ K_{i,j} + e^{i\theta} K_{i,j+N_J}, & i \ge 2N_J + 1, & 1 \le j \le N_J, \\ K_{i,j} + e^{-i\theta} K_{i+N_J,j} + e^{i\theta} K_{i,j+N_J} + K_{i+N_J,j+N_J}, & 1 \le i, j \le N_J, \end{cases}$$

and its analogue for the matrix \widetilde{M} . Now the generalized eigenvalue problem (3.3.20) can be solved for each θ_i , $1 \leq i \leq K$ which yields an approximation of the beginning of the essential spectrum of the operator A^{μ}_{ε} due to (3.3.19). We remark however one drawback of this method compared to the method based on the computation of the spectral radius of the propagation operator: we do not know a priori how many spectral bands we have to compute in order to cover the interval of frequences we are interested in. Here again the knowledge of the spectrum of the operator A on the graph can give an approximate idea, at least for ε small enough.

3.3.3 Summary of the algorithm

Let us now resume the algorithm of computation of the eigenvalues of the operator A^{μ}_{ε} and give some details of implementation in Matlab.

I Essential spectrum

- The function CalculSpectreEssentiel.m computes an approximation of the beginning of the essential spectrum of the operator A^{μ}_{ε} using the first N eigenvalues $\lambda_n(\varepsilon, \theta)$.
- The function Calc_spec_ess_R.m computes an approximation of the essential spectrum of the operator A^{μ}_{ε} on the interval $[a_0^2, b_0^2]$ using the spectral radius of the propagation operator.

Using one of these methods one finds approximations of gaps of the operator A_{ε}^{μ} . To be precise, the result provided by these functions refers to the square root of the spectral parameter. In order to pass to the spectral parameter, we should take the square of the obtained result. Let us fix one of the approximated gaps, that we denote by $]a^2, b^2[$, and look for eigenvalues inside it.

II Discrete spectrum

(a) Let us start by plotting the function $f(\alpha)$ defined in (3.3.15) on the interval]a, b[. This step can be skipped if an interval $[\xi, \eta]$ described in Section 3.3.1.5 can be found in another way.

Construction of the matrices K, M and $K^{\mu}, M^{\mu} \leftarrow KMA.m$

for $\alpha = a : pas : b \leftarrow dessin_eig.m$

compute $\varkappa^*(\alpha) \leftarrow \text{compute_eig.m}$ which consists in

- construction of the matrices $T^h_{\gamma\delta}(\alpha) \leftarrow \text{DtN_eig.m}$
 - construction of the matrices $A(\alpha)$ and $A^D(\alpha)$
 - construction of the matrices $E_0(\alpha)$, $E_1(\alpha)$
 - construction of the matrices $T_{00}^h(\alpha)$, $T_{01}^h(\alpha)$, $T_{10}^h(\alpha)$, $T_{11}^h(\alpha)$
- computation of the matrix $P^h(\alpha) \leftarrow \text{newton1.m}$ (solving the Riccati equation)
- construction of the matrices $L^{\pm}(\alpha)$
- solving the eigenvalue problem $(3.3.14) \leftarrow Matlab function eigs.m$ end
- (b) Once an interval $[\xi, \eta]$ is found, the false position method is applied on this interval.

while $(abs(f(x_n)) > tol)$ or $(abs(x_n-x_{n-1}) > tol) \leftarrow quasinewton.m$ compute x_{n+1} (cf. (3.3.16)) \leftarrow using compute_eig.m compute ξ_{n+2} , η_{n+2} (cf. (3.3.17)-(3.3.18)) end

3.4 Numerical results

In this section we present numerical results obtained with a Matlab code which implements the algorithm described above.

3.4.1 Validation of the computation of the local DtN operators

We take as a test geometry the square with a unit edge shown in figure 3.6.



Figure 3.6: Test geometry: a unit square

The first test problem is the following:

$$\begin{cases} \Delta u = 0, \\ u|_{\Gamma_2} = 1, \\ u|_{\Gamma_3} = 0, \\ \partial_n u|_{\Gamma_1 \cup \Gamma_4} = 0. \end{cases}$$
(3.4.1)

Its exact solution is $u_{ex}(x,y) = 1 - x$. We compute the matrix $E_0(0)$ with the help of the function DtN.m. It follows from the definition of the matrix $E_0(0)$ that the sum of its columns gives an approximation of u_{ex} . It is shown in figure 3.7. In this case there is no error due to the finite element approximation since the exact solution is linear.

The second test problem is

$$\begin{cases} -\Delta u = \frac{\pi^2}{4}u, \\ u|_{\Gamma_2} = \cos\left(\frac{\pi y}{2}\right), \\ u|_{\Gamma_3} = \cos\left(\frac{\pi y}{2}\right), \\ u|_{\Gamma_4} = 0, \\ \partial_n u|_{\Gamma_1} = 0. \end{cases}$$
(3.4.2)

Its exact solution is $u_{ex}(x,y) = \cos(\frac{\pi y}{2})$. Let U_{ex} be the vector of values of u_{ex} at the vertices of the grid \mathcal{T}_h :

$$U_{ex}(j) = \cos\left(\frac{\pi y(M_j)}{2}\right), \qquad 1 \le j \le N.$$



Figure 3.7: Solution of the problem (3.4.1) obtained with the function DtN.m using an unstructured grid with h = 0.1

We compute the matrices $E_0\left(\frac{\pi}{2}\right)$, $E_1\left(\frac{\pi}{2}\right)$ with the help of the function DtN.m. This yields the following approximation of U_{ex} :

$$U = \sum_{i=1}^{N_y} \left(E_0^i\left(\frac{\pi}{2}\right) \cos\left(\frac{\pi y(M_{J_0(i)})}{2}\right) + E_1^i\left(\frac{\pi}{2}\right) \cos\left(\frac{\pi y(M_{J_1(i)})}{2}\right) \right),$$

where E_0^i , E_1^i are the *i*-th columns of the matrices E_0 , E_1 respectively. The solution U is represented in figure 3.8a for an unstructured grid with h = 0.1. We study the L_2 and H^1 errors given by the relations

$$err_L2 = ((U - U_{ex})^t M (U - U_{ex}))^{1/2}, \quad err_H1 = ((U - U_{ex})^t K (U - U_{ex}))^{1/2}$$

In figure 3.8b the dependence of these errors on h is represented in a logarithmic scale. As we see, when $h \to 0$, the slopes are in agreement with the theoretic ones: 2 for the L_2 error and 1 for the H^1 error.

Next, we validate the computation of the operators $T_{\gamma\delta}$, $\gamma, \delta \in \{0, 1\}$ considering the test problems

$$\begin{cases} -\Delta u_{0} = \omega^{2} u_{0}, \\ u_{0}|_{\Gamma_{2}} = \cos(k\pi y), \\ u_{0}|_{\Gamma_{3}} = 0, \\ \partial_{n} u_{0}|_{\Gamma_{1}\cup\Gamma_{4}} = 0, \end{cases} \qquad \begin{cases} -\Delta u_{1} = \omega^{2} u_{1}, \\ u_{1}|_{\Gamma_{2}} = 0, \\ u_{1}|_{\Gamma_{3}} = \cos(k\pi y), \\ \partial_{n} u_{1}|_{\Gamma_{1}\cup\Gamma_{4}} = 0, \end{cases}$$
(3.4.3)

for $k \in \mathbb{N}$. The exact expressions for $T_{\gamma\delta}(\omega)\varphi_k$, $f_k(y) = \cos(k\pi y)$, are

$$T_{00}(\omega)f_k = T_{11}(\omega)f_k = \begin{cases} \frac{\alpha}{\tan\alpha}\cos\left(k\pi y\right), & \omega > k\pi, \\ \cos\left(k\pi y\right), & \omega = k\pi, \\ |\alpha|\frac{1+e^{2|\alpha|}}{e^{2|\alpha|}-1}\cos\left(k\pi y\right), & \omega < k\pi, \end{cases}$$

$$\int -\frac{\alpha}{\sin\alpha}\cos\left(k\pi y\right),\qquad \qquad \omega > k\pi,$$

$$T_{01}(\omega)f_k = T_{10}(\omega)f_k = \begin{cases} -\cos\left(k\pi y\right), & \omega = k\pi \\ -2|\alpha|\frac{e^{|\alpha|}}{e^{2|\alpha|} - 1}\cos\left(k\pi y\right), & \omega < k\pi \end{cases}$$

where $\alpha = \sqrt{(k\pi)^2 - \omega^2}$.



(a) Solution U obtained for an unstructured grid with h=0.1



(b) L_2 and H^1 error for different values of h

Figure 3.8: Numerical resolution of the test problem (3.4.2) using the function DtN.m

We compute the L_2 error

$$error_T_{\gamma\delta} = \left\| T^{h}_{\gamma\delta}(\omega)f_{k} - T_{\gamma\delta}(\omega)f^{h}_{k} \right\|_{L_{2}(\Gamma_{\delta+2})}$$
$$\simeq \left(\left(T^{h}_{\gamma\delta}(\omega)F_{k} - S^{(\delta)}Y^{k}_{\gamma\delta}(\omega) \right)^{t}S^{(\delta)} \left(T^{h}_{\gamma\delta}(\omega)F_{k} - S^{(\delta)}Y^{k}_{\gamma\delta}(\omega) \right) \right)^{1/2}. \quad (3.4.4)$$

Here $T_{\gamma\delta}^h(\omega)$ are the approximated local DtN operators computed with the function DtN.m, F_k is the vector of values of the function f_k at the vertices \mathcal{M}_h^{γ} , $Y_{\gamma\delta}^k(\omega)$ is the vector of values of $T_{\gamma\delta}(\omega)f_k$ at the vertices \mathcal{M}_h^{δ} and $S^{(\delta)}$ are the surface mass matrices for the boundaries $\Gamma_{\delta+2}$ defined as follows:

$$S_{i,j}^{(\delta)} = \int_{\Gamma_{\delta+2}} \varphi_i^{\delta} \varphi_j^{\delta}, \qquad 1 \leqslant i, j \leqslant N_J, \qquad \delta \in \{0,1\}.$$

In figure 3.9 we represent the dependence of the error (3.4.4) on h for $\omega = 5$, k = 2. The slope in a logarithmic scale is 2.



Figure 3.9: Dependence of the error (3.4.4) on h

In figure 3.10 we represent the dependence of the error on ω for fixed k and h. We can see that it has singularities at the points ω that belong to the set $\{\pi\sqrt{m^2 + n^2}, m \in \mathbb{N}, n \in \mathbb{N}^*\}$. For these points ω^2 is an eigenvalue of the problem

$$\begin{cases}
-\Delta u = \omega^2 u, \\
u|_{\Gamma_{2,3}} = 0, \\
\partial_n u|_{\Gamma_1 \cup \Gamma_4} = 0.
\end{cases}$$
(3.4.5)



Figure 3.10: Dependence of the error (3.4.4) on ω : as expected, singularities are observed at the points corresponding to the eigenvalues of the problem (3.4.5)

3.4.2 Numerical computation of the essential spectrum

We now compute the essential spectrum of the operator $A_{\varepsilon,s}$ for L = 2 using the function CalculSpectreEssentiel.m. In figure 3.11 we represent the essential spectrum computed for different values of ε using the first 5 spectral bands. The intervals in blue correspond to the values λ such that λ^2 belongs to the spectral bands and the intervals in white separating them correspond to the values λ such that the λ^2 belongs to the spectral bands and the spectral gaps. In what follows we mean by representing the spectrum the representation of the parameter λ (and not λ^2 which is the spectral parameter). For $\varepsilon = 0$ the spectrum of the limit operator A_s is represented.



Figure 3.11: Dependence of the essential spectrum of the operator A^s_{ε} on ε : the first 4 gaps. For each value of ε an unstructured grid has been used with h chosen in such a way that the error due to the discretization is very small compared to the effect due to ε : h = 0.00125 for $\varepsilon = 0.02$ and $\varepsilon = 0.04$, h = 0.0025 for $\varepsilon = 0.06$, $\varepsilon = 0.08$, $\varepsilon = 0.1$.

One can see that the spectrum of the operator $A_{\varepsilon,s}$ is very close to the spectrum of the limit operator for small values of ε (more precisely, the convergence is linear as it is predicted by the theory, cf. figure 3.12). However, we can notice a phenomenon that has not been studied theoretically in Chapter 1: opening of a gap near the values $\{\pi\mathbb{N}^*\}$. These are the points where the dispersion curves for the limit operator A_s touch (cf. figure 3.14). As shown in figure 3.13, the size of these gaps is also linear in h.



Figure 3.12: Dependence of the lower end of the first gap on ε : linear convergence to the lower end of the first gap of the limit operator A_s .



Figure 3.13: Opening of a gap in the neighbourhood of the point 2π .



Figure 3.14: Dispersion curves in the case L = 2 for the graph and the 2D domain with $\varepsilon = 0.1$: the points $\{\lambda = \pi n, n \in \mathbb{N}^*\}$ where two spectral bands for the graph touch become gaps for the 2D domain.

Another phenomenon concerns the eigenvalues of infinite multiplicity for the limit operator. Consider the case L = 0.5. Then, the operator A_s has the following set of eigenvalues of infinite multiplicity: $\sigma_{pp} = \{\lambda^2, \lambda = 2(2n+1)\pi, n \in \mathbb{N}\}$ (cf. property 4 of Proposition 1.3.4). An eigenvalue of infinite multiplicity can only become a spectral band in the 2D case with ε small enough. Indeed, as shown in [57], the dimension of the spectral projector on any interval is preserved for ε small enough. On the other hand, a periodic 2D operator cannot have eigenvalues. Thus, the only possibility is that the operator $A_{\varepsilon,s}$ has a small spectral band in a neighbourhood of an eigenvalue of infinite multiplicity. This situation is shown in figure 3.15.

3.4.3 Numerical computation of the discrete spectrum

We now present the results for the discrete spectrum of the operator $A^{\mu}_{\varepsilon,s}$. In figure 3.16 the function $f(\alpha)$ defined in (3.3.15) is represented in the first gap of the operator $A^{\mu}_{\varepsilon,s}$ in the case L = 2, $\mu = 0.25$, $\varepsilon = 0.1$. We see from this graph that there are exactly 2 roots of the function f in the gap which correspond to two eigenvalues. The interval $[\xi, \eta]$ can be chosen, for instance, as [1.4, 1.5] for the first eigenvalue and [1.9, 2] for the second one. The eigenvalues computed for different values of ε using the function quasinewton.m are represented in figure 3.17. In figure 3.18 we show the convergence of the eigenvalues in the first gap which is linear as predicted by the theory. In figure 3.19 the eigenfunction corresponding to the first eigenvalue of the operator $A^{\mu}_{\varepsilon,s}$ is represented.

In figure 3.20 we represent the dependence of the eigenvalues on $\mu \in]0, 1[$. As it is natural to expect, the smaller μ is (so, the stronger the perturbation is), the better the eigenvalues are separated from the essential spectrum. When μ is close to 1, the computation becomes more costly since the distance between the eigenvalue and the essential spectrum is very small.



(b) Zoom at the neighbourhood of the point $\lambda = 2\pi$.

Figure 3.15: Case L = 0.5: the size of the spectral band that appears in a neighbourhood of the point $\lambda = 2\pi$ (which is an eigenvalue of infinite multiplicity of the limit operator A_s) is linear in ε .



Figure 3.16: Function $f(\alpha) = \varkappa^{h,*}(\alpha) - \alpha^2$ in the first gap of the operator $A^{\mu}_{\varepsilon,s}$ for L = 2, $\mu = 0.25$, $\varepsilon = 0.1$.



Figure 3.17: Eigenvalues of the operator $A_{\varepsilon,s}^{\mu}$ for L = 2, $\mu = 0.25$ (red asterisks) computed with the function quasinewton.m. The values for $\varepsilon = 0$ correspond to the limit operator A_s^{μ} .



Figure 3.18: Linear convergence of the eigenvalues of the operator $A^{\mu}_{\varepsilon,s}$ in the first gap for $L = 2, \ \mu = 0.25$ as $\varepsilon \to 0$.



Figure 3.19: Eigenfunction corresponding to the first eigenvalue of the operator $A^{\mu}_{\varepsilon,s}$ for $L = 2, \ \mu = 0.25$.



Figure 3.20: Dependence of the eigenvalues in the first gap on μ for L = 2, $\varepsilon = 0.1$.

Another question for which no theoretical answer has been given for the moment is what happens for larger values of ε when the spectrum of the operator $A^{\mu}_{\varepsilon,s}$ is not close to the spectrum of the limit operator. In particular, a gap that exists for small values of ε does it still exist for any value of ε (until the obstacles disappear)? The eigenvalues inside the gaps that exist for the limit operator do they exist for any value of ε or do they immerse into the essential spectrum before the gap disappears?

In the cases that we tested numerically we saw that the gaps were present for any value of ε for which the obstacles are present, i.e. for $\varepsilon \in [0, \min\{1, L/2\}]$. In figure 3.21a we show the dependence of the first two gaps on ε in the case L = 2. Of course, when ε is close to 1 the computation becomes very costly since the size of the gaps is very small. For any discretization taken we obtain reliable results only up to a certain value of ε that is smaller than 1. However, the existence of the gaps at least up to this value (0.9 in the particular case represented in figure 3.21a) permits to conjecture that they exist for any $\varepsilon < 1$.

The behaviour of the eigenvalues is even more unclear. In figure 3.21b we represent the eigenvalues in the first gap of the operator $A^{\mu}_{\varepsilon,s}$ for L = 2, $\mu = 0.25$. It seems that the eigenvalues immerse into the essential spectrum for some values of $\varepsilon < 1$ (the second eigenvalue seems to disappear between $\varepsilon = 0.55$ and $\varepsilon = 0.6$). Again, the analysis becomes costly when the eigenvalues approach the essential spectrum. For this reason we cannot distinguish between the case when the eigenvalues do not exist any more and the case when they exist but are very close to the essential spectrum. For the second eigenvalue the computation with a mesh of size h = 0.0025 gave the following results: for $\varepsilon = 0.55$ the upper edge of the first gap $b_1 \approx 2.995$, the second eigenvalue $\lambda_2 \approx 2.987$, for $\varepsilon = 0.6$ $b_1 \approx 3.1$, the second eigenvalue was not found.



(a) Dependence of the first two gaps of the operator $A_{\varepsilon,s}$ on ε for L = 2. The gaps are present for any $\varepsilon < 1$.



(b) Eigenvalues of the operator $A_{\varepsilon,s}$ in the first gap for L = 2, $\mu = 0.25$ seem to immerse into the essential spectrum at some $\varepsilon < 1$.

Figure 3.21: Behaviour of the spectrum for large values of ε .

Comparison with the eigenvalues found from using asymptotic development

We will now compare the results obtained for the eigenvalues using the DtN operator method described in this chapter with the results obtained from the asymptotic expansion of the eigenvalue found in Chapter 2. We remind the algorithm of computation of the first n terms of the asymptotic expansion.

Algorithm of computation of $\lambda^{(n)}$

I Initialization:

• $\lambda^{(0)}$ is an eigenvalue of the limit operator A_s^{μ} . It is computed by solving the equation (1.3.55) (we remind that according to the notation of Chapter 2 λ should be replaced by $\sqrt{\lambda}$):

$$1 - \frac{\sqrt{g^2(\sqrt{\lambda}) - 1}}{|g(\sqrt{\lambda}) + \cos\sqrt{\lambda}|} = \mu.$$

• $u^{(0)}$ is the corresponding eigenfunction on the graph given by the relations (2.2.40)–(2.2.42). We represent it in the form (2.3.4)–(2.3.5):

$$\widetilde{a}_{0}^{(0)}(j) = 1, \qquad \widetilde{b}_{0}^{(0)}(j) = \frac{1}{\sin\sqrt{\lambda^{(0)}}} \left(r - \cos\sqrt{\lambda^{(0)}}\right), \qquad j \in \mathbb{N},$$
$$\widetilde{c}_{0}^{(0)}(j) = \frac{1}{\cos\left(\frac{\sqrt{\lambda^{(0)}L}}{2}\right)}, \qquad \widetilde{d}_{0}^{(0)}(j) = 0, \qquad j \in \mathbb{N}^{*}.$$

• $U_j^{(0)} = \mathbf{u}_j^{(0)}, \quad \forall j \in \mathbb{Z}.$ We represent it in the form (2.3.21):

$$\mathcal{U}_0^{(0)} = 1, \qquad j \in \mathbb{N}^*.$$

II Computation of the functions W_j^{\pm} solving the problems (2.2.18), (2.2.19) : P_1 finite elements. There are 4 problems to solve: (2.2.18) and (2.2.19) for j = 0. and for $j \neq 0$. The solutions of these problems are unique modulo a constant. In order to fix the missing constants, we add the condition

$$\int_{\Sigma_j^+} W^{\pm} = 0.$$

III For k = 1 : n

 $\left\{\lambda^{(\ell)}, u^{(\ell)}, U^{(\ell)}\right\}_{\ell=1}^{k-1}$ are known

- 1) Computation of the right-hand sides of the far field problem for $u^{(k)}$ and the near field problem for $U^{(k)}$ (see the proof of Proposition 2.3.1):
- The function $f^{(k-1)}$ in the form (2.3.24)–(2.3.25):
 - Construction of the polynomials $\left\{a_{f,\ell}^{(k-1)}, b_{f,\ell}^{(k-1)}, c_{f,\ell}^{(k-1)}, d_{f,\ell}^{(k-1)}\right\}, \ 0 \leq \ell \leq k-1$;
 - Construction of $f_0^{(k-1)}$.

- The coefficients $\left\{c_{j,p,\delta,\ell}^{(k)}, j \in \mathbb{Z}, \delta \in \{+,-,0\}, 1 \leq \ell \leq \left[\frac{k}{2}\right], 1 \leq p \leq N\right\}$ in the form (2.3.28) with some N corresponding to the truncation of the infinite series in (2.2.24)–(2.2.26):
 - Construction of the coefficients $\check{c}_{q,p,\delta,\ell}^{(k)}$, $0 \leq q \leq k$, $\delta \in \{+,-,0\}$, $1 \leq \ell \leq \left\lfloor \frac{k}{2} \right\rfloor$, $1 \leq p \leq N$;
 - Construction of $c_{0,p,\delta,\ell}^{(k)}$, $\delta \in \{+,-,0\}$, $1 \leq \ell \leq \left[\frac{k}{2}\right]$, $1 \leq p \leq N$.
- The functions $\left\{g_{j,\delta}^{(k-1)}\right\}_{j\in\mathbb{Z}}, \delta \in \{+,-,0\}$, in the form (2.3.31):
 - Construction of the functions $\left\{\check{g}_{q,\delta}^{(k-1)}, 0 \leqslant q \leqslant k-1, \delta \in \{+,-,0\}\right\};$
 - Construction of $\left\{g_{0,\delta}^{(k-1)}\right\}, \delta \in \{+, -, 0\}.$
- The functions $\left\{\Phi_{j}^{(k-1)}\right\}_{j\in\mathbb{Z}}$, in the form (2.3.34):
 - Construction of the functions $\check{\Phi}_q^{(k-1)}$, $0 \leq q \leq k-1$;
 - Construction of $\Phi_0^{(k-1)}$.
- The sequences $\left\{\Xi_{j}^{(k-1)}\right\}_{j\in\mathbb{Z}}$, $\left\{\Delta_{j,\pm}^{(k-1)}\right\}_{j\in\mathbb{Z}}$ in the form (2.3.36):
 - Construction of the polynomials $q_{\Sigma}^{(k-1)}$, $q_{\Delta_{\pm}}^{(k-1)}$;

- Construction of
$$\Xi_0^{(k-1)}$$
, $\Delta_{0,\pm}^{(k-1)}$.

2) $\lambda^{(k)}$ is given by the relation (2.2.60):

$$\lambda^{(k)} = \left\| u^{(0)} \right\|_{L_2^{\mu}(G_{\mathcal{C}})}^{-2} \left(\sum_{j \in \mathbb{Z}} \widetilde{\Xi}_j^{(k-1)} \mathbf{u}_j^{(0)} - \left(f^{(k-1)}, u^{(0)} \right)_{L_2^{\mu}(G_{\mathcal{C}})} \right).$$

- 3) Resolution of the far field problem for $u^{(k)}$:
 - We search $u^{(k)}$ in the form (2.3.4)–(2.3.5). As explained in the proof of Lemma 2.3.1, the problem reduces to solving a linear system (2.3.17) of size $k \times k$. This permits to find $u^{(k)}$ at all edges of the graph $G_{\mathcal{C}}$ except at the edge e_0 (since in (2.3.5) $j \ge 1$). To determine $u_0^{(k)}$, one can use the transmission conditions at the vertex v_0^- . Let us note that there are two transmission conditions at the vertex v_0^- : the jump condition (which is the same to the right and to the left of the vertex due to the symmetry of the problem with respect to the axis y = 0, see Lemma 2.2.6) and the Kirchhoff's condition to determine one missing parameter $\tilde{c}_0^{(k)}$ (since $\tilde{d}_0^{(k)}$ is defined by the relation (2.3.8)). One can use any of these two conditions, and the other one will be automatically satisfied since the far field problem has a solution (the other condition is in fact equivalent to the compatibility condition (2.2.60)).
- 4) Resolution of the near field problem for $U^{(k)}$: We search $U^{(k)}$ in the form (2.3.21). Thus, there are k+2 problems to solve using P_1 finite elements: k+1 problems for the functions $\left\{\mathcal{U}_q^{(k)}\right\}_{q=0}^k$ (which permits to find all the near field terms except $U_0^{(k)}$) and the problem for $U_0^{(k)}$.

In figure 3.22a we compare the results for the first eigenvalue λ_{ε} in the first gap obtained using the DtN operator method and the asymptotic expansion of the eigenvalue.



(a) Dependence on ε of the eigenvalues computed with the DtN method and of the first several terms of the asymptotic expansion.



(b) Zoom on the interval [0.45, 0.6]

Figure 3.22: Comparison of the numerical approximations of the first eigenvalue in the first gap of the operator $A^{\mu}_{\varepsilon,s}$ computed with the DtN operator method and with the asymptotic expansion of the eigenvalue for L = 2, $\mu = 0.25$, $\varepsilon \in [0, 0.6]$. The black curve is obtained using the DtN operator method and the coloured curves correspond to the first terms of the asymptotic expansion of the eigenvalue, $\sum_{k=0}^{n} \lambda^{(k)} \varepsilon^{k}$, for $1 \leq k \leq 5$.

CHAPTER 4

GUIDED MODES IN OPEN PERIODIC LINEIC WAVEGUIDES: THE 2D CASE

In this chapter we study the problem creating of guided modes in the domain $\Omega_{\varepsilon}^{\mu}$ represented in figure 4.1b. This domain, supposed homogeneous, can be seen as a perturbation of the 2D periodic domain Ω_{ε} (represented in figure 4.1a) defined by

$$\Omega_{\varepsilon} = \mathbb{R}^2 \setminus \mathcal{S}_{\varepsilon},$$
$$\mathcal{S}_{\varepsilon} = \bigcup_{(j,k) \in \mathbb{Z}^2} \left[\frac{\varepsilon}{2}, 1 - \frac{\varepsilon}{2}\right] \times \left[-\frac{L}{2} + \frac{\varepsilon}{2}, \frac{L}{2} - \frac{\varepsilon}{2}\right] + (j, kL), \qquad \varepsilon > 0, \quad L > 0,$$

 ε being a small parameter. The domain Ω_{ε} is \mathbb{R}^2 minus an infinite set of rectangular



Figure 4.1: The purely periodic domain Ω_{ε} and the perturbed domain $\Omega_{\varepsilon}^{\mu}$.

obstacles (of size $(1-\varepsilon) \times (L-\varepsilon)$) placed periodically with the period 1 in the x-direction

and the period L in the y-direction. The distance between two consecutive obstacles is ε . The domain $\Omega_{\varepsilon}^{\mu}$ is obtained from Ω_{ε} by changing the distance between two neighbour columns of obstacles from ε to $\mu\varepsilon$ with $\mu > 0$ and simultaneously modifying the width of the obstacles of these two columns from $1 - \varepsilon$ to $1 - (1 + \mu)\varepsilon/2$ (the total width of the two columns is then preserved):

$$\Omega^{\mu}_{\varepsilon} = \mathbb{R}^2 \setminus \left\{ \mathcal{S}^{\mu,+}_{\varepsilon} \cup \mathcal{S}^{\mu,-}_{\varepsilon} \right\},\,$$

$$\begin{split} \mathcal{S}^{\mu,+}_{\varepsilon} &= \bigcup_{\substack{j \in \mathbb{N} \\ k \in \mathbb{Z}}} \left[\frac{w^{\mu}_{j}\varepsilon}{2}, 1 - \frac{\varepsilon}{2} \right] \times \left[-\frac{L}{2} + \frac{\varepsilon}{2}, \frac{L}{2} - \frac{\varepsilon}{2} \right] + (j, kL), \\ \mathcal{S}^{\mu,-}_{\varepsilon} &= \bigcup_{\substack{j \in \mathbb{N} \\ k \in \mathbb{Z}}} \left[-1 + \frac{\varepsilon}{2}, -\frac{w^{\mu}_{j}\varepsilon}{2} \right] \times \left[-\frac{L}{2} + \frac{\varepsilon}{2}, \frac{L}{2} - \frac{\varepsilon}{2} \right] - (j, kL), \end{split}$$

where w_j^{μ} is the weight function defined in (1.3.1). Neumann boundary conditions are imposed on the boundaries of the obstacles.

Our goal is to prove the existence of guided modes. By a guided mode we mean, roughly speaking, a solution of the scalar wave equation,

$$\frac{\partial^2 u}{\partial t^2} = \Delta u \quad \text{in} \quad \Omega^{\mu}_{\varepsilon}, \qquad \left. \frac{\partial u}{\partial n} \right|_{\partial \Omega^{\mu}_{\varepsilon}} = 0, \tag{4.0.1}$$

which propagates along the perturbation (in the y-direction) and is confined in the transversal direction (x-direction). Thus, it should have the form of a plane wave propagating in the y-direction multiplied by some function v(x, y) periodic in the y-direction with period L:

$$u(x, y, t) = e^{i(\omega t - \beta y/L)}v(x, y), \qquad v(x, y) = v(x, y + L).$$
(4.0.2)

To express the fact that the solution is confined in x-direction we impose the condition

$$v \in L_2\left(\mathcal{B}^{\mu}_{\varepsilon}\right),$$

where $\mathcal{B}^{\mu}_{\varepsilon}$ is a periodicity band of the domain $\Omega^{\mu}_{\varepsilon}$ (cf. figure 4.2) which can be formally described by the expression

$$\mathcal{B}^{\mu}_{\varepsilon} = \left\{ \mathbb{R} \times \left] - \frac{\varepsilon}{2}, \frac{\varepsilon}{2} \right[\right\} \cup \left\{ \bigcup_{j \in \mathbb{N}} \left] j - \frac{w^{\mu}_{j}\varepsilon}{2}, j + \frac{w^{\mu}_{j}\varepsilon}{2} \right[\times \left] - \frac{L}{2}, \frac{L}{2} \right[\right\}.$$

Remark 4.0.1. Notice that the choice of the periodicity band is obviously not unique. In particular, we could have taken a periodicity band that has a ladder shape. However, the periodicity cell of the band $\mathcal{B}^{\mu}_{\varepsilon}$ which has the form of a cross turns out to be more convenient for numerical simulations.

One can separate in (4.0.2) the harmonic factor in time to get

$$u(x, y, t) = e^{i\omega t} \widetilde{v}(x, y), \qquad \widetilde{v}(x, y) = e^{-i\beta y/L} v(x, y), \qquad (4.0.3)$$



Figure 4.2: Periodicity band (grey area)

where the function \tilde{v} is β -quasiperiodic:

$$\widetilde{v}(x, y+L) = e^{-i\beta}\widetilde{v}(x, y).$$

Then, after plugging (4.0.3) into the equation (4.0.1), one gets the following problem for the function $\tilde{v} \in L_2(\mathcal{B}_{\varepsilon}^{\mu})$:

$$\begin{cases} \Delta \widetilde{v} = -\omega^2 \widetilde{v}, \\ \widetilde{v}|_{\Sigma'} = e^{-i\beta} \left. \widetilde{v} \right|_{\Sigma}, \\ \frac{\partial \widetilde{v}}{\partial n} \right|_{\partial \mathcal{B}_{\varepsilon}^{\mu} \setminus \{\Sigma \cup \Sigma'\}} = 0. \end{cases}$$

Here

$$\Sigma = \bigcup_{j \in \mathbb{N}} \left[j - \frac{w_j^{\mu}\varepsilon}{2}, j + \frac{w_j^{\mu}\varepsilon}{2} \right] \times \left\{ -\frac{L}{2} \right\}, \qquad \Sigma' = \bigcup_{j \in \mathbb{N}} \left[j - \frac{w_j^{\mu}\varepsilon}{2}, j + \frac{w_j^{\mu}\varepsilon}{2} \right] \times \left\{ \frac{L}{2} \right\}.$$

Thus, one ends up with an eigenvalue problem for the Laplacian in the periodicity band $\mathcal{B}^{\mu}_{\varepsilon}$ with β -quasiperiodic boundary conditions on the upper and lower parts of the boundary for $\beta \in [0, \pi]$. More precisely, one has to study the spectrum of the following operator:

$$\begin{split} A^{\mu}_{\varepsilon}(\beta) &: L_{2}\left(\mathcal{B}^{\mu}_{\varepsilon}\right) \to L_{2}\left(\mathcal{B}^{\mu}_{\varepsilon}\right), \qquad A^{\mu}_{\varepsilon}(\beta)v = -\Delta v, \\ D(A^{\mu}_{\varepsilon}(\beta)) &= \left\{ v \in H^{1}_{\Delta}(\mathcal{B}^{\mu}_{\varepsilon}), \quad \frac{\partial v}{\partial n} \Big|_{\partial \mathcal{B}^{\mu}_{\varepsilon} \setminus \{\Sigma \cup \Sigma'\}} = 0, \\ v|_{\Sigma'} &= e^{-i\beta}v|_{\Sigma}, \quad \frac{\partial v}{\partial y} \Big|_{\Sigma'} = e^{-i\beta} \left. \frac{\partial v}{\partial y} \right|_{\Sigma} \right\}. \end{split}$$

We will show (cf. Theorems 4.1.2, 4.1.1 and 4.1.3) that for any $\beta \in [0, \pi]$, for any $k \in \mathbb{N}$ there exists $\varepsilon_k > 0$ such that for $\varepsilon < \varepsilon_k$ the operator $A^{\mu}_{\varepsilon}(\beta)$ has at least k gaps and at least one or two eigenvalues in each of these gaps.

As for the ladder, the study is based on asymptotic analysis: as $\varepsilon \to 0$, the domain $\mathcal{B}^{\mu}_{\varepsilon}$ shrinks to a graph and the spectrum of the limit operator defined on this graph can be computed explicitly. The essential spectrum of the limit operator is discussed in Section 4.1.1 and the discrete spectrum in Section 4.1.2. This will permit to derive existence results for the operator $A^{\mu}_{\varepsilon}(\beta)$ in Section 4.1.3. In Section 4.2 we present numerical results for the operator A^{μ}_{ε} .

4.1 Existence results

When $\varepsilon \to 0$ the domain $\mathcal{B}^{\mu}_{\varepsilon}$ shrinks to the graph $G_{\mathcal{B}} = \bigcap_{\varepsilon > 0} \mathcal{B}^{\mu}_{\varepsilon}$ shown in figure 4.3.



Figure 4.3: The limit graph $G_{\mathcal{B}}$

We will use the following notation. We enumerate the vertices by the index $j \in \mathbb{Z}$, j = 0 corresponding to the perturbed vertex. The value of the function u at the j-th middle vertex is denoted by u_j , its value at the upper vertex by u_j^+ and at the lower vertex by u_j^- . The horizontal edge joining the vertices j and j + 1 is denoted by $e_{j+\frac{1}{2}}$ and the function on it is denoted by $u_{j+\frac{1}{2}}(s)$, the local variable s taking values in [0, 1]. The vertical edges above and below the vertex j are denoted by e_j^+ and e_j^- respectively and the function on this edges is denoted by $u_j^+(t)$ and $u_j^-(t)$ respectively, the local variable t taking values in $\left[-\frac{L}{2}, \frac{L}{2}\right]$ (t = 0 at the middle vertex). We denote by $\mathcal{E}_{G_{\mathcal{B}}}$ the set of the edges of $G_{\mathcal{B}}$.

Similarly to the case of the ladder, the appropriate functional spaces are

$$L_{2}^{\mu}(G_{\mathcal{B}}) = \left\{ u: \|u\|_{L_{2}^{\mu}(G_{\mathcal{B}})}^{2} < \infty \right\}, \qquad H^{2}(G_{\mathcal{B}}) = \left\{ u \in C(G_{\mathcal{B}}): \|u\|_{H^{2}(G_{\mathcal{B}})}^{2} < \infty \right\},$$
$$\|u\|_{L_{2}^{\mu}(G_{\mathcal{B}})}^{2} = \sum_{j \in \mathbb{Z}} \left(w_{j}^{\mu} \|u_{j}^{+}\|_{L_{2}(e_{j}^{+})}^{2} + w_{j}^{\mu} \|u_{j}^{-}\|_{L_{2}(e_{j}^{-})}^{2} + \|u_{j+\frac{1}{2}}\|_{L_{2}(e_{j+\frac{1}{2}})}^{2} \right),$$
$$\|u\|_{H^{2}(G_{\mathcal{B}})}^{2} = \sum_{j \in \mathbb{Z}} \left(\|u_{j}^{+}\|_{H^{2}(e_{j}^{+})}^{2} + \|u_{j}^{-}\|_{H^{2}(e_{j}^{+})}^{2} + \|u_{j+\frac{1}{2}}\|_{H^{2}(e_{j+\frac{1}{2}})}^{2} \right).$$

The limit operator $A^{\mu}(\beta) : L_{2}^{\mu}(G_{\mathcal{B}}) \to L_{2}^{\mu}(G_{\mathcal{B}})$ is now defined as

$$\left(A^{\mu}(\beta)u\right)|_{e} = -\left(u|_{e}\right)'', \qquad \forall e \in \mathcal{E}_{G_{\mathcal{B}}},$$

$$D\left(A^{\mu}(\beta)\right) = \left\{ u \in H^{2}\left(G_{\mathcal{B}}\right):$$

$$(4.1.1)$$

$$\mathbf{u}_{j}^{+} = e^{-i\beta}\mathbf{u}_{j}^{-}, \quad (u_{j}^{+})'\left(\frac{L}{2}\right) = e^{-i\beta}(u_{j}^{-})'\left(-\frac{L}{2}\right), \qquad \forall j \in \mathbb{Z}, \qquad (4.1.2)$$

$$u_{j+\frac{1}{2}}'(0) - u_{j-\frac{1}{2}}'(1) + w_{j}^{\mu}(u_{j}^{+})'(0) - w_{j}^{\mu}(u_{j}^{-})'(0) = 0, \quad \forall j \in \mathbb{Z} \Big\}.$$
(4.1.3)

The relations (4.1.2) express the quasiperiodicity and the relations (4.1.3) are the Kirchhoff's conditions at the middle vertices. Notice that by definition of $H^2(G_{\mathcal{B}})$ the condition (4.1.1) implies the continuity at the middle vertices.

As previously, we start with the explicit computation of the essential spectrum of the operator $A(\beta)$ which turns out to have infinitely many gaps for any β . The computations in this section will be very similar to the ones done for the ladder in Section 1.3. We will still repeat it in detail for the seek of completeness. Then we prove that the perturbed operator has one or two eigenvalues in each gap.

4.1.1 The essential spectrum of the limit operator

As usual, due to a compact perturbation argument the study of the essential spectrum of the operator $A^{\mu}(\beta)$ is reduced to the study of the spectrum of the non-perturbed operator $A^{1}(\beta)$ (corresponding to the case $\mu = 1$) that will be denoted by $A(\beta)$.

Proposition 4.1.1. For any $\mu > 0$, $\beta \in [0, \pi]$

$$\sigma_{ess}\left(A^{\mu}(\beta)\right) = \sigma_{ess}\left(A(\beta)\right)$$

For this reason, we start by determining the (essential) spectrum of the operator $A(\beta)$.

4.1.1.1 Computation of the spectrum of the non-perturbed limit operator

According to the Floquet-Bloch theory, in order to determine the spectrum of the operator $A(\beta)$ one has to study a set of problems on the periodicity cell $C_{\mathcal{B}}$ that consists of four edges: $I_1 = \left[-\frac{1}{2}, 0\right] \times 0$, $I_2 = \left[0, \frac{1}{2}\right] \times 0$, $I_3 = 0 \times \left[-\frac{L}{2}, 0\right]$, $I_3 = 0 \times \left[0, \frac{L}{2}\right]$. A function on the cell is defined by its four restrictions on the edges of the cell: $u = \left\{u_i\right\}_{i=1}^4$ (cf. figure 4.4).



Figure 4.4: Periodicity cell $C_{\mathcal{B}}$

The corresponding functional spaces on the cell are defined as

$$L_2(\mathcal{C}_{\mathcal{B}}) = \left\{ u : \sum_{i=1}^4 \|u_i\|_{L_2(I_i)}^2 < \infty \right\}, \quad H^2(C_{\mathcal{B}}) = \left\{ u \in C(C_{\mathcal{B}}) : \sum_{i=1}^4 \|u_i\|_{H^2(I_i)}^2 < \infty \right\}.$$

We have to determine the spectrum of the operator family $A_{\beta}(\theta) : L_2(\mathcal{C}_{\mathcal{B}}) \to L_2(\mathcal{C}_{\mathcal{B}}), \theta \in [0, \pi]$, defined as

$$A_{\beta}(\theta)u = -u'',$$

$$D(A_{\beta}(\theta)) = \left\{ u \in H^{2}(\mathcal{C}_{\mathcal{B}}) : \quad u_{4}\left(\frac{L}{2}\right) = e^{-i\beta}u_{3}\left(-\frac{L}{2}\right), \quad u_{4}'\left(\frac{L}{2}\right) = e^{-i\beta}u_{3}'\left(-\frac{L}{2}\right), \quad (4.1.4)$$

$$u_{2}\left(\frac{1}{2}\right) = e^{-i\theta}u_{1}\left(-\frac{1}{2}\right), \quad u_{2}'\left(\frac{1}{2}\right) = e^{-i\theta}u_{1}'\left(-\frac{1}{2}\right), \quad (4.1.5)$$

$$u_2'(0) - u_1'(0) + u_4'(0) - u_3'(0) = 0$$
 (4.1.6)

Proposition 4.1.2. For $\theta, \beta \in [0, \pi]$, $\lambda^2 \in \sigma(A_\beta(\theta)) \setminus \{0\}$ if and only if $\lambda > 0$ is a solution of the equation

$$\sin\lambda\left(\cos\beta - \cos\left(\lambda L\right)\right) = \sin\left(\lambda L\right)\left(\cos\lambda - \cos\theta\right). \tag{4.1.7}$$

The point $\lambda^2 = 0$ belongs to $\sigma(A_\beta(\theta))$ if and only if $\theta = \beta = 0$.

Proof. Searching the eigenvalues of the operator $A_{\beta}(\theta)$ implies solving the ordinary differential equation

$$u'' + \lambda^2 u = 0 \tag{4.1.8}$$

on each edge of the cell $\mathcal{C}_{\mathcal{B}}$. We first consider the case $\lambda \neq 0$. We have:

$$u_1(x) = c_1 e^{i\lambda x} + d_1 e^{-i\lambda x}, \qquad x \in \left[-\frac{1}{2}, 0\right],$$
(4.1.9)

$$u_2(x) = c_2 e^{i\lambda x} + d_2 e^{-i\lambda x}, \qquad x \in \left[0, \frac{1}{2}\right], \tag{4.1.10}$$

$$u_{3}(y) = c_{3}e^{i\lambda y} + d_{3}e^{-i\lambda y}, \qquad y \in \left[-\frac{L}{2}, 0\right],$$
(4.1.11)

$$u_4(y) = c_4 e^{i\lambda y} + d_4 e^{-i\lambda y}, \qquad y \in \left[0, \frac{1}{2}\right].$$
 (4.1.12)

The continuity of the eigenfunction at the vertex x = 0, y = 0 implies the relations

$$c_1 + d_1 = c_2 + d_2 = c_3 + d_3 = c_4 + d_4.$$
(4.1.13)

The quasi-periodicity conditions (4.1.4), (4.1.5) lead to the following relations:

$$c_{2}e^{i\lambda/2} + d_{2}e^{-i\lambda/2} = e^{-i\theta} \left(c_{1}e^{-i\lambda/2} + d_{1}e^{i\lambda/2} \right), \qquad (4.1.14)$$

$$c_2 e^{i\lambda/2} - d_2 e^{-i\lambda/2} = e^{-i\theta} \left(c_1 e^{-i\lambda/2} - d_1 e^{i\lambda/2} \right), \tag{4.1.15}$$

$$c_2 e^{i\lambda/2} - d_2 e^{-i\lambda/2} = e^{-i\theta} \left(c_1 e^{-i\lambda/2} - d_1 e^{i\lambda/2} \right), \tag{4.1.16}$$

$$c_4 e^{i\lambda L/2} + d_4 e^{-i\lambda L/2} = e^{-i\beta} \left(c_3 e^{-i\lambda L/2} + d_3 e^{i\lambda L/2} \right), \qquad (4.1.16)$$

$$c_4 e^{i\lambda L/2} - d_4 e^{-i\lambda L/2} = e^{-i\beta} \left(c_3 e^{-i\lambda L/2} - d_3 e^{i\lambda L/2} \right).$$
(4.1.17)

Finally, we take into account the Kirchhoff's condition (4.1.6) which can be rewritten as

$$c_2 - d_2 - c_1 + d_1 + c_4 - d_4 - c_3 + d_3 = 0.$$
(4.1.18)

Let us denote

$$\alpha = e^{i\theta}, \qquad \gamma = e^{i\beta}, \qquad z = e^{i\lambda}$$

Then, the relations (4.1.14), (4.1.15) imply that

$$c_1 = \alpha z c_2, \qquad d_1 = \alpha \overline{z} d_2. \tag{4.1.19}$$

Analogously, the relations (4.1.16), (4.1.21) imply

$$c_3 = \gamma z^L c_4, \qquad d_3 = \gamma \overline{z}^L d_4. \tag{4.1.20}$$

Taking into account (4.1.13), we get:

$$c_2\left(\alpha z - 1\right) = d_2\left(1 - \alpha \overline{z}\right),\tag{4.1.21}$$

$$c_4\left(\gamma z^L - 1\right) = d_4\left(1 - \gamma \overline{z}^L\right). \tag{4.1.22}$$

Combining the relations (4.1.13), (4.1.18), (4.1.21), (4.1.22) we arrive at the following linear system of four equations for the unknowns c_2 , d_2 , c_4 , d_4 :

$$\begin{cases} c_2 + d_2 - c_4 - d_4 = 0, \\ c_2(\alpha z - 1) + d_2(\alpha \overline{z} - 1) = 0, \\ c_4(\gamma z^L - 1) + d_4(\gamma \overline{z}^L - 1) = 0, \\ c_2(1 - \alpha z) + d_2(\alpha \overline{z} - 1) + c_4(1 - \gamma z^L) + d_4(\gamma \overline{z}^L - 1) = 0 \end{cases}$$

The criterion of existence of a non-trivial solution is:

$$D(\lambda) = 0, \qquad D(\lambda) = \begin{vmatrix} 1 & 1 & -1 & -1 \\ \alpha z - 1 & \alpha \overline{z} - 1 & 0 & 0 \\ 0 & 0 & \gamma z^{L} - 1 & \gamma \overline{z}^{L} - 1 \\ 1 - \alpha z & \alpha \overline{z} - 1 & 1 - \gamma z^{L} & \gamma \overline{z}^{L} - 1 \end{vmatrix}$$

Let us compute the determinant $D(\lambda)$. We have

$$D(\lambda) = -(\alpha z - 1) \begin{vmatrix} 1 & -1 & -1 \\ 0 & \gamma z^L - 1 & \gamma \overline{z}^L - 1 \\ \alpha \overline{z} - 1 & 1 - \gamma z^L & \gamma \overline{z}^L - 1 \end{vmatrix} + (\alpha \overline{z} - 1) \begin{vmatrix} 1 & -1 & -1 \\ 0 & \gamma z^L - 1 & \gamma \overline{z}^L - 1 \\ 1 - \alpha z & 1 - \gamma z^L & \gamma \overline{z}^L - 1 \end{vmatrix}$$
$$= (1 - \alpha z) \left(2 + 2\gamma^2 + \alpha \gamma \overline{z} \left(z^L - \overline{z}^L \right) - \gamma \left(3z^L + \overline{z}^L \right) \right)$$
$$+ (\alpha \overline{z} - 1) \left(2 + 2\gamma^2 + \alpha \gamma z \left(\overline{z}^L - z^L \right) - \gamma \left(3\overline{z}^L + z^L \right) \right),$$

$$\frac{D(\lambda)}{\gamma} = (1 - \alpha z) \left(4\Re\gamma + 2i\alpha \overline{z}\Im z^L - (3z^L + \overline{z}^L) \right) + (\alpha \overline{z} - 1) \left(4\Re\gamma - 2i\alpha z\Im z^L - (3\overline{z}^L + z^L) \right) \\ = 8i\alpha \Re\gamma\Im\overline{z} + 4i\Im\overline{z}^L + 8i\alpha\Im z^{L+1} + 4i\alpha^2\Im\overline{z}^L,$$

$$\frac{D(\lambda)}{4i\alpha\gamma} = 2\Re\gamma\Im\overline{z} + \overline{\alpha}\Im\overline{z}^L + 2\Im z^{L+1} + \alpha\Im\overline{z}^L = 2\left(\Re\gamma\Im\overline{z} + \Re\alpha\Im\overline{z}^L + \Im z^{L+1}\right).$$

Consequently, the condition $D(\lambda) = 0$ is equivalent to the relation

$$\Re\gamma\Im\overline{z} + \Re\alpha\Im\overline{z}^L + \Im z^{L+1} = 0,$$

which leads to the relation (4.1.7) for $\lambda \neq 0$. If $\lambda = 0$, instead of the relations (4.1.9)–(4.1.12) we have

$$u_1(x) = c_1 + d_1 x, \qquad x \in \left[-\frac{1}{2}, 0\right], \\ u_2(x) = c_2 + d_2 x, \qquad x \in \left[0, \frac{1}{2}\right], \\ u_3(y) = c_3 + d_3 y, \qquad y \in \left[-\frac{L}{2}, 0\right], \\ u_4(y) = c_4 + d_4 y, \qquad y \in \left[0, \frac{1}{2}\right].$$

The continuity at the vertex x = 0, y = 0 implies that $c_i = c$, $1 \leq i \leq 4$. From the quasiperiodic conditions it follows that

$$d_2 = e^{-i\theta} d_1, \quad d_4 = e^{-i\beta} d_3, \quad d_1 = c \left(1 - e^{i\theta}\right), \quad d_3 L = c \left(1 - e^{i\beta}\right). \tag{4.1.23}$$

Finally, Kirchhoff's condition gives

$$-d_1 + d_2 - d_3 + d_4 = 0. (4.1.24)$$

After injecting (4.1.23) in (4.1.24) we get

$$c\left(\cos\theta - 1 + \frac{\cos\beta - 1}{L}\right) = 0.$$

Hence, there exists a non-trivial solution if and only if $\theta = \beta = 0$. This finishes the proof.

Remark 4.1.1. One can notice that if $L \in \mathbb{Q}$, then the set $\{\lambda : \lambda^2 \in \sigma(A_\beta(\theta))\}$ is periodic. Indeed, this is due to the fact that both the left-hand side and the right-hand side of (4.1.7) are periodic with the same period.

We can now find the spectrum of the operator $A(\beta)$ due to the decomposition

$$\sigma(A(\beta)) = \bigcup_{\theta \in [0,\pi]} \sigma(A_{\beta}(\theta)).$$
(4.1.25)

Thus, the point λ^2 (different from zero) belongs to the spectrum of the operator $A(\beta)$ if and only if there exists a value of θ such that the relation (4.1.7) is verified.

In the rest of this section we will use the following notation:

 $\Sigma = \{\pi n, n \in \mathbb{N}\}, \qquad \Sigma^* = \Sigma \setminus \{0\}, \qquad \Sigma_\beta = \left\{\pm \frac{\beta}{L} + \frac{2\pi n}{L}, n \in \mathbb{N}\right\}, \qquad \Sigma^*_\beta = \Sigma_\beta \setminus \{0\}.$

Proposition 4.1.3.

- 1. $\{\lambda^2, \lambda \in \Sigma^* \cup \Sigma_\beta\} \subset \sigma(A(\beta)).$
- 2. For any $\beta \in [0, \pi]$, the operator $A(\beta)$ has infinitely many gaps whose ends tend to infinity.

Proof.

- 1. For $\lambda \neq 0$, $\lambda \in \Sigma^* \cup \Sigma_{\beta}$ the equation (4.1.7) is obviously verified for $\cos \theta = \cos \lambda$. The point $\lambda = 0$ belongs to $\Sigma^* \cup \Sigma_{\beta}$ if and only if $\beta = 0$. At the same time, according to Proposition 4.1.2, $0 \in \sigma(A(\beta))$ if and only if $\beta = 0$.
- 2. Let us consider two cases:

a) $\beta \notin \{0, \pi\}$:

Consider $\lambda_n = \frac{\pi n}{L}$, $n \in \mathbb{N}^*$. There are two possibilities:

(i) $\lambda_n \notin \Sigma^*$: then, the right-hand side of the equation (4.1.7) equals to zero, whereas the left-hand side is different from zero for any $\theta \in [0, \pi]$. Consequently, there exists a gap of the operator $A(\beta)$ containing the point λ_n^2 .

(ii) $\lambda_n \in \Sigma^*$: in this case it follows from the property 1 that $\lambda_n^2 \in \sigma(A(\beta))$. We are going to show that the point λ_n^2 is an isolated point of the spectrum of the operator $A(\beta)$, so that there exist gaps to the left and to the right of it. Setting $\lambda = \lambda_n + \delta$ in (4.1.7) we get:

$$\sin \delta \frac{\cos \beta}{\cos (\lambda_n L)} + \sin (\delta L) \frac{\cos \theta}{\cos \lambda_n} = \sin (\delta (L+1)). \tag{4.1.26}$$

If δ is small enough (but different from 0) this equation cannot be verified for any θ . Indeed, for $\delta \neq 0$ it can be rewritten as

$$\frac{\cos\beta}{\cos\left(\lambda_n L\right)} - \cos\left(\delta L\right) = \frac{\sin\left(\delta L\right)}{\sin\delta} \left(\cos\delta - \frac{\cos\theta}{\cos\lambda_n}\right)$$

Since $|\cos \beta| < 1$ and $|\cos \lambda_n| = |\cos (\lambda_n L)| = 1$, the limit of the left-hand side as $\delta \to 0$ is negative, whereas the limit of the right-hand side is non-negative for any θ with a uniform bound in θ for δ small enough:

$$\frac{\sin\left(\delta L\right)}{\sin\delta}\left(\cos\delta - \frac{\cos\theta}{\cos\lambda_n}\right) \geqslant \frac{\sin\left(\delta L\right)}{\sin\delta}\left(\cos\delta - 1\right), \qquad \forall \theta \in [0,\pi]$$

Hence, the equation (4.1.26) does not have solutions for δ small enough. This proves the existence of gaps of the form $\lambda_n^2 - l_n^-, \lambda_n^2[, \lambda_n^2, \lambda_n^2 + l_n^+]$ for some $l_n^-, l_n^+ > 0$.

Thus, we see that for any $n \in \mathbb{N}^*$ there exists a gap of the operator $A(\beta)$ of the form $]\lambda_n^2 - l_n^-, \lambda_n^2 + l_n^+[,]\lambda_n^2 - l_n^-, \lambda_n^2[\text{ or }]\lambda_n^2, \lambda_n^2 + l_n^+[\text{ with } l_n^-, l_n^+ > 0.$ We also know from the property 1 that there exists an infinite sequence of points tending to infinity and belonging to the spectrum of the operator $A(\beta)$. This proves the existence of an infinity of gaps of the operator $A(\beta)$ for $\beta \notin \{0, \pi\}$.

b) $\beta \in \{0, \pi\}$:

For $\beta = 0$ the equation (4.1.7) takes the form

$$\sin\left(\frac{\lambda L}{2}\right)\cos\left(\frac{\lambda L}{2}\right)(\cos\lambda - \cos\theta) = \sin\lambda\sin^2\left(\frac{\lambda L}{2}\right),\tag{4.1.27}$$

and for $\beta = \pi$ it takes the form

$$\sin\left(\frac{\lambda L}{2}\right)\cos\left(\frac{\lambda L}{2}\right)\left(\cos\lambda - \cos\theta\right) = -\sin\lambda\cos^2\left(\frac{\lambda L}{2}\right). \tag{4.1.28}$$

It can be shown that the operators A(0) and $A(\pi)$ have gaps that contain some neighbourhoods (or deleted neighbourhoods) of the points $\{\lambda^2, \cos\left(\frac{\lambda L}{2}\right) = 0\}$ and $\{\lambda^2, \sin\left(\frac{\lambda L}{2}\right) = 0\}$ respectively. Indeed, in the neighbourhoods of these points the relations (4.1.27) and (4.1.28) can be rewritten as

$$\cos\left(\frac{\lambda L}{2}\right)\left(\cos\lambda - \cos\theta\right) = \sin\lambda\sin\left(\frac{\lambda L}{2}\right), \qquad (\beta = 0), \qquad (4.1.29)$$

$$\sin\left(\frac{\lambda L}{2}\right)(\cos\lambda - \cos\theta) = -\sin\lambda\cos\left(\frac{\lambda L}{2}\right), \qquad (\beta = \pi). \tag{4.1.30}$$

Comparing these relations with the relations (1.3.12), (1.3.59) characterizing the spectra of the operators A_s and A_{as} respectively, we see that they coincide up to a factor 2. This factor being unimportant in the proofs of Propositions 1.3.4 (2), 1.3.12 (2), the same argument (together with the property 1) shows the existence of an infinity of gaps for the operators A(0), $A(\pi)$.

We will come back to this similarity of the relations (4.1.29), (1.3.12) and (4.1.30), (1.3.59) which implies some relation between the symmetric case and the 0-quasiperiodic one, as well as between the antisymmetric case and the π -quasiperiodic one (see Remark 4.1.6).

Proposition 4.1.4. The operator $A(\beta)$ has the following set of eigenvalues of infinite multiplicity:

$$\sigma_{pp}(A(\beta)) = \Sigma_{is}(\beta) \cup \Sigma_{emb}(\beta)$$

where

$$\Sigma_{is}(\beta) = \begin{cases} \{\lambda^2 / \lambda \in \Sigma^*, & \sin(\lambda L) = 0\}, & \beta \in]0, \pi[, \\ \{\lambda^2 / \lambda \in \Sigma^*, & \cos(\lambda L) = -1\}, & \beta = 0, \\ \{\lambda^2 / \lambda \in \Sigma^*, & \cos(\lambda L) = 1\}, & \beta = \pi, \end{cases}$$
(4.1.31)

and

$$\Sigma_{emb}(\beta) = \begin{cases} \emptyset, & \beta \in]0, \pi[, \\ \{\lambda^2 / \lambda \neq 0, \cos(\lambda L) = 1\}, & \beta = 0, \\ \{\lambda^2 / \lambda \neq 0, \cos(\lambda L) = -1\}, & \beta = \pi. \end{cases}$$
(4.1.32)

The eigenvalues of the set $\Sigma_{is}(\beta)$ are isolated points of the spectrum whereas the eigenvalues of the set $\Sigma_{emb}(\beta)$ are embedded (interior points of the spectrum).

Proof. The point λ^2 is an eigenvalue of the operator $A(\beta)$ of infinity multiplicity if and only if it is an eigenvalue of the operator $A(\beta(\theta))$ for any $\theta \in [0, \pi]$. It follows from Proposition 4.1.2 that 0 is not an eigenvalue of infinite multiplicity for any $\beta \in [0, \pi]$. For $\lambda \neq 0$ it means that the equation (4.1.7) is satisfied for any $\theta \in [0, \pi]$. This is possible in two cases:

- (a) $\sin \lambda = \sin (\lambda L) = 0$ and $\cos (\lambda L) \neq \cos \beta$;
- (b) $\cos(\lambda L) = \cos\beta$ and $\sin(\lambda L) = 0$.

In both cases (and only in this cases) the left-hand side and the right-hand side of (4.1.7) equal to zero simultaneously and independently of $\theta \in [0, \pi]$. The case (a) leads to the set $\Sigma_{is}(\beta)$. We have already shown in the proof of Proposition 4.1.3 (2) that these points are isolated points of the spectrum. The case (b) leads to the set $\Sigma_{emb}(\beta)$. Let us show that the points of this set are interior points of the spectrum. Consider $\beta \in \{0, \pi\}$ and λ_0 such that $\cos(\lambda_0 L) = \cos\beta = \pm 1$. Then, the equation (4.1.7) for $\lambda = \lambda_0 + \delta$ takes the form

$$\cos \theta = \cos \lambda - \sin \lambda \tan \left(\frac{\delta L}{2}\right).$$

It is clear that if $|\cos \lambda_0| < 1$ then this equation admits a solution when δ is small enough (since the absolute value of the right-hand side is smaller than 1 for δ small enough). If $|\cos \lambda_0| = 1$, then the last equation can be rewritten as

$$\cos\theta = \cos\lambda_0 \left(\cos\delta - \sin\delta\tan\left(\frac{\delta L}{2}\right)\right),\,$$

and again it admits a solution when δ is small enough. This shows that the points of the set $\Sigma_{emb}(\beta)$ are interior points of the spectrum.

The following assertion is an immediate consequence of the decomposition (4.1.25), Proposition 4.1.2 and property 1 of Proposition 4.1.3.

Proposition 4.1.5. $\lambda^2 \in \sigma(A(\beta))$ if and only if one of the following possibilities holds:

- (i) $\lambda \in \Sigma^* \cup \Sigma_\beta$;
- (ii) $\lambda \in \mathbb{R}_+ \setminus (\Sigma \cup \Sigma_\beta)$ is a solution of the equation

$$\phi_{L,\beta}(\lambda) = f_{\theta}(\lambda), \qquad (4.1.33)$$

for some $\theta \in [0, \pi]$. Here

$$\phi_{L,\beta}(\lambda) = \frac{\sin(\lambda L)}{\cos\beta - \cos(\lambda L)}, \qquad \lambda \notin \Sigma_{\beta},$$
$$f_{\theta}(\lambda) = \frac{\sin\lambda}{\cos\lambda - \cos\theta}, \qquad \lambda \in \{\lambda / \cos\lambda \neq \cos\theta\}.$$

As in the case of the ladder (cf. Proposition 1.3.6) a simple geometric interpretation can be given to Proposition 4.1.5. Again, the spectrum is obtained as the image by the function $x \mapsto x^2$ of the closure of the projection on the axis of positive abscissas of the intersection of the the domain \overline{D} with graph of the function $\phi_{L,\beta}$:

$$\lambda^2 \in \sigma(A(\beta)) \quad \Leftrightarrow \quad \lambda \in \overline{D_x \setminus \{0\}},$$

where

$$D_x = \left\{ x : (x, \phi_L(\lambda)) \in \overline{D} \right\}, \qquad D = \bigcup_{n \in \mathbb{N}} D_n^{\pm}$$

$$D_n^+ = \left\{ (x, y) / \ x \in]\pi n, \pi(n+1)[, \ y \in [f^+(x), +\infty[\right\} \cup (\pi n, 0), \\ D_n^- = \left\{ (x, y) / \ x \in]\pi n, \pi(n+1)[, \ y \in]-\infty, f^-(x)] \right\},$$

$$f^{+}(x) = \tan\left(\frac{x-\pi n}{2}\right), \qquad x \in [\pi n, \pi(n+1)[, f^{-}(x) = -\cot\left(\frac{x-\pi n}{2}\right), \qquad x \in]\pi n, \pi(n+1)].$$

An example illustrating this geometric interpretation is given in figure 4.5. Similarly to Proposition 1.3.8, two types of gaps can be distinguished.

Proposition 4.1.6. An interval]a, b[is a gap of the operator $A(\beta)$ for $\beta \in [0, \pi]$ if and only if $[a, b] \cap \Sigma_{\beta} = \emptyset$ and one of the following possibilities holds:

- I There exists $n \in \mathbb{N}$ such that $\pi n < \sqrt{a} < \sqrt{b} < \pi(n+1)$, and $\phi_{L,\beta}(\sqrt{a}) = f^+(\sqrt{a})$, $\phi_{L,\beta}(\sqrt{b}) = f^-(\sqrt{b})$;
- II (i) There exists $n \in \mathbb{N}$ such that $\pi n = \sqrt{a} < \sqrt{b} < \pi(n+1)$, and $\phi_{L,\beta}(\sqrt{a}) \leq 0$, $\phi_{L,\beta}(\sqrt{b}) = f^{-}(\sqrt{b});$
 - (ii) There exists $n \in \mathbb{N}$ such that $\pi n < \sqrt{a} < \sqrt{b} = \pi(n+1)$, and $\phi_{L,\beta}(\sqrt{a}) = f^+(\sqrt{a})$, $\phi_{L,\beta}(\sqrt{b}) \ge 0$.



Figure 4.5: The images of the spectral bands by the function $x \mapsto \sqrt{x}$ are given by the closure of the projection on the axis of positive abscissas of the intersections of the function $\phi_{L,\beta}$ with the domains \overline{D}_n^+ , \overline{D}_n^- .

Remark 4.1.2. If $\beta \in]0, \pi]$, then $\sigma_* = \min \sigma(A(\beta)) > 0$. According to Proposition 4.1.6, the interval $]0, \sigma_*[$ is considered as a gap of type II (it satisfies the condition II(i)). Strictly speaking, this is false: σ_* is the bottom of the essential spectrum. Being aware of this inaccuracy, we will still call the interval $]0, \sigma_*[$ a gap. This will permit us to study the number of eigenvalues of the perturbed operator inside this interval without considering it as a particular case (cf. Theorem 4.1.1).

The proof of Proposition 4.1.6 is based on the following lemma (which is an analogue of Lemma 1.3.2 for the ladder).

Lemma 4.1.1. The point λ_0^2 is the lower end of a gap of the operator $A(\beta)$ if and only if one of the following possibilities holds:

- (i) $\lambda_0 \in \mathbb{R}_+ \setminus (\Sigma \cup \Sigma_s)$ and $\phi_{L,\beta}(\lambda_0) = f^+(\lambda_0);$
- (*ii*) $\lambda_0 \in \Sigma \setminus \Sigma_\beta$ and $\phi_{L,\beta}(\lambda_0) \leq 0$.

Similarly, the point λ_0^2 is the upper end of a gap of the operator $A(\beta)$ if and only if one of the following possibilities holds:

- (iii) $\lambda_0 \in \mathbb{R}_+ \setminus (\Sigma \cup \Sigma_s)$ and $\phi_{L,\beta}(\lambda_0) = f^-(\lambda_0)$;
- (iv) $\lambda_0 \in \Sigma^* \setminus \Sigma_\beta$ and $\phi_{L,\beta}(\lambda_0) \ge 0$.

Without giving a detailed proof (which is an obvious modification of the proof of Lemma 1.3.2) we mention that it is based on the properties of the function $\phi_{L,\beta}$ which has in general the same behaviour as the function ϕ_L . More precisely, it is a $\frac{2\pi}{L}$ -periodic function defined on $\mathbb{R} \setminus \Sigma_{\beta}$. In each interval $\left] -\frac{\beta}{L} + \frac{2\pi n}{L}, \frac{\beta}{L} + \frac{2\pi n}{L}, -\frac{\beta}{L} + \frac{2\pi (n+1)}{L}\right], n \in \mathbb{Z}$, it is continuous and strictly decreasing and takes all the values in \mathbb{R} (cf. figure 4.5).

Remark 4.1.3. If $\beta \in]0, \pi]$, then according to Lemma 4.1.1, 0 is the lower end of a gap since it satisfies $(ii) : 0 \in \Sigma \setminus \Sigma_{\beta}$ and $\phi_{L,\beta}(0) = 0$. Similarly, $\sigma_* = \min \sigma(A(\beta))$ is the upper end of a gap since the cases (iii), (iv) characterize all the points λ such that $|\lambda^2 - \delta, \lambda^2| \cap \sigma(A(\beta)) = \{\lambda^2\}$ for some $\delta > 0$ (cf. Remark 4.1.2).



Figure 4.6: Illustration for Proposition 4.1.6: types of gaps.

4.1.2 The discrete spectrum of the limit operator

We now come back to the perturbed operator $A^{\mu}(\beta)$. We are interested in determining its discrete spectrum. If λ^2 is an eigenvalue of the operator $A^{\mu}(\beta)$, then the corresponding eigenfunction $u \in D(A^{\mu}(\beta))$ solves the equation $u'' + \lambda^2 u = 0$ on each edge of the graph $G_{\mathcal{B}}$.

Case $\lambda \neq 0$

In this case the eigenfunction u it has the form

$$u_{i+\frac{1}{5}}(s) = a_{i+\frac{1}{5}}\sin(\lambda s) + b_{i+\frac{1}{5}}\cos(\lambda s), \qquad s \in [0,1], \qquad j \in \mathbb{Z}, \qquad (4.1.34)$$

$$u_j^+(y) = c_j^+ \sin\left(\lambda y\right) + d_j^+ \cos\left(\lambda y\right), \qquad \qquad y \in \left[0, \frac{L}{2}\right], \qquad \qquad j \in \mathbb{Z}, \qquad (4.1.35)$$

$$u_j^-(y) = c_j^- \sin\left(\lambda y\right) + d_j^- \cos\left(\lambda y\right), \qquad \qquad y \in \left[-\frac{L}{2}, 0\right], \qquad j \in \mathbb{Z}.$$
(4.1.36)

The continuity of an eigenfunction at the vertices (j, 0) implies

$$b_{j+\frac{1}{2}} = a_{j-\frac{1}{2}} \sin \lambda + b_{j-\frac{1}{2}} \cos \lambda = d_j^+ = d_j^-, \qquad j \in \mathbb{Z}.$$
(4.1.37)

Thus, we will denote $d_j := d_j^+ = d_j^-$, $j \in \mathbb{Z}$. After plugging (4.1.35), (4.1.36) into the quasiperiodicity conditions (4.1.2), we get

$$\left(c_{j}^{+}+e^{-i\beta}c_{j}^{-}\right)\sin\left(\frac{\lambda L}{2}\right)+d_{j}\cos\left(\frac{\lambda L}{2}\right)\left(1-e^{-i\beta}\right)=0, \qquad j\in\mathbb{Z},$$

$$(4.1.38)$$

$$\left(c_j^+ - e^{-i\beta}c_j^-\right)\cos\left(\frac{\lambda L}{2}\right) - d_j\sin\left(\frac{\lambda L}{2}\right)\left(1 + e^{-i\beta}\right) = 0, \qquad j \in \mathbb{Z}.$$
(4.1.39)

Finally, from the Kirchhoff's conditions (4.1.3) we get

$$w_{j}^{\mu}\left(c_{j}^{+}-c_{j}^{-}\right)+a_{j+\frac{1}{2}}-a_{j-\frac{1}{2}}\cos\lambda+d_{j-1}\sin\lambda=0, \qquad j\in\mathbb{Z}.$$
(4.1.40)

Let us denote

$$\Sigma_L = \left\{ \frac{\pi n}{L}, \ n \in \mathbb{N}^* \right\}.$$

We will show that the set $\Sigma^* \cup \Sigma_L$ can be excluded from the consideration while searching the eigenvalues of the operator $A^{\mu}(\beta)$. Indeed, due to Proposition 4.1.4 we know that the sets $\Sigma^* \cap \Sigma_L$ for $\beta \in]0, \pi[$ and Σ_L for $\beta \in \{0, \pi\}$ correspond to eigenvalues of infinite multiplicity. The following Lemma states that the points of $\Sigma^* \cup \Sigma_L$ which are not covered by these cases do not correspond to eigenvalues of the operator $A^{\mu}(\beta)$.

Lemma 4.1.2. Suppose that one of the following assumptions holds:

- (i) $\beta \in [0, \pi], \lambda \in \Sigma^* \setminus \Sigma_L;$
- (*ii*) $\beta \in]0, \pi[, \lambda \in \Sigma_L \setminus \Sigma^*.$

Then, λ^2 is not an eigenvalue of the operator $A^{\mu}(\beta)$ for any $\mu > 0$.

Proof.

(i) Since $\sin(\lambda L) \neq 0$, one can derive from (4.1.38)–(4.1.39) the following relations:

$$c_j^+ = \alpha^+ d_j, \quad \alpha^+ = \frac{1}{2} \left(\tan\left(\frac{\lambda L}{2}\right) \left(1 + e^{-i\beta}\right) + \frac{e^{-i\beta} - 1}{\tan\left(\frac{\lambda L}{2}\right)} \right), \quad \forall j \in \mathbb{Z},$$
(4.1.41)

$$c_j^- = \alpha^- d_j, \quad \alpha^- = \frac{1}{2} \left(-\tan\left(\frac{\lambda L}{2}\right) \left(1 + e^{i\beta}\right) + \frac{1 - e^{i\beta}}{\tan\left(\frac{\lambda L}{2}\right)} \right), \quad \forall j \in \mathbb{Z}.$$
(4.1.42)

We have then

$$u_j^+(y) = d_j \left(\alpha^+ \sin\left(\lambda y\right) + \cos\left(\lambda y\right) \right), \qquad y \in \left[0, \frac{L}{2}\right], \qquad j \in \mathbb{Z}, \\ u_j^-(y) = d_j \left(\alpha^- \sin\left(\lambda y\right) + \cos\left(\lambda y\right) \right), \qquad y \in \left[-\frac{L}{2}, 0\right], \qquad j \in \mathbb{Z}.$$

On the other hand, it follows from (4.1.37) that $|b_{j+\frac{1}{2}}| = |d_j| = d$, $\forall j \in \mathbb{Z}$. Consequently, one necessarily has d = 0 (otherwise $u \notin L_2(G_{\mathcal{B}})$). Thus, $b_{j+\frac{1}{2}} = d_j = c_j^{\pm} = 0$, $\forall j \in \mathbb{Z}$. Finally, from the relation (4.1.40) we get $a_{j+\frac{1}{2}} = a_{j-\frac{1}{2}} \cos \lambda$, $\forall j \in \mathbb{Z}$, which implies $|a_{j+\frac{1}{2}}| = a$, $\forall j \in \mathbb{Z}$. If $a \neq 0$ then $u \notin L_2(G_{\mathcal{B}})$. Otherwise u = 0.

(ii) If $\sin\left(\frac{\lambda L}{2}\right) = 0$, then it follows from (4.1.38) that $d_j = 0$, $\forall j \in \mathbb{Z}$ (since $1 - e^{-i\beta} \neq 0$). Similarly, if $\cos\left(\frac{\lambda L}{2}\right) = 0$, then it follows from (4.1.39) that $d_j = 0$, $\forall j \in \mathbb{Z}$ (since $1 + e^{-i\beta} \neq 0$). Consequently, the relations (4.1.37) imply that $b_{j+\frac{1}{2}} = a_{j+\frac{1}{2}} = 0$, $\forall j \in \mathbb{Z}$ (since $\sin \lambda \neq 0$). Next, it follows from (4.1.40) that $c_j^{\pm} = c_j$, $\forall j \in \mathbb{Z}$. If $\sin\left(\frac{\lambda L}{2}\right) = 0$, then (4.1.39) implies $c_j \left(1 - e^{-i\beta}\right) = 0$, $\forall j \in \mathbb{Z}$. If $\cos\left(\frac{\lambda L}{2}\right) = 0$, then (4.1.38) implies $c_j \left(1 + e^{-i\beta}\right) = 0$, $\forall j \in \mathbb{Z}$. In both cases we get $c_j = 0$, $\forall j \in \mathbb{Z}$, and, consequently, u = 0.

For $\lambda \notin \Sigma \cup \Sigma_L$ one can express the coefficients $\left\{a_{j+\frac{1}{2}}, b_{j+\frac{1}{2}}, c_j^{\pm}, d_j\right\}$ in terms of the values $\{\mathbf{u}_j\}$:

$$a_{j+\frac{1}{2}} = \frac{1}{\sin\lambda} \left(\mathbf{u}_{j+1} - \mathbf{u}_j \cos\lambda \right), \quad b_{j+\frac{1}{2}} = d_j = \mathbf{u}_j, \quad c_j^{\pm} = \alpha^{\pm} \mathbf{u}_j, \qquad j \in \mathbb{Z}, \quad (4.1.43)$$

where α^{\pm} are defined in (4.1.41)–(4.1.42). Then, the relations (4.1.34)–(4.1.36) can be rewritten as

$$u_{j+\frac{1}{2}}(s) = \mathbf{u}_j \frac{\sin \lambda (1-s)}{\sin \lambda} x + \mathbf{u}_{j+1} \frac{\sin (\lambda s)}{\sin \lambda}, \qquad s \in [0,1], \qquad \forall j \in \mathbb{Z}, \quad (4.1.44)$$

$$u_j^+(y) = \mathbf{u}_j \frac{\sin\left(\lambda\left(\frac{L}{2} - y\right)\right)}{\sin\left(\frac{\lambda L}{2}\right)} + \mathbf{u}_j^+ \frac{\sin\left(\lambda y\right)}{\sin\left(\frac{\lambda L}{2}\right)}, \qquad y \in \left[0, \frac{L}{2}\right], \qquad \forall j \in \mathbb{Z}, \quad (4.1.45)$$

$$u_j^-(y) = \mathbf{u}_j \frac{\sin\left(\lambda\left(\frac{L}{2} + y\right)\right)}{\sin\left(\frac{\lambda L}{2}\right)} - \mathbf{u}_j^- \frac{\sin\left(\lambda y\right)}{\sin\left(\frac{\lambda L}{2}\right)}, \qquad y \in \left[-\frac{L}{2}, 0\right], \qquad \forall j \in \mathbb{Z}.$$
(4.1.46)

After plugging (4.1.43) into the relation (4.1.40) we end up with the following finite difference equation:

$$\mathbf{u}_{j+1} + 2g_{\beta}(\lambda)\mathbf{u}_j + \mathbf{u}_{j-1} = 0, \qquad \forall j \in \mathbb{Z}^*, \tag{4.1.47}$$

$$\mathbf{u}_1 + 2g^{\mu}_{\beta}(\lambda)\mathbf{u}_0 + \mathbf{u}_{-1} = 0. \tag{4.1.48}$$

Here

$$g_{\beta}(\lambda) = -\cos \lambda + \frac{\sin \lambda (\cos \beta - \cos (\lambda L))}{\sin (\lambda L)}, \qquad \lambda \notin \Sigma \cup \Sigma_L,$$
$$g_{\beta}^{\mu}(\lambda) = -\cos \lambda + \mu \frac{\sin \lambda (\cos \beta - \cos (\lambda L))}{\sin (\lambda L)}, \qquad \lambda \notin \Sigma \cup \Sigma_L.$$

Case $\lambda = 0$

If 0 is an eigenvalue of the operator $A^{\mu}(\beta)$, then the corresponding eigenfunction has the form

$$u_{j+\frac{1}{2}}(s) = \mathbf{u}_j(1-s) + \mathbf{u}_{j+1}s, \qquad s \in [0,1], \qquad \forall j \in \mathbb{Z}, \qquad (4.1.49)$$

$$u_j^+(y) = \mathbf{u}_j \left(1 - \frac{2y}{L}\right) + \frac{2\mathbf{u}_j^+}{L} y, \qquad y \in \left[0, \frac{L}{2}\right], \qquad \forall j \in \mathbb{Z}, \qquad (4.1.50)$$

$$u_j^-(y) = \mathbf{u}_j \left(1 + \frac{2y}{L} \right) - \frac{2\mathbf{u}_j^-}{L} y, \qquad y \in \left[-\frac{L}{2}, 0 \right], \qquad \forall j \in \mathbb{Z}.$$
(4.1.51)

From the quasiperiodicity conditions (4.1.2) we find

$$\mathbf{u}_j^+ = \frac{1+e^{-i\beta}}{2}\mathbf{u}_j, \qquad \mathbf{u}_j^- = \frac{1+e^{i\beta}}{2}\mathbf{u}_j, \qquad j \in \mathbb{Z}.$$

Taking into account the Kirchhoff's conditions (4.1.3), we get a finite difference equation again:

$$\mathbf{u}_{j+1} - 2\left(1 + \frac{w_j^{\mu}}{L}\left(1 - \cos\beta\right)\right)\mathbf{u}_j + \mathbf{u}_{j-1} = 0, \qquad j \in \mathbb{Z}.$$
 (4.1.52)

Finally, combining the cases $\lambda \notin \Sigma \cup \Sigma_L$ and $\lambda = 0$, we get the finite difference equation (4.1.47)–(4.1.48) with the function g_β defined as

$$g_{\beta}(\lambda) = \begin{cases} -\cos\lambda + \frac{\sin\lambda(\cos\beta - \cos(\lambda L))}{\sin(\lambda L)}, & \lambda \notin \Sigma \cup \Sigma_L, \\ \frac{\cos\beta - 1}{L} - 1, & \lambda = 0, \end{cases}$$
(4.1.53)
$$g^{\mu}_{\beta}(\lambda) = \mu g_{\beta}(\lambda) + (\mu - 1), \qquad \lambda \notin \Sigma^* \cup \Sigma_L.$$

Consider the characteristic equation associated with the system (4.1.47):

$$r^{2} + 2g_{\beta}(\lambda)r + 1 = 0, \qquad (4.1.54)$$

It has a solution such that $|r_{\beta}(\lambda)| \leq 1$:

$$r_{\beta}(\lambda) = -g_{\beta}(\lambda) + sign\left(g_{\beta}(\lambda)\right)\sqrt{g_{\beta}^{2}(\lambda)} - 1.$$
(4.1.55)

The following proposition, analogous to Proposition 1.3.10, gives the relation between the absolute value of $g_{\beta}(\lambda)$ and the nature of the point λ^2 .

Proposition 4.1.7. For $\lambda \in \mathbb{R}_+ \setminus \Sigma_L$,

$$|g_{\beta}(\lambda)| \leq 1 \quad \Leftrightarrow \quad |r_{\beta}(\lambda)| = 1 \quad \Leftrightarrow \quad \lambda^2 \in \sigma(A(\beta)). \tag{4.1.56}$$

Proof. The first equivalence follows immediately from (4.1.55). Next, $|g_{\beta}(\lambda)| \leq 1$ if and only if there exists $\theta \in [0, \pi]$ such that $g_{\beta}(\lambda) = -\cos \theta$. Taking into account the definition (4.1.53) of the function g_{β} , we get

$$\cos \lambda - \cos \theta = \frac{\sin \lambda (\cos \beta - \cos (\lambda L))}{\sin (\lambda L)}, \qquad \lambda \notin \Sigma \cup \Sigma_L. \tag{4.1.57}$$

The relation (4.1.57) is equivalent to (4.1.33) for $\lambda \in \mathbb{R}_+ \setminus (\Sigma \cup \Sigma_\beta \cup \Sigma_L)$. If $\lambda \in \{\Sigma^* \cup \Sigma_\beta^*\} \setminus \Sigma_L$, then we have $|g_\beta(\lambda)| \leq 1$ and $\lambda^2 \in \sigma(A(\beta))$ (cf. Proposition 4.1.5 (*i*)). Finally, $0 \in \sigma(A(\beta))$ if and only if $\beta = 0$. At the same time, $|g_\beta(0)| \leq 1$ if and only if $\beta = 0$.

The argument used in Section 1.3.2.2 applies in the present case to show that

$$\lambda^2 \in \sigma_d \left(A^{\mu}(\beta) \right) \quad \Leftrightarrow \quad r_{\beta}(\lambda) = -g^{\mu}_{\beta}(\lambda),$$

and the corresponding eigenfunction is

$$\mathbf{u}_j = \mathbf{u}_0 r_\beta^{|j|}(\lambda), \qquad \forall j \in \mathbb{Z}.$$

With the classification of the gaps in two types introduced in Proposition 4.1.6 we can state the following theorem.

Theorem 4.1.1. The operator $A^{\mu}(\beta)$ has no embedded eigenvalues of finite multiplicity for any $\mu > 0$, $\beta \in [0, \pi]$. For any $0 < \mu < 1$, $\beta \in [0, \pi]$ there exist two simple eigenvalues of the operator $A^{\mu}(\beta)$ in each gap of this operator of type I and one simple eigenvalue in each gap of type II. These eigenvalues are characterised as follows:

$$\lambda^2 \in \sigma_d \left(A^{\mu}(\beta) \right) \quad \Leftrightarrow \quad \lambda \text{ is a solution of the equation } \mu = F_{\beta}(\lambda),$$

where

$$F_{\beta}(\lambda) = 1 - \frac{\sqrt{g_{\beta}^2(\lambda) - 1}}{|g_{\beta}(\lambda) + \cos \lambda|}.$$

For $\mu \ge 1$ the operator $A^{\mu}(\beta)$ has no eigenvalues.

The proof is an obvious modification of the one of Theorem (1.3.1), the relation (1.3.58) being replaced by the relation

$$F_{\beta}(\lambda) = 1 - \sqrt{1 - \phi_{L,\beta}(\lambda) \left(\phi_{L,\beta}(\lambda) - \psi(\lambda)\right)}.$$

Remark 4.1.4. For $\beta \in [0, \pi]$, the interval $[0, \sigma_*[, \sigma_* = \min \sigma(A(\beta)))$, is considered as a gap of type *II* (cf. Remark 4.1.2). Thus, the operator $A^{\mu}(\beta)$ has one simple eigenvalue below the essential spectrum if $\beta \in [0, \pi]$.

4.1.3 Results for the operator $A^{\mu}_{\varepsilon}(\beta)$

We can now give the analogues of Theorems 1.4.1, 1.4.3 for the operator $A^{\mu}_{\varepsilon}(\beta)$. Let $A_{\varepsilon}(\beta)$ be the non-perturbed operator defined as $A^{\mu}_{\varepsilon}(\beta)$ with $\mu = 1$, acting in $L_2(\mathcal{B}_{\varepsilon})$, where $\mathcal{B}_{\varepsilon} = \mathcal{B}^{1}_{\varepsilon}$ is the unperturbed periodicity band.

Theorem 4.1.2 (Essential spectrum). Let $\{]a_n(\beta), b_n(\beta)[, n \in \mathbb{N}^*\}\ be the gaps of the operator <math>A(\beta)$ on the limit graph $G_{\mathcal{B}}$ for $\beta \in [0, \pi]$. Then, for each $n_0 \in \mathbb{N}^*$ there exists $\varepsilon_0(\beta) > 0$ such that if $\varepsilon < \varepsilon_0(\beta)$ the operator $A_{\varepsilon}(\beta)$ has at least n_0 gaps $\{]a_{\varepsilon,n}(\beta), b_{\varepsilon,n}(\beta)[\}_{n=1}^{n_0}$ such that

$$a_{\varepsilon,n}(\beta) = a_n(\beta) + O(\varepsilon), \quad b_{\varepsilon,n}(\beta) = b_n(\beta) + O(\varepsilon), \quad \varepsilon \to 0, \quad 1 \le n \le n_0.$$

As in the case of the ladder, the proof of this theorem is based on the reduction to the periodicity cell due to Floquet-Bloch theory and the min-max principle for the bounded domain. Next, using a standard compact perturbation argument (an analogue of Proposition 1.2.1, the proof being entirely similar), we know that

$$\sigma_{ess}(A^{\mu}_{\varepsilon}(\beta)) = \sigma_{ess}(A_{\varepsilon}(\beta)).$$

Theorem 4.1.3 (Discrete spectrum – weak version). Let $]a(\beta), b(\beta)[$ be a gap of the operator $A^{\mu}(\beta)$ on the limit graph $G_{\mathcal{B}}$ for $\beta \in [0, \pi]$ and $\lambda^{(0)}(\beta) \in]a(\beta), b(\beta)[$ a (simple) eigenvalue of this operator. Then there exists $\varepsilon_0(\beta) > 0$ such that if $\varepsilon < \varepsilon_0(\beta)$ the operator $A^{\mu}_{\varepsilon}(\beta)$ has an eigenvalue $\lambda_{\varepsilon}(\beta)$ inside a gap $]a_{\varepsilon}(\beta), b_{\varepsilon}(\beta)[$ such that:

$$\lambda_{\varepsilon}(\beta) = \lambda^{(0)}(\beta) + O\left(\sqrt{\varepsilon}\right), \qquad \varepsilon \to 0.$$

This theorem can be proved using the same argument as in the case of the ladder based on construction of an appropriate pseudo-mode obtained as an extrapolation of the eigenfunction corresponding to the eigenvalue $\lambda^{(0)}(\beta)$.

Remark 4.1.5. An analogue of Theorem 1.4.2 should also be possible to derive using matched asymptotic expansions with obvious minor modifications in the proof. This would give the optimal order of convergence ε for the eigenvalue as well as its complete asymptotic expansion:

$$\lambda_{\varepsilon}(\beta) = \sum_{k=0}^{n} \lambda^{(k)}(\beta) \varepsilon^{k} + O\left(\varepsilon^{n+1}\right), \qquad \varepsilon \to 0.$$

4.1.4 A more general geometry

We can also consider a slightly more general geometry, where the thickness of the horizontal edges is different from the thickness of the vertical edges. In other words, the distance between the obstacles is ε in the x-direction and $\nu\varepsilon$ in the y-direction for some $\nu > 0$ (cf. figure 4.7).



Figure 4.7: A more general domain: the thickness of the non-perturbed vertical edges is ε whereas the thickness of the horizontal edges is $\nu\varepsilon$.

This does not change the results qualitatively. Indeed, the only modification in the definition of the limit operator is the appearance of a factor ν in Kirchhoff's conditions. More precisely, the operator $A^{\mu}_{\nu}(\beta)$ is defined in the space

$$L_{2}^{\mu}(G_{\mathcal{B}}) = \left\{ u: \|u\|_{L_{2}^{\mu}(G_{\mathcal{B}})}^{2} < \infty \right\},$$
$$\|u\|_{L_{2}^{\mu}(G_{\mathcal{B}})}^{2} = \sum_{j \in \mathbb{Z}} \left(w_{j}^{\mu} \|u_{j}^{+}\|_{L_{2}(e_{j}^{+})}^{2} + w_{j}^{\mu} \|u_{j}^{-}\|_{L_{2}(e_{j}^{-})}^{2} + \nu \|u_{j+\frac{1}{2}}\|_{L_{2}\left(e_{j+\frac{1}{2}}\right)}^{2} \right),$$

as

$$(A^{\mu}_{\nu}(\beta)u)|_{e} = -(u|_{e})'', \qquad e \in \mathcal{E}_{G_{\mathcal{B}}},$$

$$D(A^{\mu}_{\nu}(\beta)) = \left\{ u \in H^{2}(G_{\mathcal{B}}) : \\ u^{+}_{j} = e^{-i\beta}u^{-}_{j}, \quad (u^{+}_{j})'\left(\frac{L}{2}\right) = e^{-i\beta}(u^{-}_{j})'\left(-\frac{L}{2}\right), \qquad \forall j \in \mathbb{Z}, \\ \nu u'_{j+\frac{1}{2}}(0) - \nu u'_{j-\frac{1}{2}}(1) + w^{\mu}_{j}(u^{+}_{j})'(0) - w^{\mu}_{j}(u^{-}_{j})'(0) = 0, \qquad \forall j \in \mathbb{Z} \right\}$$

This results in the same characterization of the spectrum of the operator $A_{\nu}(\beta)$ (corresponding to the non-perturbed case $\mu = 1$) as the one given in Proposition 4.1.3 (5) with the equation (4.1.33) replaced by

$$\nu \phi_{L,\beta}(\lambda) = f_{\theta}(\lambda). \tag{4.1.58}$$

All the other statements of Proposition 4.1.3 hold without any change. This means that the spectrum of the $A_{\nu}(\beta)$ for $\nu \neq 1$, does not change qualitatively from the case $\nu = 1$. Nevertheless, the parameter ν influences the size of the gaps. Clearly, the size of the image of a gap by the function $x \mapsto \sqrt{x}$ can never exceed π (since $\{\lambda^2, \lambda \in \{\pi\mathbb{N}^*\}\} \subset \sigma(A_{\nu}(\beta))$), cf. Proposition 4.1.3 (2)). However, choosing a bigger value of ν leads to decreasing of this size whereas choosing a smaller value of ν leads to increasing of this size. An example is given in figure 4.8. Roughly speaking, for ν big there is a lot of spectrum with small gaps inside it and for small ν there are big gaps separated by small pieces of spectrum.



Figure 4.8: Influence of the parameter ν on the size of the gaps for $L = \frac{\pi}{2}$: for smaller values of ν the gaps are bigger and the spectral bands are smaller.

Similarly, the results about the discrete spectrum when a perturbation is introduced hold true for this more general geometry. The equation characterization for the eigenvalues given in Theorem 4.1.1 is be replaced by

$$\mu = F_{\beta}^{\nu}(\lambda), \qquad F_{\beta}^{\nu}(\lambda) = 1 - \sqrt{1 - \nu \phi_{L,\beta}(\lambda) \left(\nu \phi_{L,\beta}(\lambda) - \psi(\lambda)\right)}.$$

Obviously, the conclusion about the number of eigenvalues inside the gaps does not change: there are still precisely one or two eigenvalues inside each gap depending on its type.

Remark 4.1.6. We see that the ladder considered in Chapter 1 is a periodicity band for the domain shown in figure 4.7 with $\nu = 2$. We remember that the equations describing the spectra of the operators A(0), $A(\pi)$ respectively, differed from the ones corresponding

to the operators A_s , A_{as} by a coefficient 2 (cf. (4.1.29), (4.1.30), (1.3.12), (1.3.59)). On the other hand, for $\nu = 2$ the spectra of the operators $A_2(0)$, $A_2(\pi)$ are described by the equation $2\phi_{L,\beta}(\lambda) = f_{\theta}(\lambda)$ (cf. (4.1.58)). This leads precisely to the relations (1.3.12), (1.3.59). However, the relations (4.1.29), (4.1.30) are not exactly the ones describing the spectra of the operators A(0), $A(\pi)$. Indeed, while obtaining them from the "true" ones, (4.1.27) and (4.1.28), we performed a division by $\sin\left(\frac{\lambda L}{2}\right)$ and $\cos\left(\frac{\lambda L}{2}\right)$ respectively. Thus, we excluded from consideration the sets $\{\lambda^2, \sin\left(\frac{\lambda L}{2}\right) = 0\}$ for $\beta = 0$ and $\{\lambda^2, \cos\left(\frac{\lambda L}{2}\right) = 0\}$ for $\beta = \pi$ which belong to the spectra of the operators A(0) and $A(\pi)$ respectively. They also belong to the spectra of the operators A_s and A_{as} respectively, as shown in Propositions 1.3.4 (1), 1.3.12 (1). Thus, we can finally conclude that

$$\sigma(A_2(0)) = \sigma(A_s), \qquad \sigma(A_2(\pi)) = \sigma(A_{as}).$$

To resume, we see that considering the symmetric (resp. antisymmetric) part of the operator with Neumann boundary conditions and the 0-quasiperiodic (resp. π -quasiperiodic) operator leads to the same limit problems on the graph.

4.2 Numerical approach

The numerical method described in Section 3.2 applies to the β -quasiperiodic case with minor modifications (the detailed description of the method in the β -quasiperiodic case can be found in [24]). Since the domain in question is now $\mathcal{B}^{\mu}_{\varepsilon}$ (cf. figure 4.2), the periodicity cell and the interior domains are the ones shown in figure 4.9. Since β -



Figure 4.9

quasiperiodic conditions are imposed on the boundaries Γ_1 , Γ_4 (resp. Γ_1^{μ} , Γ_4^{μ}), in our numerical method the meshes of these two boundaries have to be the same, and the corresponding vertices of the two meshes are coupled. This reduces the number of degrees of freedom in the corresponding functional space.

4.2.1 Numerical results

In this section we give some numerical results for the operator $A^{\mu}_{\varepsilon}(\beta)$. For a fixed β they are very similar to those shown in Section 3.4. For this reason we will concentrate on the

dependence on β . In figure 4.10 we show the dependence of the essential spectrum of the operator $A_{\varepsilon}(\beta)$ as well as the limit operator $A(\beta)$ with respect to β for $L = \frac{\pi}{2}$, $\varepsilon = 0.1$. For the limit operator it follows from Lemma 4.1.1 that any point $\lambda \in \pi \mathbb{N}^*$ (in the range represented in figure 4.10 there is only the point $\lambda = \pi$) corresponds to the lower end of a gap for some values of β and to the upper end of a gap for the others. At the point β_0 where the "transition" happens it corresponds to an interior point of the spectrum. This transition point is the point of discontinuity of the function $\phi_{L,\beta}$, i.e. the point given by the relation $\cos \beta_0 = \cos (\lambda L)$. It is interesting to notice that according to the numerical results it seems that the same property holds for the non-limit operator as well. In other words, there is a point λ which corresponds to the lower end of a gap for some values of β and to the upper end of a gap for some values of given in figure 4.10 $\lambda \approx 3.24$). In figure 4.11 the dependence on β of the eigenvalues



Figure 4.10: Dependence of the essential spectrum on β for $L = \frac{\pi}{2}$. The first 5 spectral bands are represented.

in the first gap and below the essential spectrum is shown for $L = \pi/2$, $\mu = 0.25$ in the case of the graph and for the 2D domain with $\varepsilon = 0.1$. One can remark that the eigenvalues below the essential spectrum are very close to it. In the first gap they are better separated from the essential spectrum if β is not close to π . Again, the question of existence of eigenvalues for any $\beta \in [0, \pi]$ is not answered since the computations become costly when the eigenvalue approaches the essential spectrum.



Figure 4.11: Dependence of the eigenvalues in the first gap and below the essential spectrum on β for $L = \frac{\pi}{2}$, $\mu = 0.25$.

Time-dependent simulations

As it was discussed in the beginning of this chapter, an eigenvalue ω^2 of the operator $A^{\mu}_{\varepsilon}(\beta)$ corresponds to a guided mode propagating the the *y* direction with the speed $\frac{\omega L}{\beta}$. We would like then to see such guided modes in a time-dependent simulation. The idea is to put a time-harmonic source with the frequency ω at some point of the perturbed line which would give rise to the corresponding guided mode.

Two kinds of simulations have been preformed. The first one is based on a "naive" Matlab finite elements code where a finite difference scheme in time is implemented. The infinite domain $\Omega^{\mu}_{\varepsilon}$ is replaced by a finite one, $\tilde{\Omega}^{\mu}_{\varepsilon}$, without putting any special boundary conditions that would take into account the infinite nature of the domain. For this reason, one obtains an approximation of the solution as long as the wave does not reach the boundary of the domain. The problem in question is

$$\begin{cases} \frac{\partial^2 u}{\partial t^2} = \Delta u + f_{\omega}(x, y, t) & \text{in} \quad \widetilde{\Omega}^{\mu}_{\varepsilon}, \\ u|_{t=0} = 0, & \frac{\partial u}{\partial t}\Big|_{t=0} = 0, \\ \frac{\partial u}{\partial n}\Big|_{\partial \widetilde{\Omega}^{\mu}_{\varepsilon}} = 0, \end{cases}$$

where

$$f_{\omega}(x, y, t) = \sin(\omega t)\delta(x_0, y_0).$$

For the discretization, we use a finite element method in space and an explicit finite difference scheme in time.

In the second set of simulations we used a much more sophisticated C++ code by Julien Coatléven which takes into account the infinite nature of the propagation domain (for more details see [12]). In figure 4.12 is represented the solution obtained with this code for $\omega = 2.25$ at t = 70.



Figure 4.12: Solution in the time-dependent simulation performed using the C++ code (Julien Coatléven) for L = 2, $\mu = 0.25$, $\omega = 2.25$, t = 70. A guided mode propagating along the linear defect seems to appear.

CHAPTER 5

GUIDED MODES IN OPEN PERIODIC LINEIC WAVEGUIDES: THE 3D CASE

5.1 Geometry and statement of the problem

Propagation domain

In this chapter we consider a 3D generalization of the plane waveguide considered in the previous chapters. By a 3D generalization we mean a domain obtained by fattening a 3D infinite primitive orthorhombic lattice (which is periodic in three orthogonal directions). In each direction the straight lines of the lattice are replaced by parallel infinite pipes of a constant cross-section of size of order ε which is going to be small. The cross-sections are not necessarily the same for the three directions. More precisely, let L_x , L_y , L_z be positive numbers (the periods of the domain in the x, y and z directions respectively). Let ω_x , ω_y , ω_z be bounded domains of \mathbb{R}^2 containing the origin (the normalized cross sections of the pipes parallel to the axes x, y and z respectively). Then, the propagation domain Ω_{ε} is defined as follows:

$$\Omega_{\varepsilon} = \bigcup_{(k,\ell)\in\mathbb{Z}^2} \left\{ \mathcal{P}^x_{\varepsilon,k,\ell} \cup \mathcal{P}^y_{\varepsilon,k,\ell} \cup \mathcal{P}^z_{\varepsilon,k,\ell} \right\},$$
$$\mathcal{P}^{\gamma}_{\varepsilon,k,\ell} = \left\{ \left(\frac{\alpha}{\varepsilon}, \frac{\beta}{\varepsilon} \right) \in \omega_{\gamma} + (kL_{\alpha}, \ell L_{\beta}) \right\}, \qquad (\alpha, \beta, \gamma) \in \{(x, y, z), (y, z, x), (z, x, y) \}.$$

Some examples of such domains are shown in figure 5.1.

A linear defect is introduced to the structure by changing the thickness of one pipe (for instance, parallel to the z direction): the characteristic size of its cross section is set to be $\sqrt{\mu}\varepsilon$ instead of ε . The perturbed domain $\Omega_{\varepsilon}^{\mu}$ is defined by the following relation:

$$\Omega^{\mu}_{\varepsilon} = \mathcal{P}^{x}_{\varepsilon,0,0} \cup \mathcal{P}^{y}_{\varepsilon,0,0} \cup \mathcal{P}^{z,\mu}_{\varepsilon,0,0} \cup \bigcup_{(k,\ell) \in \mathbb{Z}^{2} \setminus (0,0)} \left\{ \mathcal{P}^{x}_{\varepsilon,k,\ell} \cup \mathcal{P}^{y}_{\varepsilon,k,\ell} \cup \mathcal{P}^{z}_{\varepsilon,k,\ell} \right\},$$
$$\mathcal{P}^{z,\mu}_{\varepsilon,0,0} = \left\{ \left(\frac{x}{\sqrt{\mu\varepsilon}}, \frac{y}{\sqrt{\mu\varepsilon}} \right) \in \omega_{z} \right\}.$$

The limit problem for $\varepsilon \to 0$ will be posed on a 3D graph and its spectrum will be easy to analyse. For ε small enough the spectrum of the non-limit operator will be approximated

by the spectrum of the limit operator. In this chapter we will study essentially the limit operator. In Section 5.3 we mention the results for the non-limit operator that follow from the works [61], [47], [57] (see also [31, 32, 4] for an example of application for a Maxwell problem). We do not perform the asymptotic analysis which would permit to obtain a full asymptotic expansion for the eigenvalues. We mention that the approach used in the 2D case should be possible to apply in the 3D case as well with appropriate modifications.



Figure 5.1: Propagation domain

Guided modes

As in the 2D, case we are interested in guided modes, i.e. solutions of the wave equation

$$\frac{\partial^2 u}{\partial t^2} = \Delta u \quad \text{in} \quad \Omega^{\mu}_{\varepsilon}, \tag{5.1.1}$$

that propagate along the defect and stay confined in the transversal directions. Neumann boundary conditions are imposed on the boundaries of the obstacles. More precisely, we search solutions of the form

$$u(x, y, z, t) = v(x, y, z)e^{i(\omega t - \beta z/L_z)},$$
(5.1.2)

where v is a L_z -periodic function in z-direction confined in the periodicity band $B_{\varepsilon}^{\mu} = \Omega_{\varepsilon}^{\mu} \times \left] - \frac{L_z}{2}, \frac{L_z}{2} \right[$:

$$v(x, y, z + L_z) = v(x, y, z), \qquad v \in L_2(B^{\mu}_{\varepsilon}).$$

Clearly, we could choose any other periodicity band, but we will consider the symmetric one. After injecting (5.1.2) in the wave equation (5.1.1), taking into account Neumann boundary conditions on the boundaries of the obstacles,

$$\left.\frac{\partial u}{\partial n}\right|_{\partial\Omega^{\mu}_{\varepsilon}} = 0$$

and introducing the function

$$\widetilde{v} = v(x, y, z)e^{-i\beta z/L_z},$$

we end up with the following problem posed in the periodicity band:

$$\begin{cases} -\Delta \widetilde{v} = \omega^2 \widetilde{v} & \text{in } B^{\mu}_{\varepsilon}, \\ \widetilde{v} \in L_2 \left(B^{\mu}_{\varepsilon} \right), \\ \frac{\partial \widetilde{v}}{\partial n} \Big|_{\partial B^{\mu}_{\varepsilon} \setminus \{ \Sigma \cup \Sigma' \}} = 0, \\ \widetilde{v} & \text{is } \beta - \text{quasiperiodic (with respect to } z) \end{cases}$$

This is an eigenvalue problem for the operator $A^{\mu}_{\varepsilon}(\beta): L_2(B^{\mu}_{\varepsilon}) \to L_2(B^{\mu}_{\varepsilon})$ defined as

$$\begin{aligned} A^{\mu}_{\varepsilon}(\beta) &= -\Delta u, \\ D\left(A^{\mu}_{\varepsilon}(\beta)\right) &= \left\{ u \in H^{1}_{\Delta}\left(B^{\mu}_{\varepsilon}\right), \quad \frac{\partial u}{\partial n} \Big|_{\partial B^{\mu}_{\varepsilon} \setminus \{\Sigma \cup \Sigma'\}} = 0, \quad u|_{\Sigma'} = e^{-i\beta} \left. u|_{\Sigma}, \quad \frac{\partial u}{\partial z} \right|_{\Sigma'} = e^{-i\beta} \left. \frac{\partial u}{\partial z} \right|_{\Sigma} \right\} \end{aligned}$$

In the next section we describe the limit operator and compute its spectrum. The general strategy is analogous to the one used in the 2D case. We start by studying the nonperturbed operator using the Floquet-Bloch theory (Section 5.2.1). This leads to the equation (5.2.4) for the eigenvalues of the operator in the periodicity cell which is very similar to the one found in the 2D case. Here again we show that the limit operator has infinitely many gaps whose ends tend to infinity (Proposition 5.2.3). Next, we study the discrete spectrum of the perturbed operator (Section 5.2.2). As in the 2D case, the problem reduces the to study of a finite difference equation (Lemma 5.2.4). However, this time the equation is two-dimensional and cannot be solved via the associated characteristic equation as before. To study this equation we apply the discrete Fourier transform. We end up with the characterisation of the eigenvalues given in Proposition 5.2.9. This characterisation is more difficult to analyse than the one obtained in the 2D case. In Section 5.2.2.4 we prove the existence of at least one or two eigenvalues in each gap according to its type but contrarily to the 2D case we are not able to find the exact number of eigenvalues in each gap.

5.2 The limit operator

The limit operator is defined on the graph $G = \bigcup_{\varepsilon>0} \Omega_{\varepsilon} \cap \left\{z \in \left[-\frac{L_z}{2}, \frac{L_z}{2}\right]\right\}$ shown in figure 5.2. Let us introduce some notation for this graph. The vertices in the plane z = 0 are enumerated by two indices $k, \ell \in \mathbb{Z}$ in such a way that the coordinates of the vertex $v_{k,\ell}$ are $(kL_x, \ell L_y, 0)$. The vertex $v_{0,0}$ corresponds to the perturbed line. We denote by $v_{k,\ell}^{\pm}$ the vertices with the coordinates $(kL_x, \ell L_y, \pm L_z/2)$. The edge joining the vertices $v_{k,\ell}$ and $v_{k+1,\ell}$ is denoted by $e_{k+\frac{1}{2},\ell}$, the one joining the vertices $v_{k,\ell}$ and $v_{k,\ell+1}$ is denoted by $e_{k+\frac{1}{2},\ell}$. The network $v_{k,\ell}$ with the vertices $v_{k,\ell}$ and $v_{k,\ell+1}$ is denoted by $e_{k,\ell}$. Let u be a function on G. Then, its value at the vertex $v_{k,\ell}$ is denoted by $\mathbf{u}_{k,\ell}$ and its values at the vertices $v_{k,\ell}^{\pm}$ by $\mathbf{u}_{k,\ell}^{\pm}$. The restrictions of the function u at the edge $e_{k+\frac{1}{2},\ell}$ is denoted by $u_{k,\ell+\frac{1}{2}}(t)$, where the local variable s takes values in $[0, L_x]$. Similarly, its restriction at the edge $e_{k,\ell+\frac{1}{2}}$ is denoted by $u_{k,\ell+\frac{1}{2}}(t)$, where the local variable t takes values in $[0, L_y]$. Finally, the restrictions of u at the edges $e_{k,\ell}^{\pm}$ are denoted by $u_{k,\ell}^{\pm}(z)$.



Figure 5.2: Graph G

The following function spaces are introduced analogously to the 2D case:

$$L_{2}^{\mu}(G) = \left\{ u : \|u\|_{L_{2}^{\mu}(G)}^{2} < \infty \right\}, \qquad H^{2}(G) = \left\{ u \in C(G) : \|u\|_{H^{2}(G)}^{2} < \infty \right\},$$
$$\|u\|_{L_{2}^{\mu}(G)}^{2} = \sum_{k,\ell \in \mathbb{Z}} \left(w_{k,\ell}^{\mu} \left(\|u_{k,\ell}^{+}\|_{L_{2}(e_{k,\ell}^{+})}^{2} + \|u_{k,\ell}^{-}\|_{L_{2}(e_{k,\ell}^{-})}^{2} \right) + \|u_{k+\frac{1}{2},\ell}\|_{L_{2}(e_{k+\frac{1}{2},\ell})}^{2} + \|u_{k,\ell+\frac{1}{2}}\|_{L_{2}(e_{k,\ell+\frac{1}{2}})}^{2} \right),$$
$$\|u\|_{H^{2}(G)}^{2} = \sum_{k,\ell \in \mathbb{Z}} \left(\|u_{k,\ell}^{+}\|_{H^{2}(e_{k,\ell}^{+})}^{2} + \|u_{k,\ell}^{-}\|_{H^{2}(e_{k,\ell}^{-})}^{2} + \|u_{k+\frac{1}{2},\ell}\|_{H^{2}(e_{k+\frac{1}{2},\ell})}^{2} + \|u_{k,\ell+\frac{1}{2}}\|_{H^{2}(e_{k,\ell+\frac{1}{2}})}^{2} \right).$$

Here w^{μ} is the weight defined as

$$w_{k,\ell}^{\mu} = \begin{cases} \mu, & k = \ell = 0, \\ 1, & \text{otherwise.} \end{cases}$$

The limit operator $A^{\mu}(\beta): L_2(G) \to L_2(G)$ is then defined as follows: $A^{\mu}(\beta)u = -u'',$

$$D(A^{\mu}(\beta)) = \left\{ u \in H^{2}(G) : \forall k, \ell \in \mathbb{Z}, \quad u_{k,\ell}^{+} = e^{-i\beta}u_{k,\ell}^{-}, \quad (u_{k,\ell}^{+})'\left(\frac{L_{z}}{2}\right) = e^{-i\beta}(u_{k,\ell}^{-})'\left(-\frac{L_{z}}{2}\right), \quad (5.2.1)\right\}$$

$$u_{k+\frac{1}{2},\ell}'(0) - u_{k-\frac{1}{2},\ell}'(L_x) + u_{k,\ell+\frac{1}{2}}'(0) - u_{k,\ell-\frac{1}{2}}'(L_y) + w_{k,\ell}^{\mu} \left((u_{k,\ell}^+)'(0) - (u_{k,\ell}^-)'(0) \right) = 0 \right\}.$$
(5.2.2)

The relations (5.2.2) are Kirchhoff's conditions. As in the 2D case, they mean that the weighted sum of outgoing derivatives is zero at each vertex of the graph G. As usually, we will start by studying the non-perturbed operator $A(\beta)$ that corresponds to the case $\mu = 1$.

Proposition 5.2.1.

$$\sigma_{ess}\left(A^{\mu}(\beta)\right) = \sigma(A(\beta)).$$

The proof of this proposition is an obvious modification of the one for the 2D case if the 1D characteristic function is replaced by a 2D one (cf. Proposition 1.2.1). The periodic operator $A(\beta)$ has only essential spectrum and we compute it in the next section.

5.2.1 Computation of the essential spectrum

According to Floquet-Bloch theory in order to study the spectrum of the operator $A(\beta)$ we have to study the operators $A_{\beta}(k_x, k_y)$ defined on the periodicity cell

$$\mathcal{C} = G \cup \left\{ (x, y) \in \left[-\frac{L_x}{2}, \frac{L_x}{2} \right] \times \left[-\frac{L_y}{2}, \frac{L_y}{2} \right] \right\},\$$

shown in figure 5.3, with (k_x, k_y) -quasiperiodicity conditions:

$$A_{\beta}(k_{x},k_{y}): L_{2}(\mathcal{C}) \to L_{2}(\mathcal{C}), \qquad A_{\beta}(k_{x},k_{y})u = -u'',$$

$$D\left(A_{\beta}(k_{x},k_{y})\right) = \left\{u \in H^{2}(\mathcal{C}): \quad u_{6}\left(\frac{L_{z}}{2}\right) = e^{-i\beta}u_{6}\left(-\frac{L_{z}}{2}\right), \qquad u_{6}'\left(\frac{L_{z}}{2}\right) = e^{-i\beta}u_{5}'\left(-\frac{L_{z}}{2}\right), \\ u_{2}\left(\frac{L_{x}}{2}\right) = e^{-ik_{x}}u_{1}\left(-\frac{L_{x}}{2}\right), \qquad u_{2}'\left(\frac{L_{x}}{2}\right) = e^{-ik_{x}}u_{1}'\left(-\frac{L_{x}}{2}\right), \\ u_{4}\left(\frac{L_{y}}{2}\right) = e^{-ik_{y}}u_{3}\left(-\frac{L_{y}}{2}\right), \qquad u_{4}'\left(\frac{L_{y}}{2}\right) = e^{-ik_{y}}u_{3}'\left(-\frac{L_{y}}{2}\right), \\ u_{2}'(0) - u_{1}'(0) + u_{4}'(0) - u_{3}'(0) + u_{6}'(0) - u_{5}'(0) = 0\right\}.$$
(5.2.3)



Figure 5.3: Periodicity cell C

The function spaces $L_2(\mathcal{C})$, $H^2(\mathcal{C})$ are defined in a standard way, i.e.

$$L_{2}(\mathcal{C}) = \left\{ u: \sum_{i=1}^{6} \|u_{i}\|_{L^{2}(I_{i})}^{2} < \infty \right\}, \qquad H^{2}(\mathcal{C}) = \left\{ u \in C(\mathcal{C}): \sum_{i=1}^{6} \|u_{i}\|_{H^{2}(I_{i})}^{2} < \infty \right\}.$$

Proposition 5.2.2. For $(k_x, k_y) \in [0, \pi]^2$, $\lambda^2 \in \sigma(A_\beta(k_x, k_y)) \setminus \{0\}$ if and only if λ is a solution of the equation

$$\sin(\lambda L_y)\sin(\lambda L_z)\left(\cos(\lambda L_x) - \cos k_x\right) + \sin(\lambda L_z)\sin(\lambda L_x)\left(\cos(\lambda L_y) - \cos k_y\right) + \sin(\lambda L_x)\sin(\lambda L_y)\left(\cos(\lambda L_z) - \cos\beta\right) = 0. \quad (5.2.4)$$

The point $\lambda = 0$ belongs to $\sigma(A_{\beta}(k_x, k_y))$ if and only if $k_x = k_y = \beta = 0$.

Remark 5.2.1. One can notice that the equation (5.2.4) has a similar structure as the one found in the 2D case (cf. Proposition 4.1.2).

Proof. Let us start with the case $\lambda \neq 0$. If λ^2 is an eigenvalue of the operator $A_\beta(k_x, k_y)$, then the corresponding eigenfunction $u \in D(A_\beta(k_x, k_y))$ solves the equation

$$u'' + \lambda^2 u = 0$$

on each edge of the periodicity cell \mathcal{C} . We have then

$$\begin{split} & u_1(x) = c_1 e^{i\lambda x} + d_1 e^{-i\lambda x}, & x \in \left[-\frac{L_x}{2}, 0\right], \\ & u_2(x) = c_2 e^{i\lambda x} + d_2 e^{-i\lambda x}, & x \in \left[0, \frac{L_x}{2}\right], \\ & u_3(y) = c_3 e^{i\lambda y} + d_3 e^{-i\lambda y}, & y \in \left[-\frac{L_y}{2}, 0\right], \\ & u_4(y) = c_4 e^{i\lambda y} + d_4 e^{-i\lambda y}, & y \in \left[0, \frac{L_y}{2}\right], \\ & u_5(z) = c_5 e^{i\lambda z} + d_5 e^{-i\lambda z}, & z \in \left[-\frac{L_z}{2}, 0\right], \\ & u_6(z) = c_6 e^{i\lambda z} + d_6 e^{-i\lambda z}, & z \in \left[0, \frac{L_z}{2}\right]. \end{split}$$

From the continuity of the eigenfunction at the central vertex we get

$$c_1 + d_1 = c_2 + d_2 = c_3 + d_3 = c_4 + d_4 = c_5 + d_5 = c_6 + d_6.$$
(5.2.5)

The quasiperiodic conditions imply that

$$\left(c_1 e^{-\frac{i\lambda L_x}{2}} + d_1 e^{\frac{i\lambda L_x}{2}}\right) e^{-ik_x} = c_2 e^{\frac{i\lambda L_x}{2}} + d_2 e^{-\frac{i\lambda L_x}{2}},$$
(5.2.6)

$$\left(c_1 e^{-\frac{i\lambda L_x}{2}} - d_1 e^{\frac{i\lambda L_x}{2}}\right) e^{-ik_x} = c_2 e^{\frac{i\lambda L_x}{2}} - d_2 e^{-\frac{i\lambda L_x}{2}},\tag{5.2.7}$$

$$\left(c_3 e^{-\frac{i\lambda L_y}{2}} + d_3 e^{\frac{i\lambda L_y}{2}}\right) e^{-ik_y} = c_4 e^{\frac{i\lambda L_y}{2}} + d_4 e^{-\frac{i\lambda L_y}{2}},$$
(5.2.8)

$$\left(c_3 e^{-\frac{i\lambda L_y}{2}} - d_3 e^{\frac{i\lambda L_y}{2}}\right) e^{-ik_y} = c_4 e^{\frac{i\lambda L_y}{2}} - d_4 e^{-\frac{i\lambda L_y}{2}},\tag{5.2.9}$$

$$\left(c_5 e^{-\frac{i\lambda L_z}{2}} + d_5 e^{\frac{i\lambda L_z}{2}}\right) e^{-i\beta} = c_6 e^{\frac{i\lambda L_z}{2}} + d_6 e^{-\frac{i\lambda L_z}{2}},$$
(5.2.10)

$$\left(c_5 e^{-\frac{i\lambda L_z}{2}} - d_5 e^{\frac{i\lambda L_z}{2}}\right) e^{-i\beta} = c_6 e^{\frac{i\lambda L_z}{2}} - d_6 e^{-\frac{i\lambda L_z}{2}}.$$
(5.2.11)

Finally, Kirchhoff's condition yields

$$-c_1 + d_1 + c_2 - d_2 - c_3 + d_3 + c_4 - d_4 - c_5 + d_5 + c_6 - d_6 = 0.$$
 (5.2.12)

From the equations (5.2.6)–(5.2.11) one finds:

$$c_1 = e^{i\lambda L_x + ik_x} c_2,$$
 $c_3 = e^{i\lambda L_y + ik_y} c_4,$ $c_5 = e^{i\lambda L_z + i\beta} c_6,$ (5.2.13)

$$d_1 = e^{-i\lambda L_x + ik_x} d_2, \qquad d_3 = e^{-i\lambda L_y + ik_y} d_4, \qquad d_5 = e^{-i\lambda L_z + i\beta} d_6.$$
(5.2.14)

Due to the relations (5.2.13), (5.2.14) the system (5.2.5)-(5.2.12) reduces to the following one:

$$c_2 + d_2 = c_4 + d_4 = c_6 + d_6, (5.2.15)$$

$$c_2 \left(1 - e^{i\lambda L_x + ik_x} \right) + d_2 \left(1 - e^{-i\lambda L_x + ik_x} \right) = 0, \tag{5.2.16}$$

$$c_4 \left(1 - e^{i\lambda L_y + ik_y} \right) + d_4 \left(1 - e^{-i\lambda L_y + ik_y} \right) = 0, \tag{5.2.17}$$

$$c_6 \left(1 - e^{i\lambda L_z + i\beta}\right) + d_6 \left(1 - e^{-i\lambda L_z + i\beta}\right) = 0,$$
 (5.2.18)

$$c_2 \left(1 - e^{i\lambda L_x + ik_x} \right) + c_4 \left(1 - e^{i\lambda L_y + ik_y} \right) + c_6 \left(1 - e^{i\lambda L_z + i\beta} \right) = 0.$$
 (5.2.19)

Thus, the system (5.2.15)–(5.2.19) has a non-trivial solution if and only if

$$D(\lambda) = 0, \tag{5.2.20}$$

where

$$D(\lambda) = \left| \begin{array}{ccccccc} 1 & 1 & -1 & -1 & 0 & 0 \\ 1 & 1 & 0 & 0 & -1 & -1 \\ 1 - e^{i(\lambda L_x + k_x)} & 1 - e^{i(k_x - \lambda L_x)} & 0 & 0 & 0 \\ 0 & 0 & 1 - e^{i(\lambda L_y + k_y)} & 1 - e^{i(k_y - \lambda L_y)} & 0 & 0 \\ 0 & 0 & 0 & 0 & 1 - e^{i(\lambda L_z + \beta)} & 1 - e^{i(\beta - \lambda L_z)} \\ 1 - e^{i(\lambda L_x + k_x)} & 0 & 1 - e^{i(\lambda L_y + k_y)} & 0 & 1 - e^{i(\lambda L_z + \beta)} & 0 \end{array} \right|.$$

$$(5.2.21)$$

Computing $D(\lambda)$ leads to the condition (5.2.4) (see Lemma A.4.1 in Annexe for the details of the computation).

If $\lambda = 0$, then the eigenfunction has to be linear at each edge of the graph:

$$\begin{aligned} u_1(x) &= c_1 + d_1 x, & x \in \left[-\frac{L_x}{2}, 0 \right], \\ u_2(x) &= c_2 + d_2 x, & x \in \left[0, \frac{L_x}{2} \right], \\ u_3(y) &= c_3 + d_3 y, & y \in \left[-\frac{L_y}{2}, 0 \right], \\ u_4(y) &= c_4 + d_4 y, & y \in \left[0, \frac{L_y}{2} \right], \\ u_5(z) &= c_5 + d_5 z, & z \in \left[-\frac{L_z}{2}, 0 \right], \\ u_6(z) &= c_6 + d_6 z, & z \in \left[0, \frac{L_z}{2} \right]. \end{aligned}$$

The continuity condition implies that $c_i = c_0$, $1 \leq i \leq 6$. From the quasiperiodic conditions it follows that

$$d_2 = e^{-ik_x} d_1, \qquad d_1 = \frac{c_0}{L_x} \left(1 - e^{ik_x} \right), \qquad (5.2.22)$$

$$d_4 = e^{-ik_y} d_3, \qquad d_3 = \frac{c_0}{L_y} \left(1 - e^{ik_y} \right),$$
 (5.2.23)

$$d_6 = e^{-ik_y} d_5, \qquad d_5 = \frac{c_0}{L_z} \left(1 - e^{i\beta}\right).$$
 (5.2.24)

Finally, Kirchhoff's condition (5.2.3) gives

$$-d_1 + d_2 - d_3 + d_4 - d_5 + d_6 = 0. (5.2.25)$$

After injecting (5.2.22)–(5.2.24) in (5.2.25) we get

$$c_0\left(\frac{\cos k_x - 1}{L_x} + \frac{\cos k_y - 1}{L_y} + \frac{\cos \beta - 1}{L_z}\right) = 0.$$

Hence, there exists a non-trivial solution if and only if $k_x = k_y = \beta = 0$ (since otherwise the quantity inside the brackets is strictly negative). This finishes the proof.

Remark 5.2.2. Similarly to the 2D case, one can notice that if L_x , L_y and L_z are commensurable, then the set $\{\lambda : \lambda^2 \in \sigma(A_\beta(k_x, k_y))\}$ is periodic. Indeed, in this case the left-hand side of the equation (5.2.4) is a periodic function.

The spectrum of the operator $A(\beta)$ can now be determined due to the decomposition

$$\sigma(A(\beta)) = \bigcup_{k_x, k_y \in [0,\pi]} \sigma(A_\beta(k_x, k_y)).$$
(5.2.26)

Thus, the point λ^2 (different from zero) belongs to the spectrum of the operator $A(\beta)$ if and only if there exists a couple (k_x, k_y) such that the relation (5.2.4) is satisfied. Let us introduce some notation that we will use throughout the chapter:

$$\Sigma_{x} = \{\lambda \ge 0 : \sin(\lambda L_{x}) = 0\}, \qquad \Sigma_{x}^{*} = \Sigma_{x} \setminus \{0\},$$

$$\Sigma_{y} = \{\lambda \ge 0 : \sin(\lambda L_{y}) = 0\}, \qquad \Sigma_{y}^{*} = \Sigma_{y} \setminus \{0\},$$

$$\Sigma_{z} = \{\lambda \ge 0 : \sin(\lambda L_{z}) = 0\}, \qquad \Sigma_{z}^{*} = \Sigma_{z} \setminus \{0\},$$

$$\Sigma = \Sigma_{x} \cup \Sigma_{y} \cup \Sigma_{z}, \qquad \Sigma^{*} = \Sigma \setminus \{0\},$$

$$\Sigma_{z}(\beta) = \{\lambda \ge 0 : \cos(\lambda L_{z}) = \cos\beta\}, \qquad \Sigma_{z}^{*}(\beta) = \Sigma_{z}(\beta) \setminus \{0\},$$

$$\widetilde{\Sigma}_{z}(\beta) = \left\{\sum_{z} = \{\lambda \ge 0 : \sin(\lambda L_{z}) = 0\}, \qquad \beta \in [0, \pi[, \\ \Sigma_{z} \setminus \Sigma_{z}(\beta) = \{\lambda \ge 0 : \cos(\lambda L_{z}) = -\cos\beta\}, \qquad \beta \in \{0, \pi\},$$

$$\widetilde{\Sigma}_{z}^{*}(\beta) = \widetilde{\Sigma}_{z}(\beta) \setminus \{0\}. \qquad (5.2.28)$$

Proposition 5.2.3.

- $1. \ \left\{ \lambda^2: \ \lambda \in \Sigma^*_x \cup \Sigma^*_y \cup \Sigma_z(\beta) \right\} \subset \sigma(A(\beta)).$
- 2. For any $\beta \in [0, \pi]$, the operator $A(\beta)$ has infinitely many gaps whose ends tend to infinity.

Proof.

1. For $\lambda \neq 0$, $\lambda \in \Sigma_x^* \cup \Sigma_y^* \cup \Sigma_z(\beta)$ the equation (5.2.4) is obviously verified for $\cos k_x = \cos(\lambda L_x)$, $\cos k_y = \cos(\lambda L_y)$. The point $\lambda = 0$ belongs to the set $\Sigma_x^* \cup \Sigma_y^* \cup \Sigma_z(\beta)$ if and only if $\beta = 0$. At the same time, according to Proposition 5.2.2, $0 \in \sigma(A(\beta))$ if and only if $\beta = 0$.

2. We will prove the existence of a gap for the operator $A(\beta)$ in a (deleted) neighbourhood of each point of the set $\{\lambda^2, \lambda \in \widetilde{\Sigma}_z^*(\beta)\}$. More precisely, we will show that for any $\lambda \in \widetilde{\Sigma}_z^*(\beta)$ there exist positive numbers l^+ and l^- such that $]\lambda^2 - l^-, \lambda^2[\cap \sigma(A(\beta)) = \emptyset$ and $]\lambda^2, \lambda^2 + l^+[\cap \sigma(A(\beta)) = \emptyset$. Let $\lambda_0 \in \widetilde{\Sigma}_z^*(\beta)$. Then,

$$\frac{\cos\beta}{\cos\left(\lambda_0 L_z\right)} < 1. \tag{5.2.29}$$

Let us rewrite the relation (5.2.4) in a neighbourhood of λ_0 by putting $\lambda = \lambda_0 + \delta$:

$$\sin(\delta L_z) \left(\sin(\lambda L_x + \lambda L_y) - \cos k_x \sin(\lambda L_y) - \cos k_y \sin(\lambda L_x) \right) \\ = \sin(\lambda L_x) \sin(\lambda L_y) \left(\frac{\cos\beta}{\cos(\lambda_0 L_z)} - \cos(\delta L_z) \right). \quad (5.2.30)$$

- (i) $\lambda_0 \notin \Sigma_x \cup \Sigma_y$: the equation (5.2.30) has no solution for δ small enough. Indeed, due to the inequality (5.2.29), its right-hand side tends to a non-zero limit as $\delta \to 0$, whereas its left-hand side tends to zero uniformly in $(k_x, k_y) \in [0, \pi]^2$. This proves the existence of a gap of the operator $A(\beta)$ containing the point λ_0^2 .
- (ii) $\lambda_0 \in \Sigma_x^* \setminus \Sigma_y^*$ or $\lambda_0 \in \Sigma_y^* \setminus \Sigma_x^*$: we will consider the case $\lambda_0 \in \Sigma_x^* \setminus \Sigma_y^*$, the case $\lambda_0 \in \Sigma_y^* \setminus \Sigma_x^*$ can be considered in a similar way. For $\delta \neq 0$ small enough the relation (5.2.30) can be rewritten as

$$\sin(\delta L_x) \frac{\cos(\lambda L_y) - \cos k_y}{\sin(\lambda L_y)} + \cos(\delta L_x) - \frac{\cos k_x}{\cos(\lambda_0 L_x)} = \frac{\sin(\delta L_x)}{\sin(\delta L_z)} \left(\frac{\cos\beta}{\cos(\lambda_0 L_z)} - \cos(\delta L_z)\right). \quad (5.2.31)$$

Taking into account that $|\cos(\lambda_0 L_x)| = 1$, when δ is small enough, the left-hand side of this relation can be bounded from below uniformly in $(k_x, k_y) \in [0, \pi]^2$ by a continuous function in δ that tends to zero as $\delta \to 0$:

$$\sin\left(\delta L_x\right) \frac{\cos\left(\lambda L_y\right) - \cos k_y}{\sin\left(\lambda L_y\right)} + \cos\left(\delta L_x\right) - \frac{\cos k_x}{\cos\left(\lambda_0 L_x\right)}$$
$$\geqslant \cos\left(\delta L_x\right) - 1 - \sin\left(\delta L_x\right) \frac{|\cos\left(\lambda L_y\right)| + 1}{|\sin\left(\lambda L_y\right)|} \xrightarrow{\delta \to 0} 0, \qquad \forall (k_x, k_y) \in [0, \pi]^2.$$

The limit of the right-hand side of (5.2.31) when $\delta \to 0$ is

$$\frac{\sin\left(\delta L_x\right)}{\sin\left(\delta L_z\right)} \left(\frac{\cos\beta}{\cos\left(\lambda_0 L_z\right)} - \cos\left(\delta L_z\right)\right) \xrightarrow[\delta \to 0]{} \frac{L_x}{L_z} \left(\frac{\cos\beta}{\cos\left(\lambda_0 L_z\right)} - 1\right) < 0,$$

which is strictly negative due to the inequality (5.2.29). Hence, the equation (5.2.31) has no solution for $\delta \neq 0$ small enough.

(iii) $\lambda_0 \in \Sigma_x^* \cap \Sigma_y^*$: the relation (5.2.30) can be rewritten as

$$\frac{\sin(\delta L_z)}{\sin(\delta L_y)} \left(\cos(\delta L_y) - \frac{\cos k_y}{\cos(\lambda_0 L_y)} \right) + \frac{\sin(\delta L_z)}{\sin(\delta L_x)} \left(\cos(\delta L_x) - \frac{\cos k_x}{\cos(\lambda_0 L_x)} \right) \\ = \frac{\cos\beta}{\cos(\lambda_0 L_z)} - \cos(\delta L_z). \quad (5.2.32)$$

Again, since $|\cos(\lambda_0 L_x)| = |\cos(\lambda_0 L_y)| = 1$, when δ is small enough, the left-hand side of this relation can be bounded from below uniformly in $(k_x, k_y) \in [0, \pi]^2$ by a continuous function in δ that tends to zero as $\delta \to 0$:

$$\frac{\sin\left(\delta L_z\right)}{\sin\left(\delta L_y\right)} \left(\cos\left(\delta L_y\right) - \frac{\cos k_y}{\cos\left(\lambda_0 L_y\right)}\right) + \frac{\sin\left(\delta L_z\right)}{\sin\left(\delta L_x\right)} \left(\cos\left(\delta L_x\right) - \frac{\cos k_x}{\cos\left(\lambda_0 L_x\right)}\right)$$
$$\geqslant \frac{L_z}{L_y} \left(\cos\left(\delta L_y\right) - 1\right) + \frac{L_z}{L_x} \left(\cos\left(\delta L_x\right) - 1\right) \xrightarrow{\delta \to 0} 0, \qquad \forall (k_x, k_y) \in [0, \pi]^2.$$

The limit of the right-hand side of (5.2.32) as $\delta \to 0$ is $\cos \beta / \cos (\lambda_0 L_z) - 1 < 0$. Consequently, the equation (5.2.32) has no solution for δ small enough.

The following assertion gives a description of the set of eigenvalues of infinite multiplicity of the operator $A(\beta)$. We give its proof in Appendix.

Proposition 5.2.4. The operator $A(\beta)$ has the following set of eigenvalues of infinite multiplicity:

$$\sigma_{pp}(A(\beta)) = \left\{ \lambda^2 : \ \lambda \in \Sigma_{is}(\beta) \cup \Sigma_{emb}(\beta) \right\}$$

where

$$\Sigma_{is}(\beta) = \left(\Sigma_x^* \cup \Sigma_y^*\right) \cap \Sigma_z^*(\beta), \qquad (5.2.33)$$
$$\Sigma_{emb}(\beta) = \begin{cases} \left(\Sigma_x^* \cap \Sigma_y^*\right) \setminus \widetilde{\Sigma}_z^*(\beta), & \beta \in]0, \pi[, \\ \left(\left(\Sigma_x^* \cap \Sigma_y^*\right) \setminus \widetilde{\Sigma}_z^*(\beta)\right) \cup \Sigma_z^*(\beta), & \beta \in \{0, \pi\}. \end{cases}$$

The eigenvalues of the set $\{\lambda^2 : \lambda \in \Sigma_{is}(\beta)\}\$ are isolated points of the spectrum whereas the eigenvalues of the set $\{\lambda^2 : \lambda \in \Sigma_{emb}(\beta)\}\$ are not isolated points of the spectrum.

The following proposition gives a criterion for λ^2 to be a point of the spectrum of the operator $A(\beta)$.

Proposition 5.2.5. $\lambda^2 \in \sigma(A(\beta))$ if and only if one of the following possibilities holds:

(i) $\lambda \in \Sigma_x^* \cup \Sigma_y^*$;

(ii) $\lambda \in \mathbb{R}_+ \setminus (\Sigma_x^* \cup \Sigma_y^*)$ is a solution of the equation

$$\phi_{L_z,\beta}(\lambda) = f_{L_x,L_y,k_x,k_y}(\lambda). \tag{5.2.34}$$

for some $(k_x, k_y) \in [0, \pi]^2$. Here

$$f_{L_x,L_y,k_x,k_y}(\lambda) = \begin{cases} f_{L_x,k_x}(\lambda) + f_{L_y,k_y}(\lambda), & \lambda \in \mathbb{R}^*_+ \setminus \left(\Sigma^*_x \cup \Sigma^*_y\right), \\ 0, & (k_x,k_y,\lambda) = (0,0,0), \end{cases}$$
(5.2.35)

$$f_{L,k}(\lambda) = \frac{\cos k - \cos \left(\lambda L\right)}{\sin \left(\lambda L\right)}, \qquad \sin \left(\lambda L\right) \neq 0, \tag{5.2.36}$$

$$\phi_{L_z,\beta}(\lambda) = \begin{cases} \frac{\cos\left(\lambda L_z\right) - \cos\beta}{\sin\left(\lambda L_z\right)}, & \lambda \in \mathbb{R}^+ \setminus \Sigma_z, \\ 0, & \lambda \in \Sigma_z(\beta). \end{cases}$$
(5.2.37)

Proof. This is a consequence of the decomposition (5.2.26), Proposition 5.2.2 and property 1 of Proposition 5.2.3. Indeed, the equation (5.2.34) is equivalent to the equation (5.2.4) for $\lambda \notin \Sigma$. The case $\lambda \in \Sigma_x^* \cup \Sigma_y^*$ is treated directly: the corresponding points belong to the spectrum of the operator $A(\beta)$ for any $\beta \in [0, \pi]$. If $\lambda \in \Sigma_z \setminus (\Sigma_x^* \cup \Sigma_y^*)$, then $\lambda^2 \in \sigma(A(\beta))$ if and only if $\lambda \in \Sigma_z(\beta)$ (and hence, $\beta \in \{0, \pi\}$). Notice that this is also valid for $\lambda = 0$. As well, if $\lambda \in \Sigma_z$, then the function $\phi_{L_z,\beta}$ is defined at λ if and only if $\lambda \in \Sigma_z(\beta)$, and its value is 0. In this case, the equation (5.2.34) is satisfied for $\cos k_x = \cos(\lambda L_x), \cos k_y = \cos(\lambda L_y)$.

Remark 5.2.3. Proposition 5.2.5 is an analogue of Proposition 4.1.5 obtained in the 2D case. However, we use here the inverses of the functions $\phi_{L,\beta}$, f_{θ} used in the 2D case. This choice is done in order to have f_{L_x,L_y,k_x,k_y} given by a sum $f_{L_x,L_y,k_x,k_y} = f_{L_x,k_x} + f_{L_y,k_y}$ and not by a some of inverses of functions. Indeed, this form will be easier to represent geometrically.

As in the 2D case two types of gaps can be distinguished for the operator $A(\beta)$:

Proposition 5.2.6. An interval]a, b[is a gap of the operator $A(\beta)$ for $\beta \in [0, \pi]$ if and only if $0 \leq a < b$, $[\sqrt{a}, \sqrt{b}] \cap \Sigma_z(\beta) = \emptyset$ and one of the following possibilities holds:

$$I \{ \sqrt{a}, \sqrt{b} \} \cap \left(\Sigma_x \cup \Sigma_y \cup \widetilde{\Sigma}_z(\beta) \right) = \emptyset, \quad \phi_{L_z,\beta} \left(\sqrt{a} \right) = f^-_{L_x,L_y} \left(\sqrt{a} \right), \quad \phi_{L_z,\beta}(\sqrt{b}) = f^+_{L_x,L_y}(\sqrt{b});$$

$$II \quad (i) \quad \sqrt{a} \in \left(\Sigma_x^* \cup \Sigma_y^* \right) \setminus \widetilde{\Sigma}_z^*(\beta), \quad \phi_{L_z,\beta}(\sqrt{a}) > f^+_{L_x,L_y}(\sqrt{a}), \quad \sqrt{b} \notin \Sigma_x \cup \Sigma_y \cup \widetilde{\Sigma}_z(\beta), \quad \phi_{L_z,\beta}(\sqrt{b}) = f^+_{L_x,L_y}(\sqrt{b});$$

$$(i') \quad \sqrt{a} \in \left(\Sigma_x \cup \Sigma_y \right) \cap \widetilde{\Sigma}_z(\beta), \quad \sqrt{b} \notin \Sigma_x \cup \Sigma_y \cup \widetilde{\Sigma}_z(\beta), \quad \phi_{L_z,\beta}(\sqrt{b}) = f^+_{L_x,L_y}(\sqrt{b});$$

$$(ii) \quad \sqrt{a} \notin \Sigma_x \cup \Sigma_y \cup \widetilde{\Sigma}_z(\beta), \quad \phi_{L_z,\beta}(\sqrt{a}) = f^-_{L_x,L_y}(\sqrt{a}), \quad \sqrt{b} \in \left(\Sigma_x^* \cup \Sigma_y^* \right) \setminus \widetilde{\Sigma}_z^*(\beta), \quad \phi_{L_z,\beta}(\sqrt{a}) = f^-_{L_x,L_y}(\sqrt{b});$$

$$(ii') \quad \sqrt{a} \notin \Sigma_x \cup \Sigma_y \cup \widetilde{\Sigma}_z(\beta), \quad \phi_{L_z,\beta}(\sqrt{a}) = f^-_{L_x,L_y}(\sqrt{a}), \quad \sqrt{b} \in \left(\Sigma_x^* \cup \Sigma_y^* \right) \cap \widetilde{\Sigma}_z^*(\beta).$$

Remark 5.2.4. For $\beta \in [0, \pi]$, the bottom of the spectrum of the operator $\sigma_* = \min \sigma(A(\beta))$ is strictly positive. As in the 2D case (cf. Remark 4.1.3) we will call the interval $]0, \sigma_*[$ a gap even if it is not quite correct. As follows from Proposition 5.2.6, it is a gap of type II (it satisfies the condition II(i')).

The rest of this section is devoted to the proof of this Proposition 5.2.6. We start with a geometric interpretation of Proposition 5.2.5 which is similar to the one given in the 2D case. Now we have to consider the abscissas of the intersections of the graph of the function $\phi_{L_{z},\beta}$ with the ones of the functions $f_{L_{x},L_{y},k_{x},k_{y}}$. Let us introduce the union of the graphs of the functions $f_{L_{x},L_{y},k_{x},k_{y}}$ for $(k_{x},k_{y}) \in [0,\pi]^{2}$:

$$D_{L_x,L_y} = \bigcup_{(k_x,k_y)\in[0,\pi]^2} \left\{ \left(\lambda, f_{L_x,L_y,k_x,k_y}(\lambda)\right) / \lambda \in \mathcal{D}(f_{L_x,L_y,k_x,k_y}) \right\},\$$

where $\mathcal{D}(f)$ stands for the domain of the function f.

Lemma 5.2.1.

$$D_{L_x,L_y} = \left\{ (\lambda, p) / \lambda \in \mathbb{R}^*_+ \setminus \left(\Sigma^*_x \cup \Sigma^*_y \right), \ p \in \left[f^-_{L_x,L_y}(\lambda), f^+_{L_x,L_y}(\lambda) \right] \right\} \cup \left\{ (0,0) \right\},$$

where

,

$$f_{L_x,L_y}^+(\lambda) = f^+(\lambda L_x) + f^+(\lambda L_y), \qquad f_{L_x,L_y}^-(\lambda) = f^-(\lambda L_x) + f^-(\lambda L_y), \qquad (5.2.38)$$

$$f^+(\lambda) = \tan\left(\frac{\lambda - \pi n}{2}\right), \qquad \lambda \in [\pi n, \pi(n+1)], \qquad n \in \mathbb{N}, \qquad (5.2.39)$$

$$f^{-}(\lambda) = -\cot\left(\frac{\lambda - \pi n}{2}\right), \qquad \lambda \in \left[\pi n, \pi(n+1)\right], \qquad n \in \mathbb{N}.$$
(5.2.40)

Proof. If $\sin(\lambda L) \neq 0$, then the function $f_{L,k}$ defined in (5.2.36) is continuous in $k_x \in [0, \pi]$ and

$$\bigcup_{k_x \in [0,\pi]} f_{L,k}(\lambda) = \left[f^-(\lambda L), f^+(\lambda L) \right].$$

Hence, for $\lambda \notin \Sigma_x \cup \Sigma_y$ one has

$$\bigcup_{(k_x,k_y)\in[0,\pi]^2} f_{L_x,L_y,k_x,k_y}(\lambda) = \bigcup_{(k_x,k_y)\in[0,\pi]^2} \left(f_{L_x,k_x}(\lambda) + f_{L_y,k_y}(\lambda) \right) = \left[f^-_{L_x,L_y}(\lambda), f^+_{L_x,L_y}(\lambda) \right].$$

Finally, $0 \in \mathcal{D}(f_{L_x,L_y,k_x,k_y})$ if and only if $(k_x,k_y) = (0,0)$. In this case, $f_{L_x,L_y,0,0}(0) = 0$. \Box

The set D_{L_x,L_y} is shown in figure 5.4. Let us mention that the functions f_{L_x,L_y}^{\pm} are continuous in $\mathbb{R}^*_+ \setminus (\Sigma_x \cup \Sigma_y)$ and strictly increasing in each interval of continuity (between two neighbour points of the set $\Sigma_x \cup \Sigma_y$). Moreover, the function f_{L_x,L_y}^+ is right-continuous and the function f_{L_x,L_y}^- is left-continuous and

$$f_{L_x,L_y}^-(\lambda) \ge 0, \quad \forall \lambda \in \mathbb{R}_+, \qquad f_{L_x,L_y}^+(\lambda) > 0, \quad \forall \lambda \in \mathbb{R}_+^* \setminus \left(\Sigma_x^* \cup \Sigma_y^*\right), \tag{5.2.41}$$

$$f_{L_x,L_y}^-(\lambda) \leqslant 0, \quad \forall \lambda \in \mathbb{R}^*_+, \qquad f_{L_x,L_y}^+(\lambda) < 0, \quad \forall \lambda \in \mathbb{R}^*_+ \setminus \left(\Sigma_x^* \cup \Sigma_y^*\right). \tag{5.2.42}$$

In order to get a geometric interpretation of the spectrum of the operator $A(\beta)$, we need to consider the abscissas of the intersections of the set D_{L_x,L_y} with the graph of the function $\phi_{L_{z,\beta}}$ (cf. figure 5.5). This will correspond to the points described in (*ii*) of Proposition 5.2.5. In order to get all the points λ such that $\lambda^2 \in \sigma(A(\beta))$, we have to include the set $\Sigma_x^* \cup \Sigma_y^*$. Thus, the spectrum of the operator $A(\beta)$ can be characterized as follows:

$$\sigma(A(\beta)) = \left\{ \lambda^2 / \ \lambda \ge 0, \ \phi_{L_z,\beta}(\lambda) \in D_{L_x,L_y} \right\} \cup \left\{ \lambda^2 / \ \lambda \in \Sigma_x^* \cup \Sigma_y^* \right\}.$$

This permits to give the following description of the gaps of the operator $A(\beta)$.

Lemma 5.2.2. The point λ_0^2 is the lower end of a gap of the operator $A(\beta)$ if and only if one of the following possibilities holds:

- (i) $\lambda_0 \in \mathbb{R}^*_+ \setminus \widetilde{\Sigma}^*_z(\beta)$ and $\phi_{L_z,\beta}(\lambda_0) = f^-_{L_x,L_y}(\lambda_0) < 0;$
- (*ii*) $\lambda_0 \in \left(\Sigma_x^* \cup \Sigma_y^*\right) \setminus \widetilde{\Sigma}_z^*(\beta)$ and $\phi_{L_z,\beta}(\lambda_0) > f_{L_x,L_y}^+(\lambda_0) \ge 0;$
- (*iii*) $\lambda_0 \in (\Sigma_x \cup \Sigma_y) \cap \widetilde{\Sigma}_z(\beta).$

Similarly, the point λ_0^2 is the upper end of a gap of the operator $A(\beta)$ if and only if one of the following possibilities holds:



Figure 5.4: The set D_{L_x,L_y} . The vertical lines correspond to the points of the set $\Sigma_x \cup \Sigma_y$.

(iv) $\lambda_0 \in \mathbb{R}^*_+ \setminus \widetilde{\Sigma}^*_z(\beta)$ and $\phi_{L_z,\beta}(\lambda_0) = f^+_{L_x,L_y}(\lambda_0) > 0;$ (v) $\lambda_0 \in (\Sigma^*_x \cup \Sigma^*_y) \setminus \widetilde{\Sigma}^*_z(\beta)$ and $\phi_{L_z,\beta}(\lambda_0) < f^-_{L_x,L_y}(\lambda_0) \leq 0;$ (vi) $\lambda_0 \in (\Sigma^*_x \cup \Sigma^*_y) \cap \widetilde{\Sigma}^*_z(\beta).$

Proof. The proof is very similar to the one of Lemme 1.3.2. We repeat it with appropriate modifications for the sake of completeness. Let λ_- , λ_+ be two neighbour points of the set $\Sigma_x \cup \Sigma_y$. We will study the possible configurations of the intersections of the function $\phi_{L_z,\beta}$ with the domain D_{L_x,L_y} in the interval $[\lambda_-, \lambda_+]$. Notice that $\phi_{L_z,\beta}$ is a $2\pi/L_z$ -periodic function defined on $\mathbb{R}^+ \setminus \widetilde{\Sigma}_z(\beta)$. In each interval of continuity it is strictly decreasing and takes all the values in \mathbb{R} . We will consider separately the internal points of the interval $[\lambda_-, \lambda_+]$ and its ends.

- Case 1. Internal points points: $\lambda_0 \in]\lambda_-, \lambda_+[$
- (a) $\lambda_0 \notin \widetilde{\Sigma}_z^*(\beta)$ and the point $(\lambda_0, \phi_{L_z,\beta}(\lambda_0))$ is an interior point of the domain D_{L_x,L_y} :

$$(\lambda_0, \phi_{L_z,\beta}(\lambda_0)) \in \operatorname{int} \left(D_{L_x,L_y} \right) \implies \exists \delta > 0 \quad \text{s.t.} \quad (\lambda, \phi_{L_z,\beta}(\lambda)) \in D_{L_x,L_y}, \quad \forall \lambda : |\lambda - \lambda_0| < \delta.$$

$$(5.2.43)$$



Figure 5.5: Example: $L_x = 1.44$, $L_y = 0.9$, $L_z = 1.3$, $\beta = 0.7$. The images of the spectral bands of the operator $A(\beta)$ by the function $x \mapsto \sqrt{x}$ are given by the projections on the axis of abscissas of the intersections of the set D_{L_x,L_y} with the graph of the function $\phi_{L_z,\beta}$.

Hence, the points such that $(\lambda_0, \phi_{L_z,\beta}(\lambda_0)) \in \operatorname{int}(D_{L_x,L_y})$ correspond to the interior points of the spectrum (and not to the ends of gaps).

(b) $\lambda_0 \notin \widetilde{\Sigma}_z^*(\beta)$ and $\phi_{L_z,\beta}(\lambda_0) = f_{L_x,L_y}^+(\lambda_0)$: (In this case, $f_{L_x,L_y}^+(\lambda_0) > 0$ due to (5.2.41)).

$$\lambda_{0} \in]\lambda_{-}, \lambda_{+}[\qquad (\lambda, \phi_{L_{z},\beta}(\lambda)) \notin D_{L_{x},L_{y}}, \quad \forall \lambda \in]\lambda_{0} - \delta, \lambda_{0}[, \\ \phi_{L_{z},\beta}(\lambda_{0}) = f_{L_{x},L_{y}}^{+}(\lambda_{0}) \quad \Rightarrow \quad \exists \delta > 0 \quad \text{s.t.} \quad (\lambda, \phi_{L_{z},\beta}(\lambda)) \in D_{L_{x},L_{y}}, \quad \forall \lambda \in [\lambda_{0}, \lambda_{0} + \delta[. \\ (5.2.44)$$

(c) $\lambda_0 \notin \widetilde{\Sigma}_z^*(\beta)$ and $\phi_{L_z,\beta}(\lambda_0) = f_{L_x,L_y}^-(\lambda_0)$: (In this case, $f_{L_x,L_y}^-(\lambda_0) < 0$ due to (5.2.42)).

$$\begin{aligned} \lambda_{0} \in]\lambda_{-}, \lambda_{+}[& (\lambda, \phi_{L_{z},\beta}(\lambda)) \notin D_{L_{x},L_{y}}, \quad \forall \lambda \in]\lambda_{0}, \lambda_{0} + \delta[, \\ \phi_{L_{z},\beta}(\lambda_{0}) = f_{L_{x},L_{y}}^{-}(\lambda_{0}) & \Rightarrow \quad \exists \delta > 0 \quad \text{s.t.} \quad \begin{aligned} & (\lambda, \phi_{L_{z},\beta}(\lambda)) \notin D_{L_{x},L_{y}}, \quad \forall \lambda \in [\lambda_{0} - \delta, \lambda_{0}[. \\ (5.2.45) \end{aligned}$$

This follows from the fact that the functions f_{L_x,L_y}^{\pm} defining the boundaries of the domain D_{L_x,L_y} are strictly increasing and the function $\phi_{L_z,\beta}$ is strictly decreasing. The relation (5.2.44) means that λ_0^2 is the lower end of a gap and the relation (5.2.45) means that λ_0^2 is the upper end of a gap.

(d) Points of discontinuity of the function $\phi_{L_z,\beta}$:

$$\begin{split} \lambda_0 \in \widetilde{\Sigma}_z^*(\beta) \cap]\lambda_-, \lambda_+[&\Rightarrow \exists \delta > 0 \text{ s.t. } (\lambda, \phi_{L_z,\beta}(\lambda)) \notin D_{L_x,L_y}, \quad \forall \lambda : 0 < |\lambda - \lambda_0| < \delta. \\ (5.2.46) \end{split}$$
This follows from the continuity of the functions f_{L_x,L_y}^{\pm} inside the interval $]\lambda_-, \lambda_+[$ and the behaviour of the function $\phi_{L_z,\beta}$ in a neighbourhood of its points of discontinuity.

We see that the interior points of the interval $[\lambda_{-}, \lambda_{+}]$ are ends of a gap only in the cases (b) and (c).

• Case 2. Ends of the interval: $\lambda_0 \in \Sigma_x^* \cup \Sigma_y^*$ (In this case, $\lambda_0 \in \sigma(A(\beta))$ according to Proposition 5.2.5).

$$\lambda_{0} \in \left(\Sigma_{x}^{*} \cup \Sigma_{y}^{*}\right) \setminus \widetilde{\Sigma}_{z}^{*}(\beta) \qquad \Rightarrow \quad \exists \delta > 0 \quad \text{s.t.} \quad \begin{array}{c} \left(\lambda, \phi_{L_{z},\beta}(\lambda)\right) \notin D_{L_{x},L_{y}}, \quad \forall \lambda \in]\lambda_{0}, \lambda_{0} + \delta[, \\ \left(\lambda, \phi_{L_{z},\beta}(\lambda)\right) > f_{L_{x},L_{y}}^{+}(\lambda_{0}) \qquad \Rightarrow \quad \exists \delta > 0 \quad \text{s.t.} \quad \begin{array}{c} \left(\lambda, \phi_{L_{z},\beta}(\lambda)\right) \notin D_{L_{x},L_{y}}, \quad \forall \lambda \in]\lambda_{0} - \delta, \lambda_{0}[. \\ \left(\lambda, \phi_{L_{z},\beta}(\lambda)\right) \in D_{L_{x},L_{y}}, \quad \forall \lambda \in]\lambda_{0} - \delta, \lambda_{0}[. \\ \left(5.2.47\right) \end{bmatrix} \right)$$

$$\lambda_{0} \in \left(\Sigma_{x}^{*} \cup \Sigma_{y}^{*}\right) \setminus \widetilde{\Sigma}_{z}^{*}(\beta) \qquad \Rightarrow \quad \exists \delta > 0 \quad \text{s.t.} \quad \begin{array}{l} \left(\lambda, \phi_{L_{z},\beta}(\lambda)\right) \notin D_{L_{x},L_{y}}, \quad \forall \lambda \in]\lambda_{0} - \delta, \lambda_{0}[, \\ \left(\lambda, \phi_{L_{z},\beta}(\lambda)\right) < f_{L_{x},L_{y}}^{-}, \quad \forall \lambda \in]\lambda_{0}, \lambda_{0} + \delta[. \\ \left(\lambda, \phi_{L_{z},\beta}(\lambda)\right) \in D_{L_{x},L_{y}}, \quad \forall \lambda \in]\lambda_{0}, \lambda_{0} + \delta[. \\ \left(5.2.48\right) \\ \end{array}$$

$$\begin{aligned} \lambda_0 &\in \left(\Sigma_x^* \cup \Sigma_y^*\right) \setminus \Sigma_z^*(\beta) \\ \phi_{L_z,\beta}(\lambda_0) &\in \left[f_{L_x,L_y}^-(\lambda_0), f_{L_x,L_y}^+(\lambda_0)\right] \\ \forall \lambda : 0 < |\lambda - \lambda_0| < \delta. \end{aligned} \tag{5.2.49}$$

$$\lambda_0 \in \left(\Sigma_x^* \cup \Sigma_y^*\right) \cap \widetilde{\Sigma}_z^*(\beta) \implies \exists \delta > 0 \text{ s.t. } (\lambda, \phi_{L_z,\beta}(\lambda)) \notin D_{L_x,L_y}, \quad \forall \lambda : 0 < |\lambda - \lambda_0| < \delta.$$
(5.2.50)

Thus, the cases (5.2.47), (5.2.48), (5.2.50) correspond to ends of a gap. The case (5.2.50) corresponds to an isolated point of the spectrum which is an eigenvalue of infinite multiplicity (cf. Proposition 5.2.4).

- Case 3. $\lambda_0 = 0$
- (a) $\beta = 0$: in this case $0 \in \sigma(A(\beta))$ and $\phi_{L_z,0}(0) = 0$. Consequently, $\phi_{L_z,0}(0) \in [f_{L_x,L_y}^-(0), f_{L_x,L_y}^+(0)]$, and similarly to (5.2.49) there exists $\delta > 0$ such that $(\lambda, \phi_{L_z,\beta}(\lambda)) \in D_{L_x,L_y}$, $\forall \lambda : 0 < \lambda \lambda_0 < \delta$. Thus, 0 is not an end of a gap. It is not included in any of the conditions (i)-(vi) (it is not included in (iii) since $0 \notin \widetilde{\Sigma}_z(0)$).
- (b) $\beta \in]0,\pi]$: in this case $0 \notin \sigma(A(\beta))$ and we call it the lower end of the gap $]0,\sigma_*[$, $\sigma_* = \min \sigma(A(\beta))$. For this reason it is included in the case (*iii*) (since $0 \in \widetilde{\Sigma}_z(\beta)$ for $\beta \in]0,\pi]$).

Corollary 5.2.1. If $\lambda_0 \in \Sigma_z^*(\beta)$, then λ_0^2 is an internal point of $\sigma(A(\beta))$.

Proof. It follows from Proposition 5.2.3 that $\lambda_0^2 \in \sigma(A(\beta))$. Let us show that it is not an end of a gap. Indeed, $\lambda_0 \in \Sigma_z^*(\beta)$ implies that $\phi_{L_z,\beta}(\lambda_0) = 0$. This is not compatible with any of the possibilities (i), (ii), (iv), (v) of Lemma 5.2.2. Since $\Sigma_z^*(\beta) \cap \widetilde{\Sigma}_z(\beta) = \emptyset$, it is not compatible with (*iii*) and (vi) neither.

Proof of Proposition 5.2.6.

(a) Suppose first that]a, b[is a gap of the operator $A(\beta)$. Then, one necessarily has $0 \leq a < b$, and it follows from Corollary 5.2.1 that $[\sqrt{a}, \sqrt{b}] \cap \Sigma_z^*(\beta) = \emptyset$. Moreover, $0 \in \Sigma_z(\beta)$ if and only if $\beta = 0$. Consequently, if $\beta \in]0, \pi]$, then $[\sqrt{a}, \sqrt{b}] \cap \Sigma_z(\beta) = \emptyset$. For $\beta = 0$, we have $a \neq 0$ since 0 is not the lower end of a gap, cf. Lemma 5.2.2. So, we have $[\sqrt{a}, \sqrt{b}] \cap \Sigma_z(\beta) = \emptyset$ again.

In other words,

$$\phi_{L_z,\beta}(\lambda) \neq 0, \qquad \forall \lambda \in [\sqrt{a}, \sqrt{b}] \setminus \widetilde{\Sigma}_z(\beta).$$
 (5.2.51)

I Suppose that $\{\sqrt{a}, \sqrt{b}\} \cap \left(\Sigma_x \cup \Sigma_y \cup \widetilde{\Sigma}_z(\beta)\right) = \emptyset$. Then, due to Lemma 5.2.2 we have $\phi_{L_z,\beta}(\sqrt{a}) = f^-_{L_x,L_y}(\sqrt{a}), \ \phi_{L_z,\beta}(\sqrt{b}) = f^+_{L_x,L_y}(\sqrt{b}).$

Π

(*i-i'*) Suppose now that $\sqrt{a} \in \Sigma_x \cup \Sigma_y$. It follows from Lemma 5.2.2 that either $\sqrt{a} \in \widetilde{\Sigma}_z(\beta)$ (case (*i'*)) or $\sqrt{a} \in (\Sigma_x^* \cup \Sigma_y^*) \setminus \widetilde{\Sigma}_z^*(\beta)$ and $\phi_{L_z,\beta}(\sqrt{a}) > f_{L_x,L_y}^+(\sqrt{a}) \ge 0$ (case (*i*)). In both cases there exists $\delta > 0$ such that $\phi_{L_z,\beta}(\lambda) > 0$ for $\lambda \in]\sqrt{a}, \sqrt{a} + \delta[$. This implies that

$$]\sqrt{a},\sqrt{b}] \cap \widetilde{\Sigma}_{z}(\beta) = \emptyset$$
 and $\phi_{L_{z},\beta}(\lambda) > 0, \quad \forall \lambda \in]\sqrt{a},\sqrt{b}]$

Indeed, in the opposite case the function $\phi_{L_z,\beta}$ is not continuous in $]\sqrt{a}, \sqrt{b}]$ (if there exists a point $\lambda_0 \in]\sqrt{a}, \sqrt{b}]$ such that $\phi_{L_z,\beta}(\lambda_0) < 0$ and $\phi_{L_z,\beta}$ is continuous in $]\sqrt{a}, \sqrt{b}]$, then $]\sqrt{a}, \sqrt{b}] \cap \Sigma_z(\beta) \notin \emptyset$, which, as was shown above, is impossible). Let λ' be the closest to \sqrt{a} point of discontinuity of $\phi_{L_z,\beta}$ in $]\sqrt{a}, \sqrt{b}]$: $\lambda' \in]\sqrt{a}, \sqrt{b}] \cap$ $\widetilde{\Sigma}_z(\beta)$. Taking into account that $\lim_{\lambda \to \lambda' = 0} \phi_{L_z,\beta}(\lambda) = -\infty$, we conclude that there exists a point $\lambda'' \in]\sqrt{a}, \lambda'[$ such that $\phi_{L_z,\beta}(\lambda'') = 0$, which contradicts (5.2.51). This shows that $\sqrt{b} \notin \widetilde{\Sigma}_z(\beta)$ and $\phi_{L_z,\beta}(\sqrt{b}) > 0$, which, due to Lemma 5.2.2, implies that $\sqrt{b} \notin \Sigma_x \cup \Sigma_y$ and $\phi_{L_z,\beta}(\sqrt{b}) = f_{L_x,L_y}^+(\sqrt{b})$.

- (*ii-ii*) The case $\sqrt{b} \in \Sigma_x^* \cup \Sigma_y^*$ can be considered analogously to the previous case.
- (b) Suppose now that $0 \leq a < b$ and $[a, b] \cap \Sigma_z(\beta) = \emptyset$. We will show that if one of the conditions *I*, *II* holds, then [a, b] is a gap of the operator $A(\beta)$.
 - I It follows from Lemma 5.2.2 that a is the lower end of a gap and b is the upper end of a gap. Suppose that there exists $c \in]a, b[$ such that $c \in \sigma(A(\beta))$. Then, Proposition 5.2.5 implies that $\phi_{L_z,\beta}(\sqrt{c}) \in [f_{L_x,L_y}^-(\sqrt{c}), f_{L_x,L_y}^+(\sqrt{c})]$. The functions f_{L_x,L_y}^\pm being strictly increasing in $[\sqrt{a}, \sqrt{b}]$, we get $\phi_{L_z,\beta}(\sqrt{a}) < \phi_{L_z,\beta}(\sqrt{c}) < \phi_{L_z,\beta}(\sqrt{b})$. The function $\phi_{L_z,\beta}$ is, in turn, strictly decreasing in its intervals of continuity. Hence, there exist $\lambda' \in]\sqrt{a}, \sqrt{c}[$ and $\lambda'' \in]\sqrt{c}, \sqrt{b}[$ such that $\lambda', \lambda'' \in \widetilde{\Sigma}_z(\beta)$. Since in each interval of continuity the function $\phi_{L_z,\beta}$ takes all values in \mathbb{R} , there exists $\lambda_0 \in]\lambda', \lambda''[$ such that $\phi_{L_z,\beta}(\lambda_0) = 0$ which contradicts the assumption $[a,b] \cap \Sigma_z(\beta) = \emptyset$.

\mathbf{II}

- (*i*-*i*') It follows from Lemma 5.2.2 that *b* is the upper end of a gap. Let us denote by *a'* its lower end. Since $a \in \sigma(A(\beta))$ (cf. Proposition 5.2.5), one necessarily has $a \leq a'$. Suppose that a < a'. Then, it follows from Lemma 5.2.2 that $\sqrt{a'} \notin \widetilde{\Sigma}_z(\beta)$ and $\phi_{L_z,\beta}(\sqrt{a'}) = f_{L_x,L_y}(\sqrt{a'}) < 0$. On the other hand, both in the case (*i*) and in the case (*i'*) there exists $\delta > 0$ such that $\phi_{L_z,\beta}(\lambda) > 0$ for $\lambda \in]\sqrt{a}, \sqrt{a} + \delta[$. Consequently, there exists $\lambda' \in]\sqrt{a}, \sqrt{a'}[$ such that $\phi_{L_z,\beta}(\lambda') = 0$, which contradicts the assumption $[a, b] \cap \Sigma_z(\beta) = \emptyset$.
- (*ii-ii*') This case can be considered analogously to the previous one.

Let us mention another possible characterization of the two types of gaps. It will be used in the next section while discussing the number of eigenvalues of the perturbed operator in the gaps. **Lemma 5.2.3.** Let]a, b[be a gap of the operator $A(\beta)$. Then, it is a gap of type I if and only if there exists exactly one point c such that $c \in]a, b[$ such that $\sqrt{c} \in \widetilde{\Sigma}_z(\beta)$. It is a gap of type II if and only if $]\sqrt{a}, \sqrt{b}[\cap \widetilde{\Sigma}_z(\beta) = \emptyset$.

Proof. First, let us mention that the intersection $]\sqrt{a}, \sqrt{b}[\cap \tilde{\Sigma}_z(\beta)]$ is either reduced to one point or empty. Indeed, if there exist two distinct points $\lambda', \lambda'' \in]\sqrt{a}, \sqrt{b}[\cap \tilde{\Sigma}_z(\beta)]$, then as we have seen in the proof of Proposition 5.2.6 (b) I, there exists $\lambda_0 \in]\lambda', \lambda''[$ such that $\phi_{L_z,\beta}(\lambda_0) = 0$, which contradicts the assumption that]a, b[is a gap of the operator $A(\beta)$.

- I Let]a, b[be a gap of the operator $A(\beta)$ of type *I*. In this case $\phi_{L_{z,\beta}}(\sqrt{a}) < \phi_{L_{z,\beta}}(\sqrt{b})$. Consequently, $]\sqrt{a}, \sqrt{b}[\cap \widetilde{\Sigma}_{z}(\beta) \neq \emptyset$. As it was mentioned above, this intersection is necessarily reduced to one point.
- II Let]a, b[be a gap of the operator $A(\beta)$ of type II. Then, there exists $\delta > 0$ such that $\phi_{L_z,\beta}(\lambda')\phi_{L_z,\beta}(\lambda'') > 0$ for any $(\lambda',\lambda'') \in]\sqrt{a}, \sqrt{a} + \delta[\times]\sqrt{b} \delta, \sqrt{b}[$. Suppose that there exists $\sqrt{c} \in]\sqrt{a}, \sqrt{b}[\cap \widetilde{\Sigma}_z(\beta)]$. Then, at least in one of the intervals $]\sqrt{a}, \sqrt{c}[,]\sqrt{c}, \sqrt{b}[$ there exists λ_0 such that $\phi_{L_z,\beta}(\lambda_0) = 0$, which contradicts the assumption that]a, b[is a gap of the operator $A(\beta)$.

Examples illustrating Proposition 5.2.6, Lemma 5.2.3 and Proposition 5.2.4 are given in figures 5.7, 5.6.

5.2.2 Computation of the discrete spectrum

5.2.2.1 Reduction to a finite difference equation

Let us now determine the discrete spectrum of the operator $A^{\mu}(\beta)$. If λ^2 is an eigenvalue of this operator, then the corresponding eigenfunction $u \in D(A^{\mu}(\beta))$ solves the equation $u'' + \lambda^2 u = 0$ on each edge of the graph G.

$\underline{\mathbf{Case } \lambda > 0}$

In this case the eigenfunction u it has the form

$$u_{k+\frac{1}{2},\ell}(s) = a_{k+\frac{1}{2},\ell}\sin(\lambda s) + b_{k+\frac{1}{2},\ell}\cos(\lambda s), \quad s \in [0, L_x], \qquad (k,\ell) \in \mathbb{Z}^2, \quad (5.2.52)$$

$$u_{k,\ell+\frac{1}{2}}(t) = a_{k,\ell+\frac{1}{2}}\sin(\lambda t) + b_{k,\ell+\frac{1}{2}}\cos(\lambda t), \quad t \in [0, L_y], \quad (k,\ell) \in \mathbb{Z}^2, \quad (5.2.53)$$

$$u_{k,\ell}^{+}(z) = c_{k,\ell}^{+} \sin(\lambda z) + d_{k,\ell}^{+} \cos(\lambda z), \qquad z \in \left[0, \frac{L_z}{2}\right], \qquad (k,\ell) \in \mathbb{Z}^2, \quad (5.2.54)$$

$$u_{k,\ell}^{-}(z) = c_{k,\ell}^{-} \sin(\lambda z) + d_{k,\ell}^{-} \cos(\lambda z), \qquad z \in \left[-\frac{L_z}{2}, 0\right], \qquad (k,\ell) \in \mathbb{Z}^2.$$
(5.2.55)

The continuity of the eigenfunction u at the vertices $v_{k,l}$ of the graph G implies that

$$b_{k+\frac{1}{2},\ell} = b_{k,\ell+\frac{1}{2}} = a_{k-\frac{1}{2},\ell} \sin\left(\lambda L_x\right) + b_{k-\frac{1}{2},\ell} \cos\left(\lambda L_x\right) = a_{k,\ell-\frac{1}{2}} \sin\left(\lambda L_y\right) + b_{k,\ell-\frac{1}{2}} \cos\left(\lambda L_y\right) = d_{k,\ell}^+ = d_{k,\ell}^- =: d_{k,\ell}, \qquad (k,l) \in \mathbb{Z}^2.$$
(5.2.56)

After plugging (5.2.54), (5.2.55) into the quasiperiodicity conditions (5.2.1), we get

$$(c_{k,l}^{+} + e^{-i\beta}c_{k,l}^{-})\sin\left(\frac{\lambda L_{z}}{2}\right) + d_{k,l}\cos\left(\frac{\lambda L_{z}}{2}\right)\left(1 - e^{-i\beta}\right) = 0, \qquad (k,l) \in \mathbb{Z}^{2}, \qquad (5.2.57)$$

$$\left(c_{k,l}^{+} - e^{-i\beta}c_{k,l}^{-}\right)\cos\left(\frac{\lambda L_z}{2}\right) - d_{k,l}\sin\left(\frac{\lambda L_z}{2}\right)\left(1 + e^{-i\beta}\right) = 0, \qquad (k,l) \in \mathbb{Z}^2.$$
(5.2.58)







(a) The interval]a, b[is a gap of type I

(b) The interval]a, b[is a gap of type II(i)

(c) The interval]a, b[is a gap of type II(ii)

Figure 5.6: Types of gaps

(a) Gap of type I: $\phi_{L_z,\beta}(\sqrt{a}) = f_{L_x,L_y}^-(\sqrt{a}) < 0$, $\phi_{L_z,\beta}(\sqrt{b}) = f_{L_x,L_y}^+(\sqrt{b}) > 0$, and there is one point of discontinuity of the function $\phi_{L_z,\beta}$ in $]\sqrt{a}, \sqrt{b}[$. (b) Gap of type II(i): $\sqrt{a} \in \Sigma_y^*$, $\phi_{L_z,\beta}(\sqrt{a}) > f_{L_x,L_y}^+(\sqrt{a})$, $\phi_{L_z,\beta}(\sqrt{b}) = f_{L_x,L_y}^+(\sqrt{b}) > 0$, and the function $\phi_{L_z,\beta}$ is continuous in $]\sqrt{a}, \sqrt{b}[$. (c) Gap of type II(i): $\phi_{L_z,\beta}(\sqrt{a}) = f_{L_x,L_y}^-(\sqrt{b}) < 0$, $\sqrt{b} \in \Sigma_x^*$, $\phi_{L_z,\beta}(\sqrt{b}) < f_{L_x,L_y}^-(\sqrt{b})$, and the function $\phi_{L_z,\beta}$ is continuous in $]\sqrt{a}, \sqrt{b}[$.

From Kirchhoff's conditions (5.2.2) we find

$$w_{k,l}^{\mu} \left(c_{k,l}^{+} - c_{k,l}^{-} \right) + a_{k+\frac{1}{2},\ell} + a_{k,\ell+\frac{1}{2}} - a_{k-\frac{1}{2},\ell} \cos\left(\lambda L_{x}\right) - a_{k,\ell-\frac{1}{2}} \cos\left(\lambda L_{y}\right) + d_{k-1,\ell} \sin\left(\lambda L_{x}\right) + d_{k,\ell-1} \sin\left(\lambda L_{y}\right) = 0, \qquad (k,l) \in \mathbb{Z}^{2}.$$
(5.2.59)

The following assertion permits to exclude from the consideration the set Σ^* while searching the discrete spectrum of the operator $A^{\mu}(\beta)$. Its proof is given in Appendix.

Proposition 5.2.7. If $\lambda \in \Sigma^*$, then λ^2 is not an eigenvalue of finite multiplicity of the operator $A^{\mu}(\beta)$ for any $\beta \in [0, \pi]$, $\mu > 0$.

If $\lambda \notin \Sigma$, then the coefficients $\left\{a_{k+\frac{1}{2},\ell}, b_{k+\frac{1}{2},\ell}, a_{k,\ell+\frac{1}{2}}, b_{k,\ell+\frac{1}{2}}, c_{k,\ell}^{\pm}, d_{k,\ell}\right\}$ can be expressed in terms of the values $\{\mathbf{u}_{k,l}\}$:

$$a_{k+\frac{1}{2},\ell} = \frac{1}{\sin(\lambda L_x)} \left(\mathbf{u}_{k+1,\ell} - \mathbf{u}_{k,\ell} \cos(\lambda L_x) \right), \quad a_{k,\ell+\frac{1}{2}} = \frac{1}{\sin(\lambda L_y)} \left(\mathbf{u}_{k+1,\ell} - \mathbf{u}_{k,\ell} \cos(\lambda L_y) \right),$$
(5.2.60)

$$b_{k+\frac{1}{2},\ell} = b_{k,\ell+\frac{1}{2}} = d_{k,\ell} = \mathbf{u}_{k,\ell}, \qquad c_{k,l}^{\pm} = \alpha^{\pm} \mathbf{u}_{k,\ell}, \qquad (k,\ell) \in \mathbb{Z}^2.$$
(5.2.61)

where

$$\alpha^{+} = \frac{1}{2} \left(\tan\left(\frac{\lambda L_{z}}{2}\right) \left(1 + e^{-i\beta}\right) + \frac{e^{-i\beta} - 1}{\tan\left(\frac{\lambda L_{z}}{2}\right)} \right), \quad \alpha^{-} = \frac{1}{2} \left(\frac{1 - e^{i\beta}}{\tan\left(\frac{\lambda L_{z}}{2}\right)} - \tan\left(\frac{\lambda L_{z}}{2}\right) \left(1 + e^{i\beta}\right) \right).$$



(a) λ_0^2 is an isolated eigenvalue of infinite multiplicity: $\lambda_0 \in \Sigma_y^* \cap \widetilde{\Sigma}_z^*(\beta)$. The gap to the left of λ_0^2 is a gap of type II(ii') and the gap to the right of λ_0^2 is a gap of type II(i')



plicity: $\lambda_0 \in (\Sigma_x^* \cap \Sigma_y^*) \cap \widetilde{\Sigma}_z^*(\beta)$



(c) λ_0^2 is an embedded eigenvalue of infinite multiplicity: $\beta = 0$ and $\lambda_0 \in \Sigma_z^*(\beta)$

Figure 5.7: Eigenvalues of infinite multiplicity

Then, the relations (5.2.52)–(5.2.55) can be rewritten as

$$u_{k+\frac{1}{2},\ell}(s) = \mathbf{u}_{k,\ell} \frac{\sin\left(\lambda(L_x - s)\right)}{\sin\left(\lambda L_x\right)} + \mathbf{u}_{k+1,\ell} \frac{\sin\left(\lambda s\right)}{\sin\left(\lambda L_x\right)}, \quad s \in [0, L_x], \qquad k, \ell \in \mathbb{Z}, \quad (5.2.62)$$

$$u_{k,\ell+\frac{1}{2}}(t) = \mathbf{u}_{k,\ell} \frac{\sin\left(\lambda(L_y - t)\right)}{\sin\left(\lambda L_y\right)} + \mathbf{u}_{k,l+1} \frac{\sin\left(\lambda t\right)}{\sin\left(\lambda L_y\right)}, \quad t \in [0, L_y], \qquad k, \ell \in \mathbb{Z}, \quad (5.2.63)$$

$$u_{k,\ell}^+(z) = \mathbf{u}_{k,\ell} \frac{\sin\left(\lambda\left(\frac{L_z}{2} - z\right)\right)}{\sin\left(\frac{\lambda L_z}{2}\right)} + \mathbf{u}_{k,\ell}^+ \frac{\sin\left(\lambda z\right)}{\sin\left(\frac{\lambda L_z}{2}\right)}, \qquad z \in \left[0, \frac{L_z}{2}\right], \qquad k, \ell \in \mathbb{Z}, \quad (5.2.64)$$

$$u_{k,\ell}^{-}(z) = \mathbf{u}_{k,\ell} \frac{\sin\left(\lambda\left(\frac{L_z}{2} + z\right)\right)}{\sin\left(\frac{\lambda L_z}{2}\right)} - \mathbf{u}_{k,\ell}^{-} \frac{\sin\left(\lambda z\right)}{\sin\left(\frac{\lambda L_z}{2}\right)}, \qquad z \in \left[-\frac{L_z}{2}, 0\right], \qquad k, \ell \in \mathbb{Z}, \quad (5.2.65)$$

$$\mathbf{u}_{k,\ell}^{+} = \frac{1 + e^{-i\beta}}{2\cos\left(\frac{\lambda L_z}{2}\right)} \mathbf{u}_{k,\ell}, \qquad \mathbf{u}_{k,\ell}^{-} = \frac{1 + e^{i\beta}}{2\cos\left(\frac{\lambda L_z}{2}\right)} \mathbf{u}_{k,\ell}, \qquad (k,\ell) \in \mathbb{Z}^2.$$
(5.2.66)

After plugging (5.2.60)–(5.2.61) into (5.2.59), we get

$$\frac{\mathbf{u}_{k+1,\ell} + \mathbf{u}_{k-1,\ell}}{\sin(\lambda L_x)} + \frac{\mathbf{u}_{k,\ell+1} + \mathbf{u}_{k,\ell-1}}{\sin(\lambda L_y)} - 2g_\beta(\lambda)\mathbf{u}_{k,l} = 0, \qquad (0,0) \neq (k,\ell) \in \mathbb{Z}^2, \quad (5.2.67)$$

$$\frac{\mathbf{u}_{k+1,\ell} + \mathbf{u}_{k-1,\ell}}{\sin(\lambda L_x)} + \frac{\mathbf{u}_{k,\ell+1} + \mathbf{u}_{k,\ell-1}}{\sin(\lambda L_y)} - 2g_\beta(\lambda)\mathbf{u}_{0,0} = 2(\mu - 1)\phi_{L_z,\beta}(\lambda)\mathbf{u}_{0,0},$$
(5.2.68)

where

$$g_{\beta}(\lambda) = \frac{1}{\tan\left(\lambda L_x\right)} + \frac{1}{\tan\left(\lambda L_y\right)} + \phi_{L_z,\beta}(\lambda), \qquad \lambda \notin \Sigma, \tag{5.2.69}$$

and the function $\phi_{L_z,\beta}$ is defined in (5.2.37). Thus, we replaced the initial continuous problem for $u \in L_2(G)$ by a discrete problem for $\{\mathbf{u}_{k,l}\}_{(k,l)\in\mathbb{Z}^2} \in \ell_2(\mathbb{Z}^2)$.

Case $\lambda = 0$

If 0 is an eigenvalue of the operator $A^{\mu}(\beta)$, then the corresponding eigenfunction has the form

$$u_{k+\frac{1}{2},\ell}(s) = \mathbf{u}_{k,\ell}(1-s) + \mathbf{u}_{k+1,\ell}s, \qquad s \in [0, L_x], \qquad (k,\ell) \in \mathbb{Z}^2, \qquad (5.2.70)$$

$$u_{k,\ell+\frac{1}{2}}(t) = \mathbf{u}_{k,\ell}(1-t) + \mathbf{u}_{k,\ell+1}t, \qquad t \in [0, L_y], \qquad (k,\ell) \in \mathbb{Z}^2, \qquad (5.2.71)$$

$$u_{k,\ell}^{+}(z) = \mathbf{u}_{k,\ell} \left(1 - \frac{2z}{L_z} \right) + \frac{2\mathbf{u}_{k,\ell}^{+}}{L_z} z, \qquad z \in \left[0, \frac{L_z}{2} \right], \qquad (k,\ell) \in \mathbb{Z}^2, \qquad (5.2.72)$$

$$u_{k,\ell}^{-}(z) = \mathbf{u}_{k,\ell} \left(1 + \frac{2z}{L_z} \right) - \frac{2\mathbf{u}_{k,\ell}^{-}}{L_z} z, \qquad z \in \left[-\frac{L_z}{2}, 0 \right], \qquad (k,\ell) \in \mathbb{Z}^2.$$
(5.2.73)

The quasiperiodicity conditions (5.2.1) imply that

$$\mathbf{u}_{k,\ell}^{+} = \frac{1 + e^{-i\beta}}{2} \mathbf{u}_{k,\ell}, \qquad \mathbf{u}_{k,\ell}^{-} = \frac{1 + e^{i\beta}}{2} \mathbf{u}_{k,\ell}, \qquad (k,\ell) \in \mathbb{Z}^2.$$
(5.2.74)

After plugging (5.2.70)–(5.2.74) into the Kirchhoff's conditions (5.2.2) we find

$$\frac{\mathbf{u}_{k+1,\ell} + \mathbf{u}_{k-1,\ell}}{L_x} + \frac{\mathbf{u}_{k,\ell+1} + \mathbf{u}_{k,\ell-1}}{L_y} - 2\left(\frac{1}{L_x} + \frac{1}{L_y} + \frac{1 - \cos\beta}{L_z}\right)\mathbf{u}_{k,\ell} = 0, \qquad (k,\ell) \neq (0,0),$$
(5.2.75)

$$\frac{\mathbf{u}_{k+1,\ell} + \mathbf{u}_{k-1,\ell}}{L_x} + \frac{\mathbf{u}_{k,\ell+1} + \mathbf{u}_{k,\ell-1}}{L_y} - 2\left(\frac{1}{L_x} + \frac{1}{L_y} + \frac{1 - \cos\beta}{L_z}\right)\mathbf{u}_{k,\ell} = 2(\mu - 1)\frac{1 - \cos\beta}{L_z}\mathbf{u}_{0,0}$$
(5.2.76)

We end up with the following assertion.

Lemma 5.2.4. If λ^2 is an eigenvalue of finite multiplicity of the operator $A^{\mu}(\beta)$ and u is an eigenfunction associated with λ^2 then $\{u_{k,l}\}_{(k,l)\in\mathbb{Z}^2} \in \ell_2(\mathbb{Z}^2)$ and the sequence $\{u_{k,l}\}_{(k,l)\in\mathbb{Z}^2}$ satisfies the finite difference equation (5.2.67)–(5.2.68) for $\lambda \neq 0$ and (5.2.75)– (5.2.76) for $\lambda = 0$. Conversely, if a sequence $\{u_{k,l}\}_{(k,l)\in\mathbb{Z}^2} \in \ell_2(\mathbb{Z}^2)$ satisfies (5.2.67)– (5.2.68) (resp. (5.2.75)–(5.2.76)) then λ^2 (resp. 0) is an eigenvalue of finite multiplicity of the operator $A^{\mu}(\beta)$ and the function u defined by the relations (5.2.62)–(5.2.66) (resp.(5.2.70)–(5.2.74)) is a corresponding eigenfunction.

5.2.2.2 Absence of embedded eigenvalues

Let us apply to the sequence $\{\mathbf{u}_{k,l}\}$ the discrete Fourier transform $\mathcal{F}: \ell_2(\mathbb{Z}^2) \to L_2([0, 2\pi]^2)$ defined as

$$\mathcal{F}(\{\mathbf{v}_{k,l}\}) = \widehat{\mathbf{v}}, \qquad \widehat{\mathbf{v}}(\xi,\eta) = \sum_{(k,\ell)\in\mathbb{Z}^2} e^{i(k\xi+l\eta)} \mathbf{v}_{k,\ell}, \qquad (\xi,\eta)\in[0,2\pi]^2.$$

Then, for $\widehat{\mathbf{u}}$ we get the following equation:

For $\lambda \neq 0$,

$$\left(f_{L_x,L_y,\xi,\eta}(\lambda) - \phi_{L_z,\beta}(\lambda)\right) \widehat{\mathbf{u}}(\xi,\eta) = (\mu - 1)\phi_{L_z,\beta}(\lambda)\mathbf{u}_{0,0}, \quad \forall (\xi,\eta) \in [0,2\pi]^2, \quad (5.2.77)$$

where the function $f_{L_x,L_y,\xi,\eta}$ is defined in (5.2.35).

For $\lambda = 0$,

$$\left(\frac{\cos\xi - 1}{L_x} + \frac{\cos\eta - 1}{L_y} - \frac{1 - \cos\beta}{L_z}\right)\widehat{\mathbf{u}}(\xi, \eta) = (\mu - 1)\frac{1 - \cos\beta}{L_z}\mathbf{u}_{0,0}, \quad \forall (\xi, \eta) \in [0, 2\pi]^2.$$
(5.2.78)

We can now prove the absence of embedded eigenvalues for the operator $A^{\mu}(\beta)$.

Proposition 5.2.8. If λ^2 is an eigenvalue of finite multiplicity of the operator $A^{\mu}(\beta)$ for some $\beta \in [0, \pi]$, $\mu > 0$, then $\lambda^2 \notin \sigma(A(\beta))$.

Proof.

(a) $\lambda > 0$: suppose that $\lambda^2 \in \sigma(A(\beta))$ is an eigenvalue of finite multiplicity of the operator $A^{\mu}(\beta)$. As follows from Proposition 5.2.7, $\lambda \in \mathbb{R}^*_+ \setminus \Sigma^*$. According to Proposition 5.2.5, there exist $(k_x, k_y) \in [0, \pi]^2$ such that $\phi_{L_z,\beta}(\lambda) = f_{L_x,L_y,k_x,k_y}(\lambda)$. On the other hand, if u is an eigenfunction corresponding to λ^2 , then the equation (5.2.77) is satisfied. Its right-hand side does not depend on (ξ, η) and its left-hand side is zero for $(\xi, \eta) = (k_x, k_y)$. Thus,

$$\left(f_{L_x,L_y,\xi,\eta}(\lambda) - \phi_{L_z,\beta}(\lambda)\right) \widehat{\mathbf{u}}(\xi,\eta) = 0, \qquad \forall (\xi,\eta) \in [0,2\pi]^2.$$

Consequently, $\widehat{\mathbf{u}}(\xi, \eta) = 0$ for any $(\xi, \eta) \in [0, 2\pi]^2$ such that $f_{L_x, L_y, \xi, \eta}(\lambda) \neq f_{L_x, L_y, k_x, k_y}(\lambda)$. The set $\{(\xi, \eta) \in [0, 2\pi]^2 / f_{L_x, L_y, \xi, \eta}(\lambda) \neq f_{L_x, L_y, k_x, k_y}(\lambda)\}$ being a full-mesure set, we conclude that $\widehat{\mathbf{u}} = 0$ in $L_2([0, 2\pi]^2)$. Hence, u = 0 and λ^2 is not an eigenvalue of the operator $A^{\mu}(\beta)$.

(b) $\lambda = 0$: suppose that $0 \in \sigma(A(\beta))$. Then, one necessarily has $\beta = 0$. If u is an eigenfunction corresponding to 0, then its discrete Fourier transform satisfies the equation (5.2.78) with $\beta = 0$:

$$\left(\frac{\cos\xi - 1}{L_x} + \frac{\cos\eta - 1}{L_y}\right)\widehat{\mathbf{u}}(\xi, \eta) = 0, \qquad \forall (\xi, \eta) \in [0, 2\pi]^2.$$

Similarly to the case $\lambda \neq 0$, we find $\widehat{\mathbf{u}}(\xi, \eta) = 0$ for any $(\xi, \eta) \in [0, 2\pi] \setminus (0, 0)$, which implies that u = 0 and 0 is not an eigenvalue of the operator A(0).

5.2.2.3 Obtaining a characteristic equation for the eigenvalues

From now on we will consider λ such that $\lambda^2 \notin \sigma(A(\beta))$. Then, it follows from Proposition 5.2.3 (1) and Proposition 5.2.5 that

$$\phi_{L_z,\beta}(\lambda) \neq 0, \qquad f_{L_x,L_y,\xi,\eta}(\lambda) - \phi_{L_z,\beta}(\lambda) \neq 0, \qquad \forall (\xi,\eta) \in [0,2\pi]^2. \tag{5.2.79}$$

Consequently, for $\lambda \in \mathbb{R}_+ \setminus \Sigma^*$, $\lambda^2 \notin \sigma(A(\beta))$ the relations (5.2.77), (5.2.78) impliy

$$\widehat{\mathbf{u}}(\xi,\eta) = \frac{(1-\mu)\mathbf{u}_{0,0}}{1-\varphi_{\beta}(\xi,\eta,\lambda)}, \qquad \forall (\xi,\eta) \in [0,2\pi]^2,$$

where

$$\varphi_{\beta}(\xi,\eta,\lambda) = \begin{cases} \frac{f_{L_{x},L_{y},\xi,\eta}(\lambda)}{\phi_{c,\beta}(\lambda)}, & \lambda \in \mathbb{R}_{+} \setminus (\Sigma \cup \Sigma_{z}(\beta)), \\ \frac{L_{z}}{1-\cos\beta} \left(\frac{\cos\xi-1}{L_{x}} + \frac{\cos\eta-1}{L_{y}}\right), & \beta \in]0,\pi], \quad \lambda = 0. \end{cases}$$
(5.2.80)

Notice that for $\beta \in]0, \pi]$ the function φ_{β} is continuous at 0. Finally, applying the inverse Fourier transform, we get

$$\mathbf{u}_{k,\ell} = \frac{(1-\mu)\mathbf{u}_{0,0}}{4\pi^2} \iint_{[0,2\pi]^2} \frac{1}{1-\varphi_\beta(\xi,\eta,\lambda)} e^{-i(k\xi+\ell\eta)} d\xi d\eta, \qquad (k,\ell) \in \mathbb{Z}^2.$$
(5.2.81)

We can now give a criterion of existence of an eigenfunction.

Proposition 5.2.9. λ^2 is an eigenvalue of finite multiplicity of the operator $A^{\mu}(\beta)$ if and only if $\lambda \in \mathbb{R}_+ \setminus \Sigma^*$, $\lambda^2 \notin \sigma(A(\beta))$ and the following relation is satisfied:

$$(1-\mu)\mathcal{I}_{\beta}(\lambda) = 1, \qquad (5.2.82)$$

where

$$I_{\beta}(\lambda) = \frac{1}{\pi^2} \iint_{[0,\pi]^2} \frac{1}{1 - \varphi_{\beta}(\xi,\eta,\lambda)} d\xi d\eta, \qquad (5.2.83)$$

and the function φ_{β} is defined in (5.2.80).

Proof. If λ^2 is an eigenvalue of finite multiplicity of the operator $A^{\mu}(\beta)$, then it follows from Propositions 5.2.7 and 5.2.8 that $\lambda \in \mathbb{R}_+ \setminus \Sigma^*$ and $\lambda^2 \notin \sigma(A(\beta))$. Let u be an eigenfunction corresponding to the eigenvalue λ^2 . Then, the relation (5.2.81) is satisfied. For $k = 0, \ell = 0$ it gives

$$1 = (1-\mu) \left(\frac{1}{4\pi^2} \iint_{[0,2\pi]^2} \frac{1}{1-\varphi_{\beta}(\xi,\eta,\lambda)} e^{-i(k\xi+\ell\eta)} d\xi d\eta \right) = (1-\mu) I_{\beta}(\lambda).$$

The last equality is due to the symmetry of φ_{β} :

$$\varphi_{\beta}(\xi,\eta,\lambda) = \varphi_{\beta}(2\pi - \xi,\eta,\lambda) = \varphi_{\beta}(\xi,2\pi - \eta,\lambda), \qquad \forall (\xi,\eta) \in \mathbb{R}^{2}, \quad \lambda \in \mathbb{R}_{+} \setminus (\Sigma^{*} \cup \Sigma(\beta)).$$
(5.2.84)

Conversely, suppose that $\lambda \in \mathbb{R}_+ \setminus \Sigma^*$, $\lambda^2 \notin \sigma(A(\beta))$ and the relation (5.2.82) is satisfied. Then, $\lambda \notin \Sigma(\beta)$ and one can define $\{\mathbf{u}_{k,\ell}\}_{(k,\ell)\in\mathbb{Z}^2}$ by the relation (5.2.81). Since $\varphi_{\beta}(\xi,\eta,\lambda) \neq 1, \forall (\xi,\eta) \in [0,\pi]^2$, one has $\{\mathbf{u}_{k,\ell}\}_{(k,\ell)\in\mathbb{Z}^2} \in \ell_2(\mathbb{Z}^2)$ and the relations (5.2.62)– (5.2.66) (or (5.2.70)–(5.2.74) for $\lambda = 0$) define an eigenfunction of the operator $A^{\mu}(\beta)$ corresponding to the eigenvalue λ^2 .

5.2.2.4 Existence of eigenvalues

We will now study the number of eigenvalues in the gaps of the operator $A^{\mu}(\beta)$. With the classification of gaps introduced in Proposition 5.2.6 we can prove the following assertion.

Theorem 5.2.1. For any $0 < \mu < 1$, $\beta \in [0, \pi]$ there exist at least two simple eigenvalues of the operator $A^{\mu}(\beta)$ in each gap of type I and at least one simple eigenvalue in each gap of type II. These eigenvalues are characterized as follows:

$$\lambda^2 \in \sigma_d(A^{\mu}(\beta)) \quad \Leftrightarrow \quad \lambda \text{ is a solution of } (5.2.82).$$

For $\mu \ge 1$ the operator $A^{\mu}(\beta)$ has no eigenvalues.

Remark 5.2.5. Since in our terminology the interval $]0, \min \sigma(A(\beta))[$ for $\beta \in]0, \pi]$ is called a gap of type II (cf. Remark 5.2.4), Theorem 5.2.1 implies that the operator $A^{\mu}(\beta)$ has one simple igenvalue below the essential spectrum for $\beta \in]0, \pi], 0 < \mu < 1$.

The rest of the section is devoted to the proof of Theorem 5.2.1. The following auxiliary assertion shows that for $\mu \ge 1$ the operator $A^{\mu}(\beta)$ has no discrete spectrum.

Lemma 5.2.5. For any $\beta \in [0, \pi]$, if $\lambda \in \mathbb{R}_+ \setminus \Sigma^*$ and $\lambda^2 \notin \sigma(A(\beta))$, then $\varphi_\beta(\xi, \eta, \lambda) < 1$, $\forall (\xi, \eta) \in [0, \pi]^2$.

Proof. If $\lambda^2 \notin \sigma(A(\beta))$, then $\phi_{L_z,\beta}(\lambda) \in \mathbb{R} \setminus [f_{L_x,L_y}^-(\lambda), f_{L_x,L_y}^+(\lambda)]$. Taking into account that $f_{L_x,L_y}^-(\lambda) < 0$, $f_{L_x,L_y}^+(\lambda) > 0$, we get the result for $\lambda > 0$. Finally, for $\lambda = 0, \beta \in]0, \pi]$, it follows from (5.2.80) that $\varphi_{\beta}(\xi, \eta, \lambda) \leq 0, \forall (\xi, \eta) \in [0, \pi]^2$.

Corollary 5.2.2. For any $\beta \in [0, \pi]$, $\mu \ge 1$, the operator $A^{\mu}(\beta)$ has no eigenvalues of finite multiplicity.

Proof. As follows from Lemma 5.2.5, if $\lambda^2 \in \sigma_d(A^{\mu}(\beta))$, then $I_{\beta}(\lambda) > 0$. The relation (5.2.82) implies that $\mu < 1$.

Remark 5.2.6. Lemma 5.2.5 permits to rewrite the relation (5.2.82) as

$$\mu = 1 - F_{\beta}(\lambda), \qquad (5.2.85)$$

with

$$F_{\beta}(\lambda) = I_{\beta}(\lambda)^{-1}.$$

We will now study the behaviour of the function F_{β} inside the gaps of the operator $A^{\mu}(\beta)$ in order to determine the possible number of eigenvalues of the operator $A^{\mu}(\beta)$ in each gap. As follows from Lemma 5.2.5, the function F_{β} is continuous inside the gaps except at the points of discontinuity of the function $\phi_{L_z,\beta}$. In Lemmas 5.2.6 and 5.2.7 we describe the behaviour of F_{β} near the ends of the gap and in Lemma 5.2.8 we study its behaviour near the points of discontinuity of the function ϕ_{β,L_z} .

Lemma 5.2.6. If $\lambda_{+} \in \mathbb{R}_{+} \setminus \left(\Sigma_{x} \cup \Sigma_{y} \cup \widetilde{\Sigma}_{z}(\beta)\right)$ and $\phi_{L_{z},\beta}(\lambda_{+}) = f^{+}_{L_{x},L_{y}}(\lambda_{+})$, then $F_{\beta}(\lambda) \xrightarrow[\lambda \to \lambda_{+} \to 0]{} 0$. Similarly, if $\lambda_{-} \in \mathbb{R}_{+} \setminus \left(\Sigma_{x} \cup \Sigma_{y} \cup \widetilde{\Sigma}_{z}(\beta)\right)$ and $\phi_{L_{z},\beta}(\lambda_{-}) = f^{-}_{L_{x},L_{y}}(\lambda_{+})$, then $F_{\beta}(\lambda) \xrightarrow[\lambda \to \lambda_{-} \to 0]{} 0$.

Proof. We will prove the statement for λ_+ only, the proof for λ_- being analogous. Let ξ_0 , η_0 be the values of ξ , η such that $f_{L_x,L_y,\xi_0,\eta_0}(\lambda_+) = f^+_{L_x,L_y}(\lambda_+)$ (obviously, $\xi_0 \in \{0,\pi\}$ and $\eta_0 \in \{0,\pi\}$). Then, due to the symmetry of the function φ_β (cf. (5.2.84)), one has

$$I_{\beta}(\lambda) = \frac{1}{4\pi^2} \iint_{\substack{|\xi-\xi_0| \leqslant \pi \\ |\eta-\eta_0| \leqslant \pi}} \frac{1}{1 - \varphi_{\beta}(\xi,\eta,\lambda)} d\xi d\eta \ge \frac{1}{4\pi^2} \iint_{\sqrt{\xi'^2 + \eta'^2} \leqslant \pi} \frac{1}{1 - \varphi_{\beta}(\xi_0 + \xi',\eta_0 + \eta',\lambda)} d\xi' d\eta'.$$

In the last inequality we made a change of variables $\xi = \xi_0 + \xi'$, $\eta = \eta_0 + \eta'$. We have then

$$I_{\beta}(\lambda) \ge \frac{1}{4\pi^2} I_{\beta,r}(\lambda), \qquad \forall r < \pi, \qquad \lambda \notin \Sigma, \quad \lambda^2 \notin \sigma\left(A(\beta)\right), \tag{5.2.86}$$

where

$$I_{\beta,r}(\lambda) = \iint_{D_r} \frac{1}{1 - \varphi_{\beta}(\xi_0 + \xi, \eta_0 + \eta, \lambda)} d\xi d\eta,$$

and $D_r = \left\{ \left(\xi, \eta\right) / \sqrt{\xi^2 + \eta^2} \leqslant r \right\}$. It follows from the assumptions of the lemma that λ_+^2 is the upper end of a gap (it can also be the bottom of the essential spectrum (cf. Lemma 5.2.2 and Remark 5.2.4). Thus, we can find $\delta > 0$ such that $\lambda^2 \notin \sigma_{ess} \left(A^{\mu}(\beta)\right)$, $\forall \lambda \in]\lambda_+ - \delta, \lambda_+[$. From now on we suppose that $\lambda^2 \notin \sigma_{ess} \left(A^{\mu}(\beta)\right)$ and we study the behaviour of the integral $I_{\beta,r}(\lambda)$ as $\lambda \to \lambda_+ - 0$. Notice that the function φ_{β} is smooth in some neighbourhood U of the point $(\xi_0, \eta_0, \lambda_+)$. Indeed, by assumption $\lambda_+ \notin \Sigma_x \cup \Sigma_y \cup \widetilde{\Sigma}_z(\beta)$ which guarantees the continuity of the functions $f_{L_x,L_y,\xi,\eta}$ and $\phi_{L_z,\beta}$. Moreover, $\phi_{L_z,\beta}(\lambda_+) \neq 0$ since $f_{L_x,L_y}^+(\lambda_+) > 0$. Let ρ and δ be positive numbers such that $\varphi_{\beta} \in C^{\infty}(\overline{U}_{\rho,\delta})$, where

$$U_{\rho,\delta} = \left\{ (\xi - \xi_0)^2 + (\eta - \eta_0)^2 < \rho^2 \right\} \times]\lambda_+ - \delta, \lambda_+].$$

Let us write down the Taylor series of the function φ_{β} in $U_{\rho,\delta}$:

$$\varphi_{\beta}(\xi,\eta,\lambda) = 1 + \partial_{\lambda}\varphi_{\beta}(\xi_{0},\eta_{0},\lambda_{+})(\lambda-\lambda_{+}) + \partial_{\xi\xi}^{2}\varphi_{\beta}(\xi_{0},\eta_{0},\lambda_{+})(\xi-\xi_{0})^{2} + \partial_{\eta\eta}^{2}\varphi_{\beta}(\xi_{0},\eta_{0},\lambda_{+})(\eta-\eta_{0})^{2} + O\left((\lambda-\lambda_{+})^{2}\right) + O\left((\xi-\xi_{0})^{4}\right) + O\left((\eta-\eta_{0})^{4}\right).$$
(5.2.87)

We took into account the relations

$$\varphi_{\beta}(\xi_{0}+\xi,\eta,\lambda)=\varphi_{\beta}(\xi_{0}-\xi,\eta,\lambda),\qquad \varphi_{\beta}(\xi,\eta_{0}+\eta,\lambda)=\varphi_{\beta}(\xi,\eta_{0}-\eta,\lambda),$$

which lead to the absence of the terms with odd powers of $(\xi - \xi_0)$, $(\eta - \eta_0)$ in (5.2.87). It follows from (5.2.87) that there exists a constant $C_{\rho,\delta} > 0$ such that

$$\varphi_{\beta}(\xi,\eta,\lambda) \ge 1 - C_{\rho,\delta} \left(|\lambda - \lambda_{+}| + (\xi - \xi_{0})^{2} + (\eta - \eta_{0})^{2} \right), \qquad \forall (\xi,\eta,\lambda) \in U_{\rho,\delta}.$$

Consequently,

$$I_{\beta,\rho}(\lambda) \geqslant \iint_{D_{\rho}} \frac{C_{\rho,\delta}^{-1}}{|\lambda - \lambda_{+}| + \xi^{2} + \eta^{2}} d\xi d\eta = \pi C_{\rho,\delta}^{-1} \ln\left(1 + \frac{\rho^{2}}{|\lambda - \lambda_{+}|}\right),$$

which shows that $I_{\beta,\rho}(\lambda) \xrightarrow[\lambda \to \lambda_+ = 0]{} +\infty$. Taking into account (5.2.86) and the definition of the function F_{β} we conclude that $F_{\beta}(\lambda) \xrightarrow[\lambda \to \lambda_+ = 0]{} 0$.

Lemma 5.2.7. If $\lambda_{-} \in (\Sigma_{x}^{*} \cup \Sigma_{y}^{*}) \setminus \widetilde{\Sigma}_{z}^{*}(\beta)$ and $\phi_{L_{z},\beta}(\lambda_{-}) > f_{L_{x},L_{y}}^{+}(\lambda_{-})$ or $\lambda_{-} \in (\Sigma_{x} \cup \Sigma_{y}) \cap \widetilde{\Sigma}_{z}(\beta)$, then $\lim_{\lambda \to \lambda_{-} + 0} F_{\beta}(\lambda) \geq 1$. Similarly, if $\lambda_{+} \in (\Sigma_{x}^{*} \cup \Sigma_{y}^{*}) \setminus \widetilde{\Sigma}_{z}^{*}(\beta)$ and $\phi_{L_{z},\beta}(\lambda_{+}) < f_{L_{x},L_{y}}^{-}(\lambda_{+})$ or $\lambda_{+} \in (\Sigma_{x}^{*} \cup \Sigma_{y}^{*}) \cap \widetilde{\Sigma}_{z}^{*}(\beta)$, then $\lim_{\lambda \to \lambda_{+} - 0} F_{\beta}(\lambda) \geq 1$.

Proof. We give the proof for the case of λ_{-} only, the proof for λ_{+} being similar. We will suppose that $\lambda_{-} \in \Sigma_{x}$ (the case $\lambda_{-} \in \Sigma_{y}$ can be considered analogously). It follows from Lemma 5.2.2 that λ_{-} is the lower end of a gap (or $\lambda_{-} = 0$ and 0 is not the bottom of the spectrum of $A(\beta)$). Thus, there exists $\delta > 0$ such that $\lambda^{2} \notin \sigma(A(\beta)), \forall \lambda \in]\lambda_{-}, \lambda_{-} + \delta[$.

(i) Consider first the case $\lambda_{-} \in (\Sigma_{x}^{*} \cup \Sigma_{y}^{*}) \setminus \widetilde{\Sigma}_{z}^{*}(\beta), \ \phi_{L_{z},\beta}(\lambda_{-}) > f_{L_{x},L_{y}}^{+}(\lambda_{-}) \ge 0$. Let us study the behaviour of $I(\lambda)$ as $\lambda \to \lambda_{-} + 0$. Since $\varphi_{\beta}(\xi, \eta, \lambda) \xrightarrow[\lambda \to \lambda_{-} + 0]{} -\infty$ for any $\xi \in [0, \pi]$ such that $\cos \xi \neq \cos (\lambda_{-}L_{x})$ and any $\eta \in [0, \pi]$, we have

$$\frac{1}{1 - \varphi_{\beta}(\xi, \eta, \lambda)} \xrightarrow{\lambda \to \lambda_{-} + 0} 0^{+}, \qquad (\xi, \eta) \in [0, \pi]^{2} \setminus \{(\xi, \eta) / \cos \xi = \cos (\lambda_{-} L_{x})\}.$$

We can then apply the dominated convergence theorem. Indeed, there exists $\delta > 0$ such that $\phi_{L_z,\beta}(\lambda) > f^+_{L_x,L_y}(\lambda)$ for any $\lambda \in [\lambda_-, \lambda_- + \delta]$. Then,

$$\varphi_{\beta}(\xi,\eta,\lambda) < \frac{f_{L_{x},L_{y}}^{+}(\lambda)}{\phi_{L_{z},\beta}(\lambda)} \leqslant \frac{f_{L_{x},L_{y}}^{+}(\lambda_{-}+\delta)}{\phi_{L_{z},\beta}(\lambda_{-}+\delta)} < 1, \qquad (\xi,\eta,\lambda) \in [0,\pi]^{2} \times [\lambda_{-},\lambda_{-}+\delta].$$

$$(5.2.88)$$

Here we used the monotonicity and positivity of the functions f_{L_x,L_y}^+ and $\phi_{L_z,\beta}$ in $[\lambda_-, \lambda_- + \delta]$. Putting

$$C = \frac{f_{L_x,L_y}^+(\lambda_- + \delta)}{\phi_{L_z,\beta}(\lambda_- + \delta)} < 1,$$

we get

$$\frac{1}{1 - \varphi_{\beta}(\xi, \eta, \lambda)} \leqslant \frac{1}{1 - C}, \qquad (\xi, \eta, \lambda) \in [0, \pi]^2 \times [\lambda_{-}, \lambda_{-} + \delta].$$

Consequently, the dominated convergence theorem applies and $I_{\beta}(\lambda) \xrightarrow[\lambda \to \lambda_{-} + 0]{} + 0^{+}$, which implies $F_{\beta}(\lambda) \xrightarrow[\lambda \to \lambda_{-} + 0]{} + \infty$.

(ii) Suppose now that $\lambda_{-} \in (\Sigma_x \cup \Sigma_y) \cap \widetilde{\Sigma}_z(\beta)$. Then, if $\lambda_{-} \in \Sigma_x \setminus \Sigma_y$, we obtain

$$\varphi_{\beta}(\xi,\eta,\lambda) \xrightarrow[\lambda \to \lambda_{-} + 0]{} \frac{L_{z} \left(\frac{\cos\xi}{\cos(\lambda_{-}L_{x})} - 1\right)}{L_{x} \left(1 - \frac{\cos\beta}{\cos(\lambda_{-}L_{z})}\right)}, \qquad (\xi,\eta) \in [0,\pi]^{2}$$

Using the relations $|\cos(\lambda_{-}L_{x})| = 1$, $|\cos(\lambda_{-}L_{z})| = 1$ and $\cos\beta \neq \cos(\lambda_{-}L_{z})$ (since $\lambda_{-} \in \widetilde{\Sigma}_{z}(\beta)$), we get

$$\frac{\cos\xi}{\cos\left(\lambda_{-}L_{x}\right)} - 1 \leqslant 0, \qquad 1 - \frac{\cos\beta}{\cos\left(\lambda_{-}L_{z}\right)} > 0$$

Consequently, $\lim_{\lambda \to \lambda_- + 0} \varphi_{\beta}(\xi, \eta, \lambda) \leq 0$ for any $(\xi, \eta) \in [0, \pi]^2$.

Similarly, if $\lambda_{-} \in \Sigma_{x} \cap \Sigma_{y}$, we obtain

$$\varphi_{\beta}(\xi,\eta,\lambda) \xrightarrow[\lambda \to \lambda_{-} + 0]{} \frac{L_{z}\left(\frac{\cos\xi}{\cos(\lambda_{-}L_{x})} - 1\right)}{L_{x}\left(1 - \frac{\cos\beta}{\cos(\lambda_{-}L_{z})}\right)} + \frac{L_{z}\left(\frac{\cos\eta}{\cos(\lambda_{-}L_{x})} - 1\right)}{L_{y}\left(1 - \frac{\cos\beta}{\cos(\lambda_{-}L_{z})}\right)} \leqslant 0 \qquad (\xi,\eta) \in [0,\pi]^{2}.$$

The dominated convergence theorem can be applied again since the estimate (5.2.88) is still valid. We have $\lim_{\lambda \to \lambda_{-} \to 0} I_{\beta}(\lambda) \leq 1$, which implies $\lim_{\lambda \to \lambda_{-} \to 0} F_{\beta}(\lambda) \geq 1$.

Lemma 5.2.8. If $\lambda_0 \in \widetilde{\Sigma}_z(\beta) \setminus (\Sigma_x \cup \Sigma_y)$, then $\lim_{\lambda \to \lambda_0} F_\beta(\lambda) = 1$.

Proof. This is obvious since $\varphi_{\beta}(\xi, \eta, \lambda) \xrightarrow[\lambda \to \lambda_0]{} 0$ uniformly for $(\xi, \eta) \in [0, \pi]^2$. \Box

We can now finish the proof of Theorem 5.2.1. We will use the classification of gaps given in Lemma 5.2.3, according to which the image of each gap of type I by the function $x \mapsto \sqrt{x}$ contains exactly one point of the set $\tilde{\Sigma}_z(\beta)$ (point of discontinuity of the function $\phi_{L_z,\beta}$), whereas the image of each gap of type II does not contain such points.

Proof of Theorem 5.2.1. The result for $\mu \ge 1$ is given in Corollary 5.2.2.

(a) Let]a, b[be a gap of type I. Then, according to Proposition 5.2.6, \sqrt{a} and \sqrt{b} satisfy the hypothesis of Lemma 5.2.6 and $\lim_{\lambda \to \sqrt{a^+}} (1 - F_{\beta}(\lambda)) = \lim_{\lambda \to \sqrt{b^-}} (1 - F_{\beta}(\lambda)) = 1$. Next, it follows from Lemma 5.2.3 that there exists $c \in]a, b[$ such that \sqrt{c} satisfies the hypothesis of Lemma 5.2.8, which implies $\lim_{\lambda \to \sqrt{c}} (1 - F_{\beta}(\lambda)) = 0$. Moreover, the point \sqrt{c} being the unique point of discontinuity of the function $\phi_{L_z,\beta}$ in $]\sqrt{a}, \sqrt{b}[$, the

function F_{β} is continuous in $]\sqrt{a}, \sqrt{c}[$ and $]\sqrt{c}, \sqrt{b}[$. Consequently, for $0 < \mu < 1$ there exists at least one solution of the equation (5.2.85) in each of the intervals $]\sqrt{a}, \sqrt{c}[$ and $]\sqrt{c}, \sqrt{b}[$, which gives at least two eigenvalues of the operator $A^{\mu}(\beta)$ in the gap]a, b[.

(b) Let]a, b[be a gap of type II. In this case, it follows from Proposition 5.2.6 that one of the points \sqrt{a} , \sqrt{b} satisfies the hypothesis of Lemma 5.2.6 whereas the other one satisfies the hypothesis of Lemma 5.2.7. Suppose that \sqrt{b} satisfies the hypothesis of Lemma 5.2.6 and \sqrt{a} satisfies the hypothesis of Lemma 5.2.7 (the opposite case can be considered similarly). Then, $\lim_{\lambda \to \sqrt{b}^-} (1 - F_{\beta}(\lambda)) = 1$ and $\lim_{\lambda \to \sqrt{a}^+} (1 - F_{\beta}(\lambda)) \leq 0$. On the other hand, Lemma 5.2.3 implies that the function F_{β} is continuous in $]\sqrt{a}, \sqrt{b}[$. This proves that for any $0 < \mu < 1$ there exists at least one solution of the equation (5.2.85) in $]\sqrt{a}, \sqrt{b}[$, which gives at least one eigenvalue of the operator $A^{\mu}(\beta)$ in the gap]a, b[.

Finally, the eigenvalues of the operator $A^{\mu}(\beta)$ are simple since the corresponding eigenfunctions satisfy the relation (5.2.81), which defines an eigenfunction corresponding to a given eigenvalue in a unique way.

Remark 5.2.7. Similarly to the 2D case a characterization of the essential spectrum of the operator $A^{\mu}(\beta)$ can be given in terms of the absolute value of the function g_{β} . Indeed, comparing the equation (5.2.34) with the definition (5.2.69) of the function g_{β} we conclude that

for
$$\lambda \notin \Sigma$$
, $\lambda^2 \in \sigma_{ess} \left(A^{\mu}(\beta) \right) \iff |g_{\beta}| \leqslant \frac{1}{|\sin(\lambda L_x)|} + \frac{1}{|\sin(\lambda L_y)|}.$

Notice that if we consider the non-perturbed case $\mu = 1$, the equation for the Fourier transform $\hat{\mathbf{u}}$ takes the form

$$\left(\frac{\cos\xi}{\sin\left(\lambda L_x\right)} + \frac{\cos\eta}{\sin\left(\lambda L_y\right)} - g_\beta(\lambda)\right)\widehat{\mathbf{u}}(\xi,\eta) = 0.$$

Consequently, for $\lambda \notin \Sigma$, there exists a non-zero solution if and only if

$$\exists (\xi, \eta) \in [0, \pi]^2$$
 s.t. $\frac{\cos \xi}{\sin (\lambda L_x)} + \frac{\cos \eta}{\sin (\lambda L_y)} - g_\beta(\lambda) = 0,$

i.e., if and only if $\lambda^2 \in \sigma(A(\beta))$. This means that for $\lambda \notin \Sigma$ there exists a generalized eigenfunction of the operator $A(\beta)$ (a solution of the equation $A(\beta)u = \lambda^2 u$ of at most polynomial growth) if and only if $\lambda^2 \in \sigma(A(\beta))$.

5.3 The operator $A^{\mu}_{\varepsilon}(\beta)$

As in the 2D case, the results of the works [61], [47], [57] can be applied to get the following assertions.

Theorem 5.3.1 (Essential spectrum). Let $\{]a_n(\beta), b_n(\beta)[, n \in \mathbb{N}^*\}\ be the gaps of the operator <math>A(\beta)$ on the limit graph G for $\beta \in [0, \pi]$. Then, for each $n_0 \in \mathbb{N}^*$ there exists $\varepsilon_0(\beta) > 0$ such that if $\varepsilon < \varepsilon_0(\beta)$ the operator $A_{\varepsilon}(\beta)$ has at least n_0 gaps $\{]a_{\varepsilon,n}(\beta), b_{\varepsilon,n}(\beta)[\}_{n=1}^{n_0}$ such that

$$a_{\varepsilon,n}(\beta) = a_n(\beta) + O(\varepsilon), \quad b_{\varepsilon,n}(\beta) = b_n(\beta) + O(\varepsilon), \quad \varepsilon \to 0, \quad 1 \le n \le n_0.$$

Theorem 5.3.2 (Discrete spectrum). Let $]a(\beta), b(\beta)[$ be a gap of the operator $A^{\mu}(\beta)$ on the limit graph G for $\beta \in [0, \pi]$ and $\lambda^{(0)}(\beta) \in]a(\beta), b(\beta)[$ a (simple) eigenvalue of this operator. Then there exists $\varepsilon_0(\beta) > 0$ such that if $\varepsilon < \varepsilon_0(\beta)$ the operator $A^{\mu}_{\varepsilon}(\beta)$ has an eigenvalue $\lambda_{\varepsilon}(\beta)$ inside a gap $]a_{\varepsilon}(\beta), b_{\varepsilon}(\beta)[$ such that:

$$\lambda_{\varepsilon}(\beta) = \lambda^{(0)}(\beta) + O(\varepsilon), \qquad \varepsilon \to 0.$$

We do not construct the full asymptotic expansion of the eigenvalue here, which should be possible to do using the approach described in Chapter 2. The weak version of Theorem 5.3.2 with a suboptimal rate of convergence should also be easy to obtain by constructing a "naive" pseudo-mode and adapting the argument used in Section 1.4.3 to the 3D geometry.
APPENDIX A

A.1 Self-adjointness of the operator A^{μ}

Proof of Proposition 1.3.1. Let us show that the operator A^{μ} defined in (1.3.4)–(1.3.5) is self-adjoint. Let us show first that it is symmetric. Indeed, if $f, g \in D(A^{\mu})$ then

$$(A^{\mu}f,g) = \sum_{e \in E} \int_{e} -w^{\mu}(e)f''\bar{g}d\mathbf{x} = \sum_{e \in E} \left(\int_{e} -w^{\mu}(e)f\bar{g}''d\mathbf{x} - w^{\mu}(e)(f'\bar{g})|_{ve_{1}}^{ve_{2}} + w^{\mu}(e)(f\bar{g}')|_{ve_{1}}^{ve_{2}} \right)$$
$$= (f,A^{\mu}g) + \sum_{v \in V} \left(\sum_{e_{v} \in E_{v}} (w^{\mu}(e_{v})(f_{e_{v}})'_{ext}(v)\bar{g}(v) - f(v)w^{\mu}(e_{v})(\overline{g_{e_{v}}})'_{ext}(v)) \right) = (f,A^{\mu}g)$$

To prove that the operator A^{μ} we will show that $D((A^{\mu})^*) \subset D(A^{\mu})$. Let $f \in D((A^{\mu})^*)$. by definition there exists $h_f(=(A^{\mu})^*f) \in L_2^{\mu}(G)$ such that

$$(f, A^{\mu}g) = (h_f, g), \quad \forall g \in D(A^{\mu}), \tag{A.1.1}$$

and consequently,

$$\sum_{e \in E} \int_{e} -w^{\mu}(e) f \overline{g}'' d\mathbf{x} = \sum_{e \in E} \int_{e} w^{\mu}(e) h_{f} \overline{g} d\mathbf{x}, \quad \forall g \in D(A^{\mu}).$$
(A.1.2)

The set $D(A^{\mu})$ obviously contains the functions $C_0^{\infty}(e)$ for any e. The relation (A.1.2) applied to functions $g \in C_0^{\infty}(e)$ implies that $-f''|_e = h_f|_e$. Thus,

$$-f'' = h_f, \quad f'' \in L_2^{\mu}(G).$$
 (A.1.3)

Hence,

$$f \in H^2(e), \quad \forall e \in E; \qquad \sum_{e \in E} \|f\|_{H^2(e)}^2 < \infty; \qquad (A^{\mu})^* f = -f''.$$
 (A.1.4)

We have to verify the continuity of the function f at the vertices of the graph G and the Kirchhoff's conditions. Suppose that $g \in D(A^{\mu})$ vanishes outside some neighbourhood of a given vertex v_0 . The relation (A.1.2) yields

$$\sum_{i=1}^{3} \int_{e_{v_0}^i} -w^{\mu} (e_{v_0}^i) f \overline{g}'' d\mathbf{x} = \sum_{i=1}^{3} \int_{e_{v_0}^i} -w^{\mu} (e_{v_0}^i) f'' \overline{g} d\mathbf{x}.$$
 (A.1.5)

On the other hand, integrating by parts, we get:

$$\sum_{i=1}^{3} \int_{e_{v_0}^i} -w^{\mu}(e_{v_0}^i) f \overline{g}'' d\mathbf{x} = \sum_{i=1}^{3} \int_{e_{v_0}^i} -w^{\mu}(e_{v_0}^i) f'' \overline{g} d\mathbf{x} + \sum_{i=1}^{3} w^{\mu}(e_{v_0}^i) (f_{e_{v_0}^i})'_{ext}(v_0) \overline{g}(v_0)$$
$$- \sum_{i=1}^{3} w^{\mu}(e_{v_0}^i) f_{e_{v_0}^i}(v_0) (\overline{g_{e_{v_0}^i}})'_{ext}(v_0),$$

which implies

$$\left(\sum_{i=1}^{3} w^{\mu} \left(e_{v_{0}}^{i}\right) (f_{e_{v_{0}}^{i}})_{ext}'(v_{0})\right) \bar{g}(v_{0}) - \sum_{i=1}^{3} w^{\mu} \left(e_{v_{0}}^{i}\right) f_{e_{v_{0}}^{i}}(v_{0}) (\overline{g_{e_{v_{0}}^{i}}})_{ext}'(v_{0}) = 0.$$
(A.1.6)

Choosing a function g such that it is equal to a non-zero constant in a neighbourhood of the vertex v_0 , we get

$$\sum_{i=1}^{3} w^{\mu} \left(e_{v_0}^i \right) (f_{e_{v_0}^i})'_{ext}(v_0) = 0.$$
(A.1.7)

Let us now choose a function $g \in D(A^{\mu})$ such that

$$g(v_0) = 0, \quad (g_{e_{v_0}^i})'_{ext}(v_0) = \frac{1}{w^{\mu}(e_{v_0}^i)}, \quad (g_{e_{v_0}^j})'_{ext}(v_0) = -\frac{1}{w^{\mu}(e_{v_0}^j)}, \quad (g_{e_{v_0}^k})'_{ext}(v_0) = 0,$$

for some permutation $\{i, j, k\} = \{1, 2, 3\}$. From the relation (A.1.6) applied to the function g we obtain

$$f_{e_{v_0}^i}(v_0) = f_{e_{v_0}^j}(v_0). \tag{A.1.8}$$

This proves the continuity of the function f at the vertex v_0 . Thus, we conclude that $D((A^{\mu})^*) \subset D(A^{\mu})$ and consequently, $A^{\mu} = (A^{\mu})^*$.

A.2 Quasi-modes method

In this section we shall prove Lemma A.2.1 that provides a result existence of eigenvalues for the operator $A^{\mu}_{\varepsilon,s}$ $(A^{\mu}_{\varepsilon,as})$. It relies on a pseudo-mode method. The result may be found in [53] Lemma 4, but, for the sake of completeness we give a proof.

Let H be a Hilbert space, A a self-adjoint positive definite operator:

$$\exists \alpha > 0: \quad (Au, u) \ge \alpha \|u\|^2, \qquad \forall u \in D(A)$$

Let a be the closed positive definite sesqui-linear form which corresponds to the operator A:

$$D[a] = D(A^{1/2}), \qquad a[u, u] \ge \alpha ||u||^2, \qquad \forall u \in D[a]$$

We denote by $|\cdot|_a$ the norm in the space D[a] corresponding to the scalar product

$$\langle u, v \rangle_a = a[u, v], \qquad \forall u, v \in D[a].$$

Lemma A.2.1. Suppose that $\lambda > \alpha$. If there exists $u \in D[a]$ such that

$$|a[u,v] - \lambda(u,v)| \leq \varepsilon |u|_a |v|_a, \qquad \forall v \in D[a], \qquad \varepsilon < (\lambda+1)^{-1}, \tag{A.2.1}$$

then

$$dist(\sigma(A), \lambda) \leqslant C\varepsilon, \qquad C = \lambda + 1$$

Proof. Let us define the operator $B : D[a] \to D[a]$ as follows:

$$a[Bf, v] = (f, v), \qquad \forall v \in D[a].$$

We notice that this implies

$$Bf \in D(A), \qquad ABf = f, \qquad \forall f \in D[a].$$
 (A.2.2)

We denote by $|\cdot|_a$ the norm associated with the sesqui-linear form a: $|f|_a^2 = a[f, f]$. Let us prove the following assertion.

Proposition A.2.1.

$$\lambda \in \sigma(A) \quad <=> \quad \frac{1}{\lambda} \in \sigma(B), \qquad \forall \lambda > 0.$$

Proof.

Case of eigenvalues:

1. Let λ be an eigenvalue of the operator A: $Af = \lambda f$. On a

$$(f,v) = \lambda^{-1}(Af,v) = a[\lambda^{-1}f,v], \qquad \forall v \in D[a].$$

By definition of the operator B this implies $Bf = \lambda^{-1}f$. **2.** Let λ^{-1} be an eigenvalue of the operator B: there exists $f \in D[a]$ such that $Bf = \lambda^{-1}f$. From (A.2.2) it follows that $f \in D(A)$ and $Af = \lambda ABf = \lambda f$.

Case of continuous spectrum:

1. Suppose that $\lambda \in \sigma_c(A)$. Then, there exist a singular sequence $\{u_n\}_{n \in \mathbb{N}} \subset D(A)$ such that

- a) $\inf_{n} ||u_{n}|| > 0,$
- b) $u_n \xrightarrow{w} 0$ in H,

c)
$$||(A-\lambda)u_n|| \longrightarrow 0.$$

Let us show that $\{u_n\}$ is also singular sequence for the operator B in the space D[a] equipped with the norm $|\cdot|_a$. Thus, the property

a)
$$\inf_{n} |u_n|_a > 0$$

is obviously verified. Let us prove the weak convergence of the sequence $\{u_n\}$ in the space D[a]:

$$a[u_n, v] = (Au_n, v) = ((A - \lambda)u_n, v) + \lambda(u_n, v) \longrightarrow 0, \qquad \forall v \in D[a].$$

We used the properties (b), (c) of the sequence $\{u_n\}$. Consequently,

b)
$$u_n \xrightarrow{w} 0$$
 in $D[a]$.

The only thing to show now is that $|(B - \lambda^{-1})u_n|_a \longrightarrow 0$. Indeed,

$$\begin{aligned} \left| Bu_n - \lambda^{-1} u_n \right|_a &= a[(B - \lambda^{-1})u_n, (B - \lambda^{-1})u_n] = (A(B - \lambda^{-1})u_n, (B - \lambda^{-1})u_n) \\ &= \lambda^{-1}(\lambda u_n - Au_n, (B - \lambda^{-1})u_n) \leqslant C \|Au_n - \lambda u_n\| \longrightarrow 0. \end{aligned}$$

Thus,

c) $|(B - \lambda^{-1})u_n|_a \longrightarrow 0,$

and $\{u_n\}$ is a singular sequence for the operator B at the point λ^{-1} .

2. Suppose that $\lambda^{-1} \in \sigma_c(B)$. Then, there exists a singular sequence $\{f_n\}_{n \in \mathbb{N}} \subset D[a]$ such that

- **a)** $\inf_{n} |f_{n}|_{a} > 0,$ **b)** $f_{n} \xrightarrow{w} 0$ in D[a],
- c) $|(B \lambda^{-1})f_n|_a \longrightarrow 0.$

Let us show that there exists a singular sequence for the operator A. We put $u_n = Bf_n \in D(A)$. From the properties (a) and (c) of the sequence $\{f_n\}$ it follows that $\inf_n |u_n|_a > 0$. We have:

$$|u_n|_a^2 = (A^{1/2}Bf_n, A^{1/2}u_n) = (ABf_n, u_n) = (f_n, u_n).$$

Therefore,

$$\inf_{n} \|u_{n}\| \ge \inf_{n} \frac{|u_{n}|_{a}^{2}}{\|f_{n}\|} > 0.$$

Thus, the property

a) $\inf_{n} ||u_n|| > 0$

is verified. The sequence $\{u_n\}$ being bounded, we can extract a subsequence which converges weakly to some element $h \in H$. We keep the same notation $\{u_n\}$ for the subsequence:

$$(u_n, w) \longrightarrow (h, w), \quad \forall w \in H.$$
 (A.2.3)

For $w \in D[a]$ we have:

$$(u_n, w) = a[Bf_n, w] \longrightarrow 0, \qquad \forall w \in D[a], \tag{A.2.4}$$

where we used the property (b) of the sequence $\{f_n\}$. Thus, the relations (A.2.3) and (A.2.4) imply that

 $(h,w) = 0, \qquad \forall w \in D[a].$

Since D[a] is dense in H we conclude that h = 0 and hence

b) $u_n \xrightarrow{w} 0$ in H.

The last thing to verify is the property (c). We have:

$$\|(A-\lambda)u_n\| = \|f_n - \lambda Bf_n\| \leqslant \alpha^{-1/2} |f_n - \lambda Bf_n|_a \longrightarrow 0.$$

The property

c) $||(A - \lambda)u_n|| \longrightarrow 0$

is verified which proves that $\{u_n\}$ is a singular sequence for the operator A at the point λ .

We can now finish the proof of Lemme A.2.1. Suppose that $u \in D[a]$ is such that (A.2.1) is verified. We have:

 $|a[u,v] - \lambda(u,v)| = |a[u,v] - \lambda(ABu,v)| = |a[u,v] - \lambda[Bu,v]| = |a[u - \lambda Bu,v]| \leq \varepsilon |u|_a |v|_a,$ and hence,

$$\left| (B - \lambda^{-1}) u \right|_a \leqslant \frac{\varepsilon}{\lambda} |u|_a$$

The last relation implies that

$$dist(\sigma(B),\lambda^{-1})\leqslant \frac{\varepsilon}{\lambda},$$

and for the operator A we obtain

$$dist(\sigma(A),\lambda) \leqslant \frac{\lambda\varepsilon}{1-\varepsilon} < (\lambda+1)\varepsilon,$$

where we took into account the relation $\varepsilon(1 + \lambda) < 1$.

A.3 Some auxiliary assertions

The following assertion is very similar to Lemma 3.10 in [37].

Lemma A.3.1. For each $\alpha > 0$ there exists $\varepsilon_0 > 0$ and a constants $C_1(\alpha)$, $C_2(\alpha)$ which does not depend on ε such that

$$\|v\|_{L_1(]0,\varepsilon^{\alpha}[\times]0,\varepsilon[)} \leqslant C_1(\alpha)\varepsilon^{\alpha+1/2} \|v\|_{H^1(]0,1[\times]0,\varepsilon[)}, \qquad \forall v \in H^1(]0,1[\times]0,\varepsilon[), \quad \forall \varepsilon < \varepsilon_0,$$
(A.3.1)

$$\|v\|_{L_2(]0,\varepsilon^{\alpha}[\times]0,\varepsilon[)} \leqslant C_2(\alpha)\varepsilon^{\alpha/2} \|v\|_{H^1(]0,1[\times]0,\varepsilon[)}, \qquad \forall v \in H^1(]0,1[\times]0,\varepsilon[), \quad \forall \varepsilon < \varepsilon_0.$$
(A.3.2)

Proof. We will prove the estimate (A.3.1), the proof of the estimate (A.3.2) being analogous. Due to the density of $C^{\infty}([0,1] \times [0,\varepsilon])$ in $H^1([0,1[\times]0,\varepsilon])$ it is sufficient to show (A.3.1) for functions $v \in C^{\infty}([0,1] \times [0,\varepsilon])$. Let us introduce a function $\psi_{\alpha} \in C^{\infty}([0,1] \times [0,\varepsilon])$ such that

$$\psi_{\alpha}(x,y) = 1, \quad (x,y) \in [0,\varepsilon^{\alpha}] \times [0,\varepsilon], \qquad \psi_{\alpha}(x,y) = 0, \quad x \in \left\lfloor \frac{1}{2}, 1 \right\rfloor \times [0,\varepsilon]$$

Then, $\|v\|_{L_1(]0,\varepsilon^{\alpha}[\times]0,\varepsilon[)} = \|v\psi_{\alpha}\|_{L_1(]0,\varepsilon^{\alpha}[\times]0,\varepsilon[)}$. We have

$$(v\psi_{\alpha})(x,y) = -\int_{x}^{1} \partial_{x} (v\psi_{\alpha})(x',y)dx'.$$

Hence,

$$|(v\psi_{\alpha})(x,y)| \leq \int_0^1 |\nabla (v\psi_{\alpha})(x',y)| \, dx'.$$

Consequently,

$$\begin{aligned} \|v\|_{L_{1}(]0,\varepsilon^{\alpha}[\times]0,\varepsilon[)} &\leqslant \int_{0}^{\varepsilon^{\alpha}} \left(\int_{0}^{1} \int_{0}^{\varepsilon} |\nabla (v\psi_{\alpha}) (x',y)| \, dx' dy \right) dx \\ &\leqslant \varepsilon^{\alpha+1/2} \left(\int_{0}^{1} \int_{0}^{\varepsilon} |\nabla (v\psi_{\alpha}) (x',y)|^{2} \, dx' dy \right)^{1/2} \leqslant C(\alpha) \varepsilon^{\alpha+1/2} \, \|v\|_{H^{1}(]0,1[\times]0,\varepsilon[)} \, . \end{aligned}$$

Lemma A.3.2. Let K be the rectangle $]0, a[\times]0, b[$ for some a, b > 0 and Γ its boundary $\{0\} \times [0, b]$. Then, for any $\delta > 0$ there exists a constant $C(\delta)$ such that

$$\|v\|_{L_2(\Gamma)}^2 \leqslant \delta \|v\|_{H^1(K)}^2 + C(\delta) \|v\|_{L_2(K)}^2, \qquad \forall v \in H^1(K).$$
(A.3.3)

Proof. Applying the same density argument as in the previous Lemma, as well as the fact that the traces of a convergent sequence in $H^1(K)$ converge in $L_2(\Gamma)$ we only need to prove (A.3.3) for functions in $C^{\infty}(K)$. Thus, for $v \in C^{\infty}(K)$ we have:

$$v(0,y) = v(x,y) - \int_{0}^{x} \partial_{x} v(\widetilde{x},y) d\widetilde{x}, \qquad (x,y) \in K,$$

and

$$|v(0,y)| \leq |v(x,y)| + \int_{0}^{x} |\partial_{x}v(\widetilde{x},y)| d\widetilde{x} \leq |v(x,y)| + \left(x \int_{0}^{x} |\partial_{x}v(\widetilde{x},y)|^{2} d\widetilde{x}\right)^{1/2},$$

where we used Cauchy-Schwarz inequality. Hence,

$$|v(0,y)|^2 \leqslant C\left(|v(x,y)|^2 + x\int_0^a |\nabla v(\widetilde{x},y)|^2 d\widetilde{x}\right)$$

Integrating the last inequality over $]0, \delta[\times]0, b[$ gives:

$$\delta \|v\|_{L_2(\Gamma)}^2 \leqslant C\left(\|v\|_{L_2(K)}^2 + \frac{\delta^2}{2} \|v\|_{H^1(K)}^2\right)$$

Dividing by δ yields (A.3.3).

Lemma A.3.3. Let v^{\pm} be the solutions of the problems (3.2.25) for $\alpha^2 \notin \sigma(A_{\varepsilon}), \varphi \in H^{1/2}(\Gamma_{\varepsilon}^{\pm})$. Then, $v^{\pm} \in H^2(K^{\pm})$ and the estimate (3.2.27) holds.

Proof. Let us introduce a cut-off function $\chi \in C^{\infty}\left(\left[\frac{1}{2}, \infty\right]\right)$ such that

$$\begin{cases} \chi(x) = 1, & x \in \left\lfloor \frac{1}{2}, \frac{1}{2} + \frac{\varepsilon}{8} \right\rfloor, \\ \chi(x) = 0, & x \in \left\lfloor \frac{1}{2} + \frac{\varepsilon}{4}, \infty \right[, \\ 0 \leqslant \chi(x) \leqslant 1, & \forall x \in \left\lfloor \frac{1}{2}, \infty \right\lfloor. \end{cases} \end{cases}$$

Then, the function $\tilde{v} = \chi v^+$ solves the following problem in $K_{\varepsilon}^+ = \left] \frac{1}{2}, \frac{1}{2} + \frac{\varepsilon}{4} \right[\times \left] - \frac{L}{2}, \frac{L}{2} \right[$:

$$\begin{cases} -\Delta \widetilde{v} = \alpha^2 v^+ \chi - v^+ \Delta \chi - 2\nabla v^+ \nabla \chi + u_{\varepsilon}^+(\alpha, \varphi) & \text{in } K_{\varepsilon}^+ \\ \widetilde{v}|_{\Gamma_{\varepsilon}^+} = 0, & \frac{\partial \widetilde{v}}{\partial n} \Big|_{\partial K_{\varepsilon}^+ \setminus \Gamma_{\varepsilon}^+} = 0, \end{cases}$$

where we used the fact that $v^+ \in D(A_{\varepsilon}^+)$, cf. (3.2.2). Then, applying the regularity result ([]), we conclude that $\tilde{v} \in H^2(K_{\varepsilon}^+)$ and

$$\|\widetilde{v}\|_{H^{2}\left(K_{\varepsilon}^{+}\right)} \leqslant C \left\|\alpha^{2} v^{+} \chi - v^{+} \Delta \chi - 2\nabla v^{+} \nabla \chi + u_{\varepsilon}^{+}(\alpha,\varphi)\right\|_{L_{2}\left(K_{\varepsilon}^{+}\right)} \leqslant C(\alpha) \|\varphi\|_{H^{1/2}(\Gamma_{\varepsilon,0})},$$

where $C(\alpha)$ is a continuous function depending only on the geometry of the domain. We used the continuity of the operator $(A_{\varepsilon}^+ - \alpha^2)^{-1} : L_2(B_{\varepsilon}^+) \to H^1(B_{\varepsilon}^+)$ to estimate $\|\tilde{v}\|_{H^1(K_{\varepsilon}^+)}$ as well as (3.2.5). Finally, since $\tilde{v}|_{K^+} = v|_{K^+}$, we get (3.2.27).

A.4 Technical results associated with Chapter 5

Lemma A.4.1. The condition $D(\lambda) = 0$ with $D(\lambda)$ defined in (5.2.21) is equivalent to the relation (5.2.4).

Proof. Let us compute the determinant $D(\lambda)$. First, we subtract the second column from the first, the fourth from the third and the sixth from the fifth. Then, we multiply the third line by e^{-ik_x} , the fourth one by e^{-ik_y} and the fifth one by e^{-ik_z} . We get:

$$D(\lambda) = e^{ik_x} e^{ik_y} e^{i\beta}$$

$$\times \begin{vmatrix} 0 & 1 & 0 & -1 & 0 & 0 \\ 0 & 1 & 0 & 0 & 0 & -1 \\ -2i\sin(\lambda L_x) & e^{-ik_x} - e^{-i\lambda L_x} & 0 & 0 & 0 & 0 \\ 0 & 0 & -2i\sin(\lambda L_y) & e^{-ik_y} - e^{-i\lambda L_y} & 0 & 0 \\ 0 & 0 & 0 & 0 & -2i\sin(\lambda L_z) & e^{-i\beta} - e^{-i\lambda L_z} \\ 1 - e^{i(\lambda L_x + k_x)} & 0 & 1 - e^{i(\lambda L_y + k_y)} & 0 & 1 - e^{i(\lambda L_z + \beta)} & 0 \end{vmatrix}$$

Developing the determinant with respect to the first line, we get:

$$D(\lambda) = e^{ik_x} e^{ik_y} e^{i\beta} \left(D_1(\lambda) + D_2(\lambda) \right), \qquad (A.4.1)$$

where

$$D_1(\lambda) = - \begin{vmatrix} 0 & 0 & 0 & -1 \\ -2i\sin(\lambda L_x) & 0 & 0 & 0 \\ 0 & -2i\sin(\lambda L_y) & e^{-ik_y} - e^{-i\lambda L_y} & 0 & 0 \\ 0 & 0 & 0 & -2i\sin(\lambda L_z) & e^{-i\beta} - e^{-i\lambda L_z} \\ 1 - e^{i(\lambda L_x + k_x)} & 1 - e^{i(\lambda L_y + k_y)} & 0 & 1 - e^{i(\lambda L_z + \beta)} & 0 \end{vmatrix} ,$$

$$D_2(\lambda) = \begin{vmatrix} 0 & 1 & 0 & 0 & -1 \\ -2i\sin(\lambda L_x) & e^{-ik_x} - e^{-i\lambda L_x} & 0 & 0 & 0 \\ 0 & 0 & -2i\sin(\lambda L_y) & 0 & 0 \\ 0 & 0 & 0 & -2i\sin(\lambda L_z) & e^{-i\beta} - e^{-i\lambda L_z} \\ 1 - e^{i(\lambda L_x + k_x)} & 0 & 1 - e^{i(\lambda L_y + k_y)} & 1 - e^{i(\lambda L_z + \beta)} & 0 \end{vmatrix}$$

The determinant $D_1(\lambda)$ can be computed directly:

$$D_1(\lambda) = -4\sin(\lambda L_x)\sin(\lambda L_z)\left(1 - e^{i(\lambda L_y + k_y)}\right)\left(e^{-ik_y} - e^{-i\lambda L_y}\right)$$
$$= -8\sin(\lambda L_x)\sin(\lambda L_z)\left(\cos k_y - \cos(\lambda L_y)\right). \quad (A.4.2)$$

To compute determinant $D_2(\lambda)$, we decompose it with respect to the first line. We get:

$$D_2(\lambda) = D_{2,1}(\lambda) + D_{2,2}(\lambda),$$

$$D_{2,1}(\lambda) = - \begin{vmatrix} -2i\sin(\lambda L_x) & 0 & 0 & 0\\ 0 & -2i\sin(\lambda L_y) & 0 & 0\\ 0 & 0 & -2i\sin(\lambda L_z) & e^{-i\beta} - e^{-i\lambda L_z}\\ 1 - e^{i(\lambda L_x + k_x)} & 1 - e^{i(\lambda L_y + k_y)} & 1 - e^{i(\lambda L_z + \beta)} & 0 \end{vmatrix},$$

$$D_{2,2}(\lambda) = - \begin{vmatrix} -2i\sin(\lambda L_x) & e^{-ik_x} - e^{-i\lambda L_x} & 0 & 0\\ 0 & 0 & -2i\sin(\lambda L_y) & 0\\ 0 & 0 & 0 & -2i\sin(\lambda L_z)\\ 1 - e^{i(\lambda L_x + k_x)} & 0 & 1 - e^{i(\lambda L_y + k_y)} & 1 - e^{i(\lambda L_z + \beta)} \end{vmatrix}.$$

We find then

$$D_{2,1}(\lambda) = -4\sin(\lambda L_x)\sin(\lambda L_y)\left(1 - e^{i(\lambda L_z + k_z)}\right)\left(e^{-i\beta} - e^{-i\lambda L_z}\right)$$
$$= -8\sin(\lambda L_x)\sin(\lambda L_y)\left(\cos\beta - \cos(\lambda L_z)\right), \quad (A.4.3)$$

$$D_{2,2}(\lambda) = -4\sin(\lambda L_y)\sin(\lambda L_y)\left(1 - e^{i(\lambda L_x + k_x)}\right)\left(e^{-ik_x} - e^{-i\lambda L_x}\right)$$
$$= -8\sin(\lambda L_y)\sin(\lambda L_y)\left(\cos k_x - \cos(\lambda L_x)\right). \quad (A.4.4)$$

Combining the relations (5.2.20), (A.4.1)–(A.4.4), we find the condition (5.2.4). \Box

Proof of Proposition 5.2.4.

The point λ^2 is an eigenvalue of infinite multiplicity of the operator $A(\beta)$ if and only if it is an eigenvalue of the operator $A_{\beta}(k_x, k_y)$ for any $(k_x, k_y) \in [0, \pi]^2$. It follows from Proposition 5.2.2 that $\lambda^2 \in \sigma_{pp}(A(\beta))$ if and only if $\lambda > 0$ and the equation (5.2.4) is satisfied for any $(k_x, k_y) \in [0, \pi]^2$. The first term of in the left-hand side of (5.2.4) only depends on k_x , the second term depends only on k_y and the third term is constant. Consequently, $\lambda^2 \in \sigma_{pp}(A(\beta))$ if and only if $\lambda > 0$ and all the three terms are identically zero. If $\lambda > 0$, the condition for the first term to be identically zero is $\lambda \in \Sigma_y^* \cup \Sigma_z^*$, the condition for the second term to be identically zero is $\lambda \in \Sigma_x^* \cup \Sigma_z^*$ and the condition for the third term to be identically zero is $\lambda \in \Sigma_x^* \cup \Sigma_z^*(\beta)$. Thus,

$$\lambda^2 \in \sigma_{pp}(A(\beta)) \quad \Leftrightarrow \quad \lambda \in \Pi,$$

where

$$\Pi = \left(\Sigma_y^* \cup \Sigma_z^*\right) \cap \left(\Sigma_x^* \cup \Sigma_z^*\right) \cap \left(\Sigma_x^* \cup \Sigma_y^* \cup \Sigma_z^*(\beta)\right).$$

Since $\widetilde{\Sigma}_{z}^{*}(\beta) \subset \Sigma_{z}^{*}$ (cf. (5.2.27)–(5.2.28)), we have $\Sigma_{is}(\beta) \subset (\Sigma_{x}^{*} \cup \Sigma_{y}^{*}) \cap \Sigma_{z}^{*} \subset \Pi$. Moreover, we have seen in the proof of Proposition 5.2.3 (2) that for any point $\lambda \in \Sigma_{is}(\beta)$ there exist positive numbers l^{\pm} such that $\lambda^{2} - l^{-}, \lambda^{2}[\cap \sigma(A(\beta)) = \emptyset, \lambda^{2}, \lambda^{2} + l^{+}[\cap \sigma(A(\beta)) = \emptyset]$. Consequently, the set $\Sigma_{is}(\beta)$ corresponds to isolated points of the spectrum. Let us determine $\Pi \setminus \Sigma_{is}(\beta)$. We distinguish two cases.

(i) $\beta \in]0, \pi[: \text{ in this case } \widetilde{\Sigma}_z^*(\beta) = \Sigma_z^* \text{ and the set } \Pi \setminus \Sigma_{is}(\beta) \text{ is }$

$$\begin{split} \Pi \setminus \Sigma_{is}(\beta) &= \overline{\Sigma_{is}(\beta)} \cap \Pi = \overline{(\Sigma_x^* \cap \Sigma_z^*)} \cap \left(\Sigma_y^* \cap \Sigma_z^*\right) \cap \left(\Sigma_x^* \cup \Sigma_z^*\right) \cap \left(\Sigma_y^* \cup \Sigma_z^*\right) \cap \left(\Sigma_x^* \cup \Sigma_y^* \cup \Sigma_z^*(\beta)\right) \\ &= \left(\left(\Sigma_x^* \setminus \Sigma_z^*\right) \cup \left(\Sigma_z^* \setminus \Sigma_x^*\right)\right) \cap \left(\left(\Sigma_y^* \setminus \Sigma_z^*\right) \cup \left(\Sigma_z^* \setminus \Sigma_y^*\right)\right) \cap \left(\Sigma_x^* \cup \Sigma_y^* \cup \Sigma_z^*(\beta)\right) \\ &= \left(\left(\left(\Sigma_x^* \cap \Sigma_y^*\right) \setminus \Sigma_z^*\right) \cup \left(\Sigma_z^* \setminus \left(\Sigma_x^* \cup \Sigma_y^*\right)\right)\right) \cap \left(\Sigma_x^* \cup \Sigma_y^* \cup \Sigma_z^*(\beta)\right) \\ &= \left(\left(\left(\Sigma_x^* \cap \Sigma_y^*\right) \setminus \Sigma_z^*\right) \cap \left(\Sigma_x^* \cup \Sigma_y^* \cup \Sigma_z^*(\beta)\right)\right) \cup \left(\left(\Sigma_z^* \setminus \left(\Sigma_x^* \cup \Sigma_y^*\right)\right) \cap \left(\Sigma_x^* \cup \Sigma_y^* \cup \Sigma_z^*(\beta)\right)\right) \\ &= \left(\left(\Sigma_x^* \cap \Sigma_y^*\right) \setminus \Sigma_z^*\right) \cup \left(\left(\Sigma_z^* \cap \Sigma_z^*(\beta)\right) \setminus \left(\Sigma_x^* \cup \Sigma_y^*\right)\right) = \left(\Sigma_x^* \cap \Sigma_y^*\right) \setminus \widetilde{\Sigma}_z^*(\beta). \end{split}$$

The last equality is due to the fact that for $\beta \in]0, \pi[, \Sigma_z^* \cap \Sigma_z^*(\beta) = \emptyset$. Let us show that if $\lambda_0 \in \Pi \setminus \Sigma_{is}(\beta)$, then λ_0^2 is not an isolated point of $\sigma(A(\beta))$. The equation (5.2.4) with $\lambda = \lambda_0 + \delta$ can be rewritten as

$$\frac{\cos k_x}{\cos\left(\lambda_0 L_x\right)} = \cos\left(\delta L_x\right) + \sin\left(\delta L_x\right) \frac{\cos\left(\lambda L_z\right) - \cos\beta}{\sin\left(\lambda L_z\right)} + \frac{\sin\left(\delta L_x\right)}{\sin\left(\delta L_y\right)} \left(\cos\left(\delta L_y\right) - \frac{\cos k_y}{\cos\left(\lambda_0 L_y\right)}\right).$$
(A.4.5)

If $\lambda_0 \notin \Sigma_z^*(\beta)$, then for δ small enough

$$\cos\left(\delta L_x\right) + \sin\left(\delta L_x\right) \frac{\cos\left(\lambda L_z\right) - \cos\beta}{\sin\left(\lambda L_z\right)} = 1 + \delta L_x \frac{\cos\left(\lambda_0 L_z\right) - \cos\beta}{\sin\left(\lambda_0 L_z\right)} + O(\delta^2).$$

The quantity $(\cos(\lambda_0 L_z) - \cos\beta) / \sin(\lambda_0 L_z)$ being different from zero (so either strictly positive or strictly negative), there exists $\varepsilon > 0$ such that either for $\delta \in]0, \varepsilon[$ or for $\delta \in]-\varepsilon, 0[$ one has

$$\left|\cos\left(\delta L_x\right) + \sin\left(\delta L_x\right) \frac{\cos\left(\lambda L_z\right) - \cos\beta}{\sin\left(\lambda L_z\right)}\right| < 1.$$
(A.4.6)

Choosing, for example, $\cos k_y = \cos (\delta L_y) \cos (\lambda_0 L_y)$, one can find a k_x such that (k_x, k_y) solve the equation (A.4.5).

If $\lambda_0 \in \Sigma_z^*(\beta)$, then $\cos(\lambda_0 L_z) = \cos\beta$, and then then for δ small enough

$$\cos\left(\delta L_x\right) + \sin\left(\delta L_x\right) \frac{\cos\left(\lambda L_z\right) - \cos\beta}{\sin\left(\lambda L_z\right)} = 1 - \delta^2\left(\frac{L_x^2}{2} + L_x L_z\right) + O(\delta^3).$$

Hence, the inequality (A.4.6) holds again for δ small enough and a solution (k_x, k_y) of the equation (A.4.5) can be found as in the previous case.

(ii) $\beta \in \{0, \pi\}$: in this case $\Sigma_z^*(\beta) \subset \Sigma_z^*$. Consequently, $\Sigma_z^*(\beta) \subset \Pi$. Taking into account that $\Sigma_z^*(\beta) \cap \widetilde{\Sigma}_z^*(\beta) = \emptyset$, we get $\Sigma_z^*(\beta) \cap \Sigma_{is}(\beta) = \emptyset$. Thus,

$$\Pi \setminus \Sigma_{is}(\beta) = (\Pi \setminus (\Sigma_{is}(\beta) \cup \Sigma_z^*(\beta))) \cup \Sigma_z^*(\beta).$$
(A.4.7)

Let us compute $\Pi \setminus (\Sigma_{is}(\beta) \cup \Sigma_z^*(\beta))$:

$$\Pi \setminus (\Sigma_{is}(\beta) \cup \Sigma_z^*(\beta)) = \overline{\Sigma_{is}(\beta)} \cap \Pi \cap \overline{\Sigma_z^*(\beta)}.$$

It follows from (5.2.33) that

$$\Sigma_{is}(\beta) = \left(\Sigma_x^* \cap \widetilde{\Sigma}_z^*(\beta)\right) \cup \left(\Sigma_y^* \cap \widetilde{\Sigma}_z^*(\beta)\right), \qquad \overline{\Sigma_{is}(\beta)} = \left(\overline{\Sigma_x^* \cap \widetilde{\Sigma}_z^*(\beta)}\right) \cap \left(\overline{\Sigma_y^* \cap \widetilde{\Sigma}_z^*(\beta)}\right).$$

Then,

$$\Pi \setminus (\Sigma_{is}(\beta) \cup \Sigma_{z}^{*}(\beta)) = \left(\overline{\Sigma_{x}^{*} \cap \widetilde{\Sigma}_{z}^{*}(\beta)}\right) \cap \left(\overline{\Sigma_{y}^{*} \cap \widetilde{\Sigma}_{z}^{*}(\beta)}\right) \cap \left(\Sigma_{x}^{*} \cup \Sigma_{z}^{*}\right) \cap \left(\Sigma_{y}^{*} \cup \Sigma_{z}^{*}\right) \cap \left(\Sigma_{x}^{*} \cup \Sigma_{y}^{*} \cup \Sigma_{z}^{*}(\beta)\right) \cap \overline{\Sigma_{z}^{*}(\beta)}.$$

If $\beta \in \{0, \pi\}$, then $\Sigma_{z}^{*} \cap \overline{\Sigma_{z}^{*}(\beta)} = \widetilde{\Sigma}_{z}^{*}(\beta)$. We have then

$$\begin{aligned} \Pi \setminus (\Sigma_{is}(\beta) \cup \Sigma_{z}^{*}(\beta)) \\ &= \left(\overline{\Sigma_{x}^{*} \cap \widetilde{\Sigma}_{z}^{*}(\beta)}\right) \cap \left(\overline{\Sigma_{y}^{*} \cap \widetilde{\Sigma}_{z}^{*}(\beta)}\right) \cap \left(\Sigma_{x}^{*} \cup \widetilde{\Sigma}_{z}^{*}(\beta)\right) \cap \left(\Sigma_{y}^{*} \cup \widetilde{\Sigma}_{z}^{*}(\beta)\right) \cap \left(\Sigma_{x}^{*} \cup \Sigma_{y}^{*}\right) \cap \overline{\Sigma_{z}^{*}(\beta)} \\ &= \left(\left(\Sigma_{x}^{*} \setminus \widetilde{\Sigma}_{z}^{*}(\beta)\right) \cup \left(\widetilde{\Sigma}_{z}^{*}(\beta) \setminus \Sigma_{x}^{*}\right)\right) \cap \left(\left(\Sigma_{y}^{*} \setminus \widetilde{\Sigma}_{z}^{*}(\beta)\right) \cup \left(\widetilde{\Sigma}_{z}^{*}(\beta) \setminus \Sigma_{y}^{*}\right)\right) \cap \left(\Sigma_{x}^{*} \cup \Sigma_{y}^{*}\right) \cap \overline{\Sigma_{z}^{*}(\beta)} \\ &= \left(\left(\left(\Sigma_{x}^{*} \cap \Sigma_{y}^{*}\right) \setminus \widetilde{\Sigma}_{z}^{*}(\beta)\right) \cup \left(\widetilde{\Sigma}_{z}^{*}(\beta) \setminus \left(\Sigma_{x}^{*} \cup \Sigma_{y}^{*}\right)\right)\right) \cap \left(\Sigma_{x}^{*} \cup \Sigma_{y}^{*}\right) \cap \overline{\Sigma_{z}^{*}(\beta)} \\ &= \left(\left(\Sigma_{x}^{*} \cap \Sigma_{y}^{*}\right) \setminus \widetilde{\Sigma}_{z}^{*}(\beta)\right) \cap \overline{\Sigma_{z}^{*}(\beta)}. \end{aligned}$$

Together with the relation (A.4.7) this implies that

$$\Pi \setminus \Sigma_{is}(\beta) = \left(\left(\left(\Sigma_x^* \cap \Sigma_y^* \right) \setminus \widetilde{\Sigma}_z^*(\beta) \right) \cap \overline{\Sigma_z^*(\beta)} \right) \cup \Sigma_z^*(\beta) \\ = \left(\left(\left(\Sigma_x^* \cap \Sigma_y^* \right) \setminus \widetilde{\Sigma}_z^*(\beta) \right) \setminus \Sigma_z^*(\beta) \right) \cup \Sigma_z^*(\beta) = \left(\left(\Sigma_x^* \cap \Sigma_y^* \right) \setminus \widetilde{\Sigma}_z^*(\beta) \right) \cup \Sigma_z^*(\beta).$$

It can also be rewritten as

$$\Pi \setminus \Sigma_{is}(\beta) = \left(\left(\Sigma_x^* \cap \Sigma_y^* \right) \setminus \Sigma_z^* \right) \cup \left(\Sigma_x^* \cap \Sigma_y^* \cap \Sigma_z^*(\beta) \right) \cup \left(\Sigma_z^*(\beta) \setminus \left(\Sigma_x^* \cap \Sigma_y^* \right) \right)$$

Let us show that the points of this set do not correspond to isolated points of the spectrum.

- (a) $\lambda_0 \in (\Sigma_x^* \cap \Sigma_y^*) \setminus \Sigma_z^*$: this case can be treated in the same way as the case (i).
- (b) $\lambda_0 \in \Sigma_x^* \cap \Sigma_y^* \cap \Sigma_z^*(\beta)$: in this case, taking into account that $\cos(\lambda_0 L_z) = \cos\beta$, the equation (5.2.4) with $\lambda = \lambda_0 + \delta$ can be rewritten as

$$\cos\left(\delta L_x\right) - \frac{\cos k_x}{\cos\left(\lambda_0 L_x\right)} + \frac{\sin\left(\delta L_x\right)}{\sin\left(\delta L_y\right)} \left(\cos\left(\delta L_y\right) - \frac{\cos k_y}{\cos\left(\lambda_0 L_y\right)}\right) + \frac{\sin\left(\delta L_x\right)}{\sin\left(\delta L_z\right)} \left(\cos\left(\delta L_z\right) - 1\right) = 0$$

If $\delta \neq 0$ is small enough, then, choosing, for example, $\cos k_y = \cos (\lambda_0 L_y) \cos (\delta L_y)$, one can find a k_x such that

$$\frac{\cos k_x}{\cos\left(\lambda_0 L_x\right)} = \cos\left(\delta L_x\right) + \frac{\sin\left(\delta L_x\right)}{\sin\left(\delta L_z\right)} \left(\cos\left(\delta L_z\right) - 1\right).$$

Indeed, the right-hand side of this relation is of absolute value smaller than 1 for δ small enough. The couple (k_x, k_y) is then a solution of (5.2.4).

(c) $\lambda_0 \in \Sigma_z^*(\beta) \setminus (\Sigma_x^* \cap \Sigma_y^*)$: suppose that $\lambda_0 \notin \Sigma_y^*$ (the case $\lambda_0 \notin \Sigma_x^*$ can be considered analogously). Then, taking into account that $\cos(\lambda_0 L_z) = \cos\beta$, the equation (5.2.4) with $\lambda = \lambda_0 + \delta$ can be rewritten as

$$\cos\left(\lambda L_x\right) - \cos k_x + \frac{\sin\left(\lambda L_x\right)}{\sin\left(\lambda L_y\right)}\left(\cos\left(\lambda L_y\right) - \cos k_y\right) + \sin\left(\lambda L_x\right)\frac{\cos\left(\delta L_z\right) - 1}{\sin\left(\delta L_z\right)} = 0$$

If δ is small enough, then choosing, for example, $\cos k_y = \cos (\lambda L_y)$, a k_x can be found such that

$$\cos k_x = \cos \left(\lambda L_x\right) + \sin \left(\lambda L_x\right) \frac{\cos \left(\delta L_z\right) - 1}{\sin \left(\delta L_z\right)}$$

Indeed, if $\lambda_0 \notin \Sigma_x^*$, then

$$\cos\left(\lambda L_x\right) + \sin\left(\lambda L_x\right) \frac{\cos\left(\delta L_z\right) - 1}{\sin\left(\delta L_z\right)} = \cos\left(\lambda_0 L_x\right) + O(\delta),$$

and the absolute value of this expression is smaller that 1 for δ small enough. If $\lambda_0 \in \Sigma_x^*$, then

$$\cos\left(\lambda L_x\right) + \sin\left(\lambda L_x\right) \frac{\cos\left(\delta L_z\right) - 1}{\sin\left(\delta L_z\right)} = \cos\left(\lambda_0 L_x\right) \left(1 - \delta^2 \frac{L_x^2 + L_x L_z}{2}\right) + O(\delta^3),$$

and its absolute value is also smaller that 1 for δ small enough. Thus, we proved that the equation (5.2.4) has a solution when δ is small enough.

Proof of Proposition 5.2.7.

Some of the points of the set Σ^* correspond to eigenvalues on infinite multiplicity of the operator $A(\beta)$. In the following lemma we identify the subset of points of Σ^* that do not correspond to eigenvalues on infinite multiplicity.

Lemma A.4.2. For any $\beta \in [0, \pi]$,

$$\Sigma^* \setminus (\Sigma_{is}(\beta) \cup \Sigma_{emb}(\beta)) = \left(\Sigma^*_x \setminus \left(\Sigma^*_y \cup \Sigma^*_z\right)\right) \cup \left(\Sigma^*_y \setminus (\Sigma^*_x \cup \Sigma^*_z)\right) \cup \left(\widetilde{\Sigma}^*_z(\beta) \setminus \left(\Sigma^*_x \cup \Sigma^*_y\right)\right).$$

Proof.

(i) $\beta \in]0, \pi[:$ in this case $\widetilde{\Sigma}_z^*(\beta) = \Sigma_z^*$. Hence,

$$\Sigma^* \setminus (\Sigma_{is}(\beta) \cup \Sigma_{emb}(\beta)) = \left(\Sigma^*_x \cup \Sigma^*_y \cup \Sigma^*_z\right) \setminus \left((\Sigma^*_x \cap \Sigma^*_z) \cup \left(\Sigma^*_y \cap \Sigma^*_z\right) \cup \left(\left(\Sigma^*_x \cap \Sigma^*_y\right) \setminus \Sigma^*_z\right)\right) \\ = \left(\Sigma^*_x \cup \Sigma^*_y \cup \Sigma^*_z\right) \setminus \left((\Sigma^*_x \cap \Sigma^*_z) \cup \left(\Sigma^*_y \cap \Sigma^*_z\right) \cup \left(\Sigma^*_x \cap \Sigma^*_y\right)\right),$$

and the result follows.

(ii)
$$\beta \in \{0, \pi\}$$
: in this case $\Sigma_z^* = \Sigma_z(\beta)^* \cup \widetilde{\Sigma}_z^*(\beta)$ and $\Sigma_z^*(\beta) \cap \widetilde{\Sigma}_z^*(\beta) = \emptyset$. We have:

$$\Sigma^* \setminus (\Sigma_{is}(\beta) \cup \Sigma_{emb}(\beta))$$

= $(\Sigma^*_x \setminus (\Sigma_{is}(\beta) \cup \Sigma_{emb}(\beta))) \cup (\Sigma^*_y \setminus (\Sigma_{is}(\beta) \cup \Sigma_{emb}(\beta))) \cup (\Sigma^*_z \setminus (\Sigma_{is}(\beta) \cup \Sigma_{emb}(\beta)))$

Taking into account that

$$\Sigma_{is}(\beta) \cup \Sigma_{emb}(\beta) = \left(\Sigma_x^* \cap \widetilde{\Sigma}_z^*(\beta)\right) \cup \left(\Sigma_y^* \cap \widetilde{\Sigma}_z^*(\beta)\right) \cup \left(\left(\Sigma_x^* \cap \Sigma_y^*\right) \setminus \widetilde{\Sigma}_z^*(\beta)\right) \cup \Sigma_z^*(\beta),$$
we find the result.

We come back to the proof of Proposition 5.2.7.

We have to prove that if $\lambda \in \Sigma^* \setminus (\Sigma_{is}(\beta) \cup \Sigma_{emb}(\beta))$, then λ^2 is not an eigenvalue of the operator $A^{\mu}(\beta)$. According to Lemma A.4.2, there are 3 possible cases.

(i) $\lambda \in \Sigma_x^* \setminus (\Sigma_y^* \cup \Sigma_z^*)$: since $\sin(\lambda L_z) \neq 0$, we have $\sin(\frac{\lambda L_z}{2}) \neq 0$, $\cos(\frac{\lambda L_z}{2}) \neq 0$ and from the relations (5.2.57)–(5.2.58) we find

$$c_{k,l}^{+} = \alpha^{+} d_{k,l}, \quad \alpha^{+} = \frac{1}{2} \left(\tan\left(\frac{\lambda L_{z}}{2}\right) \left(1 + e^{-i\beta}\right) + \frac{e^{-i\beta} - 1}{\tan\left(\frac{\lambda L_{z}}{2}\right)} \right), \quad (k,l) \in \mathbb{Z}^{2}, \quad (A.4.8)$$
$$c_{k,l}^{-} = \alpha^{-} d_{k,l}, \quad \alpha^{-} = \frac{1}{2} \left(\frac{1 - e^{i\beta}}{\tan\left(\frac{\lambda L_{z}}{2}\right)} - \tan\left(\frac{\lambda L_{z}}{2}\right) \left(1 + e^{+i\beta}\right) \right), \quad (k,l) \in \mathbb{Z}^{2}. \quad (A.4.9)$$

We have then

$$u_{k,l}^+(z) = d_{k,l} \left(\alpha^+ \sin\left(\lambda z\right) + \cos\left(\lambda z\right) \right), \qquad z \in \left[0, \frac{L_z}{2}\right], \qquad (k,l) \in \mathbb{Z}^2,$$
$$u_{k,l}^-(z) = d_{k,l} \left(\alpha^- \sin\left(\lambda z\right) + \cos\left(\lambda z\right) \right), \qquad z \in \left[-\frac{L_z}{2}, 0\right], \qquad (k,l) \in \mathbb{Z}^2.$$

Taking into account that $\sin(\lambda L_x) = 0$, we get from the relation (5.2.56)

$$|b_{k+\frac{1}{2},\ell}| = |b_{k-\frac{1}{2},\ell}| = |d_{k,\ell}| =: d_l, \quad \forall (k,\ell) \in \mathbb{Z}^2.$$

Thus, one necessarily has $d_{\ell} = 0$, $\forall \ell \in \mathbb{Z}$ (otherwise $u \notin L_2(G)$). Consequently, $b_{k+\frac{1}{2},\ell} = b_{k,\ell+\frac{1}{2}} = a_{k,\ell+\frac{1}{2}} = c_{k,\ell}^{\pm} = d_{k,\ell} = 0$, $\forall (k,\ell) \in \mathbb{Z}^2$. Then, from the relation (5.2.59) we get $a_{k+\frac{1}{2},\ell} = a_{k-\frac{1}{2},\ell} \cos(\lambda L_y)$, $\forall (k,\ell) \in \mathbb{Z}^2$, which implies that $|a_{k+\frac{1}{2},\ell}| = a_{\ell}, \forall (k,\ell) \in \mathbb{Z}^2$. If $a_{\ell} \neq$ for some ℓ , then $u \notin L_2(G)$. Hence, $a_{\ell} = 0$, $\forall \ell \in \mathbb{Z}$, and u = 0. This proves that λ is not an eigenvalue of the operator $A^{\mu}(\beta)$.

- (ii) $\lambda \in \Sigma_y^* \setminus (\Sigma_x^* \cup \Sigma_z^*)$: this case is treated analogously to the previous one.
- (iii) $\lambda \in \widetilde{\Sigma}_z^*(\beta) \setminus (\Sigma_x^* \cup \Sigma_y^*)$: since $\sin(\lambda L_z) = 0$, two cases are possible here.
- (a) $\sin\left(\frac{\lambda L_z}{2}\right) = 0$: then, $\cos\left(\lambda L_z\right) = 1$ and $\beta \neq 0$ (since $\lambda \in \widetilde{\Sigma}_z^*(\beta)$). The relations (5.2.57)-(5.2.58) imply that $d_{k,l} = 0$, $c_{k,l}^+ = e^{-i\beta}c_{k,l}^-$, $\forall (k,l) \in \mathbb{Z}^2$. Then, from (5.2.56) we get $b_{k+\frac{1}{2},\ell} = b_{k,\ell+\frac{1}{2}} = 0$, $\forall (k,l) \in \mathbb{Z}^2$, and since $\sin\left(\lambda L_x\right) \neq 0$, $\sin\left(\lambda L_y\right) \neq 0$, we also have $a_{k+\frac{1}{2},\ell} = a_{k,\ell+\frac{1}{2}} = 0$, $\forall (k,l) \in \mathbb{Z}^2$. Finally, we find from the relation (5.2.59) that $c_{k,l}^+ = c_{k,l}^-$, $\forall (k,l) \in \mathbb{Z}^2$, which implies that $c_{k,l}^+ = c_{k,l}^- = 0$, $\forall (k,l) \in \mathbb{Z}^2$ since $e^{i\beta} \neq 1$. Thus, u = 0 and λ is not an eigenvalue of the operator $A(\beta)$.
- (b) $\cos\left(\frac{\lambda L_z}{2}\right) = 0$: then, $\cos\left(\lambda L_z\right) = -1$ and $\beta \neq \pi$ (since $\lambda \in \widetilde{\Sigma}_z^*(\beta)$). The same argument as in the previous case applies, with the only change that now we have $c_{k,l}^+ = -e^{-i\beta}c_{k,l}^-$, $\forall (k,l) \in \mathbb{Z}^2$ and $e^{-i\beta} \neq -1$.

BIBLIOGRAPHY

- H. Ammari and F. Santosa. Guided waves in a photonic bandgap structure with a line defect. SIAM J. Appl. Math., 64(6):2018–2033 (electronic), 2004.
- [2] Y. Avishai and J. M. Luck. Quantum percolation and ballistic conductance on a lattice of wires. *Physical Review B*, 45(3):1074, 1992.
- [3] F. L. Bakharev, S. A. Nazarov, and K. M. Ruotsalainen. A gap in the spectrum of the Neumann-Laplacian on a periodic waveguide. *Appl. Anal.*, 92(9):1889–1915, 2013.
- [4] Geoffrey Beck, Sebastien Imperiale, and Patrick Joly. Mathematical modelling of multi conductor cables. Discrete and Continuous Dynamical Systems - Series S (DCDS-S), 8, 3:521–546, 6 2015.
- [5] A. Bendali, A. Huard, A. Tizaoui, S. Tordeux, and J.P. Vila. Asymptotic expansions of the eigenvalues of a 2-D boundary-value problem relative to two cavities linked by a hole of small size. C. R. Math. Acad. Sci. Paris, 347(19-20):1147–1152, 2009.
- [6] Gregory Berkolaiko and Peter Kuchment. Introduction to quantum graphs, volume 186 of Mathematical Surveys and Monographs. American Mathematical Society, Providence, RI, 2013.
- [7] M. Sh. Birman and M. Z. Solomjak. Spectral theory of selfadjoint operators in Hilbert space. Mathematics and its Applications (Soviet Series). D. Reidel Publishing Co., Dordrecht, 1987. Translated from the 1980 Russian original by S. Khrushchëv and V. Peller.
- [8] M. Sh. Birman and T. A. Suslina. A periodic magnetic Hamiltonian with a variable metric. The problem of absolute continuity. *Algebra i Analiz*, 11(2):1–40, 1999.
- [9] G. Borg. Eine Umkehrung der Sturm-Liouvilleschen Eigenwertaufgabe. Bestimmung der Differentialgleichung durch die Eigenwerte. Acta Math., 78:1–96, 1946.
- [10] E. Cancés, V. Ehrlacher, and Y. Maday. Periodic Schrödinger operators with local defects and spectral pollution. *submitted to SIAM J. Numer. Anal.*, 2012.

- [11] Robert Carlson. Adjoint and self-adjoint differential operators on graphs. *Electronic Journal of Differential Equations*, 6(1998):1–10, 1998.
- [12] Julien Coatléven. Analyse mathématique et numérique de quelques problèmes d'ondes en milieu périodique. PhD thesis, 10 2011.
- [13] G. David and L. Salma. Supercell calculations in the reduced Hartree-Fock model for crystals with local defects. *ArXiv e-prints*, December 2015.
- [14] Zhonghai Ding. A proof of the trace theorem of Sobolev spaces on Lipschitz domains. Proc. Amer. Math. Soc., 124(2):591–600, 1996.
- [15] M. S. P. Eastham. The spectral theory of periodic differential equations. Edinburgh : Scottish Academic Press, Edinburgh-London, distributed by chatto and windus edition, 1973.
- [16] Pavel Exner. Lattice kronig-penney models. Physical review letters, 74(18):3503, 1995.
- [17] Pavel Exner. Contact interactions on graph superlattices. Journal of Physics A: Mathematical and General, 29(1):87, 1996.
- [18] A. Figotin and A. Klein. Localized classical waves created by defects. J. Statist. Phys., 86(1-2):165–177, 1997.
- [19] A. Figotin and A. Klein. Midgap defect modes in dielectric and acoustic media. SIAM J. Appl. Math., 58(6):1748–1773 (electronic), 1998.
- [20] A. Figotin and P. Kuchment. Band-gap structure of spectra of periodic dielectric and acoustic media. I. Scalar model. SIAM J. Appl. Math., 56(1):68–88, 1996.
- [21] A. Figotin and P. Kuchment. Band-gap structure of spectra of periodic dielectric and acoustic media. II. Two-dimensional photonic crystals. SIAM J. Appl. Math., 56(6):1561–1620, 1996.
- [22] S. Fliss. Etude mathématique et numérique de la propagation des ondes dans des milieux périodiques localement perturbés. PhD thesis, Ecole Polytechnique, 5 2009.
- [23] S. Fliss, P. Joly, and J.-R. Li. Exact boundary conditions for wave propagation in periodic media containing a local perturbation. Bentham Science Publishers. E-Book Series Progress in Computational Physics (PiCP), Volume 1, 12 2009.
- [24] Sonia Fliss. A Dirichlet-to-Neumann approach for the exact computation of guided modes in photonic crystal waveguides. SIAM J. Sci. Comput., 35(2):B438–B461, 2013.
- [25] L. Friedlander. On the spectrum of a class of second order periodic elliptic differential operators. Comm. Math. Phys., 229(1):49–55, 2002.
- [26] Judith D. Gardiner, Alan J. Laub, James J. Amato, and Cleve B. Moler. Solution of the Sylvester matrix equation $AXB^T + CXD^T = E$. ACM Trans. Math. Softw., 18(2):223–231, 1992.

- [27] P. Grisvard. Singularities in Boundary Value Problems. Recherches en mathématiques appliquées. Masson, 1992.
- [28] Rainer Hempel and Karsten Lienau. Spectral properties of periodic media in the large coupling limit. Communications in Partial Differential Equations, 25(7-8):1445–1470, 2000.
- [29] Rainer Hempel and Olaf Post. Spectral gaps for periodic elliptic operators with high contrast: an overview. In *Progress in analysis, Vol. I, II (Berlin, 2001)*, pages 577–587. World Sci. Publ., River Edge, NJ, 2003.
- [30] A. M. Il'in. Matching of asymptotic expansions of solutions of boundary value problems, volume 102 of Translations of Mathematical Monographs. American Mathematical Society, Providence, RI, 1992. Translated from the Russian by V. Minachin [V. V. Minakhin].
- [31] Sébastien Imperiale and Patrick Joly. Error estimates for 1D asymptotic models in coaxial cables with non-homogeneous cross-section. Advances in Applied Mathematics and Mechanics, 2012.
- [32] Sébastien Imperiale and Patrick Joly. Mathematical modeling of electromagnetic wave propagation in heterogeneous lossy coaxial cables with variable cross section. *Applied Numerical Mathematics*, 2013.
- [33] J. D. Joannopoulos, R. D. Meade, and J. N. Winn. Photonic Crystal Molding the Flow of Light. Princeton University Press, 1995.
- [34] S.G. Johnson and J. D. Joannopoulos. *Photonic Crystal The road from theory to practice*. Kluwer Acad. Publ., 2002.
- [35] P. Joly and A. Semin. Construction and analysis of improved Kirchoff conditions for acoustic wave propagation in a junction of thin slots. In *Paris-Sud Working Group* on Modelling and Scientific Computing 2007–2008, volume 25 of ESAIM Proc., pages 44–67. EDP Sci., Les Ulis, 2008.
- [36] P. Joly and S. Tordeux. Matching of asymptotic expansions for wave propagation in media with thin slots. I. The asymptotic expansion. *Multiscale Model. Simul.*, 5(1):304–336 (electronic), 2006.
- [37] P. Joly and S. Tordeux. Matching of asymptotic expansions for waves propagation in media with thin slots. II. The error estimates. M2AN Math. Model. Numer. Anal., 42(2):193–221, 2008.
- [38] D. Klindworth. On the numerical computation of photonic crystal waveguide band structures. PhD thesis, Technische Universität Berlin, 2015.
- [39] Frédéric Klopp. Etude semi-classique d'une perturbation d'un opérateur de Schrödinger périodique: cas d'un nombre fini de perturbations. Asymptot. Anal., 15(1):55–102, 1997.
- [40] P. Kuchment. Floquet theory for partial differential equations, volume 60 of Operator Theory: Advances and Applications. Birkhäuser Verlag, Basel, 1993.

- [41] P. Kuchment and B. Ong. On guided waves in photonic crystal waveguides. In Waves in periodic and random media (South Hadley, MA, 2002), volume 339 of Contemp. Math., pages 105–115. Amer. Math. Soc., Providence, RI, 2003.
- [42] Peter Kuchment. Quantum graphs. I. Some basic structures. Waves Random Media, 14(1):S107–S128, 2004. Special section on quantum graphs.
- [43] Peter Kuchment. Quantum graphs. II. Some spectral properties of quantum and combinatorial graphs. J. Phys. A, 38(22):4887–4900, 2005.
- [44] Peter Kuchment. Quantum graphs: an introduction and a brief survey. In Analysis on graphs and its applications, volume 77 of Proc. Sympos. Pure Math., pages 291–312. Amer. Math. Soc., Providence, RI, 2008.
- [45] Peter Kuchment and Sergei Levendorskiî. On the structure of spectra of periodic elliptic operators. Trans. Amer. Math. Soc., 354(2):537–569 (electronic), 2002.
- [46] Peter Kuchment and Beng-Seong Ong. On guided electromagnetic waves in photonic crystal waveguides. In Operator theory and its applications, volume 231 of Amer. Math. Soc. Transl. Ser. 2, pages 99–108. Amer. Math. Soc., Providence, RI, 2010.
- [47] Peter Kuchment and Hongbiao Zeng. Convergence of spectra of mesoscopic systems collapsing onto a graph. J. Math. Anal. Appl., 258(2):671–700, 2001.
- [48] A. A. Kutsenko. Periodic lattice with defects. ArXiv e-prints, May 2013.
- [49] A. A. Kutsenko. Wave propagation in periodic lattices with defects of smaller dimension. ArXiv e-prints, May 2013.
- [50] V. Maz'ya, S. Nazarov, and B. Plamenevskij. Asymptotic theory of elliptic boundary value problems in singularly perturbed domains. Vol. I, volume 111 of Operator Theory: Advances and Applications. Birkhäuser Verlag, Basel, 2000. Translated from the German by Georg Heinig and Christian Posthoff.
- [51] William McLean. Strongly elliptic systems and boundary integral equations. Cambridge University Press, 2000.
- [52] S. A. Nazarov. An example of multiple gaps in the spectrum of a periodic waveguide. Mat. Sb., 201(4):99–124, 2010.
- [53] S. A. Nazarov. Opening of a gap in the continuous spectrum of a periodically perturbed waveguide. *Mat. Zametki*, 87(5):764–786, 2010.
- [54] S. A. Nazarov. The asymptotic analysis of gaps in the spectrum of a waveguide perturbed with a periodic family of small voids. J. Math. Sci. (N. Y.), 186(2):247– 301, 2012. Problems in mathematical analysis. No. 66.
- [55] L. Parnovski. Bethe-Sommerfeld conjecture. Ann. Henri Poincaré, 9(3):457–508, 2008.
- [56] L. Parnovski and A. V. Sobolev. Bethe-Sommerfeld conjecture for periodic operators with strong perturbations. *Invent. Math.*, 181(3):467–540, 2010.

- [57] Olaf Post. Spectral convergence of quasi-one-dimensional spaces. Ann. Henri Poincaré, 7(5):933–973, 2006.
- [58] Olaf Post. Spectral analysis on graph-like spaces, volume 2039 of Lecture Notes in Mathematics. Springer, Heidelberg, 2012.
- [59] Anthony Ralston and Philip Rabinowitz. A first course in numerical analysis. McGraw-Hill Book Co., New York-Auckland-Bogotá, second edition, 1978. International Series in Pure and Applied Mathematics.
- [60] M. Reed and B. Simon. Methods of modern mathematical physics v. I-IV. Academic Press, New York, 1972-1978.
- [61] Jacob Rubinstein and Michelle Schatzman. Variational problems on multiply connected thin strips. I. Basic estimates and convergence of the Laplacian spectrum. *Arch. Ration. Mech. Anal.*, 160(4):271–308, 2001.
- [62] K. Schmidt and R. Kappeler. Efficient computation of photonic crystal waveguide modes with dispersive material. Opt. Express, 18(7):7307–7322, 2010.
- [63] A. Semin. Propagation d'ondes dans des jonctions de fentes minces. PhD thesis, Université Paris-Sud, 2010.
- [64] S. Soussi. Convergence of the supercell method for defect modes calculations in photonic crystals. SIAM J. Numer. Anal., 43(3):1175–1201 (electronic), 2005.
- [65] Lawrence E. Thomas. Time dependent approach to scattering from impurities in a crystal. Comm. Math. Phys., 33:335–343, 1973.
- [66] M. Van Dyke. Perturbation methods in fluid mechanics. Applied Mathematics and Mechanics, Vol. 8. Academic Press, New York, 1964.
- [67] M. Vorobets. On the Bethe-Sommerfeld conjecture for certain periodic Maxwell operators. J. Math. Anal. Appl., 377(1):370–383, 2011.

Résumé

Cette thèse porte sur la propagation des ondes acoustiques dans des milieux périodiques. Ces milieux ont des propriétés remarquables car le spectre associée à l'opérateur d'ondes dans ces milieux a une structure de bandes : il existe des plages de fréquences dans lesquelles les ondes monochromatiques ne se propagent pas. Plus intéressant encore, en introduisant des défauts linéiques dans ce type de milieux, on peut créer des modes guidés à l'intérieur de ces bandes de fréquences interdites. Dans ce manuscrit nous montrons qu'il est possible de créer de tels modes guidés dans le cas de milieux périodiques particuliers de type quadrillage : plus précisément, le domaine périodique considéré est constitué du plan \mathbb{R}^2 privé d'un ensemble infini d'obstacles rectangulaires régulièrement espacés (d'une distance ε) dans deux directions orthogonales du plan, que l'on perturbe localement en diminuant la distance entre deux colonnes d'obstacles. Les résultats sont ensuite étendus au cas 3D.

Ce travail comporte un aspect théorique et un aspect numérique. Du point de vue théorique l'analyse repose sur le fait que, comme ε est petit, le spectre de l'opérateur associé à notre problème est "proche" du spectre d'un problème posé sur le graphe obtenu comme la limite géométrique du domaine quand ε tend vers 0. Or, pour le graphe limite, il est possible de calculer explicitement le spectre. Ensuite, en utilisant des méthodes d'analyse asymptotique on étudie le spectre de l'opérateur non-limite. On illustre les résultats théoriques par des résultats numériques obtenus à l'aide d'une méthode numérique spécialement dédiée aux milieux périodiques : cette dernière est basée sur la réduction du problème de valeurs propres initial (linéaire) posé dans un domaine non-borné à un problème nonlinéaire posé dans un domaine borné (en utilisant l'opérateur de Dirichlet-to-Neumann exact).

Abstract

The present work deals with propagation of acoustic waves in periodic media. These media have particularly interesting properties since the spectrum associated with the underlying wave operator in such media has a band-gap structure: there exist intervals of frequences for which monochromatic waves do not propagate. Moreover, by introducing linear defects in this kind of media, one can create guided modes inside the bands of forbidden frequences. In this work we show that it is possible to create such guided modes in the case of particular periodic media of grid type: more precisely, the periodic domain in question is \mathbb{R}^2 minus an infinite set of rectangular obstacles periodically spaced in two orthogonal directions (the distance between two neighbour obstacles being ε), which is locally perturbed by diminishing the distance between two columns of obstacles. The results are extended to the 3D case.

This work has a theoretical and a numerical aspect. From the theoretical point of view the analysis is based on the fact that, ε being small, the spectrum of the operator associated with our problem is "close" to the spectrum of a problem posed on a graph which is a geometric limit of the domain as ε tends to 0. However, for the limit graph the spectrum can be computed explicitly. Then, we study the spectrum of the non-limit operator using asymptotic analysis. Theoretical results are illustrated by numerical computations obtained with a numerical method developed for study of periodic media: this method is based on the reduction of the initial (linear) eigenvalue problem posed in an unbounded domain to a non-linear problem posed in a bounded domain (using the exact Dirichlet-to-Neumann operator).